CANADIAN THESES ON MICROFICHE

· •

I.S.B.N.

THESES CANADIENNES SUR MICROFICHE

National Library of Canada Collections Development Branch

Canadian Theses on Microfiche Service

Ottawa, Canada K1A 0N4

Bibliothèque nationale du Canada Direction du développement des collections

Service des thèses canadiennes sur microfiche

NOTICE

The quality of this microfiche is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

If pages are missing, contact the university which granted the degree.

Some pages may have indistinct print especially if the original pages were typed with a poor typewriter ribbon or if the university sent us a poor photocopy.

Previously copyrighted materials (journal articles, published tests, etc.) are not filmed.

Reproduction in full or in part of this film is governed by the Canadian Copyright Act, R.S.C. 1970, c. C-30. Please read the authorization forms which accompany this thesis.

THIS DISSERTATION HAS BEEN MICROFILMED EXACTLY AS RECEIVED

AVIS

La qualité de cette microfiche dépend grandement de la qualité de la thèse soumise au microfilmage. Nous avons tout fait pour assurer une qualité supérieure de reproduction.

S'il manque des pages, veuillez communiquer avec l'université qui a conféré le grade.

La qualité d'impression de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de mauvaise qualité.

Les documents qui font déjà l'objet d'un droit d'auteur (articles de revue, examens publiés, etc.) ne sont pas microfilmés.

La reproduction, même partielle, de ce microfilm est soumise à la Loi canadienne sur le droit d'auteur, SRC 1970, c. C-30. Veuillez prendre connaissance des formules d'autorisation qui accompagnent cette thèse.

> LA THÈSE A ÉTÉ MICROFILMÉE TELLE QUE NOUS L'AVONS RECUE



National Library of Canada

Canadian Theses Division

Bibliothèque nationale du Canada

0-315-05936-2

Ottawa, Canada K1A 0N4

53856

PERMISSION TO MICROFILM - AUTORISATION DE MICROFILMER

Division des thèses canadiennes

Please print or type — Écrire en lestres moulées ou dactylographier

Full Name of Author -- Nom complet de l'auteur Belcher El-Nahhas, Susan Margaret Date of Birth --- Date de naissance Country of Birth - Lieu de naissence England March 27, 1948 Permanent Address - Résidence fixe 11527-92 Street. Edmonton Alberta T5G028 Title of Thesis - Titre de la thèse Education As Capital: A Critique University - Université University of Alberta Degree for which thesis was presented - Grade pour lequel cette thèse fut présentée ŧ M. Ed. Year this degree conferred — Année d'obtention de ce grade Name of Supervisor - Nom du directeur de thèse Dr. Raj S. Pannu Permission is hereby granted to the NATIONAL LIBRARY OF L'autorisation est, par la présente, accordée à la BIBLIOTHÈ-CANADA to microfilm this thesis and to lend or sell copies of QUE NATIONALE DU GANADA de microfilmer cette thèse et de the film. prêter ou de vendre des exemplaires du film. The author reserves other publication rights, and neither the L'auteur se réserve les autres droits de publication; ni la thèse thesis nor extensive extracts from it may be printed or otherni de longs extraits de celle-ci ne doivent être imprimés ou wise reproduced without the author's written permission. autrement reproduits sans l'autorisation écrite de l'auteur.

Date	Signature
Sept. 15, 1981	Shibelchen El-Nakhas
	· · · · · · · · · · · · · · · · · · ·

THE UNIVERSITY OF ALBERTA

EDUCATION AS CAPITAL: A CRITIQUE

by ,

) Susan M. Belcher El-Nahhas

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE

OF MASTER OF EDUCATION

IN

SOCIOLOGY OF EDUCATION

DEPARTMENT OF EDUCATIONAL FOUNDATIONS

EDMONTON, ALBERTA

FALL, 1981

THE UNIVERSITY OF ALBERTA.

RELEASE FORM

NAME OF AUTHOR: SUSAN MARGARET BELCHER EL-NAHHAS

TITLE OF THESIS: EDUCATION AS CAPITAL: A CRITIQUE

DEGREE FOR WHICH THESIS WAS PRESENTED: MASTER OF EDUCATION

YEAR THIS DEGREE WAS GRANTED: 1981

Permission is hereby granted to THE UNIVERSITY OF ALBERTA LIBRARY to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly, or scientific research purposes only.

The author reserves other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without the author's written permission.

SniBelcher El-habkas

PERMANENT ADDRESS:

11527 - 92 Street

Edmonton, Alberta

Canada T5G0Z8

ept. 15, 1981 DATE:

.

THE UNIVERSITY OF ALBERTA

FACULTY OF GRADUATE STUDIES AND RESEARCH

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research, for acceptance, a thesis entitled EDUCATION AS CAPITAL: A CRITIQUE submitted by Susan Margaret Belcher El-Nahhas in partial fulfillment of the requirements for the degree of Master of Education in Sociology of Education.

Supervisor

Date.

 \mathbf{C}

Whatever there is of value in this thesis is dedicated to my parents, Norman and Amy Belcher, for their encouragement and support for my education, and to my daughters, Amy and Jasmine El-Nahhas.

f

iv

ABSTRACT

The idea that inherited and acquired capabilities (knowledge and skills) possessed by workers constitute capital in the same sense as material capital (e.g., money, land) was reintroduced to neo-classical economics by the American economist. Theodore William Schultz in 1960 . and has had a revolutionary effect upon educational ideology, policies, curricula, pedagogy, and organizational structure in North America, Western Europe, and much of the Third World. Human Capital Theory, as developed and formulated by Schultz, posits that both the quantity and quality of human capital are dependent upon several factors, such as standard of health, availability and mobility of labor, on-the-job training, and education.

Education, according to Schultz, is the one factor of which individuals in modern industrialized capitalist countries have not acquired enough in order to maximize their productive capabilities (and hence, their income); and it is one of several factors of human capital which is grossly underdeveloped in much of the Third World. This idea, like the "post-industrial" society theory in sociology, asserts that since World War II, there have been revolutionary changes in technology resulting in a drastic reduction and eventual elimination of unskilled occupations, accompanied by a proliferation of new and up-graded occupations requiring workers with sophisticated scientific and technological skills. The demand for highly educated workers was thought to be infinitely elastic, riging ever faster than the supply.

Human capital theory, therefore, shifted the traditional emphasis on physical capital accumulation to human capital formation for individual and national economic growth. Human capital theorists, like their counterparts in sociology, the technological functionalists, called upon governments and educators to implement changes in educational ideology. policy, curricula, pedagogy, and organizational structure so that systems of public education could function more effectively and efficiently in meeting the "needs," of a modern industrialized society -- as well as competing more effectively with the Soviets. Individuals and society should, thus, invest in more and better quality schooling in order to increase human capital accumulation, so that both would realize increased income from greater productivity.

The emphasis placed upon the development of human capital (to generate a stream of income and greater productivity benefiting both the individuals themselves and society as a whole) gained popularity on a mass level during the early 1960's as a panaces for many social ills ---including poverty and gross social inequality -- and renewed faith in democracy.

In this thesis, we attempt to show the theoretical and empirical inadequacies of human capital theory and to unveil the ideological underpinnings of it. It is our contention that human capital theory has served to obfuscate the class structure of capitalist society.

The relationship between education, accupation, and labor income from the human capital perspective, as specified by the rate-of-return analysis and various neo-classical economic assumptions, depends, ultimately, on the supply and demand of both labor and jobs, although human capital theory is primarily a supply theory (of labor). We have demonstrated that high levels of human capital in capitalist societies does not necessarily mean it can be productively employed. It is not the inadequacies of individuals that are responsible for their unemployment, but rather the structure and dynamics of labor markets in capitalist societies. Human

vi . '*

oapital does not guarantee one a competitive rate of return, unless it is scarce and in high demand. Human capital has not superceded physical capital as the primary source of wealth and power. Very few workers are self-employed. Employment is still subject to the power of capitalists.

The nature of the changes which have occurred in the educational system reflects the need for hierarchical differentiation of individuals in the labor force. The increased stratification and differentiation of schools by type and level facilitates the efficient allocation of individuals into the highly stratified, segmented labor markets.

Rather than workers requiring more and better quality schooling for job adequacy in a knowledge/skill-intensive occupational structure, we conclude that the primary rationale was a need for a more overt socialization for the job market, which was becoming increasingly deskilled, bureaucratic, alienating, and boring; as well as allowing differences in education, which became much more differentiated in type and level during the 1960's, to continue to be a primary basis for differential allocation in the occupational structure, and hence, maintaining socioeconomic inequality and the continuation of a class society.

In addition to providing certification for allocation to the labor force, the educational system serves to stabilize the <u>status quo</u>, reproduce the class system, and to legitimize it by socializing individuals to accept their place in society. Rather than serving the needs of the non-elite, schooling serves the needs of the elite. Thus, it is the reproduction of the class structure which sets the limits of educational policy and reform.

Vii

ACKNOWLEDCMENTS

I wish to thank the members of my committee, Drs. R.S. Pannu, A.M. Decore, and A.K. Davis, for their interest, encouragement, and effort which have contributed to the completion of this thears. A special debt of gratitude is extended to my thesis supervisor, Dr. Raj Pannu, for his patience and understanding during in production period, as well as for his constructive suprestions; and to Dr. A.M. Decore for the time and help she has so generatively provided.

This thesis is the product of long journey. It could not have been written without the financial support of my husband, Y.T. El-Nahhas; the understanding and co-operation of my daughters; and the psychological and intellectual support of friends. I wish to acknowledge the teachers, professors, and friends who have, directly or indirectly, contributed to my pursuit of graduate studies and the eventual writing of this thesis: E. Doligalski, T. Miller, J. Miller, G. Christen, C. Beatty, N. Schultz, J. Schmelzinger, D. Snowden, D. Karrels, Dr. S. Mendenhall, I. White, H. Getahun, K.A. Latigo Olal, M. Lind, Georges Tatepo, Q. Alfarhan, Dr. Louise Sweet, L. Crawford, L. Reid, P. Schmidt, Tamara Riabov, Shanna Farough, Susan Schaeffer, Mary Ann King, Margaret Hodgson, Margareta Ortega, Rachad Antonius, Alan Segal, Dr. J. Bulcock, Dr. H. Garfinkle, Dr. R.S. Patterson, Dr. M. Lupul, John Young, Dr. E.H. Shaffer, C. and R. Nunn, Swee Hin Toh, David Leadbeater, Cheryl Hume, Emma Pivato, Dr. C.N. Suchowersky, Dr. R.A. Davis, Dr. M. She'alan, Dr. M. Badawi, Dr. P.J. Miller, Friedl and Sharraf Khattab, Dr. Zoltan Berkes, Robert Gibson, Leslie Savage, Wendy Kersteen, Dr. Kwok-choy Lee, Dr. Uriel Dimer, Dr. R.W. Thatcher, Max Innes, Julian Tanner, and Judith Martin.

viii

Thanks are also expressed for the Alberta Government Graduate Scholarships which I have received and for the opportunity to utilize good quality day care. Finally, credit and thanks are extended to my parents and to Eric and Joyce Rowland for their sustained support throughout.

ix

ĺ

:

TABLE OF CONTENTS

CHAPTER		PAGE			
I.	Introduction	、 1			
II.	Occupation, and Labor Income From the Perspective of				
	Econômics	26			
	A. Introduction	26			
	B. Macro-Level: National Growth/Development Approach	32			
	C. Micro-Level: Investment Approach	51			
III.	The Influence of Human Capital Theory on Public Policy	84			
	A. Scale and Scope of Educational Expansion at All Levels B. Rate-of-Return Analysis to Determine Efficient and	. 86			
	Equitable Allocation of Resources	104			
IV.	The Technical-Function Theory of Education: The Relationship Between Education, Occupation, and Labor Income From the Perspective of Sociology				
		121			
	A. Introduction	121			
	B. The Relationship Between Education, Occupation, and				
	Labor Income as Posited by the Technical-Function Theory				
	of Education and Technological Functionalism in Sociology C. Equality of Educational Opportunity and the Technocratic-	127			
	Meritocratic Ideology	131			
v.	The Influence of the Concept of Equality of Educational				
	Opportunity on Educational Policies	137			
	A. Revitalization of Classical Liberalism in Education	137			
	B. Baby Boom The Demographic Factor.	144			
	C. Sputnik Syndrome The Ideological Factor	145			
	D. Modern Liberalism and Education	156			
	1. Individual Welfare and Equality of Opportunity				
	2. The School as an Instrument of Social Reform				
	3. Education and the "needs" of Society E. Disillusionment with Modern Liberal Educational Policies				
	E. Disillusionment with Modern Liberal Educational Policies 1. Rise of Radical and Marxist Alternatives	159			
	2. Resurgence of Conservatism				

		, ,			;	
		23		•	,	
	•		•		PAGE	
VI.		itique of Human Capits	1 Theory and the Tech	nical-	· •	
	Fu	nction Theory of Educe	ition		164	
-	. A.		l Productivity	- · · · ·	164	й .
	В. С.	Assumption of a Comp	etitive Labor Market		165	
	υ.	Quality (of Education	and the Affect of Se	x, Race, 🔅	·	1
	- ^	in Education.	ate of Return to Imve	stments		
	D.			•	168	4 N
		Education '	ty Through Investment	in		
	Ε.		ity of Education and		170	1 -
		in the Distribution	of Personal Income	inequalities	1 7 -	•
	F.	Assumption of a Cont	inually Growing Econo		172	
		Unemployment, and an	Infinitely Elastic D.	my, LOW antend Éam		- "A"
		Highly Educated Works	era	emand for	174	•
	G.		are State and Governme	ont	1.14	
		Regulation of the Eco	on on v	enc ,	176	
	Н.	Underemployment and 1	Unemployment of the H	ighly-	170	•
		Educated	4	~ e ··· • y	177	ı
• •	I:	Manpower Training Pro	Dgtams		182	-
	J.	The War on Poverty in	n the United States		187	
	K.	Gross Inequalities in	the Distribution of	Personal -		
ŧ		Income	· · · · · · · · · · · · · · · · · · ·		189	ε.
• * .	L.	•	Analysis to Balance H	Economic .	······································	
		Efficiency and Social	Equity	o	198	,
	•		•	<u>ہ</u>		
VII.	Sum	mary and Conclusion	<i>.</i>	v	210	
	000 41				· · · · · · · · · · · · · · · · · · ·	
BIBLI	UGKA	rni -	•	ų	258 h	N
		Articles		A	1997 - 1997 -	💭 🖉 🖓
•	A. B.	Articles 🔬		•	258	
	ь. C.				294	
i	υ.	Interviews	•		326	A

xi

LIST OF TABLES

TABLE DESCRIPTION PAGE 2.1: The Effect of Education on Labor Income of Workers 36 2.2: Sources of Growth in the U.S., 1929-57 and 1950-62 44 2.3: Social and Private Rates of Return by Educational Level and Country 69 2.4: Distribution of Total Resources Devoted to Education by Level and Country 71 2.5: Social Rates of Return to Investment in Physical Capital in Certain Countries 73 2.6: Average Rates of Return by Educational Level and Per Capital Income 74 2.7: Overall Social Rate of Return to Education and Per Capita 75 Income by Country 2.8: Overall Social Rate of Return to Education and the Level of Economic Development 77 3.1: Enrollment Ratios for the 20-24 Year Age Group, Selected Countries 90 · · · · · · · · 3.2: Number of Ph.D.'s Earned Per 10,000 Persons Aged 23 and 01der 93 3.3: Share of GNP Devoted to Public Expenditures on Education 94 3.4: Enrollment Ratios (of Attendance) in Post-Secondary Education 97

xii

'. ∽i

CHAPTER I

INTRODUCTION

Public systems of schooling for all children in a society is a relatively new concept in human history. Not only is it new, it is controversial. Rather than developing consensus on the role and function of public systems of schooling, educators seem to be more divided now than ever on its purpose. Heated debate and dissention seems to be accepted as part of the process of developing educational policies. That debate carries over into the political arena when local, provincial, state, and federal governments are involved in the administration and/or funding of these public systems of schooling.

The purpose of this study is to examine and explain the major shifts in educational policies during the period 1945-1980. There are broad similarities in these shifts across North America, Western Europe, and much of the Third World. In particular, it is the sudden and explosive expansion of both the public systems of education and educational participation at all levels of schooling which will be examined and explained. In the words of Karabel and Halsey (1977B:12), "(t)he rate of growth of educational systems after World War II dwarfed even the earlier lurch of European societies in the nineteenth century into universal primary education."

Both the scale and pattern of this educational change have been the subject of study by sociologists, economists, and other social scientists. In fact, a vigorous debate concerning the explanation of the post-World War II educational expansion, particularly in the Western

industrialized capitalist countries, has been going on among social scientists of different theoretical persuasions. Technical-functionalist, Weberian, neo-Weberian; Marxist, neo-Marxist, and human capital theorists have located the causes of this change in different forces and characteristics of contemporary industrial capitalism. But has social science theorizing about education played any discernible role in the expansion of educational systems, perhaps by providing social scientific rationale for expansionist policies? This question, it seems, has not received sustained attention. It seems highly appropriate to ask how social science knowledge of education, particularly since World War II, has influenced changes in education which have occurred in Western industrialized capitalist countries as well as in the underdeveloped countries.

Although this phenomenon cannot be explained entirely by one factor alone, it is posited that the development and subsequent acceptance of human capital theory by economists, and soon after by educators and politicians, has had overwhelming influence in precipitating a major shift in educational policies during the late 1950's. Furthermore, this was reinforced by the dominance of technological functionalism in sociology and its influence on sociologists and educators; and in particular, by the dominance of the technical-function theory of education. Thus, human capital theory and the technical-function theory of education developed within the paradigm of functionalism, which was the dominant paradigm encompassing all the social sciences in the United States, Canada, France, Germany, Scandinavis, and Japan (Karabel and Halsey, 1977B:4).

The major shift in educational policies during the late 1950's and early 1960's can be conceptualized in terms of a rejection of elitist education and conservative philosophy and the movement towards

modern liberalism and "mass" education. The reasons for this shift are multiple and complex. Two major factors which set the stage for this development are the post-World War II baby boom and the "Sputnik syndrome" of the Cold War. However, it is posited that "...sociology provided a convincing theoretical rationale for the expansion and differentiation of education..." and human capital theory provided the economic rationale for increased educational investment and "...exerted a considerable influence not only in the academy but also on the development of educational systems throughout the world" (Karabel and Halsey, 1977B:11-12). This thesis focuses on the theoretical character of this rationale.

This thesis is a historical one in the sociology of knowledge tradition utilizing the sociology of sociology approach to explain the development of human capital theory and its subsequent influence on education and public policy in most capitalist countries, both developed and underdeveloped.

Thus, knowledge of the social and historical context is a necessary part of the explanation. The framework employed is that of critical, conflict sociology, because we feel that by focusing on conflict better explanations of social reality are gained than by focusing on consensus. Therefore, we reject the functionalist framework which is based on value consensus and social harmony.

In rejecting the functionalist framework, we also reject the notion that social science is or ever can be value-free or neutral. The values underlying this study are not veiled. Unlike many rather naive functionalists of the 1950's,¹ there is no delusion here that the approach used, let alone this study in particular, will solve any

problems on the level of praxis. However, critical assessment using the historical, comparative approach is a powerful tool for understanding social reality. We are a product of our past; and although we may be ignorant of it, we cannot escape its influence.

The purpose of this thesis is to understand why and how we are at the present state of affairs in education. Although ideological currents and conflicts are a factor in this process, the intent of the thesis is to illustrate the need for a meta-theoretical synthesis with possibilities for praxis rather than a purely ideological socialaction approach.

The limitations of the thesis derive primarily from the fact that this is an information-rich field of inquiry. Theoretical and empirical research in economics and sociology has a long, rich history. Educational research, though much more recent, has grown in volumn and importance since the 1930's. Karabel and Halsey (1977B:1) remark that "(o)ver the last generation educational research has come from the humblest origins of the social sciences to occupy a central position in sociology, as well as to receive considerable attention from economists, historians, and anthropologists."

Consequently, the sheer volumn of literature makes it impossible for one to deal with it all. Not only does one have to be selective, but because of the plethors of dimensions to the issues, it is virtually impossible to be truly comprehensive.

The human capital literature in economics is not only massive, it is split into differing sub-schools. It was necessary, therefore, to deal with human capital theory as formulated by Theodore W. Schultz. The Schultzian model of human capital theory was chosen because there

is general consensus among economists that Schultz is the founder of the modern form of human capital theory.² Since Schultz's announcement of this theory in 1960, other economists have developed variants but the Schultzian model has remained dominant. Furthermore, it is the Schultzian model of human capital which has had great influence in public policy formation around the world. The fact that Schultz was awarded the 1980 Nobel Prize in Economics for his work on human capital theory is evidence of its importance.

Ĉ)

Structural functionalism in sociology, and the technical-function theory of education in particular, has not been marked by internal dissention. The theory of social stratification developed by Davis and Moore (1945A) is the accepted theoretical foundation of this school. Theoretical and empirical studies anchored in the technical-function theory of education tradition are plentiful. Consequently, only some of the well-known literature is used.

Because this thesis draws attention to the fact that human capital theory, and to a lesser extent the technical-function theory of education, have had world-wide impact on systems of public schooling and policy formation, literature other than the primarily American material is also used. An attempt is made to discuss the Canadian situation and in broad general terms the situation in Western Europeand the Third World. Here too, it is impossible to deal with all the literature. Literature was selected on the basis of it being both well-known and reasonably thorough.

In addition to the massive volumn of literature relevant to this thesis, there is another limitation -- one which is more difficult to deal with -- posed by the fact that "...the relation between

thought and political action is one of unusual if not unprecedented contention" (Karabel and Halsey, 1977B:1). Our rejection of value-free social science has already been mentioned.

The complexity of modern society makes it virtually impossible for governments to develop policies without relying on information made available through social science research. The fact that much of the social science research dealing with education is normative, politics becomes a very important aspect of it. Even "empirical" research is colored by values in that one's value-orientation determines to a great extent what questions are posed and what constitutes "fact".

Research has been viewed by both economists and sociologists as being both useful and important to public policy formation. The use of such research for educational policy formation is no less important. Schultz is very explicit about the fact that one of his main aims in conducting human capital research is to actively influence public policy formation.

Human capital research has important policy implications, as I shall show.

My purpose in this survey is to examine some of the interactions between policy and research with special reference to human capital. As I proceed, I shall attempt to appraise the more important parts of this research -- with a view to assessing the need for additional work in this area that may be useful in making policy decisions...

In short, then, the central purpose of my ended is to winnow the research opportunities in the area of human capital. It ally, I would like to identify these opportunities and rate them in accordance with the otential value of their contributions the private and public decisions, But I shell mettle for less because of the limitation of my knowledge and because of

the uncertainty of the nature of the advances in knowledge that can be achieved by means of research (Schultz, 1972A:1).

In discussing the setting and scope of a survey Schultz con-

ducted, he remarked,

(1) astly, and most importantly, I shall enter upon the research opportunities in substantial detail...Because of the magnitude of our educational activities, it is understandable that it should be a major social question. National concern about this problem is bound to increase; especially so in congressional committees and in public discussions pertaining to the allocation of federal funds to education (Schultz, 1972A:2;24).

Advances in research technology and quantitative methodology are considered positive, necessary developments by most social scientists, not only in understanding social reality but for influencing public policy with the intention of changing society. It is not only Marxists who state that the object is to change society, not just to understand it.³

Schultz (1972A:67;23-24) provides us with some insights into this mentality of change within the modern liberal perspective.

The research opportunities are now neatly arranged in two lots. One is large with many different policy issues on display, and the other contains a few puzzles. In deciding how much to pay for any of these offerings, it is the better part of wisdom to make sure that the economist who would be saddled with it is equipped to do it. Obsolescence has not been gentle in its treatment of economists, as is clear from the recent changes in the state of economics. Some of these advances have already been considered, mainly in connection with the various classes of policy issues. What remains to be done, therefore, is to consider some of the additional properties of these advances in analyzing the role of human capital in the economic system and to present some of the hypotheses that have emerged.

Despite the progress in dealing with social questions, it still is true that the hallmark of economic analysis is economic efficiency. It stands for rigorous analytical workmanship when it is not encumbered by the problem of equity, i.e., the distribution of personal income. But it is seldom that the realization of additional economic efficiency is neutral in its effects upon the distribution of personal income. As yet, the tradeoff choices between efficiency and equity are not clearly established, and when they become known the choice that matters depends on the values that determine social preferences. What is often overlooked, however, is the fact that there are policy choices where additional economic efficiency will also contribute to the social goal of reducing the inequality in the distribution of personal income. This is frequently true in the area of human capital.

Human capital, like material capital, is a complex heterogeneous aggregate. It is conceptualized and quantified in terms of <u>quality</u> of labor. This aggregate human capital can be broken down into separate factors, all of which require investment in order to develop the human capital. Schultz (1971B:8) classifies these sets of factors as follows: formal schooling and higher education, on-the-job training (specific and general), migration (labor mobility), health (physical and mental), economic information, personal distri-/ bution of income, international trade, international movement of skilled persons, the allocation of resources in the production of educational services, the effects of discrimination upon motivation in school performance, the treatment of "educated labor" under the production function, family planning, earnings foregone as part of the theory of allocation of time, and research.

Schultz emphasizes the inadequacies in classical and neo-classical capital theory which conceptualizes material capital as homogeneous.

Since the early 1940's, Schultz has been actively seeking an explanation for the sources of the gains in productivity in the United States. He became critical of the acceptance and use of homogeneous material capital in economic growth models, and resurrected Irving Fisher's idea of an all-inclusive concept of capital which conceived of capital as being vastly heterogeneous including both human and non-human form. To underscore the importance of this expanded concept of capital, Schultz (1971B:vii) remarks that "(i)f it were possible analytically to aggregate all of the different forms of human capital, it would exceed by a wide margin all non-human capital."

Schultz and his followers have been primarily concerned with the effects of education on human capital formation, since there appears to be adequate investment in health and labor mobility (migration) in the United States. On-the-job training has been researched principally by Jacob Mincer and this has been incorporated into the Schultzian model of human capital development. Canadian and Western European human capital economists have, likewise, concentrated their research efforts on the relationship between education and human capital formation (and correlatively between education and economic growth). Emphasis on education as an important investment for human capital formation in Third World countries is prevalent, but Schultz warns that other factors in human capital development which are taken for granted in developed countries, such as health and labor mobility, are not adequately developed. Thus, underdeveloped countries require investment in many factors of human capital. Emphasis on education. without the development of these other factors, is therefore a great mistake. However, this thesis focuses on the factor of education.

a 173

Attention is therefore drawn to the limitation of this focus alone when discussing underdeveloped countries.

Schultz (1971B: vii) states that "(t)he thrust of my studies has been primarily to clarify the investment processes and opportunities that provide the incentive to invest in human capital." He is concerned with influencing economic thinking so as to include human capital, with influencing governments to develop economically rational and efficient budgets and policies, as well as influencing individuals in making economically rational and efficient decisions in the development of their own human capital. These three elements run through most of Schultz's work. He calls for an investment approach to tackle problems of both national (macro-) economic growth and individual (micro-) economic growth. Although he feels that the investment approach and micro-economic analysis can provide the tools for theoretical and empirical advances in human capital theory, Schultz does not focus solely on the micro-level. It would be a great mistake to assume that Schultz's work is primarily micro-analytic, as he is equally concerned with macro-economic growth and public policy. Schultz's (Ibid.: 4-5) own explanation of his investment approach is contained in the following quotation.

1

An investment approach, I am convinced, is required in thinking about economic growth. In this approach the stock of capital is augmented by investment, and the productive services of the additional capital increase income, which is the essence of economic growth. It is a major step toward a general theory when all investment resources are encompassed and allocated in accordance with the meaningful economic standard established by the relative rates of return to alternative investment opportunities. Thus, in theory, this approach is grounded on an all-inclusive concept of investment and an ac-

counting of all additional investments gives a complete and consistent explanation of the marginal changes in the stock of capital, of the marginal changes in the productive services from capital, and of the marginal changes in income and, accordingly, of growth...

It is my contention that economic thinking has neglected two classes of investment that are of critical importance under modern circumstances. They are investment in man and in research, both private and public.

Quantification of the investment in education and of the return on that investment is simpler on the individual (private) level than it is on the national (societal) level. Thus, Schultz thinks that it is more productive to start with micro-economic analysis and estimate the private rates of return to investment in education and then try to develop a better measure of the social rates of return than the present one.

Furthermore, not only are the social rates of return based upon and related to the private rates of return, but the private rates of return can be utilized by policy-makers "...to induce individuals to adapt their own behavior pattern to the socially desirable one" (Hanoch, 1967A:329). Therefore, the prevailing focus on private rates of return to investments in education by human capital (especially American) economists during the past fifteen years should not be viewed as neglect or lack of interest in the social rates of return nor in public policy formation. Hanoch (<u>Ibid</u>.) argues that data from private rates-of-return analysis serve as a necessary, albeit modest, basis for a more complete analysis of as well as social decisions about education. Hence, it is clear that the individual private rates and the associated private demand functions for schooling are essential data for the evaluation and determination of social decisions. The present study should thus be construed as a modest basis for a more complete analysis of income-determination and of decisions about education, from both the private and the social points of view.

• >

n

Human capital theory has greatly influenced the perceptions of individuals towards education and socio-economic mobility which has in turn affected their decisions for increased educational participation. It has likewise influenced educators and politicians as reflected in the significant changes in educational policies. Basically, it is the relationship between education, occupation, and labor income on the individual (micro-) level and the relationship between education, occupation, and national income on the national (macro-) level as posited by human capital theory, as well as the technical-function theory of educational change, which provided the impetus for changes in individual decision-making and public policy. It is the examination of this relationship as formulated in human capital theory and the technical-function theory of education which will become the primary subject of critical analysis in this thesis.

Changes in the structure of public systems of education (including such attributes of it as accessibility and transferability), the administration of those systems, the financing of them, as well as the internal curricula changes will be discussed. The discussion is intended to be broad and to include the United States, Western Europe, and the Third World as well as Canada.

Although it is important to distinguish between the <u>actual</u> theoretical/empirical analysis developed by human capital economists

and its use as an evaluative tool for public policy formation and policy choices, it is also the case that this distinction often looses meaning in much of the human capital literature. This problem arises because of the importance Schultz and other human capital economists' put on actively influencing public policy. Thus, while admitting "...it still is true that the hallmark of economic analysis is economic efficiency", in searching for explanations of economic puzzles and paradoxes, Schultz (1972A:23) realizes that the effects of economic research on public policy are not neutral.

Since much of the economic research in the United States is conducted by non-profit agencies, it is directed towards the allocation of funds administered by those agencies and so is programoriented rather than policy-oriented --- and, therefore, is "...less than optimum in terms of social benefits" (Schultz, 1972A:22).

Schultz is strongly committed to the development of informed public policy which incorporates the latest theoretical and empirical advances in economics. Since he believes that "(h)uman capital research has important policy implications...", and "...it is not always clear how the new information derived from them [economic studies] can serve those who are making the policy decisions that determine the allocation of resources to the many forms of human capital" (Schultz, 1972A:1;5), Schultz (Ibid.:5) stresses the need for "...decoding and interpreting this information with a view of making it meaningful in arriving at policy decisions." There is also the added dimension of there being both a short-term and a long-term view in developing and utilizing new information which affects private and public decision-making.

Social policy, however, is not concerned solely with economic efficiency. Public policy is developed by politicians to meet perceived social, political, and psychological needs, as well as economic ones. Furthermore, as in education, there seem to be many social goals, some of which are mutually contradictory, and there is often also a lack of consensus on both the priorities and the methods to achieve them. Values underlying these social goals also affect the range and type of options considered. The ranking of social goals in terms of priorities is also based upon value-orientation. This is a political problem and is exasperated by budgetary constraints.

Economic and sociological research itself is premised on values. Hence, the use and effect of social science research in public policy formation is inherently value-laden. This often presents problems in terms of socio-economic policy choices, since such choices often are not neutral in their effects and often conflict with other social goals.

Schultz (1972A:60) describes some of these problems and his hope for some progress through the contribution of advances in economic theory and empirical methodologies.

> Meanwhile, endeavors to develop useful social goals for the purpose of determining national policy are reaching an impasse for lack of a meaningful common denominator for evaluating and rating the various social goals. Social theory provides no acceptable standard for determining the relative value to society of alternative social goals. The work by the panel and staff of the Department of Health, Education, and Welfare that entered into the preparation of Toward a Social Report identifies a wide array of social goals: better health and less illness, opportunity for social mobility, improvement in the physical environment, reduction of poverty, less crime and

> >

more safety, advances in the sciences, arts, and learning, and still others. But social theory and measurement do not suffice to rate the value of a given achievement with respect to any one of these social goals relative to that of the others. It is my contention that the extensions of economics as a consequence of the recent developments in economic theory and empirical analysis already referred to make it possible to bring some of these social goals into our national accounting and thus develop a more all-inclusive GNP.

The main dilemna faced in economic policy choices is in the problems faced by questions of efficiency and equity. By equity, Schultz and other human capital economists refer to the distribution of personal income. It is much easier to establish agreement on the specific efficiency of a particular program or of the general efficiency of a particular program on the entire economy than it is to evaluate the effect that choice has upon equity. Schultz (1972A:23) remarks that

(a)s yet, the tradeoff choices between efficiency and equity are not clearly established, and when they become known the choice that matters depends on the values that determine social preferences. What is overlooked, however, is the fact that there are policy choices where additional economic efficiency will also contribute to the social goal of reducing the inequality in the distribution of personal income. This is frequently true in the area of human capital.

Schultz (1972A:5) makes it quite clear that "(h)uman capital is strictly an economic concept" and that "(a)lthough it pertains to particular attributes of man, it is not intended to serve those who are engaged in analyzing psychological, social, or cultural behavior."

This conceptual stipulation sets definite, but perhaps undue, limitations on the meaning and applicability of human capital research and even more so in regards to the development of individual and

public decision-making since it is very difficult, if not impossible, to separate the economic from the political, sociological, and psychological dimensions.

Furthermore, Schultz (1972A:9) points out that

(t)he core of economic analysis rests on the assumption that preferences are given and that it is the function of economic activities to serve these preferences as best they can with the human and nonhuman capital at the disposal of the economy to obtain income streams and by transforming a part of this income by means of investment into additional forms of capital.

Schultz (Ibid.) contends that

...physical, biological, psychological, and cultural [attributes] ...account for both the social values that determine preferences and the economic value of the producer and consumer services that a people render, whether they come to them as earnings or directly as personal satisfactions.

However, the values people hold and the structure of social institutions are in dynamic relationship with each other and with technological change and economic growth. Change in any one element necessarily affects the others, though there often are lags in adjustment to such changes. Schultz (1971B:7) is aware of this, but points out that much of the literature in economics "...is naive with respect to the manner in which the economy adjusts and people gain (lose) as workers and consumers."

One of Schultz's goals in developing human capital theory is to "...bring technical changes into the corpus of economic theory" by transforming it into <u>capital</u> for the purposes of economic analysis and "...to treat much of scientific research, and also a large part of education and other skill-producing activities, as 'industries' producing new forms of capital that are more efficient than particular old forms of capital" (Schultz, 1971B:19-20).

Thus, any entities (including technological techniques) that contribute to production have an economic value and "...for purposes of economic analysis,...a technique is no more or less than a unit of capital,... a set of techniques representing a technology is a capital structure, and...a technical change is an alteration of a capital structure" (Schultz, 1971B:20). Therefore, "...the longstanding endeavor to distinguish between techniques and capital goes by the board" (Ibid.).

In addition to including technological change and acquired human skills and knowledge in the capital category along with material capital, Schultz (1971B:253) differs from most classical and contemporary economists in that he includes institutions "...as variables within the economic domain, variables that respond to the dynamics of economic growth." Thus, education as an institution is brought into the realm of economic analysis as a factor in the explanation of capital accumulation and economic growth. Schools provide a public service. Supply and demand analysis can be utilized to determine the economic value of that service (Ibid.:254).

Consequently, part of what Schultz (1971B:255) is attempting to explain with human capital theory are "...those changes in institutions that occur in response to the dynamics of economic growth" since "...the process of growth alters the demand for the service and... this alteration in the demand brings about a disequilibrium between the demand and supply measure in terms of longer run costs and returns." Although Schultz (1971B:255) realizes that the supply of the service of a particular institution (such as education) can be altered independently of economic growth dynamics (and that this cannot be explained by the theory), such a change in supply will result in economic effects which can be explained by the theory.

Human capital theory attempts, in part, to account for the ""institutional stresses and strains" produced by economic growth as well as the institutional lags. "Since we can specify and identify these institutional lags, we can also analyze the benefits in terms of efficiency and welfare that could result from reducing these lags" (Schultz, 1971B:262). Here, then, are some of the normative elements of the human capital analysis which link this analysis to that of the contribution of education to social mobility and economic growth --the latter being of central concern to functionalist sociologists as well as human capital economists.

The main cause of institutional stresses and strains in Western industrialized capitalist societies stems from advances in the production process and the interrelated changes in the occupational structure which human capital economists and technological functionalists consider to be inherent in "...our type of economic growth" (Schultz, 1971B:163). This is, in fact, the common ground shared by economists and sociologists, as well as by educators and politicians, during the era when functionalism enjoyed unchallenged domination in the social sciences -- during the 1950's and early 1960's.

Human capital economists refer to the sudden high demand for highly educated, skilled workers witnessed after World War II and the concurrent increase in the value of such services (manifested in

rising salaries and wages). Sociologists writing in the technological functionalist vein focused on the increasing differentiation, specialization, and hierarchization of the labor force in an occupational structure which was undergoing fundamental changes which they credited to advances in technology.

The concept of "post-industrial" society which developed in sociology provides the common Weltanschauung for human capital economics, as well as for the technical-function theory of education -- despite the fact that economists, sociologists, and educators ask different questions and have different focuses. Convergence of views among the three groups is reflected in their shared belief that 1) knowledge and skills required for job adequacy in modern industrialized countries are increasing due to continuous technological advances; 2) the proportion of occupations requiring more knowledge and skills is rising significantly with an accompanying decrease in the proportion of those occupations requiring little knowledge and skill; and 3) increased amounts of formal schooling are required because 4) educational requirements for most occupations are constantly increasing to keep up with technological change (Collins, 1971A: 1004).

Questions asked by human capital economists and which are part of this thesis stem originally from

(t)he growth problem, in terms of economic decisions, [and] requires an investment approach to determine the allocation of investment resources in accordance with the priorities set by the relative rates of return on alternative investment opportunities. It is applicable not only to private decisions but also to public decisions guided by economic planning (Schultz, 1971B:195).

These questions deal with education as a public institution which produces and distributes a form of human capital (knowledge and skills) which economists measure in terms of differential earnings. The following questions, then, are important in this thesis: 1) What are the private and social rates of return to investment in different' levels of schooling? 2) How reliable a measure are private and social rates of return to investment in education for individual decisionmaking and social policy? 3) "What are the effects of schooling and higher education upon the distribution of personal income?" (Schultz, 1972A:24). 4) What is the relationship between the educational system and the occupational structure? 5) What is the relationship between the educational system and 'the labor market (and specifically, between education, occupation, and labor income)? 5a) Do schools teach particular knowledge and technical skills required for job adequacy? 5b) Are workers with more years of schooling (of even general education) better able to adjust to changes in skills and knowledge required on the job than those with less years of schooling? Why? 5c) What is the relationship between productivity and earnings? 6) What is the importance of quality versus quantity of education in determining efficient and equitable allocation of resources?

These questions are all concerned in one way or another with assessing the <u>value of education</u> to the individual and/or to society. From that common ground, there is a divergence between the goal of economic efficiency and social equity (reducing the inequality in the distribution of personal income). Schultz (1972A:24) asserts that human capital theory has much to contribute to a better understanding of investment alternatives and efficient allocation of resources; and

that, even more importantly, human capital theory is useful in determining "...policy choices where additional economic efficiency will also contribute to the social goal of reducing the inequality in the distribution of personal income."

Chapter II provides a general discussion of human capital theory as it pertains to education as well as a more detailed discussion of the macro- and micro-level approaches used to determine the relationship between education, occupation, and labor income.

Chapter III discusses the influence that human capital theory has had on educational policies and the organizational structure of systems of public education at all levels of education in Canada, the United States, Western Europe, and the underdeveloped countries of the "Third World". A brief description of the influence human capital theory has had on research is also included.

Chapter IV outlines the technical-function theory of education and how the relationship between education, occupation, and labor income is viewed from the sociological perspective of technological functionalism. The concepts of equality of educational opportunity and of meritocracy which constitute a part of this perspective are also discussed.

Chapter V deals with the influence which the concept of equality of educational opportunity has had on educational policies and organizational structure of systems of public education in Canada, the United States, Western Europe, and underdeveloped countries. A brief discussion of the influence of this concept on research is also included.

In Chapter VI, a critique of human capital theory and of the technical-function theory of education is presented. These theories are critiqued on both theoretical and methodological grounds. Chapter VII presents conclusions drawn and offers possible alternative explanations for the relationship between education, occupation, and labor income -- and, more importantly, on the function of schooling in Western capitalist societies.

r

J

22

ú
CHAPTER I

Footnotes

¹Karabel and Halsey (1977B:3) remark that

(i)t would hardly seem an exaggeration to say that researchers in the field, armed with the dual weapons of structural functionalist theory and scientific method, envisaged few problems that would ultimately prove intractable.

²When Schultz (1959A:110) remarked that "...the main#tream of modern economics has by-passed undertaking any systematic analysis of human wealth", he is referring to the period when Alfred Marshall's ideas dominated economics from the late 19th century to the early 1940's. (This came about by misinterpretation and misunderstanding of Marshall's ideas on human capital as shall be discussed later.) Thus, Schultz uses the word modern to refer to relatively current times. Kiker (1966A:488) states that "...modern economics began with Sir William Petty ... " (in the 17th century). It is important to understand Schultz's use of the word "modern" because many economists have considered human beings and/or their skills as capital and recognized the importance of investing in such capital (human beings) in order to increase their productivity. Considering human skill as capital was part of the English Classical School and was "...adapted by economists interested in the distribution of income and the theory of production" (Kiker, op. cit.: 496). Kiker (Ibid.:482) reports that "(o)ne of the first attempts to estimate the money value of a human being was made around 1691 by Sir William Petty." Other well-known modern (both past and current) economists who consider human skill as capital and quantified it in monetary terms include: Adam Smith (1776); Jean Baptiste Say (1821); William Farr (1853); Theodor Wittstein (1867); J.R. McCulloch (1870); Friedrich Kapp (1870); Johann H. von Thunen (1875); William Roscher (1878); Sir Robert Giffen (1880); Henry D. Macleod (1881); Ernst Engel (1883); Alfred Marshall (1890); J. Shield Nicholson (1891); Irving Fisher (1897+); Alfred de Foville (1900); Henry Sidgwick (1901); Miles M. Dawson (1904); A. Barriol (1908); John Stuart Mill (1909); Yves Guyot (1914); S.S. Huebner (1914); Edgar Crammond (1915); Harold Boag (1916); Ernest Bogart (1919); William S. Rossiter (1919); Edward A. Woods and Clarence B. Metzger (1927); Friedrich List (1928); Louis Dublin and Alfred Lotka (1930); John M. Clark (1931); J.R. Walsh (1935); Nassau Senior (1939); Walter Bagehot (1953); Leon Walras (1954). The underlined dates refer to those authors who are well-known in the history of economic thought. See Kiker (1966A) for a full discussion.

Kiker is somewhat critical of the ignorance Schultz and other current human capital economists have about the history of human capital research in economics. Kiker's (1966A:497) essay was written as a reference source "(s) ince many of them [current human capital economists]

۰**۳۱**.

fail to cite predecessors..." and to point out that

(t)he human-capital concept was also used by past writers to demonstrate the power of a nation, propose new tax schemes, determine the total cost of war, emphasize the economic significance of human life, and aid courts in making decisions in cases dealing with compensation for personal injury and death. These uses may suggest interesting additional problems to contemporary economists.

Cohn (1972B:22-31) attempts to clarify both the wideapread misunderstanding of Alfred Marshall's views on human capital and the current "...allegations raised by Schultz and others about the neglect of human capital analysis by early economists..." which have caused a great deal of confusion in current human capital literature. Cohn (Ibid.:23) points out, as already illustrated, that there was "...extensive recognition of the importance of human capital among early economists." Cohn (Ibid.) relates this to Schultz's assertion that human capital has been neglected by economists in the past by stating that "...the bulk of their presentation was preoccupied with material wealth and material capital." Furthermore, "...it appears that current ideas in the economics of education are not novel nor are they new; rather, it is the refinement of tools, data, and concepts -as well as the intensification of study in the area of human resources -that has so revolutionized modern economic thought" (Ibid.:31).

Schultz (1959Ab:111-112) presents his view of how and why Alfred Marshall and his followers cast human capital theory to the distant margins of economics in the early years of the 20th century and the effect this has had on "mainstream" economics for nearly fifty years.

Another and more compelling reason why economists have neglected human wealth has. arisen from a conventional restriction on the concept of capital. Instead of developing and using an all-inclusive concept of capital, economists have restricted their definition of capital to include only those classes of wealth that are commonly bought and sold in the market place. Irving Fisher, in a series of papers published just before the turn of the century and then in his excellent but grossly neglected book, The Nature of Capital and Income, clearly and cogently presented an all-inclusive concept of capital. But the prestige of Alfred Marshall was too great; his ideas on this issue prevailed as his students and many followers entered upon the stage. Marshall dismissed Fisher's comprehensive concept of capital in these words: "Regarded

from an abstract and mathematical point of view, his position is incontestable But he seems to take too little account of the necessity for keeping realistic discussions in touch with the language of the market-place." Marshall concluded his appendix on "Definitions of Capital" by again making it clear that "...we are seeking a definition that will keep realistic economics in touch with the market-place..."

Free men are not for sale, and, thus, Marshall's market-place concept of capital had the effect of excluding all capital that becomes an integral part of a people. Granted that for particular purposes it is appropriate to restrict the concept of capital that one uses and granted also that the particular restriction imposed by the notion of the "market place" is altogether too narrow for many of the purposes to which I am addressing myself, Marshall's view of capital, nevertheless, could have been given a much broader interpretation than it has received. Obviously, labor is in touch with the market place as wages and salaries are presently determined. These wages and salaries represent income streams, and, like the income streams from property, they too may be discounted, and appropriate capital values may be imputed to each of them. Furthermore, investments in man, like investments in property, are ways of establishing additional income streams. Inasmuch as it is possible to observe their rates of return, these investments in man would appear to fall within the scope of Marshall's realistic economics. It is of course far from easy to estimate these rates from such statistics as are now being collected; useful statistics, however, could be had if we were to set our hand to the task.

I would be remiss if I did not acknowledge the fact that, in spite of the restriction that Marshall placed on his concept of capital, his writings are studded with gems of wisdom on the economic importance of adequate food, housing, and health and of developing the skills of workers. He, also, stressed the value of scientific training and the diffusion of scientific knowledge and looked upon education as a "national investment" (See Marshall, Book IV:216-217).

³There is, of course, a fundamental difference between how Marxists and liberals use the word "change". The change in society which Marxist envision is a radical, total restructuring of society; whereas liberals conceptualize change as evolutionary, piece-meal, and reformist.

CHAPTER II

HUMAN CAPITAL THEORY: THE RELATIONSHIP BETWEEN EDUCATION, OCCUPATION, AND LABOR INCOME FROM THE PERSPECTIVE OF ECONOMICS

A. Introduction

The purpose of this chapter is 1) to trace the development of human capital theory; and 2) to identify its basic assumptions (including the nature of the relationship between the education of workers and the occupational structure of modern industrialized capitalist societies). The influence of this theory on educational policies in Canada as well as the United States, Western Europe, and the Third World will also be discussed, but in chapter III. Finally, in chapter VI, a comprehensive critique of human capital theory will expose some of its major theoretical and empirical shortcomings.

Human capital theory was launched into international limelight at the seventy-third annual meeting of the American Economic Association in St. Louis by Theodore W. Schultz in his presidential address on December 28, 1960. It is a theory formulated and promoted primarily by neo-classical economists to explain the relationship between education, occupation, and labor income on the individual (micro-) level; and between education and economic growth/development on the societal (macro-) level. Thus, Schultz has drawn together two very different areas of neo-classical economics in his formulation of human capital theory. One strand is that of micro-economic investment theory which focuses on individual decision-making. The other is macro-economics, focusing on economic theories of growth.

Although one of the original aims of human capital theory was to provide a "rational" framework for policy decisions regarding

efficient investment in the American educational system (which has expanded at a phenomenal rate since World War II), 'it achieved almost immediate acceptance and hegemony throughout the world as the model which international agencies, governments, and universities would use in plaining and implementing national and international programs for social and economic development.

In a nutshell, human capital theory states that increased investment in human capital increases individual productivity and income and lays the technical base for the type of labor force necessary for economic growth in modern industrialized capitalist societies. In Schultz's (1963B:x-xi) words:

...people enhance their capabilities as producers and as consumers by investing in themselves and... schooling is the largest investment in human capital. This proposition implies that most of the economic capabilities of people are not given at birth or at the time when children enter upon their schooling... they these acquired capabilities ... alter the structure of wages and salaries and the amount of earnings from work relative to the amount of income from property.. There are long-standing puzzles about economic growth, changes in the structure of wages and salaries, and changes in the personal distribution of income that can be substantially resolved by taking account of investment in human capital.

Basic to the development of human capital theory is the concept of all-inclusive capital, discussed in depth by Harry G. Johnson in <u>The Residual Factor and Economic Growth</u> (OECD, 1964Bb). According to Schultz (1968A: 278).

(t)he logical basis of an all-inclusive concept of capital, which includes human capital, was established by Irving Fisher (1896, 1906). This concept treats all sources of income streams as forms of capital. These include not only such material forms as natural resources and reproducible producer and consumer goods and commodities but also such human forms as the inherited and acquired abilities of producers and consumers. ... 27

Fisher's definition of capital is "...simply any stock existing at a given instant that yields a stream of services over time, all flows of 'income' therefore being the product of some item of 'capital' whose value is calculated by capitalizing the income flow at an appropriate discount rate" (Blaug, 1970 b:5).

The concept of all-inclusive capital makes possible a generalized capital accumulation approach to economic development. The key idea here is that labor is treated as a produced means of production (Bowles and Gintis, 1975Ab:74). This means that skilled labor is really capital, i.e., capable of generating a stream of income. Human capital is <u>human</u>, according to Schultz (1971B:48) because "...it is embodied in man, and it is <u>capital</u> because it is a source of future satisfactions, or of future earnings, or of both."

The idea of education as an investment in man's productive capabilities, as capital, goes back at least to Adam Smith (1776, 1965B, Bk 1, chpt. 10, pt. 1), who stated that

(w)hen any expensive machine is erected, the extraordinary work to be performed by it before it is worn out, it must be expected, will replace the capital laid out upon it, with at least the ordinary profits. A man educated at the expense a of much labour and time to any of those employments which require extraordinary dexterity and skill, may be compared to one of those expensive machines. The work which he learns to perform, it must be expected, over and above the usual wages of common labour, will replace him the whole expense of his 12 education, with at least the ordinary profits of angequally valuable capital. It must do this too in a reasonable time, regard being had to the very Uncertain duration of human life, in the same manner as the more certain duration of the machine. The difference between the wages of skilled labour and those of common labour is founded upon this principle.

÷.

Therefore, according to human capital economists, only unskilled labor would remain in the analytical category "Labor". Skilled labor becomes part of the analytical category "Capital". However, because human capital theorists 'slieve that unskilled occupations are being eliminated from the occupational structure in advanced industrialized capitalist societies by the inevitable process of mechanization and automation, we are left with only skilled occupations. Hence, unskilled labor or "Labor" disappears and we are left with physical and human (skilled labor) "Capital". Thus, the distinction between labor and capital is eliminated and class is transcended.

Most economists view capital accumulation as the key requirement for economic growth. Human capital theory shifted the traditional emphasis on physical capital accumulation to a form of capital accumulation which had been relatively ignored during the first half of the 20th century -- human capital.

If Adam Smith recognized the importance of education as an investment in labor productivity as far back as 1776, why has the idea been more or less neglected until Schultz resurrected it with his formulation of human capital theory in 1960? Economic historians generally credit Alfred Marshall with the demise of the concept of human capital. In his <u>Principles of Economics</u> (1890, 1961B), Marshall agreed with Adam Smith that educated labor could be thought of as expensive machinery, but he said the concept of human capital is unrealistic and impractical, and that the acquired skills and knowledge of workers could not be included in the measurement of wealth or capital of a country (Blaug, 1970Bb:2). In response to Fisher's (1906B) call for an all-inclusive concept of capital, Marshall (1890; 1961B) responded

(r)egarded from the abstract and mathematical point of view, his position is incontestable. But he seems to take too little account of the necessity for keeping realistic discussions in touch with the language of the market-place... we are seeking a definition that will keep realistic economics in touch with the market place...

Thus, because Marshall felt that it would be "out of touch with the market-place"¹ to treat human beings as a form of capital,"(i)nvestment in human beings has accordingly seldom been incorporated in the formal core of economics... and has fostered the retention of the classical notion of labor as a capacity to do manual work requiring little knowledge and skill..." (Schultz, 1961Ac:3).

There are other reasons why economists have shied away from the idea of viewing human beings as a form of capital. The primary one is that modern economies lack capital markets for labor and, as Blaug (1970Bb:7) points out,

...that the abolition of slavery necessarily implies that human beings are not produced means of production in the full sense that capital goods are. After all, capital markets are simply markets which convert income into capital and capital into income, that is, convert the promise of a flow of future payments into a single advance payment and vice versa. Because non-slave society prohibits people from contracting to deliver their future services, "free" people must keep their human wealth tied up in the form of labour services and cannot hedge against unforeseen changes in the future demand for their services.

Schultz (1961Ac:2) points out some moral and philosophical objections to viewing human beings as wealth or capital:

Free men are first and foremost the end to be served by economic endeavor; they are not property or marketable assets...Our values and beliefs inhibit us from looking upon human beings as capital

that

goods, except in slavery, and this we abhor... Hence, to treat human beings as wealth that can be augmented by investment runs counter to deeply held values. It seems to reduce man once again to a mere material component, to something akin to property...J.S. Mill at one time insisted that the people of a country should not be looked upon as wealth because wealth existed only for the sake of people.

Schultz and other human capital theorists maintain a minority position in neo-classical economics with regard to their concept of capital. The overwhelming majority of neo-classical economists view capital as material capital only; inherited and acquired capabilities of people are not included in this view. On the other hand, Keynesian economists view education strictly as consumption.

Thus, national income accounting, which is an outgrowth of Keynesian macroeconomics, treats educational expenditures as final consumption, without making any allowance for the fact that the education acquired by the labor force depreciates over time and gradually becomes obsolete in much the same way as machines with which the labour force is equipped. That is to say, we inconsistently include in the net national product or national income the net additions to the stock of physical capital but not the gross additions to the stock of human capital, although both types of capital are used up in the process of contributing to output (Blaug, 1970Bb:18).

Keynesian economists do not view education as having any role to play in growth. Blaug (1970Bb:18) states that "(u)nfortunately, the Keynesian view of education as consumption logically inhibits any consideration of the contribution of education to economic growth." Consequently, human capital theorists assert that neo-classical economics, and Keynesian economics in particular, has overemphasized physical capital as the mainspring of economic growth (Ibid.). Human capital theory marks an advance in neo-classical economic theory for two other reasons. As Bowles and Gintis (1975Ab:74) point out, human capital theory "...rejects the simplistic assumption of homogeneous labor and centers attention on the differentiation of the labor force" as well as bringing "...basic social intitutions [such as schooling and the family], previously relegated to the purely cultural and superstructural realm, into the realm of economic analysis."

B. Macro-Level: National Growth/Development Approach

The growth of human capital theory literature can be traced back to the works of several economists, including: Abramovitz, 1956A; Becker, 1962A, 1964B, 1966A, 1967B; Bowman, 1962A, 1964A; Denison, 1962B, 1967B, 1974B; Fabricant, 1959; Kendrick, 1961; Mynt, 1965B; Solow, 1957; Schultz, 1958A, 1959A, 1960A, 1961A, 1962A, 1963B; Vaizey, 1958, 1961; and Bertram, 1966B. Some are empirical studies focusing on the microlevel, others focus on the macro-level. In a number of empirical studies, economists have investigated the sources of modern economic growth in the United States and grappled with the problem of measuring them. This is referred to as the national growth/development approach. It represents the macro-level strand of human capital theory.

Basically, the national growth/development approach is commended with accounting for all the sources of economic growth on a national scale. Consequently, empirical studies into the sources of past growth have been conducted and projections made on that basis for future probable growth rates. However, when calculations are made regarding the proportion of growth in GNP attributable to conventional inputs of capital and labor, a large residual is left unexplained. It was the

discovery of this residual which eventually led to the systematic focus on the role of differentiated labor skills (e.g., human capital) in economic growth.

It was Moses Abramovitz who announced the "residual" with the publication of <u>Resources and Output Trends in the United States Since</u> <u>1879</u> in 1956 (Schultz, 1972A:17). As Schultz (Ibid.) remarked,

> (t)here it was, [the residual] yet by all the canons of economics, it could not be accepted, except as a measure of our ignorance. Why was the rate of output increase so much higher than that of the inputs? What had been ignored? No wonder then that this residual became the takeoff for so much research to discover the sources of the unexplained increases in output.

This sort of research stimulated interest in the development of a comprehensive human capital theory in order to explain, among many other puzzling economic phenomena, this "residual" or this difference between the rate of increase in output and the rate of increase in input (physical capital and labor). The idea that improvement in the <u>quality</u> of inputs, especially labor, could be responsible for at least part of the increase in output prompted a flurry of research into the elements thought responsible -- namely, health and education.

The publication of Denison's work in 1962 ushered in a debate which is still raging in the literature. That debate centers on the "residual" and whether or not it can be explained. Edward F. Denison is generally credited with having made the first attempt to explain the residual (Schultz, 1971B:136).

Denison's study (1962B) is actually an implicit estimate of the aggregate production function of the American economy for the period 1910-1960 as well as an attempt to explain the residual (Blaug, 1970Bb:89). The model used is a linear aggregate production function of the Cobb-Douglas type and neutrality of technical progress is assumed. He uses the principle of marginal productivity (<u>Ibid</u>.:97). Denison also assumes that

(i)n the productive sphere, substitutions are possible between various categories of labor and the professional classes; and competition exists between employers so that the wage paid to an employee equals his marginal productivity. Entry into the idependent professions is free, and since they operate in a competitive society the renumeration of persons who are not wageearners also equals their marginal productivity (OECD, 1964Bb:62).

Other assumptions include labor market competition and substitution elasticities as well as constant returns to scale (Ibid.:69).

According to Denison (1962B), 23% of the growth in real national income in the United States between 192 and 1957 was due to increases in the length of formal education² of the labor force, measured in average years of schooling. Another 20% was due to increases in "technological and managerial knowledge", whereas the increase in inputs of physical capital accounted for about 15% of the increase in real national income for the same period (Schultz, 1971B:136). The remaining 8% was attributable to increasing returns to scale (Phelps, 1962A: 509). Denison concluded that improvements in skills of workers was due to increased education (Schultz, 1961Ac; Denison, 1965A). Except for increases of the labor force itself, this makes education the largest single source of growth in the United States for the period 1930-1960 (Blaug, 1970Bb:89). Denison divided his study into two parts, one for the period 1900-1929 and the other for the period 1930-1957. It should be noted that in his 1962 study, Denison included only formal schooling and measured it only quantitatively (days and

years spent in school). Various forms of adult education and on-the-job training were not measured or included, "...for lack of information. I am not even sure this increased or decreased, per worker, during the periods..." (Denison, 1962A:127).

Apparently, Denison attributed the residual to education rather arbitrarily on the basis of the fact that persons with higher levels of education received higher incomes. Denison assumed that higher wages reflect greater productivity (Collins, 1971A:1005). For example, Denison writes that "...earnings differences between groups of males of similar age, classified by education, are taken to represent differences in their contributions to production or quality" (Denison, 1962 #:125). Denison (1962B:68) states that

(r)eliance, for broad groups, on the marginal productivity explanation of the distribution of income permits us to treat differentials in average earnings among these groups as a measure of differentials in the average contribution to production made by the individuals comprising them.

Table 2.1 includes Tables 8 and 9 from Denisons's 1962 study, reporting his figures for the contribution of education to increased worker productivity and income. Denison (1962B:71) has also remarked that

(i)t is reasonable to suppose that increasing the number of days spent in school per year raises a man's contribution to production just as much as will an equal percentage increase in the number of years spent in school.

Denison, however, did not attribute all the income differential between mersons with different levels of education to the difference in amount of education. He attributed 2/5 of the income differentials between persons with more schooling to natural ability and family

Years of School Completed	(1) Mean Income as % of Mean Income of Eighth Grade Graduates	(2) Mean Income Differentials Used to Represent Effect of Education (% of Income of Eighth Grade Graduates)
None	50	70
Elementary School: 1 to 4 years 5 to 7 years 8 years	65 80 100	79 88 100
High School: 1 to 3 years 4 years	115 140	109 124
College: 1 to 3 years 4 years or more	165 235	139 181

Table 8

Source: Denison, 1962B:68.

Table 9

Calculation of the Effect of Longer Education on Labor Earnings Per Man'

			Per	r Cent Chan	ge		· · · · · ·	Annual Rate of Change (per cent)
			(1)	(2)	(3) Average	(4)	(5)	(6)
P	eric	ba	Lobor output per man consid- ering only years of education	Average number of years of school completed	number of days of school ncr year of school completed	Average total number of days of school attended	Labor output per man based on iotal days of education	Labor output per man based on total days of education
1920 1930 1940 1950	to to to to to	1920 1930 1940 1950 1950 1950 1950	2.7 3.3 4.1 4.4 • 5.1 4.7 • 48	9.0 8.9 10.2 10.4 9.8 8.8 8.2	6.7 8.8 10.8 10.9 9.3 8.1 7.0	16.3 18.4 92.0 22.4 20.0 17.8 15.7	4.9 6.9 8.8 10.4 10.3 9.4 9.1	0.48 0.67 0.85 1.00 0.99 0.90 0.83
1010 1930 1960	to	1960	8.1 14.2 9.7	18.6 33.6 17.6	16.1 34.2 15.6	37.8 79.3 36.0	12.1 32.6 19.4	0.57 0.94 0.89
1910 1910	to to	1960 1980	21.2 32.9	58.5 88.5	55.8 80.1	147.0 236.0	48. 6 77.4	0.79 0.82

1 Based on males 25 years of age or older.

Source: Denison, 1962B:72.

£5 a background. Therefore, 3/5 of the income differentials between workers of similar age was the result of differences in level of schooling. In Denison's (1962B:68) own words,

> ... the difference in education received and the associated loss of experience are not the only characteristics that distinguish the groups of workers of similar age so that the differences in average earnings cannot be used without adjustment to measure earnings differences that are due to differences in education and associated loss of experience -- the information that our investigation requires.

Clearly, the reported income differentials overstate these differences. Individuals of greater ability are more likely to continue their education. Wolfle states that most of those who do not enter high school come from the lower half of the ability distribution. The average score on the Army General Classification Test of those who enter high school is 105, of those who graduate from high school 110, fof those who enter college 115, of those who graduate from college 121, and of those who receive the Ph.D. degree 130; however, there is a wide dispersion at each level. Among individuals with similar AGCT scores, those with better school grades are more likely to continue their schooling. Insofar as this may reflect grea 🌑 energy, application or motivation that carries over into later life, the income differentials shown reflect these attributes as well as differences in length of education as such.

Since Denison first reported that approximately one-third of the gross earnings differentials between college and high school graduates disappear when one standardizes for differences in father's occupation, rank in high school class, and I.Q. scores, Becker (1964B:80-88) and Weisbrod and Karpoff (1968) have independently confirmed these results (Blaug, 1970Bb:50).

Then in 1964, OECD published Denison's essay, "Measuring the Contribution of Education (and the Residual) to Economic Growth" (OECD,

1964Bb). The focus of this paper is upon what factors have been the sources of past growth in the United States. It is an outgrowth of his earlier study (1962B). Denison distinguishes between the contribution of increases in factor inputs and increases in output per unit of input. His results are virtually the same as in the earlier study.

My conclusion...is that, from 1929 to 1957, the increase in education raised the average quality of the labor force at an average annual rate of 0.93 per cent a year. What this rate implies is that an increase of 80 per cent in the average amount of schooling raised the average quality of labor by 30 per cent (OECD, 1964Bb:16).

Denison (OECD, 1964Bb:36) explains the contribution of education to economic growth in the following quotation:

It is important to understand why improvements in the quality of the labor force is estimated to have made such a large contribution to the growth rate. Essentially it is the combination of two facts. First, improving the quality of labor, education affected a productive resource that, on the average from 1929 to 1957, represented 73 per cent of total input of all factors of production (labor, land, capital, and entrepreneurship, or whatever classification one prefers). This is referred from the labor share of national income. Second, the increase in the amount of education was very large. The amount of education per worker (in days) was increasing almost 2 per cent a year and I estimate that this was raising the quality of labor by an amount equivalent to an increase in its quantity of almost 1 per cent a year.

It is these two facts which explain why the increase in education per worker is estimated to have increased real national income per person employed by 0.67 percentage points while, for example, the increase in the quantity of capital per person employed contributed only 0.15 percentage points. Capital represented (at most) only 22.5 per cent of total factor inputs, and the input of capital per person employed increased only 0.55 per cent a year. The main specific assumption underlying my estimates of the contribution of additional education to growth through its effect on the quality of labor is, as stressed, that differentials in labor earnings due to differences in education equal 60 per cent of observed differentials in money income among adult males of the same age classified by years of education. The effect of any alternative assumption can be easily approximated by multiplying my results by the ratio of the desired percentage to 60 per cent.

Thus, Denison is the first economist to measure the economic value of education in respect to its contribution to national economic growth. The measure used is actually earnings differentials attributable to education. Denison is the only economist, prior to 1970 at least, to assign a cardinal number to the contribution of education to national income (Blaug, 1970Bb:89).

Denison hypothesized that education could contribute to economic growth in two distinct ways. Firstly, increasing the educational level of workers would raise the <u>quality</u> of the labor force (inclusive of all occupations), and

(t)his may be presumed to increase labor productivity, independently of any tendency for a large number of educated people to speed the enlargement of the society's stock of knowledge relevant to production

and

(s)econd(ly), an upgrading of the educational background of the population may accelerate the rate at which society's stock of knowledge itself advances (OECD, 1964Bb:22).

Denison takes great pain to distinguish conceptually and operationally between increases in the education of the labor force and "advances in knowledge".

It is essential to distinguish between: (a) society's stock of knowledge relevant to production, which (along with other conditions) governs the output obtained with given inputs; and (b) the quantity and quality of inputs (including the education of the labor force) that govern the output obtained with a given stock of knowledge...Increases in subput resulting from advances in society's stock of knowledge are credited in my classification to the growth source, "advances of knowledge" (Denison, 1967B:79).

Denison's category of "advances in knowledge" includes advances in technological knowledge as well as managerial and organizational knowledge. He assumes that advances in technological knowledge are are about equally as important as advances in managerial and organizational knowledge. In contrast, other economists tend to stress technological knowledge overomanagerial and organizational knowledge. Denison includes both of these categories of knowledge in one measure, which he labels "advances in knowledge" (OECD, 1964Bb:54).

In 1967, Denison published another major work along the same lines as the 1962 study of the American economy. It was a comparison of growth rates of nine Western European and North American countries. The study uses a framework very similar to that employed in the 1962 one. Denison compared and attempted to explain the differences in growth rates of the following countries: Belgium, Denmark, France, Germany, Italy, the Netherlands, Norway, the United Kingdom, and the United States. The time period used is 1950 to 1962, broken into two sub-periods at 1955.

The results of the 1967 study are not as clear-cut as of the previous one (1962). In Denison's (1967B:78) own words, "(e)ducation does not help to explain why growth rates in Europe (1950 to 1962) have been higher than in the United States, but on the contrary, adds to 40

CA:

the difference that must be explained by other sources" (emphasis added). This conclusion is rather curious since Denison expected to get similar results to his American study. He did, however, find that per person employed, "...education does help to explain the differences between Europe and the United States in the 1960 levels of national income..." (Ibid.).

Among the results of Denison's 1967 study of these nine Western countries was the fact that between 1950 and 1962, education "...contributed one-half a percentage point to the growth rate of national income" in the United States and "...raised the average quality of the American labor force by 9 percent (or 0.7 per cent a year)..." Denison (1967B:78). However, in Northwest Europe, the contribution of education to improvement in the quality of the labor force was only half as great as in the United States, "...and the increase in actual amount of education was proportionately even less" (Ibid.).

The contribution of education to national growth varied widely from country to country. For instance, education was found to contribute 13 per cent of the annual growth rate in the United Kingdom but only 1.4 per cent in Germany. Despite this, Britain's growth rate has been slow relative to the rest of Europe.

Differences in the quality of full-time schooling between the countries was not taken into account because it would be difficult, if not impossible, to measure. Denison concluded that other factors were probably more important than education in explaining growth rates in Europe. One probable factor is the "...national differences in the intensity of work..." (Blaug, 1970Bb:98).

Then, in 1974, Denison published another study, <u>Accounting for</u> <u>United States Economic Growth, 1929-1969</u>. This was motivated, in part, by a desire to explain the rapidly escalating growth rate experienced in the United States during the 1960's. Denison reports that his estimating procedures were greatly improved by better statistical implementation of the approach used in his earlier studies and that he had access to greater frequency of more detailed data every year from 1947 to 1969 (Denison, 1974B:2). He also changed the base year for prices from 1954 to 1958. Denison's (Ibid.:vii) conclusions included

changes in seven categories were chiefly responsible for long-term growth and for variations in the growth rate: the number of employed persons and their demographic composition; working hours, including the proportion of part-time workers; the education of employed persons; the size of the capital stock; the state of knowledge; the proportion of labor allocated to inefficient uees; the size of markets; and the strength and pattern of short-term demand pressures. Advances in knowledge were the biggest single source of growth. As in Denison's previous studies, lengthier education of the labor force appears as a major source of growth in the United States, especially in the growth of output per worker. (Emphasis added.)

The 1974 study bears out the basic thesis of the earlier studies conducted by Denison, that is that education of workers is still one of the most, if not the most, important source of growth in the United States. Increasing human capital on a national level in the form of increasing years of school attendance has a pay-off to the nation (in the form of increased growth rates) and to the individual (in the form of higher monetary wage income).

Extending the school leaving age and actively encouraging people to pursue more and more education is viewed by Denison, and human capital

theorists in general, as something desirable. Thus, policy formation and influencing policy implementation is an important component of their work. (Denison spent about half of his 1962 study discussing possible strategies for increasing the United States' growth rate in the future.) Results of the 1974 study are reported in Table 2.2, which reports Denison's Tables S-1, S-2, and S-3 comparing the results of the 1974 study with the results of the 1967 study of nine Western countries and the results of the 1962 study. Of interest to us is the change in the percentage point contribution of education to national 🏞 growth. There is some difference between the three studies, the greatest being the decline from 0.67 reported in the 1962 study to the 0.40 reported in the 1974 study. Both figures refer to the period 1929-1957 in the United States. The difference between the contribution of education to the growth rate in percentage points reported in the 1967 study (0.49) and that of the 1974 study (0.40) is insignificant. This, however, refers to the 1950-1962 period. Contribution of Education to Growth Rates in the U.S., 1929-1957

(in percentage points)

 1974 Study
 1962 Study

 0.40
 0.67

Contribution of Education to Growth Rates in the U.S., 1950-1962 (in percentage points)

 1974 Study
 1967 Study

 0.42
 0.49

Source: Denison, 1974B: 344, 345, 346.

There is actually very little difference in these figures cause, as Denison (1974B:547) points out, the drop is a result 43

の一日でなるのである

TABLE 2.2: SOURCES OF GROWTH IN THE U.S., 1929-57 and 1950-62

Table S-1. Sources of Growth of Total Actual National Income, 1950-62 and 1929-57 Contributions to growth rate in percentage points

•

	1950-62	1929-57
National Income,	3.39	3.02
Total factor leput	1.85	1.68
athor	I.II	1.32
Binpioyment	5	•
Averate hours		
Interarous shift offset	2.0	
Are-sex composition	10.01	0.01
tion	0.42	04
Unallocated	0.02	0.11
anital	0.74	91.0
Laventories	0.10	
Nonreidential surretures and		
equipment	0.31	0.13
Dwellings	0.21	0.14
International assets	0.03	0.0
and.	0.0	0.0
whet per unit of laput	1.54	1.34
Advances in knowledge and n.e.c.	1.15	0.80
manowed treasure allocation	0.32	01.0
	0.27	0.28
Nonfarm self-employment	0.02	0.02
wellings occupancy ratio	-0.02	0.01
Economics of scale	0.38	0.33
rregular factors	-0.29	-0.10
Weather in farming	10.0	-0.01
2.	83	0.0
Intendity of occurate	-0.27	60.0
Source: Calculated in the same way at	Table 9.4. c	polymas 1, 2,
a. not. Net elsevhern classified.		
Source: Denison, 1974B: 344	: 344 .	.1
	• • • • • •	

Table S-2. Present and Previous Estimates of Total Actual National Income, 1950-62 of Sources of Growth

From Coatributions to growth rate in percentage points

	· · · ·	From	Why Growth
riady D al larcome and a start of the start of a start of a start of the start of		present	Rates
rel larcone 3.39 actor inpute 1.41 doyment 1.43 doyment 0.19 r 0.19 r 0.19 r 0.10 lissifion 0.02 land dwellings occupancy 0.72 nonces 0.02 nonces 0.02 dio 0.10 retidential atructures and 0.00 ber unit of tapute 0.00 bertion of international trade 0.00 inter in farming -0.20 ther in farming -0.20 the		study	Differ
actor inpute actor inpute a hymean a hymean	National Income	3.39	3.32
rs factors and receiver and receiver and receiver and dwellings and dwellings accupancy and accupancy ratio 0.03 -	Total factor input ^a	1.83	1.95
ploymeat 0.89 urs composition 0.41 ker accomposition 0.01 discrition 0.02 Jaallocated 0.02 al and dwellings occupancy 0.02 entories 0.10 entories 0.00 entories 0.00 there allo companey ratio 0.10 companey ratio 0.10 companey ratio 0.10 companey ratio 0.10 companey ratio 0.10 benefings and occupaney ratio 0.00 there allo companey ratio 0.00 at per unit of lapoute 1.156 m 0.21 m 0.21 m 0.22 benefication 0.22 m 0.23 m 0.23 m 0.23 allo factors 0.00 barrier 0 for allo companey 1.15 m 0.23 allo factors 0.00 allo factors 0.00 al	Labor	1.11	1.12
us -0.19 us 0.41 fuscrition 0.03 Jaulocated 0.03 Jaulocated 0.03 Jaulocated 0.03 al and dwellings occupancy 0.10 attice 0.10 benches 0.26 benches 0.28 benches 0.28 benches 0.28 benches 0.03 benches 1.15 wed resource allocation 0.32 m 0.21 m 0.21 bencion of international trade 0.03 bencion of scale 0.00 bencion 0.00 benc	Employment	61.0	0.90
Algorition -0.03 -0.03 Jaallocated 0.02 -0.03 Jaallocated 0.02 -0.03 al and dwellings occupancy 0.10 -0.25 atio 0.10 0.10 atio 0.10 -0.25 atio 0.10 0.26 entories 0.10 0.26 entories 0.10 0.26 entories 0.10 0.26 vellings and occupancy ratio 0.26 Vellings 0.00 0.26 Vellings 0.00 0.26 Vellings 0.00 0.00 tratitional assets 0.00 there all occution 0.31 wed resource allocation 0.32 m 0.33 m 0.33 fuertion 0.33 m 0.33 mretion 0.33 mretion 0.33 barriers 0.00 of disputes 0.00 of disputes 0.00	Hours Char	61.0-	-0.17
Albestilion 0.02 Jaallocated 0.02 al and dwellings occupancy 0.10 atio 0.11 culptored 0.26 breakings 0.26 breakings 0.02 breakings 0.05 breakings 0.05 breakings 0.05 breakings 0.05 breakings 0.05 breakings 0.11 breakings 0.32 breakings 0.33 breakings 0.33 m 0.33 m 0.33 breakings 0.00 breakings 0.00 breakings 0.02 breakin	American commention	4 8	0.39
Jaallocated 0.02 al and dwellings occupancy 0.72 atio dwellings occupancy 0.10 arcsidomal atructures and 0.11 creational atructures and 0.11 chilings and occupancy ratio 0.26 bwellings and occupancy ratio 0.26 bwellings and occupancy ratio 0.26 bwellings and occupancy ratio 0.28 bwellings and occupancy ratio 0.28 bwellings and occupancy 1.15 weed resource allocation 0.32 moved resource allocation 0.32 moved resource allocation 0.32 moved resource allocation 0.33 moved resource allocation 0.3	Education	0.42	0.49
al and dwellings occupancy atio entories contrest entories contrest contrest contrest contrest contrest contrest breupancy ratio contrest breupancy ratio contrest breupancy ratio contrest breupancy ratio contrest breupancy ratio contrest	Unallocated	0.02	8.0
atio atio entories entories contrational atructures and contrologian contrational atructures and contrologian breakings breakings breakings creational assets breakings breakings contrologian assets contrologian	Capital and dwellings occupancy		
actiones actiones actional atructures and culporent atructures and culporent 0.26 bwellings becupancy ratio becupancy ratio becupancy ratio becupancy 1.56 0.00 a per unit of laput ⁶ 1.15 0.00 a per unit of laput ⁶ 1.15 0.00 0.00 0.32 0.00 0.32 0.00 0.33 0.00 0.33 0.00 0.33 0.000000	ratio	0.72	0.83
equipment cultings and occupancy ratio wellings becupancy ratio crastional assets becupancy atto crastional assets become allocation m m m m m berriers berrian berriers beriers berrie	Inventories Noorselduntial atministration	0.10	0.10
clinits and occupancy ratio 0.26 Dwellings and occupancy ratio 0.28 Decupancy ratio 0.03 Ernational assets 0.00 A per unit of lapust 1.15 more in knowledge and n.e.c. ^b 1.15 wed resource allocation 0.32 more in knowledge and n.e.c. ^b 1.15 nord fraction 0.32 more discriminational trade 0.00 barriers 0 acale 0.38 Mar factors -0.22 alter in farming -0.22 and in farming -0.22 more disputes -0.22	routeneur automet and couloment	11.0	0.41
Dwellings 0.28 Decupancy ratio 0.02 frantional assets 0.03 frantional assets 0.05 frantional assets 0.03 match 0.15 match 0.13 fraction 0.13 fraction 0.13 barriers 0.03 anter in farming 0.02 anter in farming 0.02 anter in farming 0.02 anter in farming 0.00		0.26	0.25
Accupancy ratio renational assets 0.05 at per unit of lapust 0.00 there in knowledge and n.c.c. ^b 1.15 wed resource allocation 0.32 m 0.27 marin self-employment 0.05 inclion of international trade 0.00 barriers 0 acale 0.38 anter in farming -0.22 ather in far	Dwellings	0.28	- The second sec
at per unit of lapust the per unit of lapust modes in knowledge and n.c.c. ^b 1.15 wed resource allocation m m m m farm self-employment 0.27 literion of international trade barriers 0.00 barriers 0.02 0.03 0.03 1.15 0.32 0.03 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 0.32 1.15 1.15 0.32 1.15 0.32 1.15 0.00 0.32 1.15 0.00 0.32 1.15 0.00 0.32 1.15 0.00 0.32 1.15 0.000 0.0000 0.000 0.000 0.00000 0.0000 0.00000 0.0000 0.0000 0.0000 0.0000 0.000000 0.00000 0.000000		55 - -	4 4 6 6
at per unit of lapust ^a 0.00 at per unit of lapust ^a 1.15 modes in knowledge and n.c.c. ^b 1.15 wed resource allocation 0.32 m marm self-employment 0.05 luction of international trade 0.00 barriers of scale 0.00 miles of scale 0.02 alter in farming -0.22 alter in farming -0.22 alter of the farmin		5	3.
af laquets 1.56 wiedge and n.e.c. ^b 1.15 eraployment 0.12 international trade 0.00 ale 0.38 ale -0.29 mand -0.02	Tang	0.0	8 .0
wiedge and a.e.c. ^b 1.15 we allocation 0.32 employment 0.27 international trade 0.00 ale 0.38 ale 0.02 ale -0.02 mand -0.27 -0.02	Output per unit of input ^a	1.56	1.37
te allocation 0.32 ° cmployment 0.05 0.27 cmployment 0.05 0.05 0.00 ala 0.00 ala 0.00 ala 0.00 -0.29 - ming -0.29 - 0.27 - 0.20 - 0.20 - 0.20 - 0.27	Advances in knowledge and n.e.c. ^b	1.15	0.76
employment international trade 0.05 0.05 0.05 0.08 0.38 -0.29 ming -0.27 -0.27 -0.27 -0.27 -0.27 -0.27 -0.02 -0.02 -0.02 -0.02 -0.02 -0.02 -0.02 -0.02 -0.03 -0.05 -0.	Improved resource allocation	0.32	0.29
employment 0.05 international trade 0.00 ala 0.38 -0.29 - ming -0.02 b -0.27 -	Farm	0.27	0.25
0.00 16 0.38 -0.29 - -0.02 -0.02 -0.02 -0.00	Nonfarm self-employment Reduction of international trade	0.05	5 0
ale 0.38 0.29 muing0.02 0.27	barriers	0.00	0.00
-0.29 - -0.02 - 6 -0.02 -	Economies of scale	0.38	0.36
- 0.02 - 0.00 - 0.27	Irregular factors	-0.29	-0.0
-0.27	Weather in farming	-0.02	0.00
	Labor disputes Intensity of demand	-0.27	82

ars. Not avaltable. A.s. Not avaltable. A. For comparability, "dwettings occupancy ratio" is classified in total factor input in fits table. b. a.e.c. Not elsewhere classified.

Source: Denison, 1974B: 345.

Table S-3. Present and Previous Estimates of Sources of Growth

of Total Actual National Income, 1929-57 Contributions to growth rate in ------

National income Total factor input Labor Emforment	From	
fational income otal factor input abor Emulorment	N esent study	Sources of Economic Growth
ohal factor input abor Emmiorment	3.02	2.93
abor Emniorment	1.69	2.00
Employment	1 17	1 23
	8	8
nour Are-ser comosition	10.1	-0.20
1000 moderno	0.40	-0.01
Increased experience and better utilization of women workers Unatiocated	0.0 8.1	
Capital	0.36	U A1
Inventories Nonresidential structures and	0.0	0.08
and a	0.13	0.28*
Dwelliags International actes	1.0	500
		0.0
Output per unit of input	1.34	Q.23
dvances in knowledge and n.e.c.b	0.80	0.59
Restrictions against optimum use		
of resources	n .n	-0.07
Improved resource allocation Parm	0.30	0.07
Nonfarm self-employment	0.02	0.0
Dwellings occupancy ratio	0.01	A.A.
Economics of scale	0.33	0.34
Irregular factors	-0.10	0.0
W calleer in Jarming Labor ditoutes	5.0	88
Intensity of demand	88.0	88.0
Sources: Table S-1 and The Sources of E	Eronomic Gr	Growth, p. 266
n.a. Not available. a. finckudet farm dwelkings.		

at 0.01. at the inclused waste of labor in agriculture" and "laduatry abilit from agriculture."

Source: Denison, 1974B: 346.

"...mainly from a reduction in the allowance for the number of days attended per school year and a reduction in the estimate of bias in Census data."

Another change of interest to us is the significant increase in the estimate for the contribution of advances in knowledge and n.e.c. (not elsewhere classified) reported in the 1974 study. It rose to 0.80 percentage points from the 0.59 reported in the 1962 study for the period 1929-1957. For the 1950-1962 period, the estimate rose from 0.76 in the 1967 study to 1.15 in the 1974 study. Denison reports that errors which were discovered in the 1967 study resulted in an underestimation of the estimates for advances in knowledge and n.e.c. (which, in any case was obtained as a residual in both studies). Therefore, the 1974 study revised this particular estimate upwards 0.23 percentage points "...for estimating the effects of fluctuations in the intensity of demand, 0.12 points by the revised data for nonresidential structures and equipment, and 0.07 points by the upward revision of the growth rate itself" (Denison, 1974B:345).

In fact, the total labor contribution to national economic growth in the United States for the 1929-1957 period was dropped 0.25 percentage points from 1.57 percentage points reported in the 1962 study to 1.32 percentage points reported in the 1974 study. This particular drop seems to be primarily a result of the reduction of the education estimate; a more minor component of the reduction is elimination of an estimate for "increased experience and better utilization of women workers" which proved to be of trivial size (Denison, 1974B:347).

For the 1950-1962 period, however, the estimates reported for total labor contribution to growth in the United States were nearly identical in the 1967 and the 1974 studies (1.12 and 1.11 respectively).

Since the mid-1960's, several economists have applied Denison's model to other national studies (Bertram, 1966-Canada; Lithwick, et. al., 1967-Canada; Bowles, 1969-Greece; Selowsky, 1967-Mexico, Chile, India; Psacharopoulos, 1969-Hawaii; Williamson, 1969-Philippines; Bergson, 1968-U.S.S.R.)(See Blaug, 1970Bb:98).

Gordon W. Bertram (1966B), in a study prepared for the Economic Council of Canada, estimated the contribution which education made to economic development (growth) in Canada between 1911 and 1961. For comparison purposes, he calculated estimates for the period 1929-1957 and compared them with Denison's (1962B) estimates for the American economy for the same period.

91

During the 1929-1957 period, the estimates for Canada are substantially lower than those Denison reported for the United States. Bertram's estimates indicate that education made a much smaller contribution to economic growth and worker productivity in Canada than it did in the United States for this period. The reason for this, according to Bertram, is because Canada's investment in education was lower than the American investment in education.

Bertram used Denison's model for the study. Bertram's study, like Denison's, was confined to male workers, but Bertram only included workers aged 25-64 years old, whereas Denison included all male workers 25 years old and older. Education is measured solely in terms of years of schooling, that is purely quantitatively. Adult education and various forms of on-the-job training were not included. Bertram employed Denison's assumption that only 3/5 of the income differentials between workers of the same age category is due to differences in education.

The remaining 2/5 is due to natural ability and family background.

Bertram (1966B:61-62) reports that

(a) ccumulating evidence and analysis point more and more to education as a pervasive and basic element contributing to the real earnings potential of people, and therefore also of a whole economy or society. Past improvements in education have raised this potential considerably by increasing the quality and skills of the labour force. Estimates have shown that better education appears to have raised labour earnings per man by about 30 per cent from 1911 to 1961 in Canada, and that this has contributed almost one quarter to the rise in output growth per employed person.

Furthermore, the "(a)verage income per man in the male labour force 25-64 years of age rose some 12 per cent due to improved education" for the period (1911-1961) (Ibid.:50).

Bertram (1966B:55) measures labor productivity by worker income (wages and salaries), as indicated in the statement that

... labor productivity (income) rose by 0.52 per cent per year over this period (1911-1961) as a result of improved education. Consequently, total productivity per man rose by 0.40 per cent per annum as a result of improved education (76 per cent of 0.52). Educational improvements thus accounted for almost one quarter of the rise in productivity per person employed from 1911-1961.³

For the period 1929-1957, improvements in education contributed some 20 per cent to productivity per employed person in Canada — which is less than half the corresponding figure (42 per cent) for the United States reported by Denison (Ibid.).

The difference between the estimates of the contribution of education to growth for Canada and the United States are significant. The fact that Denison and Bertram place great importance on education as a source (factor responsible for) growth further emphasizes the 47

"backwardness" of Canada in respect to education of the labor force.

The calculation of the contribution of education to the growth rate of total national income [in Canada] involves a method similar to that used in the...contribution of education to the growth in output per employed person. Improvements in the quality of labour input are considered equivalent to an increase in the quantity of labour input and the rate of growth in real national income in Canada for the period 1911-1961 and 1929-1957 are again approximated by the rate of growth in GNP (Bertram, 1966B:55).

Bertram (1966B:56) found the contribution of education as accounting for 12 per cent of the growth in national income for the period 1911-1961, whereas it accounted for 11.4 per cent for the period 1929-1957. The contribution of education to growth of national income for the period 1929-1957 was estimated at 23 per cent by Denison for the United States -- approximately double the Canadian figure (Ibid.).

Another significant difference between Canada and the United States is the fact that per capita income in Canada "...has remained persistently about one-quarter below that of the United States since the turn of the century" (Bertram, 1966B:57). According to Bertram (<u>Ibid</u>.) this is due to "...differences in quality and quantity of inputs and the efficiency with which they are used in the Canadian economy."

If the gap in real income per member of the labour force between Canada and the United States is taken as 20 per cent in 1961, then the estimates made in this study concerning the relative differences in the educational attainment or quality of the labour force in the two countries indicate that the quality difference is a source of about a third of the 20 per cent difference (Bertram, 1966B:58).

Universal elementary education was strongly promoted in Canada from the late nineteenth century through to the First World War,

(b)ut the record of the Canadian educational achievement seems to taper off by 1920. It appears as though the democratic spirit which had promoted universal elementary education in Canada failed to operate to the same extent in secondary education efforts, and education beyond the elementary school continued to be more the prerogative of an elite. The high schools were excellent, but their output was small. With a limited production of high school students, university education was even less available to large numbers who were potential university students.

Thus, despite the fact that there has been a significant long-term rise in the educational attainments of the labour force, the extent of this rise has been considerably below that for the United States since the First World War. This has resulted in the emergence of a widening educational gap between the two countries... For example, by the beginning of the 1960's, the median year of schooling of the Canadian male labour force was about two years below that of the United States male labour force, at least for the age group 25-64 which constitutes the bulk of the existing labour force. This appears to reflect, in particular, the widening of the gap at the secondary and university level. In relation to the United States, Canada began to lose ground in its efforts to prepare high school students in the inter-war years, and university students in both the inter-war and the post-war years (Bertram, 1966B:61).

Bertram (1966B:61) remarked that "(e)ven under very favourable assumptions, it would require many years to eliminate the differences in the average levels of education between Canada and the United States." He ascribes about one-third of the income differences per man between the United States and Canada during the early 1960's to be due to lower Canadian educational attainments. The fact that the per cent increase in labor earnings per man for the 1911-1961 period in Canada is only one-half the increase in the United States "...reflect(s) the widening gap in the average level of schooling in terms of economic growth", according to Bertram (Ibid.:62).

Bertram (1966B) refers to a comparison of social and private rates of return to high school and university education in Canada (based on Podoluk's work) and in the United States (based on Becker's work). Becker (1964B:121) estimated the social rate of return on university education in the United States to be between 8 and 11 per cent per year, which falls within the range of rates of return on all business capital. For the private rate of return to university education in the United States, Becker (<u>Ibid</u>.:114) calculated a 12+ rate of return per year (adjusted for differential ability). Bertram (<u>op</u>. <u>cit</u>.:63) reports that

(c)alculations for Canada have shown that private returns on the human investment in high school and university education are in the range of 15 to 20 per cent per year, with slightly higher rates for an investment in a university education than in a high school education (Podoluk, 1965B:61)... Canadian rates of return to education compare favourably, not only with those on alternative returns typically accruing from total capital investment in physical and financial assets, but also with the similar U.S. rates of return to adducation.4

Bertram, like Denison, is interested in influencing government policy. In that vein, Bertram (1966B:64) remarks that "...not only... [is] education a significant factor in raising productivity and living standards, but also that a relative increase in expenditures on education would contribute to an efficient allocation of resources." More bluntly,

(t)he major policy implication is that increasing efforts in the area of education are a prerequisite not only for the maintenance or acceleration of productivity growth in Canada, but also for the narrowing of the existing differences in the absolute level of productivity, and therefore the living standard, between Canada and the U.S. (Ibid.).

C. Micro-Level: Investment Approach

Despite the results of empirical studies on the macro-economic level, Schultz himself feels very strongly that macro-economic theory has failed to provide adequate answers to many long-standing economic puzzles and paradoxes. Therefore, Schultz called for what he calls "an investment approach" based on micro-economics, and investment theory in particular, to resolve some of these problems and to provide theoretical and empirical advances. This investment approach would link the macro- to the micro-levels and provide a theoretical and empirical rbasis for pragmatic and rational decision-making and policy formation for investment choices on the individual as well as on the national levels. While the national growth/development approach, sometimes referred to as the national income approach, operates on the macroeconomic level; the investment approach operates on the micro-economic level. According to Schultz (1961Ac), the investment approach, although its focus is on individual rates of return on investment in education, also enables one to estimate the social benefits of education by aggregation of data on individuals. Schultz believes that not only could this be done, but that it should be done.

Schultz is concerned with such things as individual decisionmaking and private rates of return to investment in education. In expressing the need for an investment approach to human capital formation (in addition to the national growth/development approach developed by Denison), Schultz (1968A:283) remarked that

(w)hile this knowledge of the sources of economic growth (provided by Denison) is indeed useful..., it is not an approach to determining the underlying costs and returns to investment

that produced the additional sources that account for this part of economic growth. The important matter of an optimum allocation of total net savings among investment opportunities is not a part of the aim of this national growth/development approach.

Schultz praised those economists who were already doing some work in this area and attempted to incorporate this branch of microeconomics (e.g., investment theory) into the corpus of human capital theory.

I find it intuitively plausible that advances mining to this part will come icro-analysis, mainly, in for example Telser,'s specific human capital and h by firms in his search for the dependents of the differences in the rates of Peturn in manufacturing. Thinking in terms of the activities of the household, it may prove especially rewarding in coping with human capital formation by the family to approach it as a part of the production activities of the household, and also, in getting at the satisfactions that it renders to the family in consumption. The differences in the motivation of students in their school work associated with the differences in jobmarket discrimination, following the approach of Welch, is another case in point (Schultz, 1970Ab: 305-306).

Schultz (1970Ab:297) credits Gary S. Becker with the development of the theoretical analysis of investment in human capital (in Becker, 1962A). According to Schultz (1970Ab:305), Becker's work constitutes

a major breakthrough:

(c)onsider the original theoretical analysis of of investment in human capital by Becker. I think it is fair to say that he started with the sim of estimating the returns to college and high school education in the United States. In pursuing this sim, he discovered that the investment activities associated with education were akin to other investments in people and that all these activities had basic attributes in common for which received theory, tailored to investment in structures and equipment, required reformulation. Then, later in pursuing the many implications of earnings foregone, he discovered the problem that could be solved by a theory of the allocation of time.

Gary S. Becker set out to develop a theory of income distribution which would "...both articulate well with general economic theory and... would be useful in explaining differences among regions, countries and time periods" (Becker, 1966A:358). Becker believes that an analysis of investment in human capital would and does provide a sis for such a theory. As he points out himself, the distribution of personal income has long been of interest to economists, including Adam Smith; yet despite this historical interest and the plethora of empirical information which has accumulated, economists for the past generation have neglected the study of the distribution of personal income (Ibid.).

Becker set out in the early 1960's to provide a theory on the distribution of income which was based on an analysis of investment in human capital. In Becker's (1966A: 369) own words.

(t)he body of economic analysis desperately needed a reliable theory of the distribution of income...our approach seems to offer considerable promise of filling that need. In any case we hope to have demonstrated that a theory of income distribution need not be a patchwork of Pareto distributions, ability vectors, and the toss of a coin, but can wear clothing as neat as that worn by the theory of households and firms.

Becker's theory states that "(t)he total earnings of any person after he has finished investing in human capital can be said to equal the sum of the returns on his investments and the earnings from his 'original' human capital" (Becker, 1966A: 359). Like Denison, Becker measures investment in human capital solely by number of years of formal schooling. And also like Denison, Becker does not consider all the differential in earnings between workers of the same age category to be due to differences in education alone.

Each person is assumed in effect to maximize his economic welfare by investing an appropriate amount in human capital, and the distribution of earnings is determined by the distribution of investments and their rates of return. These determinants are in turn related to various "institutional" factors which also play an important part in our theory: inheritance of property income, equality of opportunity, distribution of abilities, subsidies to education, and other human capital, etc. (Becker, 1966A: 368).

Like the macro-level strand of human capital theory, the microlevel (investment approach) has as one of its basic assumptions the neo-classical principle of marginal productivity. This means that differences in earnings of workers (generally in the same age category); are assumed to reflect differences in their productivity. Becker utilizes this principle in his work; so does Schultz. "For human agents it is fairly straightforward to the extent that there is a linkage between what they contribute to production and what they earn in wages and salaries..." (Schultz, 1968Aa:283).

Human capital theory as Schultz formulates it is posted partly on the neo-classical idea that investment should flow into those sectors of the economy which have a high rate of return (because high rates of return indicate under-investment). Both individuals and society would benefit by such an investment. Conversely, a sector which yields low rates of return would be a sector which has been over-invested in. It would be considered wasteful for governments, as well as for cor-

ര

porate groups and individuals, to invest in a sector which yields a low rate of return. It would also be considered wasteful not to invest in a sector which yields a high rate of return. The ultimate aim of neo-classical investment economics is to equalize the rates of return in all sectors of the economy; presumabley, this would eliminate under- and over-investment in any one sector of the economy and maximize economic efficiency and equilibrium.

Schultz and other human capital economists advocate investment in human capital because it yields high rates of return to both the individual and society. "...(T)he quality of human effort can be greatly improved and its productivity enhanced...such investment in human capital accounts for most of the impressive rise in the real earnings per worker" (Schultz, 1971B:25). Schultz (<u>Ibid</u>.:35) states that "(i)nsofar as expenditures to enhance such capabilities also increase the vafue of productivity of human effort [labor] they will yield a positive rate of return."

The method employed by the investment approach is that of rateof-return analysis. Rate-of-return analysis for investment in education distinguishes between two types of returns: private (or individual) rate of feturn and social rate of return. Individuals tend to be concerned with the former, whereas governments are moth concerned with the latter.

The private rate of return to education⁵ is the financial return in the form of increased monetary earnings which the individual gains as a result of investment in more education. The costs, both direct and indirect, to the individual and/or her/his family constitute the investment. The subsequent increase in earnings as a result of 55 H

acquiring the additional education minus the investment costs constitute the return on the investment. 56

It is important to point out that the private rate of return to education in economic terms (monetary earnings) is not the same as the <u>real</u> or <u>full</u> private mate of return. Becker (1964B:121-122)

(f)ull or real returns and costs⁰ would be the sum of monetary and psychic ones, and the real gain would depend on the relation between these real returns and costs. The psychic gain from college, like the monetary gain, probably differs considerably between the typical college and high school graduate. For presumably the former does and the latter does not go to college partly because of a difference in expected psychic gains...

Quantities ve estimates of psychic gains are neveralizectly available and are usually computed Terridually as the difference between independent estimates of monetary and real gains. Unfortunately, independent estimates of the real gains to college graduates are not available.

in

There are some strong conceptual and statistical difficulties

...estimating the private rate of return on education from income differentials between persons differing in education...Simply worded, the argument is that the true rate of return on education is grossly overestimated because persons differing in education also differ in many characteristics⁷ that cause their incomes to differ systematically (Becker, 1964B:79).

This leads us to a discussion of how private returns to education are measured and calculated. There are divergent approaches to measuring the private returns. Basically, the difference between the approaches binges on whether or not it is the "...measuring of lifetime incomes by levels of education..." or "the "...rates of return on investment in education" which is being calculated (Fodoluk, 1965B:54-65).

Ċ.

Schultz initially used the measure of lifetime earnings by levels of education approach. So did Herman Miller (1960A), whose data were used by Schultz. This approach has been abandoned by Schultz in favor of the other approach of estimating present value of income streams by discounting them at various interest rates. The switch was made because of a number of difficulties with the earlier approach.⁸ 57

 \bigcirc

Because of the lifetime earnings by levels of education approach to the return on education has been largely abandoned, we shall focus on the private rate-of-return analysis as formulated by Gary S. Becker (1964 B: 37) who developed the theory and technique of deriving the rate of return; the smount invested, and the investment period from het earnings.

Becker (1962A:49) states that "...the total amount invested in a generalized concept of human capital and its rate of return can be estimated from information on earnings alone." Therefore, Becker's theoretical and empirical work in human capital deals with <u>all</u> activities and characteristics which have the effect of raising earnings., Our focus, however, is on odly one of these factors -- education. According to most human capital theorists, education is the single most important investment factor (at least in developed countries). Furthermore, what is really being referred to by the term education is formal schooling. Becker utilizes the general economic belief that the incentive for investment is the expected rate of return, and that this is true whether one is investing in physical (material) capital or in human capital (people). Becker (19624:37), shows his own work as contributing to general economic theory in that "...an investment approach to human resources is a powerful and simple tool capable of explaining a wide range of phenomena, including much that has either been ignored or given ad hoc interpretations."

Belief that individuals do in fact maximize their investment opportunities in a similar fashion as individual firms is an inherent element of the investment approach to human capital, as it is to micro-economic analyses in general. "...(T)he amount invested in human capital results from optimizing behavior: each person results approach in effect to invest an amount that maximizes his economic welfare" (Becker, 1966A:359).

Rational behavior, however, is not entirely necessary to come out with the same results. Becker (1976B:153-168) "...postulates impulsive or random behavior...and shows that, betause of the constraint of scarcity, negatively inclined demand curves result from these assumptions as well as from rational behavior" (MacRae, Jr., 1978A:1255).

Thus, Becker has developed a micro-economic model to explain⁹ the profits on investment in human capital as measured by the rate of return individuals earn on

...a unit of human capital (given by the slope of its demand curve) adjusted for the "rate of interest" on investment opportunities (given the slope of its supply curve). In principle, every individual has a unique demand-supply function thich accounts for observed differences in earnings and levels of human capital accumulation (Sawyer, 1978A:1266). Individuals with lower supply curves or higher demand curves or both invest more than others. It is important also to understand that Becker \$1966A:368) sees his approach as fitting into the mainstream

of general neo-classical economic theory, unlike other approaches to
the distribution of personal income; so that in addition to developing an investment approach to human capital formation, he has developed a theory of the distribution of income.

> ... (U)nlike most other approaches to income distribution, it does not consist mainly of mechanical curve fitting or ad hoc probability mechanisms, but rather relies fundamentally on maximizing behavior, the basic assumption of general economic theory. Each person is assumed in effect to maximize his economic welfare by investing an appropriate amount in human capital, and the distribution of earnings is determined by the distribution of investments and their rates of return. These determinants are in turn related to various "institutional" factors which also play an important part in our theory: inheritance of property income, equality of opportunity, distribution of abilities, subsidies to education, and other human capital, etc

Furthermore, Becker (1966A: 369) complains that "(t)he body of analysis desperately needs a reliable theory of the distribution of income."

Focusing on individual investment in education, Becker (1964B) calculates the private rate of return to college gradingtion "...as the internal rate, or rate of discount, that makes the gries of differences between college and high school graduates sum to zero, where the costs of college (including foregone earnings) gives rise to negative differences" (Rees, 1965A:959).

Becker treats the costs of schooling as negative income during periods of school attendance.¹⁰ The internal rate of return¹¹ is "that discount rate which yields a present value of zero for the net income streams derived from the additional education" (Podoluk, 1965B:60). Thus, the earnings differentials for college graduates are between those males with college graduation and those males who have completed

Name of Street Street, South

high school. For male high school graduates, the earnings differentials are between the male high school graduates and males having completed five to eight years of elementary education.

What, then, is the private rate of return to investment in education? Becker (1964B:76), as already noted, relies primarily on the internal rate of return.¹² The rate of return to an average college entrant is 10 to Trant cent per year; with a higher rate for urban, white, male college graduates, and a lower rate for non-whites, women, rural persons, and college drop-outs, and a variance within any particular subgroup according to ability (Ibid.).

The private rate of return for white, male college graduates (adjusted for ability) is "...more than 12 per cent..."¹³ (Becker, 1964B:114). The rate of return to investments in smaller corporate manufacturing firms is "...of the same general magnitude as that from college education...The average rate of return... at a little over 7 per cent¹⁴ (<u>Ibid</u>.:115).

Becker (1964B:112) points out that "...a dynamic competitive economy produces considerable variation in the gain from capital..." whether physical or human. Despite the difficulty in estimating the rate of return to college education due to known factor differences (such as race, age, aex, region, ability, urban/rural) and to the fact that the pay-off takes much longer than that for most other capital investments,¹⁵ a liberal college education yields a higher private rate of return than any other investment.

Realizing that the private rate of return to college education is lower than the real rate, such investment becomes more appealing than investment by Physical Capital. This is reflected in the data.

Becker (1964B:122) cites data that

...indicates that the gross investment in college education rose from about 2.5 per cent of that in physical capital in 1920 to about 8 per cent in 1949 to 12 per cent in 1956. Foregone earnings, which are a rough measure of private investment, rose no less rapidly. So the private real rate of return has apparently been higher on college education than on physical capital.¹⁶

Before concluding that it is more profitable for relatively young individuals to invest in college education than in tangible capital, attention must be given the fact that Becker's data was for white, urban males. Becker (1971A: 109) subsequently reveals that this group of college graduates constitutes only 45% of all college graduates. It is a very different story for the 55% of all college graduates who are rural, non-white, or female. Even though Becker made no estimates of returns to the latter groups of college graduates, the average returns to such groups are probably appreciably lower than to white, urban males. Consequently, Becker (<u>Ibid</u>.) sets the average private rate of return to 11 college graduates at less than 9%, even though a definite figure has not been calculated.

This casts an entirely different picture to the question regarding which investment yields the greatest private return, that in human capital or that in tangible capital. Becker (1971A:109-110) has refined his estimates of returns to tangible capital as well. "...(T)he average return on capital owned by business enterprises" is his latest conception of what the average return to college education should be compared (Ibid.). The 7% figure previously quoted refers to the rate of return to amaller corporate manufacturing firms (1938-1954) after corporate income taxes, in Since the return on college graduation is

computed before tax, the 12% (before-tax) return rather than the 7% should be used (Ibid.).¹⁷

After estimating the return to unincorporated business enterprises at 5 to 8 per cent, and the ratio of corporate to unincorporated capital at 6 to 4, Becker (1971A:110) comes up with the average return to <u>all</u> business capital at 8 per cent.

Therefore, there is very little difference in the average private rate of return to college education and to business capital (Becker, 1971A:110). The conclusion drawn by Becker is that there has not been an over-all under-investment in college education based on direct private returns.¹⁸

The very scanty data that are available for the first forty years of the 20th century indicates that in the United States there was a general decline in the absolute earnings differentials as well as the private rate of return to college graduates vis-a-vis high school graduates. The real costs of schooling rose substantially during the same period. However, Becker (1964B:131;134) cautions against firm conclusions being drawn because the information for the period prior to 1939 "...is extremely scanty and unreliable..." but that it is "...sufficiently important that much more attention should be paid to the historical evidence."

Becker found higher private rates of return on high school education than on college education. For the 1939 urban, native-white, male cohort of high school graduates, the best private rate of return is 16 per cent, unadjusted for ability. For the 1949 all white, male cohort of high school graduates, the private rate of return is 20 per cent, again unadjusted for ability. Thus, Becker puts the average private rate of return at 16 per cent, unadjusted for ability.¹⁹ This is several percentage points higher than the returns to college graduates.

Even though the adjusted private rate of return to high school graduates is probably much lower than the unadjusted rate, it "...is probably considerable" (Becker, 1964B:127). Becker (<u>Ibid</u>.) states that although high school education has been compulsory in the United States for many years, "(t)he very rapid secular growth in high school education in the United States...is probably ultimately more directly related to anticipated private and social real rates of return."

Between 1939 and 1958, the private rate of return²⁰ to elementary education has been higher than high school graduation, which in turn has been higher than the return to college graduation. However, one must keep in mind that even the lowest average private rate of return (that to college graduation) is about 12 per cent per year. That is about the same as the after-tax rate of return in industry and much higher than the rate of return to low risk, liquid capital investment.

The private rate of return to college graduation seclined about 1.5 percentage points from 1939 to 1949. Despite questionable statistical significance of the calculation, Becker (1964B: 129) points out that this "...is consistent with extensive evidence of a general narrowing of skill differentials during the 1940's..." Since 1956, there has been a slight increase in the private rate of return to college graduates, "...consistent with the slight general widening of skill differentials during the 1950's" (Ibid.). During the entire twenty-three year period under study by Becker (1939-1962), the private 63

1.1

rate of return to male college graduates changed very little.

In contrast, the private rate of return to high school graduates increased by 4 points from 1939-1949, and by 8 points after 1949, for a total of 12 points over the twenty-three years²¹ (Becker, 1964B:129). The changes in the rates since 1939, according to Becker (Ibid.) are a "...net result of several changes with different effects."²² Becker (Ibid.:131) thinks that, "(i)f adjusted rates behaved similarly to unadjusted ones, the rate of return from college did not change on balance and that from high school increased substantially after 1939." Becker's (Ibid.) apparent explanation for this phenomenon is rather curious:

Therefore, advances in technology and other forces increasing the demand for educated persons must have offset the increase in college graduates and more than offset the increase in high-school graduates. Consequently, technological advance and other changes apparently increased the demand for high-school graduates more than that for college graduates.

The traditional view in neo-classical economics regarding the secular increase in education, especially since World War II, is that earnings differentials and rates of return on education will decline due to increased supply. However, Becker's (1964B:156) work "...indicates that this rapid growth in the number of high-school and college graduates has not reduced their economic position." An alternative view has developed from the empirical and theoretical research done by Becker and others to the effect that increased educational attainments "...in good part adjust to, as well as influence, the demands of the economic system" (Ibid.). Becker (19648:156) also found that "...investment in education in fact steepens and increases the concavity of age-earnings profiles... and that earnings do not tend to drop when persons reach their late forties or fifties. Furthermore, Becker (Ibid.) states that

... the steepness of age-wealth profiles -the relation between age and the discounted value of subsequent earnings -- is also increased by investment in education and other human capital. It is suggested that the apparent large secular increase in the peak wealth age in the United States resulted from a secular increase in the amount invested in such capital.

Becker (1964B:157) did not deal with the returns to specialized programs of education in high school or college, nor with the possible differences in returns to bachelor degrees, master degrees, and doctorates. His data were undifferentiated aggregates in terms of the type of program of education. Differences in <u>quality</u> between high schools and between colleges were not adjusted for either, even though Becker recognizes that differences do exist and that they are important. This is because a workable technique for quantifying quality of education has yet to be developed.

Becker's (1964B) study deals only with the United States, though he thinks it is reasonable to assume that there are similar, at least positive, private returns to education in European as well as underdeveloped countries. Others have done work on the returns to education in Europe and the underdeveloped countries.

The social rate of return to investment in different levels of education is a much more contentious and tenuous thing to deal with than is the private rate of return. The true, total social rate of return is the sum of all the direct and indirect social costs subtracted

from the total social benefit fully measured in terms of national income growth). The true (real or full) social rate of return, like the real private rate of return, is practically impossible to quantify. Economists calculate the social rate of return by taking pre-tax earnings differentials by level of education as a percentage yield on the total resource costs of education (Blaug, 1970Bb:204). Thus, the social rate of return to education encompasses the costs and benefits to individuals and/or her/his family as well as to the entire society.

Social rates of return were calculated by including all costs and all benefits attributable to obtaining an education, both those accruing to the individual and/or his family and to society at large, whereas private rate-of-return estimates include only the costs and benefits to a student and/or his family (Rogers and Ruchlin, 1971B:171).

Since World War II, governments have been increasely concerned with eveloping socio-economic policies which will minute business cycle fluctuations and promote sustained economic growth. With the realization that increased education of the labor force has been responsible for a significant (nearly 25%) portion of the growth in GNP in the United States between 1929 and 1957, other countries have calculated the private and social rates of meturn to different levels of education in order to develop more effective policies. Bertram's study in Canada has already been discussed. Before discussing social rates of return to different levels of education in various countries, private rates of return will be compared for different countries.

Wilkinson (1966B) and Podoluk (1965B) estimated the private . net present values of lifetime earnings for different occupations and the private rate of return to education (pre-tax income) respectively

for males in Canada from 1961 census data. Podoluk (Ibid.:61) calculated a 16.3% private rate of return to completion of four to five years of secondary school, and 19.7% private rate of return to college graduation.²³ Thus, returns to college graduation are significantly higher than those to high school graduation in Canada -- the reverse of what was found in the United States. Podoluk (Ibid.:62-63) points out that

...there are significant differences in the age-earnings profiles of the various educational groups in the two countries. For example, in the age group 25 to 34 there are much greater relative differences between the average earnings of university graduates and the average earnings of those with secondary or elementary school diplomas in Canada than in the United States. A higher education is a scarcer attribute in Canada and as a vesult, younger entrants into the labor force with university degrees may have a greater ismediate salary advantage relative to high school graduates and older university graduates than is the case in the United States.

Podoluk (1965B) hypothesizes that there has been competition among employers in Canada for the limited supply of young university graduates in the labor market, raising salaries rapidly. The shapes of the age-earnings curves are also different between Canada and the United States. There are greater differences between earnings of younger and older age groups in the United States than in Canada, though age-earnings patterns are similar for high school graduates in both countries. Podoluk (Ibid.:64) explains part of the sharp differences in age-earnings profiles for college graduates in the United States as compared to Canada is due to the fact that a much higher proportion of the United States college graduates had post-sucondary training and/or graduate degress.

Psacharopoulos (1973B) reported rates of return to secondary and post-secondary education from studies conducted in thirty-two countries across Europe, Africa, Asia, and the Americas. The rates of return reported are unadjusted and are for completed levels of education. See Table 2.3 for data.

For Great Britain, the private rate of return to secondary education is 6.2% and 12% to college education. In Norway, the figures were 7.4% and 7.7% respectively. The Netherlands have 8.5% return to secondary education and 10.4% return to college education. Turkey has a private rate of return to secondary education of 24% and 26% to college graduation. New Zealand, in contrast, has a 20% return to secondary education and 14.7% return to college education. The Philippines have a private wate of return of 28% to secondary education and 12.5% to college education, whereas Ghans has a 17% private rate of return to secondary education and 37% rate of return to college education. Kenya has 30% private rate of return to secondary education and 27.4% private rate of return to college education.

These figures are given as illustrative examples of the great variance in the private rates of return to education at both the secondary and college levels as well as between developed and underdeveloped countries. Table 2.3, taken from Psacharopoulos (1973B:62), reports the private and social rates of return for thirty-two countries. Primary education generally has the highest private and social rates of return because it is free in most countries and the foregone earnings are "virtually zero". Many empirical studies indicate a general decline in the private and social rates of return to education as one progresses from elementary to secondary to college education.²⁴

TABLE 2.3: SOCIAL AND PRIVATE RATES OF RETURN BY EDUCATIONAL LEVEL AND COUNTRY

TABLE 4.1

2

SOCIAL AND PRIVATE RATES OF RETURN BY EDUCATIONAL LEVEL AND COUNTRY (per cent)

		Social			Privale	• •		
Country (1)	Year (2)	Primary (3)	Secondar (4)	Higher (5)	Primary (6)	Secondary (7)	Higher (S)	
United States	1959	17.8	14.0	9.7	155.1	19.5	13.6	-1
Canada	1961	••	11.7	14.0		16.3	19.7	
Puerto Rico	1959	17.1	21.7	16.5	> 100.0	23.4	27.9	
Mexico	1963	25.0	17.0	23.0	32.0	23.0	29.0	
Venezuela	1957	82.0	17.0	23.0	• •	18.0	27.0	
Colombia	1966	40.0	24.0	8.0	> 50.0	32.0		
Chile	1959	24:0	16.9	12.2			15.5	
Brazil	1962	10.7	17.2	14.5	11.3	21.4		
Great Britain	1966		3.6	8.2		41.4 6.2	38.1	
Norway	1966	••	7.2	7.5	••	6.2 7.4	12.0	
Sweden	296 7	••	10.5		u 1 4			
Denmerk	1964	••		9.2 7.8	••• 🌾	••	10.3	
The Netherlands		••	 5.2	1.8 5.5	• • - S ₁		10.0	
Belgium	1967				••	8.5	10.4	
Germany	1964	••	•••	9.3	• • • • A . • •	/ ••	17.0 4.6	
Breece	1004						4. U	
urkey	1964	••	3:0	8.0	•••	• 5.0	14.0	
Rael	1968		•••	8.5	•• .	24.0	26.0	
ndia h	1958	16.5	6.9	6.6	27.0	6.9	8.0	
	1960	20.2	16.8	12.7	24.7	19.2	14.3	
lalaysia in	196	9.3	12.3	10.7 -		i		
ingapore	1966	6.6	17.6	14.6	•• •	20.0	25.4	
he Philippines	1966	7.0	21.0	41.0	7.5	28.0	12.5	
ipan 📉	1961	••	5.0	6.0	**	6.0	9.0	
Korea	1967	12.0	9.0	5.0		0	0.0	
hailand	1970	30.5	13.0	11.0		14.5	14.0	
awaii	1959	24.1	4.4		> 100.0	_	11.0	
igeria	1966	23.0	12.8	17.0			34.0°	•
hana	1967	18.0	13.0	-	34.5 g	,		
	1968	21.7	19.2				37.0	٠
1 · · ·	1965	-66.0	28.6	19.4	92.1	30.0	27.4	
	1960	12.4				••	••	
	1966		19:4			••	••	

Source: Appendix C

Note: For explanation of symbol notations are Appendix I.

Source: Psacharopoulos, 19738:62.

. **6**9

s

Another general observation is that private rates of return tend to be larger than social rates of return. This is because the social costs are much higher than the private costs since education is highly subsidized by most states. Generally, higher education receives more public subsidy than secondary education. Psacharopoulos (1973B:68) points out, interestingly, that

... the less developed the country the higher the discrepancy between private and social rates at both levels under comparison secondary and higher education . This is surprising as one would expect a priori that the more developed the country the more the heavily subsidized would be its educational system.

8,

Psachapopoulos (1973B: 73) reports that the differences between the private and social fates of return between such level of education are much greater in underdeveloped countries and in the developed The rates of return to education in g idped countries ones. tend to be higher than in developed countries. The higher rate of return in underdeveloped countries accurately mellectimite relative scarcity of human capital there relative to developed countries. "(t)he large variation in the mathe of return, particularly However, in the less-developed countries; was seen to indicate large inefficiencies; in the past allocation of researces and relative under-investment in primary education" (Ibid.: 86). See Table 2.4 me data on the distribution of total pesources devoted to education by level and country. Data for the United States, Great Britain, Hawaii, and New Zealand indigate a u-shaped patters of private, rates of return to bachelor, mater, and Ph.D. degrees (Psacharopoulos, 1973B:71).25

Psacharopoulos (1973B:70) tried to determine the difference in rates of return between general secondary education and technical

TABLE 2.4: DISTRIBUTION SOF TOTAL RESOURCES DEVOTED . TO EDUCATION BY LEVEL AND COUNTRY

Υ.

71 ·

				an a	÷	
Country		Primary	Secor	ndary High	ler .	····· • ···· • ···
The state of the state			·			
Uniced States		32	31	37	` .	** • •
Wew Zealand	•	27	54 -	19		k *
Great Min	-9 -	10	- 70	20	-	
Israel	ر میں -	29 💦	4		``````````````````````````````````````	
Mexico	and and a			** 24		•
\$\$7 °		53	39. 34 5. 1. 6	4.12		. 4
Chile 🔶		46	38	To		
Colombia		37	38	2-1-4-85		
Maleysia	٩	44 **	¥ 47		•	
The Philippine		33	.31	36		•
Ghana-	¥	33 .**	51	10		
outh Korea		29		12		•
eny ž		30				
		B ₁		23		
Jganda	,	18	25	· 2.	a a	 ▲
lige ria		51	38	ħ.		
ndie		31	56	13		· · · · ·

SOURCE: Peachane 1973B:125

 \mathcal{A}

secondary education. Only four studies were available -- Colombia, the Philippines, Thailand, and Turkey. Due to the low number of cases, and the conflicting results, no over-all conclusions could be drawn.

To desermine efficient and rational investment on both the private and sofietal levels between human and non-human capital, one must know the relevant rates of return. See Table 2.5 for data which Psacharopoulos (1973B:82) reports. In comparing the aver-all social rate of return to education (all levels) and the social rate of return to physical capital for thirteen countries, Psacharopoulos (1973B:84) reports that generally, "siteman capital is a superior investment in developing countries but in developed countries..." His conclusions were that

... returns to both physical and human capital are higher in developing countries than in developed countries; and second, that whereas in developing countries returns are greatest to human capital, in the developed countries the greatest fairns are to forbical capital investments (Ibid.: 86).

This leads to mother hypothesis prevalent in the development, literature -- that there is a positive (some say tausal) relationship between education and economic growth (rate of growth of income). To teststhis hypothesis, Fsacharopoulos (1973B:91) categorized thirty-

two compared this with their over-all social rames of return to

education. See Tables 2.6 and 2.7 for data.

1

What stands out in this properties is a complete discontinuity in the relationship between those countries with per capita incomes below SUS 600 and those above SUS 1,000. Parcharopoulos (19738:94) draws the conclusion offer the releasof securit ecrease as, the level

Country	Year	Rate of Return	· .
United States	1959	9.7	••••••••••••••••••••••••••••••••••••••
Cenada	1957	12.9	
Maxico	1962	14.0	Ø
Venezuela	1958	16.7	و4
Chile	1940-61	15.0	
Great Britain	1955-59	8.6	• `-" ₩ 48 (•
Netherlands	1955-61 *	16.8	
leigium	1955-61	4.4	
armany:	1955-59	10.4	۽ ر و ت، ۽
	1960	12.5	
la pen	ه 1957 ۰	20:4	
hana	1962	8.0	

TABLE 2.5: SOCIAL RATES OF, RETURN TO INVESTMENT IN PHYSICAL CAPITAL IN CERTAIN COUNTRIES

å

.

Ø

SOURCE: Fracheropoulos, 1973B:82.

¥

Note: The rates of return are before tax except for the 0.5, where the rate is after profile pares, and probably also Mexico.

Per Capita Income Level	a Vel	Social		• •	řtivate	•
(\$n\$)	. Primery	. Secondary		Primary _	Secondary	Higher
C C	(2)	(3)	3 (6) 3	2	(9)	(1)
. 8	28.7 (5)	(4) 7 61	12.6-(4)	29.1 (3)	21.1 (3)	25.2 (3)
200	J5.6 (5)	14.6 (5)	11.6 (5)	24.8 (4)	20.2 (4)	25.4 (4)
350	24.6 (4)	17.5 (4)	13.5 (4)	32.0 (1)	27.5 (2)	22.2 (2)
650	38.5 (3)	(2) JUNE (2)	12.0 (5)	27.0 (1)	* 11.9 (5)	17.2 (5)
.600		6.8 (3)	, 9.0 (5)		10.3 (3)	12:3 (6)
2,200	21.0 (2)	n.1 (S	9.8 (5)	155.1 (1)	13.0 (4)	11.5 (5)

77,**97**1

•

•

•.

۰

74

-

TABLE 2.7: OVERALL SOCIAL RATE GE RETURN TO EDUCATION AND PER CAPITA INCOME BY COUNTRY

TABLE 5.2

OVERALL SOCIAL RATE OF RETURN AND PER CAPITA INCOME BY

. 1

75

そうでは

् देवे

COUNTRI 7	<u> </u>	<u> </u>		
Country	Rate of return	Per capita income s (\$ US)	• •	
United States	13.6	2,361		
Canada	12.4	1,774		
Puerto Rico	19.6	.761	y i .	
Mexico	21.9	· · · · · · · · · · · · · · · · · · ·	· -	
Veñezuela	42.0	776	·. •	J
Colombia	°25.9	320		3
Chile	19.3	365,		р Р°д - •
Brazil	13.7 🖤 🐪	261		
Giget Britain	4,6	1,669 📽		
Norway	7.2	.,1,831		,
Sweden	, ⁴ 10.3	2.500		, i i i i i i i i i i i i i i i i i i i
Denmark .	7.8 •	1,651 (1997)	* ¥ ¥ *	
Netherlands die	5.3	1,490		
Belgium	9.3	1,777	•	e
Germany	4.6	1,4 9 0 - 34	•	
Grèsce	3.9	478		. 4
Istael Par .	9.6	704 🕉		
India -	. 17.2	73		• •
Malaysia	10.6	280,		•
The Philippines	12.7	● 250		
Japan	5.1	464	-	•
South Korea	9.4	146	1	
Thailand	25.2	150	•	
Hawaii	12.5	2,495		•
Nigeria	18.5	75		and the second
Ghana	15,1	233		
Kenya	18.9	111	1. S. S. S.	
Uganda N. Dhadaata	39.0	. 84		
N. Rhodesia	12.4	144		•
New Zealand	17.6	1,931		

Source: Appendix D and United Nations, Yearbook of National Accounts Statistics (1958). Note: Rates for Denmark, Belgium and Germany refer to higher education only. Rate for N. Rhodesia refers to primary level only. Refer for Germany is private. 3

Source: Peacharopoulos, 1973B:85

Personal Antonia

of per capita income increases, until some particular stage of economic development is reached and from that point the profitability of education increases alongside increases in per capita income." See Table 2.8 for the graphic display of data.

These results are confirmed by our own multiple regression analysis of the relationary between schooling and economic development for 136 countries.²⁶ The slightly negative relationship between education and development found for the poorest group of countries is explained by the assumption that a certain literacy rate threshold is a necessary but not a sufficient condition for development (Anderson, 19654: 347). Education does seem to be an important factor in economic development for those countries which have reached a certain middle point of development. Elementary secondary school enrotiment rates seem to be more important than enrollment rates in higher education and me middle level of development, whereas for the developed countries enrollment rates in higher education are more deeful for predicting GNP per capits.

Psacharopoulos (19798) utilized Schultz-type (fental values) and Denison-type (production function, time series)" calculations to ascertain the contribution of edocation to the rate of sconomic growth in countries at different levels of development. He was unable to come to any conclusions bacquess "(t) he empirical estimates were too crude for any general development to be made" (Thid.: 123)

There continue to the controversies in the sconomics of human oppital concerning minoretical, mathodological, and expirital problems. These concerning minoretical include basic superions supplies whether or not investments (whether private or public) in soditions include bing



have yielded a rate of return comparable to other investment opportunities; whether or not it is indeed possible to calculate a social rate of return to investment in schooling; whether or not it is meaningful to compare rates of return to a particular level of education between developed and/or underdeveloped countries because . of the great differences in systems of education and in the structure of national labor markets and conomies; and, whether or not major breakthroughs in methodological problems will occur with technological advances (such as in econometrics).

Economists tend to ignore sociological data and considerations, even though the latter are directly relevant to social reality. Economists and sociologists ask different questions and focus on different variables. Until recently, human capital literature has been in the domain of economics, with very little in it that has not been written by an economist.

Despite all of the controversies within human capital economics. and the mounting criticisms of human capital theory by sociologists as well as economists, human capital theory has had and continues to have tremendous influence on the policies of governments and organizations around the world as well as on much current social science research. The awarding of the 1980 Nobel Prize in Economics to Professor Theodore W. Schultz for his work in human capital theory attes to its importance.

CHAPTER II



¹See Footnote 2, Chapter I.

²Denison was measuring quantity and not quality of education.

 3 0.40 as a per cent of 1.67.

⁴All calculations freat all the costs of education as investments. If any were considered consumption, the rates of return would be higher. There are other reasons also related to how rates of return are calculated which tend to underestimate them (See Bertram, 1966B:63-64).

There are other forms of investment in human capital besides the increased formal education, such as on-the-job training. (both general and specific), information (about prices, wages, consumption and production possibilities, political system, social system, job opportunities, job searching, etc.,), emotional and physical health. Not all of these investments yield positive rates of return.

⁶"'Real' earnings are the sum of mometary earnings and the mometary equivalent of psychic earnings" (Becker, 1964B:38). Accordingly, individuals with comparation stocks of them capital may have different return rates" (different montrophe job termings) if some prefer status and other psychic benefit to o one tary benefits. Thus, profit in investment in education can take the form of money or psychic returns. Becker (Ibid.:121) does not deal with the full private, rate of return because it is "...exceedingly difficult..." and "...far beyond the scope of this study." Consequently, the private rates of return on a particular level of education will always be less than the real or full rate, and hence underestimated (Schultz, 1968A:285).

⁷These characteristics include such things as sex, age, race, urban/rural status, and ability according to Becker; as well as employment status, affinity for mobility, modevation, class, and region of " habitation according to Schultz.

See Podoluk (1965B:54-60) for a detailed discussion.

Net earnings is used by Becker (1964B:38) to mean "...gross earnings during any period minus tuition costs during the same period."

10 It can be pointed out that Hansen (1963A), used a similar approach but treated the costs as an investment rather than a negative income, gliminating the use of negative numbers., "The rate of return then Was calculated to be that rate of interest which made the present value of the investment equal to the present value of the income stream" (Podoluk, 1965B:61).

"The rates of return estimates are made for gross earnings before taxes. The analysis is based on cross-sectional data at a particular point in time. Thus, the rates of return that are calculated are based on a static situation, though Becker has made adjustments for the probable secular growth in earnings as well as for the cross-sectional data for rising earnings (Podoluk, 1965B:65).

¹²Becker discounts both the streams of costs and returns to adjust for inflation (to keep dollars constant). The internal rate of return "...is simply a rate of discount that makes the series of absolute earnings differentials between college and high school graduates sum to zero" (Becker, 1964Bi76). It does not necessarily equate the present values of returns and costs (Ibid.). Becker (ibid.) does make use of the other method, namely calculating "...present value of the monetary gain...sum of all absolute differentials after they have been discounted at appropriate market interest rates...", though emphasis is on the former method.

¹³Becker (1964B:111) states, 'Downer, that "(i)f the rates of return were normally distributed that If the coefficient of variation equaled one, about one-third of the memors would receive rates either above 24 or below 0 per cent." Furthermore, "(t)he existence of many low and even negative raturns has been returned by there from the wide overlapping of the distributions of the manning of the distributions of the distributions of the distribution of the distributi

¹⁴"...(F) rom 1938-1957, a rate of return was defined for all corporate manufacturing firms as the ratio of steer-tax profits to total capital. The simple average of these ratio equals about 7 per cent..." (Becker, 1964B:115). The coefficient of variation in returns to investment in smaller corporate manufacturing firms is "...somewhere between one and two" (Ibid.:112). This indicates that such investments are with high risk and illiquidity -- qualities akin to investment in numan capital (bid.:115). The rate of return to low.risk, liquid investments are several percentage mints lower than the 7 per cent return to smaller corporate manufacturing firms (Ibid.)".

¹⁵While business investments are often said to pay off within five or ten years, the pay-off from college takes much longer: the unadjusted rate of return, to the 1949 cohogt of white male graduates is about 13 per cent; yet ten years after graduation it would grafil be negative and after a full fifteen years only about 6 per cent (Becker, 1964B:112). ¹⁶Becker states that gross investment in education may beve risen faster because costs not quantity rose faster. He states the most reactuable measure of quantity of education is the number of persons receiving college education. "Since 1940 the number of college graduates in the Tabor force has much more than doubled while the real value of the capital stock has increased by less than 70 per cent..." (Becker, 1964B:123).

¹⁷Becker adjusted the 12% return to include incorporated non-manufacturing firms (for which he did not have an estimate on the return). Thus, Becker set the before-tax return to all incorporated firms (manufacturing and non-manufacturing) at 10%.

18"...(D)irect returns alone cannot justify a large increase on college education relative to expenditures on business capital".^A (Becker, 197IA:110). However, indirect, or external returns to college education have not been taken into account, largely because an economist has "...few techniques for measuring them, and he usually does not even think that he knows much about them (Ibid.:111). Since not much is known about the possible external returns to college education, "...a firm judgment about the extent of under-investment in college education is not possible" (Ibid.:115). However, Becker (Ibid.) does point out the way of increasing the private rate of return to college education. That is to improve the <u>quality</u> of (the) college students.

19"Unfortunately, adjusted high-school rates cannot be estimated very easily" (Becker, 1964B 125) regester (Ibid.:126-127) adds that

. 9

(t)he unadjusted rate of return to white male high-school graduates is greater than that to college graduates and the unadjusted rate to elementary-school graduates would be still greater. Such evidence might well suggest "diminishing returns" or "diminishing marginal product" from additional years of schooling. Adjustments for differential ability, however, seem to reduce the apparent rate more to high-school than to college graduates, and, I may add, probably still more to elementary-school graduates. So the appearance of dimishing returns results at least in part from the nature of the correlation between ability and education. Fully adjusted rates, therefore, might show no diminishing returns and might even show "increasing. returns" to additional years of schooling. ,

²⁰In the discussion of comparisons of rates of return to elementary education, high school graduation and college graduation are all unadjusted for differential ability.

"Becker (1964E:129) warns us to note that since these rates of return are unadjusted for differential ability, the true rates would be different if the correlation between ability and education changed. It saws that differential ability of high school graduates probably has rised over the time period "...because now only the physically handicapped, dullards, or least motivated persons fail to go to high school" (Ihid.). Becker (Ibid.) goes on to state that others have found.

... a large ratio of adjusted to unadjusted earnings differentials between high-school and elementary-school graduates at younger than at older ages. The ratios between college and high-school graduates, on the other hand, are smaller at younger ages.

Thus, the "true" rates for high school graduates after adjusting for ability may not be the start than those for college graduates, since ability seems to differ more between high-school and elementary-school. Undents than between college and high-school students (Ibid.: 155). "A similar qualification applies to the crude evidence indicating that rates on elementary-school addication are the highest of sill" (Ibid.).

See Becker (1964B:129) for discussion was

23 These figures represent the increment in sernings over the earnings of persons with five to eight years schooling, and over the earnings of persons with four to five years of secondary schooling respectively.

4. 24 Schults (1961 - U.S.; Hansen, 1963 - U.S.; Becker, 1966 - U.S.; Hansch, 1967 - U.S.; Midoluk, 1965 - Ganada; Wilkinson, 1966 - Canada; Carnoy, 1967 - Latin Maerica; Carnov, 1970 - Puerto Mico; Selowsky, 1968 - Colombie: Mawlete, 1970 - Minute, at. al., 1967 - Great Britain; De Wolff and Rurter, 1968 - Metherlands; Kim, 1968 - South Kores; Blaug, 1971 - Theiland). Some studies, however, have indicated the reverse tiend.

* 25 ma private rais of maximum to a Makras degree is termined as 7.57 and 0 15 for a M.D. in the Unified Marga, based of Margariter and Mooney (1969) anappingt (1983) However, Schultz (1890, 1990) states that the 7% private rate of return to graduate companying the by Hendler (1987) anappingt to doubled because and us permitted and the for eachings of graduate statements for the Margarit and the for eachings of graduate statements where a double for the Margariter and the to graduate students and the private rate do the second for the Margariter is graduate from 7% to be marked the private rate do the second for the Margariter and the education from 7% to be marked the private rate do the states that the 700 light is growsly undergravingt. The does, however, dimense Tanks Martin 1968) 26 Data was taken from the World Handbook of Political and Social Indicators II which was collected under the direction of Charles Lewis Taylor and Michael C. Hudson and published by the Inter-University Consortium for Political Research in Ann Arbor, Michigan, 1973. The peper was "World Davelopment Project: Multiple Regression Excercise" by S.M. Belcher El-Nahhas, University of Alberta, May 1977.

· .

· . .

•د

.83

CHAPTER III

THE INFLUENCE OF HUMAN CAPITAL THEORY ON PUBLIC POLICY

Human capital theory has had great world-wide influence on public policy since the late 1950's. The idea that education is capital in the same sense as material capital has had tremendous impact on educational participation rates, the proportion of national budgets allocated to education, the administration of public systems of education, the size of public systems of education and their organizational structure, as well as the curricula and pedagogy.

In short, human capital theory has had a profound impact on the structure, content, and methods of schooling. Not only has the <u>size</u> of the educational system been expanded, the <u>pattern</u> of that system has been transformed. It is the influence of human capital theory as a major force responsible for the development of public policy behind this transformation which is the focus of this chapter.

Schultz (1972A:28) states that

(i)t is fair to say that we do not have an economic policy that encompasses education; nor do we have even a partial economic policy to optimize the many different classes of educational expenditures. Although the total annual expenditures on regular schooling and 🕔 higher education in the United States now exceeds \$70 billion, an accounting which does not include the earnings foregone by students nor the expenditures on programs complementary to education, there is no policy to integrate in terms of costs and benefits ools these educational expenditures among a and among programs to which funds are illocated, despite the magnitude of the expenditures and the heterogeneity of the educational enterprises. Even if every public program were efficient in terms of its specific purpose, it would be astonishing if we were to discover that each and every educational program met

the test of general economic efficiency. On the contrary, we would expect and we are beginning to identify some of the malallocations within the educational complex. But the task of investigating the sources and the extent of the lack of general economic efficiency in this area presents a large set of research opportunities that will continue to go by the board if they are left to the research units of major school systems, state educational bodies, and the many ongoing federal programs.

In response to Schultz, perhaps it is fair to say that human capital theory has over-all had far-reaching influence on public policy decisions regarding education as well as on research. However, there are many instances of inequality and inefficiency. This may well be due to conflicting social goals, and hence, lack of a comprehensive, well-planned, well-executed and ably-administered social policy. Changes in the educational system have certainly been piecemeal and incomplete. Despite all this, there has been a definite, decisive shift in public policy in the late 1950's as a result of social science research. Human capital theory has played no small part in this shift.

In discussing the influence of research on public policy and the public, Richard A. Dershimer¹ states that "(i)n education there are fewer well-known examples but notable ones" and then goes on to point out three, of which one is "...research on the economics of education by Schultz that altered 'the prevailing view that schools consume capital to the view that schools produce capital'" (Weaver, 1975B:66-67).

Psacharopoulos (1973B:1-2) [/]remarks that "...there has been an almost 180° shift of emphasis in development planning, the emphasis changing from physical to human capital as the major source of growth" .85

once education had been seen as an investment.

A. Scale and Scope of Educational Expansion at all Levels

The size and the cost of education has grown dramatically since the 1950's. This has caused considerable concern on the part of governments as to the efficiency and rationality of such growth. Human capital theory reassured governments that such expenditures and expansion was indeed economically justifiable and even necessary to maximize socio-economic growth for the nation and the individual.

Rivlin (Schultz, 1972A:85-86) remarks that policy decisions in the United States have been influenced by human capital research since the 1950's, that it

> ...played a role in increasing federal funding for education, health services, manpower training, and other human investment programs, especially for the poor, in the middle 1960's...It was this fundamental view of education, training, and health services as investments in people that influenced the thinking, not the rate of return estimates themselves. The strategists of the war on poverty picked up the concept of human investment...

This chapter is less concerned with focusing on the cost and size of educational expansion than it is with the pattern of that expansion. However, a few illustrations of the size and cost of educational expansion will help the reader appreciate its magnitude.

The growth of enrollments in various levels of education for the OECD countries between 1950 and 1965 are indicative of the scale of expansion. OECD (1970Bb:32) reports an increase of 32% for primary education, 87% for secondary education, and 150% for higher education. The largest growth in absolute numbers was in primary education, due mainly to demographic factors; and in secondary education, 86

 \sim

due mainly to democratization. The highest rate of growth was in higher education.

During the 1950's, higher education for a minority was stressed. Social demand for higher education was increasing in the developed countries, where the need for highly educated manpower was rising rapidly. Engineering and science were stressed. With most underdeveloped countries gaining political independence in the late 1950's and early 1960's, their development policies were geared towards copying the structures of modern, industrialized countries. This led them to invest in physical capital-intensive projects and the higher education of a minority elite. Secondary education was improved, but primary education was neglected and pre-school education virtually nonexistent. There was little, if any, vocational and technical education.

Underdeveloped countries began establishing their own uni-• versities and expanding those already in existence. A great proportion of education budgets was spent to this end.

By the early 1960's, the push for democratization of secondary education in Western Europe and Canada resulted in the development of comprehensive secondary schools. Besides building large comprehensive secondary schools, Canada had also begun to build large vocational high schools. Such vocational and technical secondary schools were already a long established part of the dual secondary system in Western Europe.

The World Bank supported an education project for the first time in 1962, and the following year issued its first memorandum on its policies regarding education (Williams, 19-76B:94). Employing primarily economists, the World Bank was influenced by human capital

theory. World Bank loans to underdeveloped countries for education were geared primarily to the development of middle-level skilled manpower. Thus, between 1963 and 1971, World Bank lending was allocated to primary, intermediate, and higher education in the proportion of 5:72:23 (Ibid.).

This reflects a lag in terms of developments in the developed countries, where educational expansion occurred at the elementary level first. However, it also reflects the lag of acceptance into economic development theory (or more appropriately, the lag in its acceptance by development planners), of the importance of human capital formation from the very beginning, e.g., pre-school age years.

Full-time enrollment in elementary, secondary, and postsecondary education combined increased 45.7% between 1960/61 and 1971/72 in Canada (OECD, 1975Bc:#29). The retention rate of secondary school students (as measured by Grade 12 enrollment related to Grade 2 enrollment ten years earlier) doubled between 1961/62 and 1972/73 from 36.4% to 71% (Ibid.:#31).

Canada stands out as having achieved a greater change in its educational system than anyother country since World War II. During the 1940's, Canada was "...one of the less developed (educationally) of the great democracies" (OECD, 1976B:22). Enrollment in Canadian universities, for instance, increased nearly eighteen times between 1952 and 1972, from 5.4% of the 20-24 year olds to 18.1% (Fleming, 1974B:88).

The expansion of elementary school (kindergarten through Grade 8) enrollment rose from 1950 through the 1960's in Canada reflects the sharp increase in school the population. In 1965/66, Canada had a record four million children enrolled in elementary schools, 96.1%

of which were in public schools, 2.8% in private schools, and 1.1% in federal (Indian, Northern Administration, and Overseas) schools (Illing and Zsigmond, 1967B:5-6).

Enrollment in secondary schools more than tripled in Canada between 1950 and 1965/66, from less than 400,000 to over 1.2 million students, with the peak rate of increase during the late 1950's, and early 1960's. It continued to grow into the late 1970's. This growth in secondary school enrollment is the result of an increase in population and an increase in educational participation rates, both roughly of equal importance (Illing and Zsigmond, 1967B:21).

The growth of enrollment in non-university post-secondary education in Canada betwen 1951 and 1965 has been tenfold (2,800 in 1951, 12,000 in 1960, 26,000 in 1965). Whereas the 18-24 year age group in Canada increased by less than one-third between 1955/56 and 1965/66, the proportion of that group enrolled more than tripled (Illing and Zsigmond, 1967B:69).

Enrollment rates for post-secondary education experienced the largest increase in most developed countries. The percentage of 20-24 year olds enrolled in post-secondary education in Canada jumped from 13.5% in 1960 to 20.9% in 1965 to 25.5% in 1969 compared to 32.2%, 40.4%, and 48.4% in the United States and 5.8%, 9.2%, and 12.1% in West Germany (OECD, 1975Bc: #33). See Table 3.1 for other comparisons.

Growth of post-secondary enrollments in OECD countries between, 1960 and 1965 was 9.1%. It dropped to 7.2% between 1965 and 1970 (OECD, 1974Bc:15). The number of students enrolled in post-secondary education between 1960 and 1970 tripled in France, Sweden, and Greece,

			·	
Country	1960	1965	1969	
Canada	13.5%	20.9%	25.5%	
France	7.4%	13.9%	15.9%	
Japan	8.6%	11.9%	15.8%	•
England and Wales	6.2%	8.7%	9.8 %	
U.S.A.	32.2%	40.4%	48.4 X [*]	
U.S.S.R. 11	11.0%*	29.5 % *	26.5%*	
West Germany	5.8%	9.2%	12.1%	
•				

TABLE 3.1: ENROLLMENT RATIOS FOR THE 20-24 YEAR AGE GROUP, SELECTED COUNTRIES

*Includes evening and correspondence students.

SOURCE: UNESCO Statistical Yearbook, 1972, Table 2.7.

while it doubled in fifteen other OECD countries (mostly between 1962 and 1967). In Canada, the growth in post-secondary enrollment occurred at the end of the 1960's, <u>after</u> the peak of the baby boom (Ibid.:16).

Increasing enrollment in Canadian universities since 1950 has also reflected more an increase in participation rates than population. Enrollment rose from 72,737 full-time university students in 1955/56 to 113,857 in 1960/61 to 205,888 in 1965/66. This represents an increase of 57% for the former period and 81% for the latter. This can be compared to the over-all population increase of 18-24 year olds of one-third to realize the substantial increase in educational participation rates at this level (Illing and Zsigmond, 1967B:34).

The full-time university enrollment as a percentage of the 18-24 year old population increased in Canada from 4.2% in 1951/52 to 4.7% in 1955/56 to 6.7% in 1960/61 to 10.1% in 1965/66 (in which year British Columbia reached 13.5%)(Illing and Zsigmond, 1967B:37).

Graduate enrollment in Canadian universities declined during the early 1950's. However, it rose from 47 per 1,000 undergraduates in 1955 to 58 in 1960 and then to 84 in 1965. As Illing and Zsigmond (1967B:48) point out, "(b)y 1965, there were 75% more undergraduate but 160% more graduate students than five years earlier."

Despite the large increase in relative terms, the absolute number of graduate degrees earned in Canada between 1950 and 1965 is low. In 1951/52, less than 234 Ph.D.'s were granted by Canadian universities.² In 1965/66, the number had risen to about 700. Canada lagged behind the United States and the U.S.S.R. and was only slightly shead of Western Europe in terms of the number of Ph.D.'s earned per 10,000 people aged 23 years (Illing and Zsigmond, 1967B:64). See Table 3.2 for comparisons.

At the master's degree and licence level, 1,601 were granted by Canadian colleges and universities in 1951/52. This dropped in 1955/56 to 1,459 and rose to 2,447 in 1960/61 and to 5,000 in 1965/66 (Illing and Zsigmond, 1967B:66).

For the period up to the mid-1960's, efforts to expand education, not only to accommodate the increased number of school-age children but to increase the rate of educational participation at the secondary and post-secondary levels, exceeded those in Western Europe, "...but lag [ged] substantially behind those of the United States, and probably also lag [ged] behind those of the U.S.S.R:" (Illing and Zsigmond, 1967B:58).

On a world level, enrollment increased 65% in pre-school education, 33% in primary education, 65% in secondary education, and 107% in higher education between 1960 and 1968 (UNESCO, 1972B:37).

Another way of gaining an appreciation of the size of educational expansion since World War II is the increase in percentage of GNP spent on education. Canada spent 4.4% of its GNP on education in 1960. This increased to 9.0% in 1970 (OECD, 1975Bc:941). Table 3.3 gives comparisons for some OECD countries.

UNESCO (1972B:42) reports that "...average world-wide public spending on education is increasing more rapidly than the average world-wide gross national product³...", rising from 3.02% of world average GNP in 1960 to 4.24% in 1968. Proportion of state budgets spent-on education.also reveal the importance placed on education as an instrument of economic growth and development. Between 1960 and

·		AGED 23 YEA	SAND OLDER
Year	Country	Number	Degree
1965/66	Canada	26 3	Phil
1965/66	U.S.A.	67	Ph.D.
1963	Canada	18 🌿 👘	Ph.D.
1963	U.S.A.	55	Ph.D.
1963	U.S.S.R.	30	Kandidat Nauk
1961	France	15	Niveau doctorat de 3 ^e cycle

TABLE 3.2: NUMBER OF PH.D.'S EARNED PER 10,000 PERSONS

١

\$

SOURCE: Illing and Zsigmond, 1967B:64.

.

.

.

93

.

Country	1961	1969	(`
Canada	4.6 %	8.3%	
France	2.4%*	4.5%	
Japan	4.17*	4.0% 🖉	
United Kingdom	4.3 X	5.6%	
Jnited States	4.0 X +	6.32	
J.S.S.R.	5.9 %*	7.3%	
lest Germany	2.9%	3.6%	

TABLE 3.3: SHARE OF GNP DEVOTED TO PUBLIC EXPENDITURES ON EDUCATION

*Figure applies to 1960

⁺Figure applies to 1959

SOURCE: UNESCO Statistical Yearbook, 1972, Table 2.24.

.

.
1965, this proportion rose from 14.5% to **T6.4%** in Africa, 15.6% to 17.6% in North America, 12.6% to 15.4% in Latin America, 11.8% to 13.2% in Asia, 13.5% to 15% in Europe and the Soviet Union, and 10.4% to 15.7% in Oceania (Ibid.:41).

While these world-wide growth rates in educational expenditures and enrollments have increased substantially between the 1950's and mid-1960's, there are marked disparities between the developed and the underdeveloped countries in terms of percentage of school-age children actually in school and in terms of the quality of education. "With about one-third of the population and only one-quarter of the young people in the world, industrialized countries spent ten times more money on education than the developing countries" (UNESCO, 1972B:50). The differences, furthermore, are getting larger.

Between 1960 and 1968, the industrialized countries increased their spending on education by 145%, whereas that of the underdeveloped countries increased by 130% (UNESCO, 1972B:50).

Between 1960 and 1968, one-half of the world's school enrollment of 5-24 year olds were in the developed countries, despite the fact that the underdeveloped countries have three times as many children as the developed countries⁴ (UNESCO, 1972B:50). Most underdevel d countries still have not reached universal elementary school enrollment, despite having set that as a priority goal in 1960. In Asia, the adjusted enrollment ratio for Grades 1-5 was 83.2% in 1970. In 1965, it was 44% in Africa. The Arab states reported 62.1% for 1967. Only Latin America appears to be coming close to reaching the target, reporting 100% enrollment in 1968 (with the exception of a few countries) (UNESCO, 1972B:53).

Even these figures mask the reality of the lack of elementary education for a significant proportion of school-age children. In 1968, 60% of school-age children were not in school in Africa, 50% in the Arab states, 45% in Asia, and 25% in Latin America (UNESCO, 1972B:54).

Despite the phenomenal expansion of education in the underdeveloped countries since 1960, the provision of education still lags at least thirty years behind Europe. Table 3.4 indicates the ratio of attendance in post-secondary education for six world regions.

During the 1960's, the OECD countries achieved universal enrollment for all children at least through the first stages of secondary education, extended the period of compulsory schooling, and increased their investment in education at twice the GNP growth rate (OECD, 1970Bb:12)

The financial burden on the underdeveloped countries is much greater than it is on the developed ones. Even though most underdeveloped countries have been spending about the same share of their income on education as the developed countries, the costs to the former are relatively much greater.

...(A)t the primary levels, it costs most low-income countries, relative to their national income, five to ten times as much to provide primary schooling for a given proportion of the school-age population as it does the high-income countries, relative to their national income. It usually costs a poor country proportionately very much more at secondary and higher levels ("National Strategies for Development" Mimeo, n.d.)

It is not surprising, then, that enrollment rates are much lower than in the developed countries.

Shifting to the qualitative, structural changes in the educational system, there were many changes in organization, content, and methods of schooling during the late 1950's and 1960's which can be traced

TABLE 3.4: ENROLLMENT RATIOS (OF ATTENDANCE) IN POST-SECONDARY EDUCATION (1960's)

	Region	Ratio
	North America	1 - 8
	Europe	1 - 20
	Asia	1 - 38
•	Arab States	1 🔆 45
ħ	Latin America	1 - 49
	Africa	1 -490

SOURCE: UNESCO, 1972B: 52.

٠

÷

.

97

.

. :

to a discernible need for public policy changes resulting from social science research. The influence of social science research has been both direct and indirect. Popularizing knowledge so that the public accepts it and then pressures institutions for a change is an indirect effect. Calling on experts for advice and funding specific research projects are direct ways in which governments are influenced by social scientists. Social scientists themselves are often explicitly interested in influencing public policy and make their views known to political decision-makers. The "...connections between government and social science have been strongly institutionalized" in the past generation, and although responsibility for policy rests with government, politicians rely on research findings of policy-oreinted studies (Karabel and Halsey, 1977B:5).

Human capital theory research has been particularly influential on public policy. The opening up of higher education and the development of alternative post-secondary institutions in North America and Western Europe are examples of structural changes justified, if not entirely induced, by human capital theory.

These post-secondary institutions generally fall into the category of "short-cycle higher education establishments" since they offer two year programs and have much lower entrance requirements than the traditional four year universities and colleges. Their establishment is viewed as part of the democratization and massification of higher education as well as providing a second chance to individuals who have dropped out of education to return and succeed. Thus, equality of educational opportunity has been extended beyond the traditional school attendance years. It fits in with the strategy

to reduce wastage of talent.

Along with this diversification in post-secondary education is the provision for part-time study. Part-time higher education has become increasingly popula since the end of the 1960's in Western Europe and North America. In fact, part-time courses in universities have developed more rapidly than full-time higher education in Western Europe since 1965. The United Kingdom has developed the greatest variety of types of post-secondary education in the 1960's as well as the largest percentage of students enrolled in part-time vis-a-vis full-time post-secondary education courses (OECD, 1974Bc:20-21).

Acceptance of the concept of common comprehensive secondary schools rather than narrow, specialized education at the secondary level, especially in the underdeveloped countries and Western Europe, is another such development. A greater flexibility in curricula, sensitive to the changing needs of technology and to the varying needs and interests of individual students, also occurred. Along with this was the postponement of educational specialization and vocational choice, at least to age 14. A common school experience for all children was seen as a positive step towards equality of educational opportunity. There was a move towards greater individualization of instruction within the classroom and a no-fail system (elimination of lock-step progress) in elementary and secondary schools in Western Europe and a move in that direction by the underdeveloped countries. Transferability from one program to another at both the secondary and the post-secondary levels was another change aimed at providing equality of educational opportunity.

Increasing the period of compulsory education to at least age 14 in Western Europe and some underdeveloped countries was a decision influenced by human capital theory. It was, however, dwarfed in reality by the fact that pressure of voluntary enrollment in European secondary schools was increasing the period of common educational experience beyond the compulsory period anyway. In May 1969, the European Ministers of Education stated in a formal resolution "that the period of education should be extended to 11 or 12 years for all and that education be based on a broad common curriculm," (OECD, 1970Bb:95).

The emphasis on pre-school education and the improvement and availability of kindergarten (integration into elementary schooling) in North America and Western Europe and to a lesser extent the underdeveloped countries are other examples of human capital theory influence on public policy regarding education. Development of human capital during the pre-school years is especially important for those children from "disadvantaged" groups who would enter school with significant handicaps affecting academic achievement vis-a-vis those children from more advantaged groups.

These pre-school education programs are related to a variety of compensatory educational programs for elementary and secondary school students from disadvantaged groups. The concept behind such programs was that wastage of talent could be minimized, if not eliminated, by developing the abilities of students from disadvantaged groups. Compensatory programs were designed to prevent dropping out, and in a wider sense, to provide equality of educational opportunity. These programs often included nutritional supplements and/or hot

meals as well as training for the mothers to off-set disadvantages in the home environment which impede the academic achievement and motivation of children from poor families.

Schultz (1972A:46) states that "...there are strong reasons for believing that preschool investment ranks high, even higher than that pertaining to elementary schooling, both in terms of rates of return and of equity." Programs designed to provide equality of educational opportunity at the entry point in the educational system are thus seen as being more promising than intervening later at higher levels after the children have experienced academic failure and reinforcement of psychological factors detrimental to achievement. Furthermore, investment in pre-school human capital is a dual investment -- since it is also "...a means of increasing the skills and knowledge of mothers with low levels of schooling..." (<u>Ibid</u>.:47).

In the underdeveloped countries where other factors required for human capital development are also lacking, pre-school years are even more important in the determination of life chances -- for physical survival as well as for academic achievement. Human capital theorists have stressed this level since the mid-1960's.

Any strategy designed to provide equality of educational opportunity fit into the category of change in organization or structure of education. The cost of continued and/or increasing inequality is higher to society than the cost of rationally designed, efficiently administered programs to equalize opportunity:

"Bussing" in the United States is an example of a nation-wide strategy to desegregate education in order to equalize opportunity. Since segregation by race and class was found to be detrimental to

the educational and life chances of poor children from racial minorities, the courts have ordered bussing of poor Negro children and middle-class white children to obtain a balance of each in the classroom. Bussing s has become institutionalized in the United States.⁵

Another strategy developed to provide equality of educational opportunity was the direct financing of education by the federal governments in the United States, and to a lesser extent in Canada.⁶ This was done, among other reasons, to try to equalize financial resources and physical facilities available to schools across neighborhoods and regions. The great discrepancies in material resources available to schools had created vast inequalities in the quality of education and in the life chances of the students.

Because education is locally controlled in the United States, there are great problems of disparities in resources available to school boards. The tax base differs as much as 2,000 to 1 between school districts in the United States (Carlson, 1974A:115). Differences in tax base are urban/suburban, rural/suburban, and even suburban/suburban in the same state -- not to mention gross differences between states and regions.

As the California Supreme Court pointed out in the Serrano case, "affluent districts can have their cake and eat it too; they can provide a high quality education for their children while paying lower taxes. Poor districts, by contrast, have no cake at all (Greenbaum, 1971:514)." The same can be said for the states. It should be borne in mind that taxes hurt the poor more than the rich, because the poor are already at a subsistence standard of living. Moreover, the property tax is especially regressive because it is not based on ability to pay (Carlson, 1974A:115).

0

In addition, it has been shown that the wealthier neighborhoods in a particular school district have more money per child spent in

school than do the poorer neighborhoods in the same school district. It even goes as far as within a school, more money is spent per child on children in the higher tracks and ability groupings. It is generally children from the middle and upper middle class families who are in the higher ability groupings and tracks (Rothstein, 1971A; Shaefer, et. 'al., 1970; and Porter et. al., 1973B)(Carlson, 1974A:116).

Carlson (1974A:115) notes that it is not only the absolute <u>wealth</u> between school districts and states which differs but also the <u>effort</u>. The following table indicates the discrepancies between wealth and effort for four districts in the United States (Ibid.)

CATEGORY	DISTRICT	WEALTH*	EFFORT+
urban	Newark, N.J.	\$20,338	\$3.69
suburban	Millburn, N.J.	\$92,856	\$1.43
rural	Baldwin Park, Ca.	\$ 3,706	\$5.48
suburban	Beverly Hills, Ca.	\$50,885	\$2.38

* Property value per pupil + Tax rate

+ lax rat

"What all this shows — and it is typical for the nation as a whole is that poor communities spend less money per pupil but they make a greater effort to support education than do rich communities" (<u>Ibid</u>.). The direct intervention in education by the federal government since the late 1950's in the United States has helped diminish some of these gross inequalities between school boards. Federal funds to school boards, and even to state governments, have been tied to requirements specified by the U.S. government in an effort to provide equality of educational opportunity throughout the country.

The establishment of student counselling services in secondary schools and post-secondary institutions is also consonant with human

capital theory. Such services are seen to be necessary by human capital economists to provide students with information about their own capabilities and interests and also information about how and where to best pursue their development.

Human capital theory focused attention on the need for improvement in the quality of elementary and secondary schooling; to the development of pre-school human capital formation in order to provide equality of educational opportunity to all children; to provide equality of educational opportunity at all levels of education (including a variety of direct and indirect aid programs and services for students); and to change the school curriculum (emphasis on general education and principles as well as development of problem-solving skills) and pedagogical style (emphasis on flexibility, individualized instruction, independent problem-solving); and to utilize rate-of-return analysis for the efficient allocation of public funds and resources to the educational system as well as between levels within it.

B. <u>Rate-of-Return Analysis to Determine Efficient and Equitable Allocation</u> of Resources

Other policy concerns of human capital theory regarding education which were not implemented in public policy during the 1960's have been accepted in the 1970's. Among some of these changes is the provision of financial aid allocated directly to students rather than indirectly through educational institutions. Another is the reduction in public expenditure on higher education in view of the lower rate of return vis-a-vis elementary and secondary education as well as in view of the socially regressive implications of highly subsidized postsecondary education on the distribution of personal income. Another

major change has been the full-fledged acceptance of continuing and recurrent adult education. This is due in part to the increasing social demand for higher education (which itself is partly responsible for the democratization of secondary education in Canada and Western Europe as well as partly a result of that process). Social demand for higher education has been high in the United States since World War II.

Acceptance of recurrent education as part of the post-secondary education system is yet another example of the important structural changes taking place in post-secondary education since the 1960's. A more comprehensive post-secondary educational system has been evolving in the 1970's, "...based on new structures designed to promote diversity and permitting much greater flexibility than has hitherto been possible" (OECD, 1970Bb:96).

It also reflects a movement away from the emphasis in education on youth. Recurrent education as an alternative for achieving equality of educational opportunity has been adopted. Adult education has been incorporated into national strategies to develop human capital and provide for equality of opportunity. Louis Emmerij (OECD, 1970Bb:66) states that

> Recurrent education would be instrumental in meeting the objectives of equity, occupational flexibility, and integration of education with active life. This fast objective should be noted in particular. With the steep increase in enrollments and the average time the individual remains in the educational system, the dichotomy between the educational sector and the other sectors of human activity, and between educational span and the active working life of the individual, has become so marked as to raise serious social problems. An educational strategy which would diminish the existing distinction between education

and other socio-economic sectors would at the same time diminish the dangers of a growing irrelevance of one system to the other and, eventually, of a growing gap between the younger and the older generations. As such, recurrent education, also because its introduction would imply much more stress on individualization, appears promising...

Papadopoulos (OECD, 1970Bb:97) sums up:

Æ.

1

By spreading out the formal educational period over the later years of life, recurrent education would have serious implications not only for the structure, content and teaching methods of the present secondary and higher education systems but also for the organisation and practices of industrial firms and other employers who would see a significant portion of their personnel return to full-time education. In thus reducing the dichotomy between active life and education and in giving adults a "second chance" in education. recurrent education would help bring about greater occupational flexibility, social equality, individual satisfaction and understanding between generations. It could therefore represent a major alternative strategy in educational policy in the years to come with far-reaching consequences for the development of the educational system and of society as a whole.

Thus, having established comprehensive secondary education, we are now establishing comprehensive post-secondary education in the developed countries.

At about the same time that recurrent education was being introduced in the developed countries, the concept of non-formal education was cautiously being discussed in the underdeveloped countries. After more than a decade of pushing for universal elementary education (100% enrollment), most underdeveloped countries were unable to carry the financial burden (exasperated by lack of international aid to elementary education during the 1960's) and were discouraged by the rising number of illiterates. Despite a lowering of the illiteracy rate, there was an increase in absolute number due to high rates of population increase and the very high ratio of youths to adults (40%-50% in many underdeveloped countries).

UNESCO Regional Conferences on Education held in Karachi (1960), Beirut (1960), Addis Ababa (1961), Santiago (1962), Tokyo (1962), Bangkok (1965), Buenos Aires (1966), Tripoli (1966), Vienna (1967), Nairobi (1968), and Marrakesh (1970), all placed prime importance on reaching universal elementary enrollment by 1970 or 1980.

Most underdeveloped countries allocate 20%-25% of all their expenditures on education and it is virtually impossible for them to spend more. Thus, cost is one important factor.

Another is the question of equity. With between 40%-70% of all elementary school-age children enrolled (and with extremely high drop-out and repeating), equity is a real problem.

Other, perhaps more immediately compelling, reasons are the high unemployment rates of the educated and partly-educated as well as the seeming irrelevance of education to the social, political, and economic realities in underdeveloped countries (Bacchus, 1979A: 3-6).

Since the poor in underdeveloped countries are mainly rural and constitute 70%-80% of the population, it appears that the existing educational system is not serving the needs of that population, nor of development (Bacchus, 1979A:7).

The publication of Coomb's <u>The World Educational Crisis</u> in 1968⁷ crystallized world attention, especially that of experts in the developed countries, on the failure of existing policies and the need for a wholesale revision.

As already indicated, the World Bank and other international development aid agencies provided aid to education in underdeveloped countries since the early 1960's. However, their emphasis then was on aid to higher education and secondary schools to educate professionals and technocrats as well as to train middle-level skilled workers and technicians for the modern sector of the economy.

Since the World Bank "...has what is probably the largest fund for employing specialists and consultants" and "(i)ts files contain by far the most information (including confidential intelligence) about educational systems (i)ts sector papers will be widely interpreted as offering well-grounded and theoretically up-to-date assessments of policies for change in many sectors..." (Williams, 1976B:15). In other words, policies developed by the World Bank have had tremendous influence on underdeveloped countries both negatively and positively and those policies have not been in tandem with some of the development objectives of the underdeveloped countries themselves -- especially as we have shown in regards to educational policies of the 1960's.

There was a total re-orientation of development strategy by most donor agencies after 1971 (Williams, 1976B:47). The World Bank issued an <u>Education Sector Working Paper</u> in 1971, followed by another in 1974. There was a prevailing focus on the equity issue and access to the educational system in order to move towards equality of educational opportunity in the underdeveloped countries.

Robert S. McNamara, President of the World Bank, made this major policy shift clear in an address to the Governors of the World Bank Group in Nairobi on September 24, 1973 (Williams, 1976B:47).

> Disparities in income will simply widen unless action is taken which will directly benefit the poorest. In my view, therefore, there is no viable alternative to increasing the productivity of small-scale agriculture if any significant advance is to be made in

solving the problems of absolute poverty in the rural areas.

The absolute poor are not merely a tiny minority of unfortunates -- a miscellaneous collection of the losers in life -- a regrettable but insignificant exception to the rule. On the contrary, they constitute roughly 40 per cent of the nearly two billion individuals living in the developing nations.

Some of the absolute poor are in urban slums, but the vast bulk of them are in the rural areas. And it is there -- in the countryside -- that we must confront their poverty.

Between 1972 and 1974, the proportion of World Bank aid to education was distributed 11:48:41 between elementary, secondary, and higher education. In 1974, there was an intensification of the policy, with educational aid allocated by the World Bank distributed 27:43:30.

In 1975, the World Bank issued the <u>Rural Development Sector</u> <u>Policy Paper</u> in which the following statement appears on page 3 (Williams, 1976B:47):

> Education systems have been irrelevant to the needs of developing countries during the last two decades because education policies were often keeping company with over-all developstrategies which were themselves irrelevant. Emphasis on the development of the modern economic sector, providing employment to a small and intensively trained elite, leads to the neglect of the 60-80 per cent of the population living in sectors characterized by traditionally lower productivity.

Sheffield (Williams, 1976B:47) lists five basic issues which shaped this policy shift. Thus, the changes envisaged were:

> ...(a) the development of functionally relevant skills, integrated with over-all development strategies; (b) mass participation in education and development through the integrated use of

Ð

Ø

expanded primary schooling and complementary non-formal education programmes; (c) greater equity through equalizing educational opportunities and linking these to broader social policies; (d) increasing efficiency by defining objectives more specifically and making qualitative improvements to reduce wastage; (e) improving management and planning including changes in the organization and finance of educational systems.

Thus, the shift from higher education, secondary education, and technical training for a relatively small elite to "basic education for the rural masses" which took place in the educational policy of aid agencies in the developed countries has been completed by 1974.

The World Bank defines basic education as

...an attempt, despite severe resource constraints, to meet the needs of substantial portions of the population who do not have access to even minimum educational opportunities. It is a supplement, not a rival to the formal education system, and is intended to provide a functional, flexible, and low-cost education for those whom the formal system cannot yet reach or has already passed by. Although the primary cycle may be its principal vehicle in many countries, it differs from the conventional concept of "universal primary education" in three major aspects:

(i) the objectives and content of basic education are <u>functionally</u> defined in terms of "minimum learning needs" of especially identified groups, and not as steps in the educational hierarchy...

(ii) the "target groups" of basic education are not necessarily school-age children. they may vary according to age, and socio-economic characteristics...

(iii) the "delivery system" of basic education will take different forms in different countries (restructured primary schools, nonformal programs, or various combinations of the two) adopted to the needs of different clienteles and to constraints upon resources (Williams, 1976B:48).

This shift in emphasis to basic education for the rural masses has led educators and politicians to seriously consider nonformal education as a viable supplement, if not outright alternative, to existing efforts.

International conferences and reports of agencies reiterate this focus on basic education for the rural masses, including the UNESCO Faure Commission (1972B), the Bellagio Conferences (1972, 1973, 1975), US-AID commissioned studies by the African-American Institute and Michigan State University (1972, 1973), UNESCO Conference in Nairobi (June 1974), UNESCO/UNICEF Conferences in Nairobi (August and October 1974), and the Canadian International Development Research Centre's commissioned study review of education and development (1974). There was also a Commonwealth Specialists Conference on Non-Formal Education for Development in New Dehli, India in January 1979.

The entire structure, organization, content, and method of educational systems were being questioned. Human capital economists stress the balance of maximum economic efficiency with equity. In view of the scarce resources available in most underdeveloped countries and the fact that significant proportions of the population are illiterate, it is much more difficult for underdeveloped countries to make such decisions than it is for the developed countries.

The same shifts occurring in educational policies of developed countries in the 1970's, discussed above, were also occurring in the underdeveloped countries -- only the relative and absolute differences in provision, quality, and equity of the educational systems as well as in the distribution of personal income and the differences in

life chances between regions, race, classes, sexes, and ages are astronomical.

The effects of some policy proposals on underdeveloped countries, likewise, are quite different than on developed countries. One example is the policy of lowering, if not eliminating, public subsidization to post-secondary education. This is part of the more recent attempt to be more efficient in the allocation of public resources to education.

As Porter, et. al. (1973B) point out, unless rising private costs of post-secondary education are balanced with a comprehensive financial aid scheme to provide grants to students from low-income families, the results are highly regressive and defeat the equity goals behind the removal of subsidies.

Schultz⁸ (1970A) insists that just as the full cost of education at the post-secondary level should be private, not publically subsidized, those students from low-income homes should receive grants to enable them to attend. Students from middle and high-income families should be able to borrow money for education in the same way that they can borrow money to obtain consumer goods or physical tcapital.

Schultz (1970A:64) and other human capital economists disagree profoundly with the conclusions of the Carnegie Commission on the Future of Higher Education that universal college education (though not universal attendance) is a right and a necessity for the provision of equality of educational opportunity because it would require a "...vast increase in public funds which would not be neutral in its effect on the distribution of personal income."

Inequalities in the distribution of personal income are much greater in the underdeveloped countries than in the developed ones. Mark Blaug (1967, quoted in Williams, 1976B:77) observed that

> The striking fact about underdeveloped countries is not so much their low level of income per head, but rather the extreme inequality in the distribution of nearly everything: income, population, education, housing, health, and the like. Not only is income per head fore unequally distributed than in developed countries, but regional variations in income within national boundaries far exceed anything found in Europe or America.

Reduction in public subsidies to higher education in the underdeveloped countries by itself would not have much effect on the situation. A switch in emphasis from higher education to elementary and pre-school human capital development would be more efficient and more equitable from the point of view of human capital economists.

The World Bank Workshop on the Economics of Education held in Washington, D.C. in October 1973 dealt with precisely these issues. Marcelo Selowsky⁹ presented a paper at that workshop entitled, "A Note on Preschool-Age Investment in Human Capital in Developing Countries" in which the opening paragraph is a quotation from Jencks' (1972B:256) study on inequality.

It is true that schools have "inputs" and "outputs" and that one of their nominal purposes is to take human "raw material" (i.e., children) and convert it into something more "valuable" (i.e., employable adults). Our research suggests, however, that the character of a school's output depends largely on a single input, namely the characteristics of the entering children. Everything else -- the school budget, its policies, the characteristics of the teachers -- is either secondary or completely irrelevant.

The poor performance of children from poor families in underdeveloped countries is rooted in physical (nutrition and health) as well as more intangible (social, psychological) factors. Therefore, these physical factors must be dealt with if these children are going to achieve in school, or for that matter in society.

The "economic problem" becomes therefore how preschool-age investment in human capital can contribute to a "least cost solution" in the production of cognitive and noncognitive skills at adult age (Selowsky, 1976B:709).

The main problem appears to be lack of resources to adequately provide a program of pre-school human capital development in the poor countries. Educational programs for mothers with emphasis on nutrition and child-rearing practices appear to be the most needed; however, such programs have not been developed in all poor countries and the ones which do exist are not available to all who need it.

Bacchus (1979A) points out that non-formal education has been tried in various poor countries but has only been relatively successful in those few countries which have also instituted radical structural changes in the economy and social structure to drastically reduce the inequality in the distribution of personal income.

The ruling elite in most underdeveloped countries reject the idea of shifting resources from the formal educational system (especially academic secondary education and higher education) and the modern sector of the economy to non-formal education and the rural sector of the economy (Leys, 1975B:258-271). What really seems to be occurring in the 1970's in the poor countries is experimenting and small-acale innovating in education, with continued emphasis on elementary education and with more international aid given to elementary level education.

As already discussed above, there has been dramatic growth of pre-school education in Canada and Western Europe and an extension ~ of it in the United States (OECD, 1970Bb:94).

In terms of educational policy per se, the aim of national governments' educational policies during the late 1960's in Western Europe and North America was to reach the goal of equality of educational opportunity. Reforms in the organizational structure, contents, and methods of schooling were all part of their over-all strategy (OECD, 1970Bb:14).

While public policy in general, and educational policy in particular, has become increasingly programmatic and comprehensive since the mid-1960's, it has also been narrowly quantitative and reactive. A basic assumption has been that the future would not differ much from the present, that the economy would continue to grow as would the need for more highly educated manpower.

In the late 1960's and 1970's, the economies of most developed countries experienced a severe downswing. The contraction seems to be long-lasting and so the large proportion of public budgets ear-marked for education are increasingly being viewed as an expense we cannot afford. There has been a renewed surge of interest in economic efficiency within the educational sector.

The fiscal crisis of the state has forced governments to cut down on government spending. A general drop in the standard of living for many people has shaken popular belief in education as a vehicle for social mobility and economic growth. A general feeling that subsidized education has benefited the wealthy at the expense of the poor has meant less public resistance to major cuts in public subsidies to post-secondary educational institutions.

Schultz has been urging such changes for reasons of economic efficiency and social welfare as well as equity for many years. The regressive effects of public subsidies to post-secondary institutions on the distribution of personal income and the resulting increase in inequality has been documented by Hansen and Weisbrod (1969A, 1969B) in their studies of post-secondary education in California and Wisconsin.

The reduction of public subsidies to higher education would bring the treatment of human capital in line with that of non-human capital. Physical capital formed by public investment is not transferred to particular individuals as a gift -- neither should public investments in human capital be transferred to individuals as a gift, according to human capital economists.

Schultz (1970A:57) states that "...the ideal price for the educational services that students obtain should be neither more nor less than the real cost of producing these services." Removal of subsidies would dramatically increase the private costs. Since Schultz says that prices should reflect the true cost and quality of education, prices will vary between institutions. The development and dissemination of better information about post-secondary programs and institutions is a necessary prerequisite for rational decision-making. A national information disseminating service should co-ordinate and publicize

information and services provided by the institutions. Schultz also stresses the need for counselling services within educational institutions to become more aggressive and "public".

In order to ensure and promote equality of educational opportunity, loans and grants have been made available by governments. This has become particularly common in the United States during the 1960's and in Canada during the 1970's.

Loans, grants, and other measures of financial aid were made available to able students who would otherwise not be able to attend university on a national level by the National Defense Education Act of 1958 in the United States. It marked the first whole-sale involvement of the federal government in education. Since then, changes have been made to promote equality of educational opportunity and later equality of educational achievement between groups. The earlier programs were designed to provide financial help to needy high achievers. Later programs were designed to help poor children increase their achievement and continue their education.

Schultz has been pressing for the development of a program of loans and grants to students from low-income homes to reimburse them for the increasing costs of post-secondary education. Students from middle- and high-income families should be able to borrow money for education in the same way that they can borrow money for material goods. Loans should, according to Schultz (1970A:55) "...rest on the level of the income stream of the family..." and even though this is "...still in its infancy..., with the relatively high level of income of so large a part of U.S. families, it carries considerable promise." Tangible assets of students' families' could also be held as collateral

on student loans.

The public debate about what kinds of changes to the educational system are desirable and practical continues. Educational policies during the past three decades have been profoundly influenced by human capital theory. The transformation of the educational system from a restrictive, elitist model to an open, mass model occurred in a very short time period. Acceptance of the idea that schools produce capital rather than consume it marked a definite shift in public policy and social science research which has been attributed to the efforts of Theodore W. Schultz in developing human capital theory. Whatever the current disagreements about public policy, concern for efficiency and equity are now integral to the discussions. Human capital economics has indeed permeated public policy decisions regarding education and social welfare as well as social science research around the world. Schultz's acceptance of the 1980 Nobel Prize in Economics is symbolic of the influence of human capital theory.

CHAPTER III

Footnotes

¹Dershimer was Executive Director of the American Education Research Association.

²This included non-residents but does not include Canadians studying abroad.

³Excluding the People's Republic of China.

⁴According to UNESCO (1972:50), "22% of the world's children under 15 are in developed countries; 78% are in developing countries."

In attempting to provide equality of educational opportunity in the United States, the issue became clouded with the question of race and ethnicity. It was generally believed by parments and experts alike that the problem was one of racial and ethnic segregation and that the solution was racial and ethnic desegregation -- when in fact, the problem was really one of class. When working class white children were integrated with poor black children, academic achievement of both groups remained low. However, when poor black children were integrated with middle class white children, academic achievement of the poor black children rose. With the average ethnic minority (except of Orientals) scoring as much as one standard deviation below the majority students' test scores in grade one, the gap in academic achievement between non-Oriental ethnic and racial minority children and white middle class children gets larger through to grade twelve. Unfortunately, many of the desegregation orders were between poor white and poor black neighborhoods. It has also been established that poor quality schooling has greater dampening effects on academic achievement of working class children than it does on those from the middle class (Gordon, 1974A:179).

^oEducation in Europe and most Third World countries is very centralized, controlled by the national government. However, education is under state control in the United States and provincial control in Canada. Education is even more decentralized in the United States, with local school boards having autonomy and control over local schools.

⁷This was preceeded by the Kericho Conference on Education, Employment, and Rural Development, held in Kenya in 1966 as well as the World Crisis in Education Conference held at Williamsburg, Virginia, U.S.A. in 1967. The publication of <u>Education for Self-Reliance</u> by Julius K. Nyerere in 1967 represents the first major policy statement in this direction, as well as one of the few policy statements developed by a Third World country rather than an aid agency in the developed countries. Philip Coombs, Director of the International Council for Educational Development, has had a key role in influencing "experts" in other development aid agencies to shift emphasis to basic education for the rural masses and away from higher education and academic secondary education for a minority elite.

⁸Schultz's position is based on the conclusions of studies done by Hansen and Weisbrod on higher education in California and Wisconsin and its effects on the distribution of personal income.

⁹This was subsequently published in <u>Economic Development and</u> <u>Cultural Change</u>, Vol. 24, No. 4, July 1976:707-720.

- 200

CHAPTER IV

THE TECHNICAL-FUNCTION THEORY OF EDUCATION: THE RELATIONSHIP BETWEEN EDUCATION, OCCUPATION, AND LABOR INCOME FROM THE PERSPECTIVE OF SOCIOLOGY

A. Introduction

The dominant paradigm in all the social sciences in the 1950's was functionalism. This was true for sociology as it was for economics. Sociology in North America, as well as in France, Germany Scandinavia, and Japan, was dominated by Parsonian structural functionalism. The most popular version of structural functionalism in the sociology of education during this period was technological functionalism. This school of functionalism stressed both the technical function of education and the efficient use of human resources. Thus, the technicalfunction theory of education, a subtype of the structural functionalist theory of stratification developed by Davis and Moore (1945A), provided the theoretical framework for sociology of education. The two main premises of the functional theory of stratification, which it shares with the technical-function theory of education, are: 1) "that occupational positions require particular kinds of skilled performance; and 2) that positions must be filled with persons who have either the native ability, or who have acquired the training necessary for the performance of the given occupational role" (Collins, 1971A:1004).

More specific to the technical-theory of education are the premises that: 1) knowledge and skills required for job adequacy in modern, industrialized societies are increasing due to continuous technological advances. This results in an increasing proportion of occupations requiring a high level of knowledge and skill accompanied by a decreasing proportion of occupations requiring little knowledge

and skill. In addition, the amount of knowledge and skill required for a given occupation increases over time; and 2) increased levels of formal education are required to provide workers with the training in specific skills and general capacities necessary for job adequacy, especially in the more highly skilled jobs; and therefore, 3) educational requirements for occupations are constantly increasing and so people tend to spend more years in institutions of formal schooling (Collins, 1971A:1004).

Advocates of technological functionalism assert that there have been significant, fundamental structural changes in the social division of labor of industrialized societies since World War II which have been determined by technological advancement and which have resulted in • increasing differentiation, specialization, and hierarchization of the labor force in the occupational structure as well as in all social institutions. At the same time, increased government involvement in and control over socio-economic institutions has also grown considerably.

Industrial innovations (including increased size and complexity of industrial and commercial firms) were said to be responsible for the tremendous upgrading of the labor force since World War II -- the increase in the number of jobs requiring more knowledge and skills, great expansion of white collar occupations in general, and for the increase in the number of technical and professional jobs. For example, in 1941, 25.2% of the Canadian labor force was employed in white collar jobs. By 1961, this percentage had increased to 38.6%. The increase in number of workers employed in professional and technical occupations was less impressive -- from 6.7% in 1941 to 10.0% in 1961 (Rinehart, 1975B:84). During the 1940's, for the first time in Canadian history

the percentage of workers employed in manufacturing exceeded the percentage employed in agriculture (<u>Ibid</u>.:50). It was this tremendous expansion of white collar occupations, however, which created a great deal of social mobility and with it the renewed belief in the popular idea that education is the great equalizer -- through schooling, everyone would be made equal (or at least less unequal).

These socio-economic changes in occupations and in the occupational structure were viewed as the <u>inevitable</u> outcome of technological change. The question of what causes technological change, however, is ignored. Clark Kerr, et. al. (1964B) present this viewpoint of technological determinism in their book, <u>Industrialism and Industrial Man</u>. As Rinehart (1975B:21) points out, proponents of technological functionalism

...almost always attributed to sophisticated technology...centralization of knowledge and authority and a detailed division of labor... [and that] a given level of technology...bears organizational imperstives that operate independent of the social, economic, and political milieu in which technology is embedded.

Thus, technological determinism fosters the idea that the (interrelated) processes of industrialization and bureaucratization inevitably mold all societies along a unilinear development path. One result of this evolutionary process is that the social division of labor is "de-socialized" into simply a technical division of labor (Johnson, 1976A:6). Social classes are transcended and society becomes more "open" and "fluid" and "mobile" -- classless. Differences between the socialist and capitalist countries dwindle to the point of convergence. There is an end of ideology. This is referred to as the Convergence Theory. The technical-function theory of education, then, basically states that as a result of technological innovations, there is a need and consequently a demand for increased cognitive and technical skills in most occupations. Therefore, educational requirements for most occupations are also rising. Thus, the phenomenon of rising educational fualifications required for most occupations is seen as an inevitable result of modern industrialization and bureaucratization (experienced in the socialist as well as the capitalist countries).

It is therefore posited by the technical-function theory of education that cognitive and technical skills needed by workers are learned in school and that the acquisition of them will make those workers more productive than their less-educated counterparts. These are very important premises of the theory. Thus, expansion of the educational system, especially at the tertiary level was seen as "funtionally" necessary during the 1950's to provide workers with knowledge and skills required for job adequacy in a rapidly expanding and changing economy.

Technological functionalism provided the framework for much of the sociology of education research done in North America during the 1930's and 1960's. With its stress on the maximization of talent in the allocation of individuals to the occupational hierarchy, technological functionalists criticized the existing system of education as elitist. It formed part of the liberal criticism of conservative ideology and policies which dominated the system of education in Western Europe and North America prior to the late 1950's. The basis of the liberal criticism is that elitist systems of education are technically inefficient because of the wastage of talent from the lower classes. Furthermore,

124

uiche a

liberal argument claimed that gross social inequality was neither rational nor just in times of fierce technological and scientific competition with the Soviet Union. Formal education, according to the functionalist argument, needed to be assigned an important role "...in the evolution of industrial, and particularly American, societies from particularism and ascription to universalism and achievement" (Murphy, 1979B:13).

So, by the late 1950's, one witnessed the traditional conservative conceptions of society coming under heavy criticism and a revitalizing of classical liberal notions of equality of opportunity and equalization of life chances. Conservatives, who believe that any society has only "...a limited pool of educable ability" (Karabel and Halsey, 1977B:9), and that social inequality is the natural and inevitable outcome of socially significant, genetically determined differences in intelligence between individuals (manifested in rigid class lines), lost ground to liberals during the 1960's -- but not without prolonged, heated public debate.¹

• The concern over availability of talent and its utilization, although it had been the subject of empirical research since World War I, took on a sense of urgency with the development of the Gold-War in the 1950's and the subsequent ideological competition between the Western capitalist countries and the Soviet Union. It was the launching of Sputnik in 1957, however, which threw the West into a virtual race against time to upgrade the North American labor force as well as reassess and reorganize the educational system.

R,

3

Consequently, new national and international surveys were undertaken during the 1950's to ascertain what the "available pool of

talent" was and how to maximize the development of talent as efficiently as possible so that workers would be as productive as possible and the national growth rates would be as high as possible in attempting to catch up with the Soviets. However, even before the on-set of the Sputnik era, interest in the maximal development of human resources had surfaced in North America, and the Commission on Human Resources and Advanced Training was appointed in the United States in 1949 to sponsor such a national survey. The results were published in 1954 by Wolfle. Similar studies done during the 1920's, 1930's, and 1940's in Europe were re-examined and used for comparison purposes (Husén, 1975B).

The conservatives, who supported elitist educational philosophy, urged a "...more rigidly administered system of ability grouping" (Sexton, 1961B:XIV), all educational resources spent on the development of an ability elite, and an early segregation of the gifted from the other students. This line of argument is found in <u>Education and Freedom</u> by Vice Admiral Rickover and in <u>Educational Wasteland</u> by Arthur Bestor.

Liberals invoked mass belief in the early American populist and egalitarian tradition (characterized by Jefferson's ability aristocracy and Horace Mann's school as the great equalizer) to call for democratization of higher education.

By the 1950's, the United States had already achieved mass, egalitarian universal secondary education and was approaching the stage of mass higher education. In 1958, the United States Congress passed the National Defense Education Act (NDEA) to provide massive federal support to higher education (Husén, 1974Bb:40). Western Europe and Canada were in the process of democratizing access to secondary education;

higher education was still highly elitist (Ibid.:4).

With the publication of the Coleman Report in 1966, however, the main issue became equality of educational opportunity at all levels of schooling. During the 1960's, with the Civil Rights movement underway in the United States, "...the debate [in the U.S.] focused on the plight of the ethnic minorities and the socially and economically underpriviledged" (Husén, 1974Bb:4). In Western Europe, research on the "reserve of talent" was the result of increased public demand for "...extending opportunities for academic secondary education to lower class children" (Ibid.). In Canada, the educational system was more elitist than egalitarian, though not as rigid as the European system. Higher education was not democratized. According to Porter (1965B:165), the educational system was inadequate to cope with the increasing complexity of the occupational structure in Canada during the 1950's. The Canadian system of education was unable to "...train people in sufficient quality and quantity for occupational roles" 🔔 (Ibid.). Canada was dependent upon the immigration of skilled manpower because Canadian education was elitist and class-bound (Ibid.: 166).

B. The Relationship Between Education, Occupation, and Labor Income as Posited by the Technical-Function Theory of Education and Technological Functionalism in Sociology.

The relationship between education, occupation, and labor income is a very strong, positive one according to the technological functionalist perspective. The functional theory of stratification (Davis and Moore, 1945A) posits a purely theoretical sociological explanation of why some workers receive a higher labor income than others. This,

in turn, provides the basis for the technical-function theory of education. Mainstream sociologists traditionally have attempted to explain "... the determination of a worker's wages in terms of the role of the worker's occupation in society" (Stolzenberg, 1975A:645). Davis and Moore (Op. cit.) attempted to explain the phenomenon of social inequality by arguing in functionalist terms the universal necessity for stratification in any and all societies. They assert that no society is (or was) classless or unstratified; social inequality is necessary, functional, and inevitable. This, they insist, is because in all societies there are some positions (occupations) which are functionally more important than others: some require special talent; some require lengthy, costly training programs; some are more agreeable than others. Hence, societies must develop a system of rewards to induce individuals to fill these positions otherwise there would be a serious scarcity of talented and/or trained personnel for them. These rewards must be distributed differentially according to the functional importance of the position as well as according to the scarcity of talented and/or trained personnel. The conclusion made by Davis and Moore (1963A:48) is that

(i)f the rights and perquisites of different positions in a society must be unequal, then the society must be stratified, because that is precisely what stratification means. Social inequality is thus an unconsciously evolved device by which societies insure that the most important positions are conscientiously filled by the most qualified people. Hence every society, no matter how simple or complex, must differentiate persons in terms of both prestige and esteem, and must therefore possess a certain amount of institutionalized inequality.

6

Societies, according to Davis and Moore (1963A:48), have three kinds of rewards which must be allocated differentially to individuals according to their occupation. They are: 1) "...the things that contribute to sustenance and comfort"; 2) "...the things that contribute to humor and diversion"; and 3) "...the things that contribute to self - respect and ego expansion."

What positions (occupations) in society are the most rewarded? Davis and Moore assert that those positions which are functionally important and require scarce talent and/or costly, lengthy training so that qualified individuals are scarce (scarcity of personnel) are the positions which should and must be most rewarded. Otherwise, it will be difficult for "society to fill" them. In the words of Davis and Moore (1963A:49),

... if the skills required are scarce by reason of the rarity of talent or the costliness of training, the position, if functionally important, must have an attractive power that will draw the necessary skills in competition with other positions. This means, in effect, that the position must be high in the social scale -must command great prestige, high salary, ample leisure, and the like.

Thus, according to the functional theory of stratification, labor income is a function of the scarcity of personnel to fill functionally important positions (occupations). Power and prestige, however, are not the results of high labor income.

It should be stressed, however, that a position does not bring power and prestige <u>because</u> it draws a high income. Rather, it draws a high income because it is functionally important and the available personnel is for one reason or another scarce. It is therefore superficial and erroneous to regard high income as the cause of a man's power and prestige, just as it is erroneous to think that a man's fever is the cause of his disease (Davis and Moore, 1963A:50).

The "purpose" of high Tabor income, therefore, is to "...induce people to compete for the position" <u>not</u> to "...confer prestige on a position..." (Davis and Moore, 1963A:51).

David and Moore state that most individuals in any society are capable (have the inherent or innate talent) of performing adequately in most occupations. However, there are <u>some</u> functionally important occupations which do require special talent which is scarce. Likewise, there are some functionally important occupations which, while the talent they require is not scarce, require costly, lengthy training which makes qualified individuals scarce. It is those functionally important occupations for which there is a scarcity of talented and/or trained personnel which must be the most highly rewarded. Differential rewards are used to recruit individuals to various occupations. The relative amount of rewards is subject to the laws of supply and demand. The natural scarcity of talent and skills, however, can be artificially manipulated to produce an over-supply or an under-supply which will, in turn, affect rewards.

... the efficiency of a stratified structure may be affected by the modes of recruitment for positions. The social order itself, however, sets limits to the inflation or deflation of the prestige of experts: an over-supply tends to debase the rewards and discourage recruitment or produce revolution, whereas an under-supply tends to increase the rewards or weaken the society in competition with other societies (Davis and Moore, 1963A:51).

The relationship between occupation and labor income, according to the functional theory of stratification, is one of functional causality based on the principle of supply and demand. Labor income, then, is a function of one's occupation. The technical-function theory
of education provides the technological functionalist explanation of the relationship education has with occupational attainment and labor income. It is asserted that because the knowledge and technical skill requirements for most occupations have been rising, especially since World War II, individuals must spend longer periods of time in school in order to acquire that knowledge and skill. Therefore, it is not the absolute amount of education that an individual has which ascertains what the labor income from a particular occupation will be, but rather the amount of education one has relative to everyone else. That is, when most individuals finished secondary school but only a small percentage went on to higher education, those who went on to higher education had greater chances of employment in an occupation which paid a high lappr income.

In general, individuals with relatively high labor income have more education than those with relatively low labor income. It is assumed that special cognitive and technical skills required for job adequacy are learned in school. In accordance with the functional theory of stratification, then, those functionally important occupations requiring costly, lengthy training and/or scarce talent will be the occupations which confer the highest labor income. Thus, one of the primary roles of schooling is to allocate individuals differentially into various occupations in the hierarchical division of labor on the basis of differential achievement and to a lesser extent personality attributes.

Equality of Educational Opportunity and the Technocratic-Meritocratic

С.

The technological functionalist (technocratic) view of production

according to which a hierarchical division of labor or ranking of occupations -- in other words, a stratified society -- is universally necessary and inevitable, ties in with what Bowles and Gintis (1976B:105) call the "meritocratic view of hiring". The American Dream of a meritocracy in which everyone, regardless of class or region, would have equal opportunity to develop their innate talents to their fullest, which dates back to the times of Thomas Jefferson, gained official status during the 1950's when the need for white collar workers, managers, technicians, engineers, and scientists was much greater than the supply. For the first time in the history of industrialized societies, there was full-scale social mobility. Many workers up-graded their education in order to benefit from the sudden explosion of employment opportunities. Since educational credentials were becoming the main criterion for employment, governments encouraged workers to do so. In the United States and Canada, for instance, the federal governments devised programs which would pay returned war veterans to pursue higher education.

A meritocratic system was politically popular because, being based on classical liberalism, it wought to replace an "ascriptive aristocracy" with a "natural aristocracy" or an achieved, allity elite. This classical liberalism is a type of Social Darwinism which says that everyone should be treated equally, regardless of class or region. Those who are innately more able and motivated will achieve and be more successful than those who are less well endowed. Compensatory programs for disadvantaged groups as well as a diversified curriculum were definitely not acceptable according to this outlook. The curriculum must be the same for all individuals through the period of compulsory schooling,

Michael Young (1958B) presents a schematic conceptualization of the meritocracy as M(eritocracy) = E(ffort) X A(bility). Thus, individuals would experience upward mobility or downward mobility as a function of their innate ability and motivation rather than being born into a particular status and class. In a meritocracy, social status and occupation would be achieved on the basis of merit, not ascribed characteristics. "Differential status and differential income are based on technical skills and higher education" (Bell, quoted in Karabel and Halsey, 1977B:607).

The Western European and North American countries embraced the technocratic-meritocratic ideology after the Second World War to ensure that talent necessary for manpower requirements would be identified and developed. The demand for white collar, technical, and professional workers far exceeded the supply. There was general dissatisfaction with the existing elitist system, which seemed to be responsible for a great deal of wastage of talent.

Talented individuals from the lower classes were not achieving in school. Hence, 'they could not compete for the new job opportunities. Since World War I, there have been many studies done in Western Europe and North America to ascertain the cause of this wastage of talent. Lack of equal educational opportunity for all groups in society was identified as the major reason for this wastage of talent. Studies of educational participation show a direct relationship between a student's social class origin and the amount and type of education received (Husén, 1975B; OECD, 1975Bb vols. 1 and 2; Boudon, 1973B; Fägerlind, 1975B; Glass, 1954B; Floud, 1956B; Halsey, et. al., 1980B; Goldthorpe, 1980B; Levin, 1976A; and Porter, et. al., 1973B).

With the added stimulus of the Cold War and its consequent competition between the Western capitalist countries and the Soviet Union, there was great concern that gross social inequality and wastage of talent due to elitist systems of education were not technically rational nor efficient. A meritocratic system with guaranteed equality of educational opportunity was deemed necessary to eliminate gross inequality and wastage of talent. Social justice and efficiency, then, would make more talent available to be utifized in catching up with Soviet education and technology.

The technocratic-meritocratic ideology which permeated the social sciences during the 1950's and early 1960's, emerged in the field of education as the liberal theory of equality of educational opportunity. This theory, which rapidly became more of a doctrine in education, like the idea of a meritocracy, stems from classical liberalism.

It is important to understand the relationship between meritocracy and equality of educational opportunity because both accept socio-economic inequality as inevitable and even just. Thus, neither seeks to eliminate this inequality. Rather, the basis of socio-economic inequality is shifted from ascribed traits to achieved ones based on individual merit. Bell (quoted in Karabel and Halsey, 1977B:616) describes the concept of equality of opportunity in general:

The principle of equality of opportunity derives from a fundamental tenet of classical liberalism: that the individual -- not the family, community, or the state -- is the singular unit of society, and that the purposes of societal arrangements

is to allow the individual the freedom to fulfill his own purposes -- by his labor to gain property, by exchange to satisfy his wants, by upward mobility to achieve a place commensurate with his talents...

As a principle, equality of opportunity denies the precedence of birth, of nepotism, of patronage, or any other than fair competition open equally to talent and ambition. It asserts, in the terms of Talcott Parsons, universalism over particularism, achievement over ascription. It is an ideal derived directly from the Enlightenment as codified by Kant, the principle of individual merit generalized as a categorical imperative...

ŝ

Technical skill becomes a condition of operative power, and higher education the means of obtaining technical skill. As a result, there has been a shift in the slope of power as, in key institutions, technical competence becomes the overriding consideration...it is the codification of a new social order based, in principle, on the priority of educated talent.

In social fact, the meritocracy is thus the displacement of one principle of stratification by another, of achievement for ascription.

CHAPTER IV

Footnotes

.

, ...

.

¹The debate between conservatives and liberals, however, is still not over -- witness the resurgence of conservatism after the publication of Jensen's work and the more recent lack of support for education in the 1970's and 1980's.

CHAPTER V

THE INFLUENCE OF THE CONCEPT OF EQUALITY OF EDUCATIONAL OPPORTUNITY ON EDUCATIONAL POLICIES

The doctrine of equality of educational opportunity became so well entrenched in popular opinion that it quickly became a national myth in the United States during the late 1950's and early 1960's. It became increasingly important in Canada as well as Western Europe. The main target for this in the United States was higher education, which was approaching the stage of mass attendance. This has been referred to as the "massification" of higher education. In Western Europe, the main target for equality of educational opportunity was secondary education, which prior to World War II was extremely elitist. Canada was mid-way between the United States and Western Europe, with problems at both the secondary and post-secondary levels of education. However, during the mid-1660's, it was realized that children from "disadvantaged" ethnic and racial minority groups and from the working class in general did not have the same opportunity for academic achievement as did those from the dominant and middle Therefore, concern for equality of educational opportunity class. became an issue in elementary and secondary schools throughout Western Europe and North America. It was the number one educational policy concern of the 1960's.

A. Revitalization of Classics! Liberalism in Education

In pursuing a policy of providing equality of educational opportunity, what policy-makers were ostensibly attempting to do was to eliminate all <u>external</u> barriers¹ to entrance at all levels of

schooling. According to this classical liberal philosophy, schooling up to the mandatory level should be uniform and the same for all students and no one should be excluded from free attendance. Furthermore, facilities should be the same between schools. Beyond mandatory schooling, access to further schooling must be limited only by <u>internal</u> barriers (i.e., innate ability and motivation). This is what is referred to as equality of educational opportunity in the formal sense.

The roots of this educational philosophy based on classical liberalism goes back to the American and French Revolutions. The 19th century writings of Horace Mann, an American educator, provide a good example of how liberals sought to eliminate gross social inequality through schooling. Equalization of life chances was the main goal of the doctrine of equality of educational opportunity as conceived by liberals. Thus, equality of opportunity for social and economic rewards was to be achieved through a more egalitarian system of education. The school was to become the great equalizer.

This classical liberal educational philosophy was revitalized during the late 1950's by the general concern throughout the Western capitalist world (though it was most pronounced in the United States) that a more equitable (but not equal) and just social order was not only morally desirable but absolutely necessary for more efficient production, rapid technological advancement, and economic growth. Decreasing social inequality would also increase social harmony and stability -- which was deemed necessary during the height of ideological and technological competition characteristic of the Cold War. A more educated work force was also believed necessary to raise worker productivity, increase the rate of surplus value, and make a greater rate of exploitation seem the reverse because of greater absolute affluence for all.

Liberals and conservatives alike, being against the idea of direct income transfer from rich to poor, turned to educational reform to accomplish the task of redistributing personal income. Equality of educational opportunity would provide everyone with an equal chance to compete for a piece of the pie. This equal chance to "...develop to the fullest extent of his (her) abilities, and by eliminating priviledge in access to all openings, would mean that inequality and injustice will be abolished" (Patterson, 1978A:22).

Thus, education is viewed as a means of solving problems of gross socio-economic inequality and inefficiency (wastage of talent). The school would be an instrument of social reform -- an idea that goes back to the early 19th century. People, especially in North America, came to believe that education would change the structure of the social order so that it would be more open, create more mobility, and be more egalitarian (Patterson, 1978A:26).

The doctrine of equality of educational opportunity, however, does not seek to substantially change the existing structure of socioeconomic inequality found outside the schools. "...(I)t merely seeks to increase the circulation up and down the ladder of success or failure" (Patterson, 1978A:23). The basis of the doctrine is utilitarian in that, by equalizing life chances for everyone, the most competent and qualified will be selected and allocated to the occupations commanding the highest prestige, income, and status; while the least competent will be allocated to those occupations with inferior status,

income, and prestige (Ibid.:22).

The proponents of the ideology of equality of opportunity did, however, believe that the proportion of highly rewarded occupations would increase ever more rapidly than the supply of highly educated manpower and that there would be an accompanying drastic reduction in the proportion of unskilled occupations so that there would be a natural redistribution of income resulting in a more egalitarian or less grossly unequal society.

How would this equality of educational opportunity be achieved? Through the relatively new social science of psychology, scientific, objective methods were to be devised and used to measure talent and ability. Thus, the criteria for academic achievement would be ability and motivation as measured by school grades, scores on "objective" tests, I.Q. and other standardized tests, and examinations. Educational w counsellors would be important for conducting I.Q. and other standardized tests in order to identify potential talent and channel it into the "educational manpower pool" (Husén, 1975B:74).

Access to formal education beyond mandatory schooling was to be democratized and entrance based on objective and fair measurement of merit (ability and motivation): There was also a move towards universal accessibility to higher education. Thus, success or failure in school would only be the result of individual innate ability and motivation. This has been referred to as the individualizat. The failure (or success).

According to technological functionalists, one of the primary functions of schooling is to allocate individuals differentially into various occupations in the hierarchical division of labor on the basis of differential school achievement and to a lesser extent on the basis of personality attributes. This allocation is done ostensibly on the basis of differences in achievement and merit -- which are a function of differences in innate ability and motivation only -- and appears to be objective and fair, especially if the idea of equality of educational opportunity for all is accepted by most members of society as the official policy.

This technocratic-meritocratic ideology is the classic liberal response to the traditional conservatives who maintain that social inequality is the natural and inevitable outcome of socially significant, genetically determined differences in intelligence between individuals (manifested in rigid class lines). In other words, conservatives believe that everyone gets what they deserve: there are genetic differences between classes. Liberals, however, do not believe that everyone gets what they deserve because of social structural obstacles to the development of talent. They do not believe that talent is class-bound and so they want all <u>external</u> (socio-economic) barriers to the development of talent eliminated.

This conservative-liberal clash was very pronounced during the late 1950's when it became evident that Soviet education and technology were much more advanced than Western capitalist countries. A debate ensued between advocates of elitist and egalitarian (mass) education. The elitists, like Vice Admiral Rickover, wanted a "...more rigidly administered system of ability grouping" (Sexton, 1961B:XIV) so that the gifted could progress as quickly as possible and not be delayed or discouraged by being in heterogeneous (undifferentiated) classes. Elitists felt that only the more advantaged students should receive

more education. Proponents of egalitarian education felt that the harmful effects of elitist educational philosophy (perpetuation of social inequality) more than offset the private gains to individuals so that education, especially higher education, should be viewed as a force for reducing income inequality by providing a means for social mobility for all qualified people, regardless of class.

The liberal cause was helped a great deal by the 1957 publication of a study on Soviet education by the United States Office of Education which stated that Soviet education was more productive than American education and that the Soviets did not separate the gifted from other students. I.Q. tests were not used in the Soviet Union because the Soviets claimed that intelligence is a product more of environment than genetics. As Patricia Cayo Sexton (1961B:XVIII) stated, "...the worker's child in the U.S.S.R. has <u>twice</u> as good a chance of going to college as his U.S. counterpart (<u>Harvard Education</u> Review, Spring 1957)..."

Thus, liberals were saying that

an industrial society, if it is to grow at top speed, as ours must, should call on the full intellectual potential of all the people in that society -- not just those in Rickover's ability elice. It must, in other words, engage in allout mass education, with full equality of opportunity extended to all students. This our nation [the U.S.] has failed to do (Sexton, 1961B:XVIII-XIX).

Sexton (1961B:XIX) goes on to state that "(t)he knowledge, understanding, and participation required to reach democratic goals and efficiency, and the skills required for peak industrial performance, remain in very large part latent and underdeveloped." In her call for a new approach to education, a discarding of elitist education, Sexton (<u>Ibid</u>.:XX) sums up the liberal support of egalitarian education in stating that

...perhaps then we can mount an educational assault which will be democratic and at the same time effectively competitive with Soviet authoritarian education. Through mass education, perhaps we can break down some of the walls of class and status...which seem totally out of place in a democratic egalitarian society such as we want ours to be. With the removal of these barriers, perhaps we can make a closer approach to the greatest of all democratic dreams -- equal educational opportunity for all, without regard to race, religion, class, or status.

This is the Dream.

In the United States, the primary focus of educational policies reform during the late 1950's and early 1960's was to provide equality of educational opportunity at the level of higher education and to build strong graduate schools (Clark, 1976Aa:123). Having already achieved mass, universal secondary education, the United States began pursuing a policy of democratizing higher education and moving towards universal, mass higher education. A college education was becoming a prerequisite for middle level office and technical workers. Physics, math, and engineering programs were expanded and promoted in an effort to "catch up" with the Soviets. Consequently, in 1958 with the passing of the National Defense Education Act (NDEA), the United States launched a program of massive financial support for higher education to ensure that talented individuals, especially those from "disadvantaged" backgrounds, would be able to pursue higher education.

In the 1961 presidential address to the American Sociological Association, Robert Faris (1961A:839) stated that institutions of

higher education have become

a potent instrument for raising the ability level of the population...What is happening... is that the nation is quietly lifting itself by the bootstraps to an importantly higher level of general ability -- an achievement which, though less dramatic than a space voyage to the moon and less measurable than the Gross National Product, may mean more to the national future than either.

Thus, there are three separate sets of phenomena which created a virtual explosion in mass demand for higher education in North America after World War II, ultimately leading to the dramatic expansion in post-secondary education witnessed during the 1960's. One is the shift in the occupational structure, creating demand for professional and paraprofessional workers. This resulted in massive up-grading of occupations and inflation in educational credentials. This was further complicated by the educational benefit schemes extended to returning war veterans in Canada and the United States. Many attended universities, obtained jobs with high salaries and status, and served as models for young people -- reinforcing their mobility, aspirations and belief in education as the vehicle for social mobility.

B. Baby Boom - The Demographic Factor

The second factor in explaining the expansion of education during the late 1950's is the post-World War II baby boom. The sudden increase in population put pressure on the educational system to expand -build more schools and train more teachers. By the 1960's, when these youngsters were ready to attend university, there was a virtual explosion in the demand for teachers and professors.

C. Sputnik Syndrome - The Ideological Factor

The third factor is the "Sputnik syndrome" — the competition with the Soviet Union during the Cold War era, as already discussed.

Although education is a state responsibility in the United States, the Congress passed eight major federal aid programs for undergraduate college and university students between 1958 and 1965. Such measures certainly illustrate the urgency and importance attached to educational reform by the United States government. These programs provided indirect and/or direct payments to students, scholarships, fellowships, subsidies, loans, work-study jobs, and tax benefits in an effort to equalize access to higher education for all groups (Finn, Jr., 1976/77A).

The long-term policy for higher education in the United States, then, was to pash for universal, mass higher education during the 1960's. The fact that women, members of subordinate groups (especially blacks), and working class people were disproportionately underrepresented in higher education led to policies of positive discrimination (affirmative action) for such groups as well as additional "compensatory" programs. The United States government set "targets" or "goals" for ethnic groups' representation in higher education during the mid-1960's (Riesman and Grant, 1973A:310). Fixed requirements for undergraduate and graduate programs were more often than not dropped and more "relevant" courses and programs were added to the gost-secondary curriculum. Black Studies, Women's Studies, Independent Studies, and Ecological and Environmental Studies are examples of such new developments.

The two main educational policy objectives in Western Europe during the 1960's were to abolish structural parallelism and to achieve

universal mass secondary school attendance. The dual system of academically-oriented and vocationally-oriented secondary schools was considered detrimental to the academic achievement and subsequent social mobility of talented working class students. The trends towards one system of comprehensive secondary schools and universal mass attendance which started during the 1940's, became a political issue during the 1950's with great popular support. The enrollment growth rate for secondary schools in Western Europe rose more rapidly relative to the United States (which had already achieved universal, mass secondary school attendance) right into the 1970's (Husén, 1974Bb:84).

The two main obstacles to the achievement of equality of educational opportunity in Western Europe were the early selectivity and the lack of openness of the educational system. In most of Western Europe, students are channeled into different types of schools at about the age of eleven years. Once in a non-academic program, the students are unable to get into an academic one. The proportion of working class students in vocationally-oriented schools is high, whereas it is very low in the academically-oriented schools. Concern for the "wastage of talent" as well as social justice prompted governments to introduce policy changes aimed at reversing this loss of talent, making schools more efficient and just. Providing equality of educational opportunity estails eliminating as many of the conditions and structures which impede its realization as possible. In the words of Torsten Husén (1974Bb; 142):

The greater the selectivity which allegedly promotes academic excellence, the more the opportunities among low status pupils to move ahead in the system are reduced. The more strict the requirements of access, promotion, examination, etc., the more biased the system is against lower class pupils.

The abolition of the dual system and its replacement with comprehensive secondary schools was the goal. However, although many schools were converted to the comprehensive type, remnants of the dual system still linger. The changeover has not been complete and there is some support for the dual system which has not been overcome. As Husén (OECD, 1975Bb vol. 1:334) has remarked,

(t)he two systems have emerged from certain historical and social conditions; the compulsory, elementary school being a product mainly of the 19th century industrialization, and the academic secondary school with a long tradition going back to the Middle Ages preparing for the learned professions. This dualism is a product of a society that was almost entirely ascriptive in its allocation of social status. The selection and/or transfer at an early age to academic secondary education by and large determines the subsequent occupational career; decides whether it will fall within the blue collar or white collar bracket. The built-in flexibility is almost nil.

Thus, in Western Europe, the focus of educational policies aimed at providing equality of educational opportunity was on secondary education. The model used was the American high school. Consequently, by the mid-1960's, in attempting to promote equality of cognitive attainment by social class (OECD, 1975Bb vol. 1:437), secondary curricula were made more flexible and open so that no one would be precluded from academic ("mainstream") secondary education. A diversity of optional courses unrelated to one's academic performance were included in the curriculum (Ibid.:334-335).

Canada falls somewhere between Western Europe and the United States in terms of the development (its structure and content) of its school system. Like the United States, and unlike some Western European countries, education in Canada is not a federal responsibility; it

D

is under provincial authority. However, like the United States, the federal government became involved directly as well as indirectly in education during the late 1950's and 1960's.²

During the 1950's, the educational system in Canada was elitist and unsuited to meet the needs of the rapidly changing and expanding occupational structure. Consequently, Canadians had not been trained in sufficient numbers to fill professional, middle management, technician, and skilled worker occupations. Therefore, according to John Porter (1965B), there has been minimal social mobility for Canadians and the society has remained relatively conservative, class-bound, and elitist.

The demand for professional, technical, and other highly skilled manpower in Canada since World War II has been met by heavy immigration. According to Lawr and Gidney (1973B:239), between 1953 and 1963, just over 80,000 such workers immigrated to Canada, three-fifths of them being British. Thus, "(t)he dependence on external recruitment has created the filusion of adequacy" of the educational system. "It has also permitted the continuity of class-bound education as exemplified by the classical college system in Quebec and academic collegiate system in Ontario" (Porter. 1965B:166).

Porter (1965B:47) points out that

(u)nlike the United States and the United Kingdom, Canada had not by the 1960's moved towards the creation of fully democratized education, either in its formal educational systems or through upgrading schemes in industry. Hence, Canadians did not have the same opportunities for upward mobility, either at home or elsewhere, as did populations of these other countries.

In addition to the need for a "modern" educational system more suited to the manpower needs of a post-industrial society, Porter (1965B:49) lamented the lack of mobility opportunity for Canadians: The necessity in the 1950's of importing skills from abroad to meet the labor force increment in skilled occupations suggests that Canadian institutions -- particularly educational and industrial -- were not geared to provide mobility opportunities. International migrations which have come with industrialization have been processes of social mobility as well as movements of labor as a factor of production...Where Canadian immigration policy seeks skilled and professional workers as an alternative to educational reforms, mobility deprivation for Canadians continues.

Porter (1965B:168-169) cites social and psychological barriers to equal onportunity in Canadian education and society. These include:

economic of the indirect costs of education), family size, redioner ties in quality and quantity of education, and state the of religion on educational policies. In criticizing Canadian equcation, Porter (Ibid.:173) states that "...little has been done to remove the barriers imposed by social conditions on the individual's educational opportunity."

Technological functionalist ideas permeate Porter's critique of Canadian education. His main concern is that the educational system be radically restructured to meet the "needs" of the labor market in a "post-industrial" society, with the principle aim being elimination of wastage of talent. "(E)ducational policy could remove from "the social environment those conditions which smother ability" (Porter, 1965B:173).

Porter's work in the late 1950's and 1960's represents the modern liberal perspective in Canada. Porter (1965B:167) wanted a meritocratic model adopted in Canada, on the greater societal level as well as within the educational system, so that Canada can become

... a society in which the allocation of individuals to social tasks and access to educational resources is determined by ability. Thus two ends are served: the occupational structure will reflect a more rational allocation of ability; and individuals will have the greatest opportunity to develop their talents and make their contribution to the social good. Where those who survive the upper levels of the educational system are less able than many who drop out of it, the investment in educational plant is being wasted and the most valuable resource of human talent is being squandered. A. society which refuses to remove barriers to educational opportunity is falling short of the democratic ideal. The principle of equality and the principle of rational use of economic resources thus have a mutually reinforcing function... A system which does not provide equality of opportunity is also inefficient.

By the mid-1960's, Canada was in step with the United States and Western Europe in striving for equality of educational opportunity in the belief that it was a prerequisite for economic efficiency and social justice. Because it was recognized that there were in fact socio-economic and psychological barriers to equality of educational opportunity, much of the educational policy reforms were made at minimizing, if not eliminating, these obstacles.

It was also recognized that so long as socio-economic barriers to equality of educational opportunity existed "...we can speak of a class-determined educational system" and by extension, a class society (Porter, 1965B:172). Thus, because "(c)lass may be a major impediment to Canadian educational systems meeting their social function of supplying the needs of a diversified occupational structure," modern liberals view the elimination of class as an inevitable outcome of our evolution into the "post-industrial" society (Ibid.:179). Thus, the prevailing idea during the 1960's that various social institutions, but particularly education, were lagging behind -indeed were out of step -- with the changes in technology and the economy was part of the functionalist paradigm that dominated all the social sciences. The rationale for educational reform was to restore social equilibrium, which would at the same time make society more efficient (and so more competitive with the Soviets) as well as more socially just (and consequently more harmonius).

Porter (1965B:196) reflects these values and concerns when he reported that one of the conclusions of the Paulend study in Canada was "...that two-thirds of the students that went to university were less than brilliant and that one-fifth of those who were brilliant reached university." The fact that "...going to university is largely a result of class and family position, (and) those who receive this training are not always the most intelligent" was perhaps the focal point around which most of the debate and reform in education turned (Ibid.:195).

In concluding this section on changes in educational policies in Western Europe and North America during the 1960's, it is fitting that a Canadian sociologist, John Porter, sum up the dominant argument in the following quotation from The Vertical Mosaic (1965B:197-198):

...educational policy cannot be viewed apart from social policy in general, and...educational equality can probably not develop without corresponding advances in other areas of the social system. No society in the modern period can afford to ignore the ability which lies in the lower strata. Whatever may be said about average intelligence and social class the fact remains that in absolute numbers there is more of the highly intelligent in lower classes than in the higher.

20.00

85

If the principle of efficiency and equality are to be upheld, Canada must be prepared to put a great deal more money into educational research than it has up to the 1960's. Not only could accessibility to educational institutions be greatly enlarged, but efforts could be made to overcome those psychological barriers which cut so many young people off from both the material and the spiritual benefits of education. Without such policies intergenerational continuity of class will remain, mobility deprivation will continue, and external recruitment will still be required to meet the needs of a complex occupational structure.

One of the most important educational policy shifts of the late 1950's was the sudden emphasis on post-secondary non-university education in the United States and Canada. After 1960, community colleges sprang up across North America at a phenomenal rate. In 1960, there were 656 community colleges in the United States and 29 in Canada (Pincus, 1974A:19; Gayfer, 1978B:29), By 1971, there more than 1,100 community colleges in the United States and 119 member institutions of the Association of Canadian Community Colleges (Pincus, Op. cit.; Fleming, 1974B:86). Enrollments, likewise, skyrocketed -increasing by five-fold in the United States between 1958 and 1973 (525,000 to 2,917,000) and roughly four-fold in Canada between 1961 and 1973 (53,000 to 202,000) (Pincus, Op. cit.; OECD, 1975Bc).³ The increases have been extremely high in various regions, such as Alberta, where "... the Alberta College Commission estimates the growth in enrolment in college-level programmes from 1964/65 to 1969/70 to have been approximately 130 per cent" (Watson, 1972B:86).

Community college education is aimed at the 17-19 year old group as well as at adults who may or may not have completed high school. It is intermediate between secondary schools and the labor

market or between secondary schools and university, depending upon program of enrollment. The traditional 8-4-4 pattern of education was undergoing a conscious restructuring to 8-4-2-2 or 6-3-3-2-2.

Ľ,

Community colleges generally provide three kinds of programs: 1) <u>academic</u> - two year, university equivalent general education courses transferable to a university; 2) <u>technical</u> - two or three year vocational/ technical job-related programs leading directly to employment (primarily in middle level, paraprofessional technician jobs); and 3) <u>continuing</u> <u>education</u> short-term (one week to one year) manpower retraining programs to upgrade skills of unemployed and/or underemployed people; remedial programs to help students develop academic skills, in which they are deficient; and in general and recreational courses not necessarily leading to employment or a diploms; and finally, programs to enable students to complete high school equivalency.

Although there is "...great diversity in propose, program, student population, administrative structure, and philosophical base" between community colleges across Canada, we will generalize using the abovementioned descriptions (Campbell, 1971B: 3).⁴

Community colleges are administered by their own boards (whether appointed or elected) and supported directly through taxes and in some provinces also indirectly through school board grants. Size of community colleges varies from a few hundred statents to a few thousand, with optimum size being 2,000 to 5,000 students.

According to Pincus (1974A:19) there are five public goals of community colleges: 1) comprehensive curriculum; 2) open-door admissions policy; 3) convenient location; 4) an attempt to give students a second chance (provide equality of educational opportunity for post-

secondary education); and 5) a community orientation. Community colleges are much cheaper to attend than university, having much lower tuition fees and being close to home. They are smaller than universities, offering greater inter-personal contact, more highly structured courses and assignments, and more comprehensive aggressive counselling. The emphasis is on teaching, not research. Teachers are expected to work closely with their students.

itudents no attend community colleges tend to have lower in the school grades and lower standardized test scores than those who extend universities. They also come from a lower social class. A significant proportion of community college students have been out of high school for ten years, whereas most university freshmen attend directly from high school. About fifty per cent of community college students are part-time students, and about half of the full-time students work more than fifteen hours per week.

Community colleges have been promoted by politicians and educators as providing an increased opportunity for higher education for those students unable or unwilling to attend university. Thus, the expansion of community colleges is viewed as part of the process of democratization of higher education. Distinction must be made here between higher education and post-secondary education. Higher education refers to university there as post-secondary education refers to a diverse array of institutions and programs for non-baccalaureate study (Medsker and Tillery, 1971B:138). Community colleges offer a variety of nonuniversity programs and courses which are employment directed or for recreation, which, prior to 1960, have not been universally available for adults. In addition, community colleges provide increased op-

154

Û

portunity for members of "disadvantaged" groups to pursue higher education, thereby democratizing access.

The sudden growth in community college enrollment is dramatic indeed when compared to the growth rate of university enrollment. While enrollment in American community colleges rose five fold between 1958 and 1973, enrollment in all other institutions of higher education rose by two-and-one-half times. Furthermore, "(o). <u>If all</u> <u>freshmen in public institutions</u> of higher learning are currently enrolled in community colleges, and some predict this figure will rise to 70% in 1980 in the United States (Munroe, 1972; Chronicle, 1974 Carnegie Commission, 1971)" (Pincus, 1974A: 19).

In Canada, full-time undergraduate university enrollment increased two-and-one-half times between 1960/61 and 1972/73 (114,000 to 285,000), where s full-time community college enrollment rose four-fold from 1961/62 to 1972/73 (53,000 to 191,000) (OECD, 1976B:237; OECD, 1995Bc).

The gap between full-time enrollment at universities and community colleges was narrowing considerably by the early 1970's. Whereas the 1961 full-time university encollment was one-and-one-half times that of community colleges, by 1975, it was reduced to one-and-one-half times (OECD, 1975Bc).

Although the sudden expansion of post-secondary non-university education is a more important and ideologically more significant development than the expansion of universities in the 1960's, it was the latter on which public attention was focused.

This expansion in Canada was more of a revolutionary nature than the American one, in that the system of secondary and post-secondary education went from small, closed, and elitist to large and open.

The transformation of post-secondary education in Canada has had more dramatic after effects on the country than have the changes which have occurred in any of the other Western industrialized countries, including the United States.

To gain a better perspective of the changes in Canadian postsecondary education, as well as secondary education, the highlights of similar developments in the United States and Western Europe during the 1960's and 1970's will be outlined.

This can best be done by examining the attempts made to achieve equality of educational opportunity through successive phases of educational policy shifts in the 1960's and 1970's.

The first phase, which has already been discussed, some length is the position of the classical liberals. That is, there should be equality of educational resources, curricula, and treatment of in students with the same measured ability. Access for all children to local schools through the period of compulsory schooling should be guaranteed and free. There should be no differences in school facilities between schools. The belief was that universal accessibility to equal schools was all that was necessary to ensure equality of educational opportunity. This was the view of the late 1950's.

D. Modern Liberalism and Education

By the early 1960's, it had become commonly accepted by politicians, educators, and sociologists that there were children from various "disadvantaged" groups in society who started school with distinct handicaps regarding achievement vis-a-vis children from "advantaged" groups. Thus, in the race for academic achievement, some

children were not even up to the starting line when they began school. Many dropped out before completing the period of mandatory attendance. Of those who finished the period of compulsory schooling, many did not graduate nor continue their education. 157

The consequent establishment of pre-school educational programs with an emphasis on compensatory programs in elementary schools along with the acceptance of counselling programs signalled a shift in educational policy away from classical liberalism and towards the position of modern liberals. In fact, positive discrimination (or affirmative action) affough compensatory educational programs came to typify this period. The idea was to bring the children from "disadvantaged" groups up to the same level of achievement as those from "advantaged" groups when they start their schooling."

With this shift in philosophy and policy, the school was "...perceived to be a positive agent for social reform, combatting. patterns of social and economic inequality..." (Pike, 1978/79A: 22). Efforts to eliminate, or at least overcome, the psychological and socio-economic barriers to achievement described by Porter became the objective of this stage of educational policy.

The third phase of educational policy shifts in the pursuit of equality of educational opportunity is that of providing educational opportunities for adults. The concept of life-long education, or recurrent education, developed in the early 1970's. Optario and Alberta were the first to propose this policy shift in Canada with the publication of the Worth Report (Choice of Futures) in Alberta and the Wright Report (The Leathing Society) in Ontario in 1972 (Pike, 1978/79A:36). The thrust of this policy is to bring everyone, regardless of age, into the

 \mathfrak{O}

realm of public education.

The resulting trend to eliminate dead-end and terminal programs, and to establish open admission are part of this third phase, which is designed to provide people with a second chance at education. This has rejuvenated adult education at both the secondary and post-secondary levels. The continuing growth of community colleges and adult vocational programs is evidence of this. The success of Athabasca University in Alberta, the Open University in Britain, and adult high schools in Denmark are some well-known examples of this.

By the mid-1970's, the aim of educational policy was to provide equality of educational opportunity from the cradle to the grave, through the use of whatever programs deemed necessary and affordable. The difference is, at least theoretically, that if one does not achieve the first time around, the door is always open to try again. The emphasis was removed from more equal achievement across groups the first time around.

There is another aspect withis last shift in educational policy which has important ramifications for everyone. That is, the emphasis has been shifted from viewing education as an instrument for individual socio-economic reward to a commodity for individual consumption. Higher education is being promoted for the inherent cultural and personal enrichment value. Hence, students are being encouraged to attend post-secondary institutions but not to expect it to lead directly and immediately to employment.

Thus, for the first time in nearly twenty years, the public is being told, by educators as well as by politicians, that post-secondary education will not necessarily have an economic pay-off for the individual. The investment approach to education has been abandoned in favor of the view that education is for consumptive purposes.

E. Disillusionment with Modern Liberal Educational Policies

This latest stage in educational policy is the result of widespread disillusionment on the part of educational experts and the public with the apparent failure of social action programs designed to provide equality of educational opportunity for children from "disadvantaged" groups.

There has been a general reassessment of the approach taken by previous policy and a rejection of the view that the educational system is or can be a vehicle for social change (Pike, 1978/79A:30). Since most of the factors responsible for differential achievement between groups of children lie outside the educational system, changes must be made outside the schools which will allow for equality of educational opportunity for all groups.

20

----;-

The OECD (1976B:39) examiners remarked that "...even the most comprehensive program of in-school efforts has not alone sufficed to eliminate the many-sided forms of disadvantage and inequality of opportunity that children experience." Therefore, it became accepted in educational circles that the school could not provide programs to overcome the many external causan for failure to achieve in school. Family and neighborhood were recognized as the primary influences in socialisation.

The realization that inequalities of educational opportunity are based on prior inequalities of condition (income, power, prestige) marked a mejor shift in educational policy between the second and the third phase of policy development. The logic'al conclusion of this line of thinking is that if we really want to implement equality of educational opportunity, then inequalities of condition must be reduced through progressive income redistribution and other measures (Pike, 1978/79A:33; National Council of Welfare, 1975B).

The school alone cannot remedy the socio-economic and psychological barriers to equality of educational opportunity; and unless there are radical socio-economic changes reducing inequalities of condition, the school cannot hope to cope with school-related problems of children (such as reading).

This has led to the rise of conservatism in education and in government. If, the current argument goes, the educational system is severely limited in its ability to equalize educational opportunity through various sophisticated programs, then why spend a lot of money on such strategies, especially when the economy is contracting and experiencing high unemployment and inflation.

The "back to basics" movement is an example of such thinking. If all children are provided with quality education in the basics, the results will be better than the previous phase where there were a variety of programs aimed at different groups of children and a lot of options in the curriculum. Parents began to feel that children achieve through hearning basic skills and that was being lost in the plethora of "progressive" programs which many parents viewed as "frills".

The fact that education was becoming extremely expensive for the public purse and at the same time not producing any significant change in differential achievement between groups of children provided the basis for public support of cuts in expenditures on education. State March 1

فلمتما

With serious unemployment and/or underemployment, inflation, and a contracting economy, governments were faced with a generalized financial crisis. It could no longer afford to devote such large portions of the budget on education. Raising tuition fees for postsecondary education during the 1970's and 1980's is one reaction to this situation.

The change in emphasis of responsibility for achievement in school was shifted from the educational system back to the individual. This third phase of educational policy puts the onus₀ on individuals to "drop-in" to the educational system (whether full-time or parttime) if they have previously dropped-out.

Thus, if there are no institutional obstructions to continuing education, an individual has the opportunity to pursue an education at any point in her/his life. Education becomes a life-long process and one never looses the opportunity for it. Young people are not "written off" as failures by the institutions; the door is always open to return.

The individual takes control of and responsibility for her/his education and is given

...considerable choice in deciding when, where, and how he will seek to learn. Such choice has always been a hall-mark of the elite in various societies. Ontario and Alberta are significant in seriously proposing to make the traditional priviledge a universal opportunity and in regarding it as a social necessity (Molyneux, 1974A:119).

Although the practical impact of this latest stage in educational policy seems to be limited, it marks a change in direction of modes and channels of schooling and in the approach taken to

facilitate equality of educational opportunity. Despite the definite shift towards conservatism evident in this development, the modern liberal view is still able to struggle through. For despite existing inequalities of condition between groups, the basic rules for allocating rewards in society are based on merit and not inheritance. Thus, such rules, and the inequalities of rewards, are legitimized and seen as equitable. The belief is still there that every effort at providing equality of educational opportunity is being made.

The notion that education is the great equalizer, the modern liberal rationale for using the school as an agent of socio-economic reform, was put on the back-burner -- along with the human capital motion that education is an economically regenerative force with a financial pay-off to the individual and to society.

Along with these changes came the move to centralize and formalize education. This was true in the United States as well as in Canada. Education always has been fairly centralized in Europe. State and provincial governments in North America became more stringent in the allocation of funds to all levals of education since the early 1970's.

There was also a marked shift in emphasis from university to community colleges and technical institutes for manpower training and educational up-grading for employment. University graduation was increasingly de-emphasized as a means of credentials for white collar employment. Student loans and other financial assistance for O post-secondary education became more difficult to get while at the same time tuition fees and the cost of living sky-rocketed. The result was shrinking university enrollments in the 1970's and 1980's.

CHAPTER V

 (\cdot)

1

¹This includes class, sex, region, ethnicity, race, religion and anyother ascribed, external characteristic.

²The federal government in Canada, however, does have the sole responsibility for the education of armed forces personnel and their dependents as well as native peoples.

We find a great deal of variation in enrollment statistics. Some sources specify full-time and some do not (leaving one to assume that the figures include part-time students); some include both. Some sources specify that their enrollment figures are for public community colleges; others include both public and private community colleges; others specify all two year colfege programs; others exclude technical institutes; others exclude junior colleges (which have no vocational/ technical component); and worst of all, some do not specify what they mean at all! Gayfer reports a full-time student enrollment in community colleges in Canada for 1960/61 of 9,000 students and states that these colleges were mostly existing institutes of technology, whereas total post-secondary non-university full-time enrollment for that year (including hospital schools of nursing and teachers' colleges) was 49,300. It is rather confusing. Furthermore, the OECD 1975 External Examiners' Report on Educational Policy in Ganada, apported that 53,000 full-time and part-time students attended all appreciation on-university institutions in Canada during 1961/62. Hospital schools of nursing and teachers' colleges as well as specialized technical institutes and junior (academic, two year) colleges would all be included in the term postsecondary non-university as distinct from community colleges. Terminology and data reporting are often vague and confusing.

⁴Community college is meant to designate comprehensive two to three year colleges which have both academic and technical/vocational programs; however, it is often used very loosely in the literature, sometimes including purely technical institutes, junior (academic only) colleges, hospital schools of nursing, and teachers' colleges. Strictly speaking, these specialized institutions by themselves do not constitute a community college.

CHAPTER VI

CRITIQUE OF HUMAN CAPITAL THEORY AND THE TECHNICAL-FUNCTION THEORY OF EDUCATION

A. Principle of Marginal Productivity

Of all the possible criticisms levelled at human capital theory, there is one which both strikes at the heart of its economic analysis and invites sociological alternatives. The principle of marginal productivity which underlies human capital theory, indeed provides the foundation for neo-classical economics, is perhaps the key to its undoing.

Marginal productivity means that individuals are paid their marginal product. It is not necessary that each individual worker receive wages equal to his/her marginal product. However, workers as a group in a particular category of labor (such as plumbers) must be paid their marginal product even though all workers in that category will not necessarily be equally productive.

Thus, "(i)f labor is paid its marginal product, earnings reflect productive capacities. If labor is not paid its marginal product, earnings do not reflect productive capacities...[and].the concept of human capital loses its economic meaning" (Thurow, 1970B:17-18).

Denison (1962B:69) states that

(r)eliance, for broad groups, on the marginal productivity explanation of the distribution of income permits us to treat differentials in average earnings among these groups as a measure of differentials in the average contribution to production made by the individuals comprising them.

Economists assume that higher wages reflect greater productivity even when other factors are held constant. For example, Denison (1962A-1961)

writes that "...earnings differences between groups of males of similar age classified by education, are taken to represent differences in their contributions to production or quality."

Lester Thurow (1970B) wrote a book on human capital, even though he has no reason to believe that labor is paid its marginal product. Thurow (Ibid.:20-21) explains that

(t)here is practically no direct information on whether or not labor is paid its marginal product. Economists take it as an article of faith or else claim that it is the best null hypothesis, and economic theory is based on the assumption that labor is indeed paid its marginal product. Without this assumption, much of economic analysis falls apart. The theory of production certainly does. The convenience of the hypothesis for economists, however, does not make it correct.

Whether or not the segucture of wages in the United States, or anyother country, is determined by differences in productivity or by other factors, including non-economic ones -- a question usually ignored by economists -- has been an issue for sociologists and psychologists. Social stratification theories focus on the distribution of personal income and status. Innate intelligence and motivation as well as early socialization are importnat factors in the psychological theories.

Economists, as already mentioned, seem content to accept the principle of marginal productivity on faith, acknowledging that, although other factors may well be influencing the structure of wages, there still is a tendency for wages to move towards equilibrium with marginal productivity. Denison (OECD, 1964Bb:81) explains how he views the problem:

My approach, like all others, requires that there be some equilibrium distribution of resources toward which the actual distribution tends to move, even if slowly and imperfectly. If everything is simply random, not only my economics but all economics has nothing to offer...

My analysis of the contribution of education requires that, however income differentials are established, there is some tendency to hire employees with different amounts of education up to the point where their marginal product equals their salary. If this does not absorb everyone in some educational categories, either their salary comes down or some of them will be persistently unemployed. In the United States unemployment is greatest among the least educated, suggesting that the least, not the most, educated are relatively overpaid in. comparison with a salary structure that would employ everyone...

I might just add that the market for new college graduates is in fact a highly competitice one, with recruiters on the campus, and prices for graduates in different fields quite responsive to the laws of supply and demand. The bidding away of experienced individuals is commonplace.

B. Assumption of a Competitive Labor Market

Neo-classical economists tend to fall back on another neoclassical assumption -- that the labor market is competitive -- to reinforce the principle of marginal productivity. If wages do not reflect productivity, it is usually explained by economists in terms of labor market imperfections. These are usually seen as relatively small, exceptional cases which do not negative rule.

For example, the earnings of physicians are not true reflections of their productivity because entrance into the occupation as well as their fees for service are monopolistically controlled by the prac-
titioners themselves.

The opposite phenomenon also occurs, whereby monopsonies (firms that are single buyers of a particular factor) pay highly skilled employees less than their marginal product because of lack of effective competition in the market place and lack of bargaining power of the employees.

Most economists assume that "(o)n balance, the two might cancel each other. In any event, more internation is needed before any serious adjustments could be made" (Cohn, 1972B:155).

Thurow (1970B:20) states that, in addition, "(c)ustom, habit, and ideas of fairness may also determine wages. Consequently, industrial wage structures may have little fation to productivity."

If we reject the assumption the weat reflect productivity, we have effectively killed human capital theory. No longer do ear reflect their true contribution to national product and growth. Thus social rates of return to investments in human capital can not be

calculated. Individual earnings, and consequently national income, would not be dependent primarily upon human skills and knowledge. Productive capacity could no longer be measured by capitalized earnnings. As Thurow (19705:18) points out, "(i)f changes in capitalized earnings do not reflect changes in productivity, human capital cannot be used to explain the sources of economic growth." Furthermore, investment in human capital would have no real effect on the unequal distribution of personal income.

If we reject the principle of marginal productivity, we remove the normative basis legitimating inequality of rewards in society. Both human capital theory and the technical-function theory of education accept the meritocratic concept of society. Schultz and other human capital economists, however, insist that federal government intervention is necessary to tedress certain socio-economic and psychological barriers which prevent individuals from certain groups and classes from averting in and developing their human capital so that there is a group of opportunity for all. Thus, Berner (1974A:248) remarks that "...functionalist theories rejected by sociologists, have appeared through the back door of the economics of education." Imperfections of the labor, market include an array of factors, such as the influence of unions and collective bargaining on wages, unvillingness of the unemployed or uneversity of migrate to other regions with bettar employment opportunities, ick information about employment, skill labeling by employers (Leibenstein, 1965A, 1969A), lack of or inadequate Counselling services, lengthy period of

168

education, and high proportion of well-educated ampower employed in the public sector (civil service, military) -- usually not competi-

tive with the private sector, in any case, not profit-maximizing.

C. Quality of Education and the Affect of Sex, Race, Age, Class and Region on Rate of Return to Investments in Education

Schultz (1971B:196-201) has remarked that human capital theory appears to be "sex-specific" and so perhaps should be renamed male capital three. We might add ethnicity, class, region, and age to this -- since white, middle-class, urban, "mature" (middle-age) males

have a much higher return on their human capital than do non-whites, females, the young, the aged, the rural, and the poor.

Becker (1964B) bays that "traditional builts practices" may be

partly irrational and discriminatory. A lower rate of return to

5.

investment in education by the "disadvantaged" groups, however, "is just lied by Becker because of the supposed lower costs and quality of that education.

Becker (1975B:172-178) states that the direct and indirect costs of college are lower for non-whites than for whites because they attend cheaper, "lower quality" colleges and earn lower wages than whites. Therefore, it is "natural" that the fate of return to their investment in college education will be lower than that of whites women have lower direct and indirect costs as well, in compari-

son to men, according to Becker's Hence, women receive a lower rate of return to their in college education. This is further mystified by Becker's (1975B:199) remark that ", women's to college partly to increase the probability of marrying I more desirable man." Somehow we are supposed to come to the "logical conclusion" that women marry men but men retain their high rate of return to college education whereas women do not. In Becker's (Ibid.) own words,

> ... the gain to women from additional schooling should be determined by family earnings classified by the wife edducation rather than by personal earning classified, and the full money gain to women may be much higher than previous estimates have indicated.

Lower rates of return to college education of rural persons is also attributed to lower direct and indirect costs and lower quality education. Rural students earn more while in college than urban students, so their foregone earnings are lower. However, fewer rural persons attend university apparently because the lower rate of return is a more forceful negative factor than is the positive lower cost (Becker, 1975B:180-181).

\$¥.

Schultz emphasizes the inferior quality of the education received by most non-whites and rural persons, asowell as many females. Therefore, improvement in quality of education should raise the rates of return. However, the importance of quality of stems from the causality which human capital economists attribute to the relationship between education and productivity and between productivity and earnings. Schultz sees a definite causal relationship between poor quality education, low worker productivity, low income, and lack of opportunity. The poor, the rural, and non-whites are generally lacking the human capital (technical skills and knowledge) necessary to raise their groductivity and income, according to Schultz. Such people must be willing to migrate to regions where Suployment opportunities are greater, and to where educational opportunities are better. Schultz considers poverty as detrimental both to the individual and to mociety. He quotes the same line from Faulkner's Intruder in the Dust in many of his books and articles when discussing poverty.

That line is, in Schultz's (1971B:47) paraphrase: "The man without skills and knowledge leaning terrifically against nothing." Schultz (<u>Ibid</u>.) consistently stresses that "...the most distinctive feature of our economic system is the growth in human capital. Without it there would only be hard, manual work and poverty, except for those who have income from property.".

D. Eradication of Poverty through Investment in Education

Instanting the quality of education at the pre-school, elementary,

at those levels for children from "disadvantaged" groups has been one of Schultz's policy recommendations to the United States and other governments as a key element in the improvement of human capital and the eradication of poverty. Such improvement in quality of education would be a major step towards the provision of equality of educational opportunity which most liberals, including human capital economists, consider to be the most significant way to establish equality of opportunity in society. Thus, Schultz has consistently decried the umphasis put on <u>quantity</u> of education and drawn attention to <u>quality</u>.

... the condition of the stock of educational capital is adversely affected by an overemphasis on quantity of schooling relative to the emphasis given to quality... there has been a marked reduction in the difference and days of school attendance on the part of pupils in different parts of the United States. But the differences in the quality of schooling are great, and they are the heart of one of our most serious problems, expecially so in elementary schooling. It is in terms of quality that many rural children and many children from the homes of nonwhite families are at a marked disadvantage (Schultz, 1972A: 37).

In a study done to quantify the contribution of education to the War on Poverty in the United States, Thomas I. Ribich (1968B:60) states that "...more intensive education at an earlier age results in more and better learning, dollar for dollar, than does prolongation of education." Even though Bibich (Ibid.:101) concluded that "...the results of his study raise serious questions about the payoffs in investment in education as an antipoverty weapon", he does support. Schultz's matenaich, there is a serious need for a significant ation is thus an efficient use of educational expenditures.

E. Inequalities in Quality of Education and Inequalities in the Distribution of Personal Income

Schultz relates inequalities in the quality of schooling to inequalities in the distribution of personal income. If inequalities in the quality of schooling can be overcome, it will sufficiently affect the distribution of personal income and contribute directly to equality of opportunity. In Schultz's (1974:36) own words:

No small part of the inequality in the distribution of schooling srises from the inequality in the distribution of personal income. Children of poor people acquire not Inly somewhat less chooling but, what is much more important, the schooling they obtain is, as a rule, much lower in quality than that acquired by children of families (in communities) with high incomes. There have been some useful reforms in Lobal and some finance. Bringing the capital market into play to provide the necessary additional resources ire people are . poor is not solving this problem. Taxexempt school bonds are not the solution. The fact of the matter is that schooling is neither free nor equal. The two common expressions "free public schooling" and "equality of educational opportunity" are in this context empty phrases. Schooling is inescapably an expensive enterprise, privately and publically. The term "opportunities" is most ambiguous. Nevertheless, as already noted, the policy implications of the inequality in schooling that is associated with the inequality in the distribution of personal income are strong and clear. Relatively more of the investment resources entering into education should be allocated purely on economic efficiency grounds in favor of the children from poor homes:

of Bligrs have been spent on compensatory educational programs for

children from "disadvantaged" (low income) groups at the pre-school,

elementary, and secondary school levels — more than \$10 billion from 1965 to 1974 (Gordon, 1974Aa:309). The Elementary and Secondary Education Act, passed in 1965, provided nearly \$1 billion for such programs that year. The amount spent reached a high of \$1.5 billion prior to 1972, when it was \$1.4 billion (at which time it involved 14,200 school districts enrolling 7.3 million of the 8.1 million qualifying children) (Plotnick and Skidmore, 1975B:22).

Operation Head Start was designed to narrow the gap in cognitive achievement between four and five year olds from "disadvantaged" (low income) groups and "average" middle-class children. School lunches and physical examinations were also provided in an attempt to improve the nutrition and health of the children in the program.

There has been a change of emphasis from higher education to pre-school and elementary education in most poor countries since the early 1970's. This is in reocanition of the fact that human capital development requires a certain minimum quality and quantity input in the pre-School years in order for later development to occur. In the peor countries the problem of malnutrition alone is monumental. Thus, development of human capital in those countries requires earlier and more comprehensive intervention than in the rich countries. International aid agencies, including the World Bank, have been involved in this policy change.

In Weitern Europe, emphasis has tended to be at the secondary level, in providing common education -- or st, least common facilities -for children eleven years and older, through the years of compulsory attendance. The establishment of comprehensive secondary schools (the secondary modern school) is the product of these efforts. Streaming, more flexible curricula, and various special roomans were established to improve the educational achievement of working class children. The ultimate goal of this was to enable and to encourage celented" children from the working class to achieve in school and to continue on in post-secondary education.

Despite the tremendous in expenditures on these educational programs around the world, there seems to be consensus that the goals have not been met.¹ As to why this has been the case, there is no consensus. The apparent failure of educational programs premised on human capital economics and political liberalism has contributed to the resurgence of neo-conservatism and the growing belief by members of the advantaged groups that poverty and unemployment are due to innate attributes of the individuals and consequently immune to any social economic, or political intervention.

American from discoventaged groups have been simed at either changing of children from discoventaged groups have been simed at either changing attributes of the individual students (such as language, conceptualization, attitudes, study habits, eating habits, etc.,) and/or providing more resources and better facilities (or their equivalent) to the schools which they attend.² These programs are, within broad parameters, attempting to improve the quality of education which children from disadvantaged groups receive.

Assumption of a Continually Growing Economy, Low Unemployment, and an Infinitely Elastic Depand for Highly-Educated Norkers

Hunden cepital theory also assumes a continually growing, expanding sconomy with nearly full employment. The capitalist economies of Western Europe and North America were booming during most of the 1950's and 1960's.

Since World War II, the economics of all the Western industrialized capitalist countries have been expanding rapidly, with increased automation, large increases in the growth of the public sector, increases in the labor force (baby boom), and relatively low unemption t. Investment in labor point has also grown considerably.³

This has not been the case in the 1970's and 1980's. Since the late 1960's, there has been world-wide inflation and depression, with high and rising unemployment. Gonick (1978B) Tefers to this as stagflation -- a concept completely unimaginable let alone explainable in contemporary liberalism. The result has been a general backlash against liberalism and a noticeable resurgence of neo-conservatism. "While hiberal economics has reached its limits, its collapse has sparked the rebirth of a new conservatism" (Ibid.:148).

This shift has occurred on goth the governmental level in terms of public policies as well as on the grass roots level in terms of public opinion.

The capitalist economy world-wide experienced the worst recession since the Dirty Thinging by 1974/75 (Gonick, 1978B:108). The downswing began in 1968; when isfilition was 4.0% and the unemployment fite was 4.8% in Canada (Ibid.:114). The peak of economic prosperity was 1966, when the unemployment rate in Canada was 3.6% and the inflation rate was rising by 2% annually (early 1960's) to 4% annually (IBid.:115).

The 1969 recession was aborted by government spending. The economy was kept afloat with an increase in the money supply and runaway inflation appeared by 1971.⁴ Stagflation was firmy established throughout the Western capitalist world. Thus, "(1)ike the Great Depression, this crisis is universal" (Gonick, 1978B:146).

cut b ack expenditures on social ervices — unemployment insurance, medical care, education, etc. Restraint and austerity were the new catch words. Retrenchment "...and to a degree, repression" were the new reality of social policy (Gonick, 1978B:126). The fiscal crisis of the state reached fruition.

In Canada, Gonick (1978B:120) reports that in the June 23, 1975 budget, "...planned public expenditure growth was cut by a billion dollars, and there were cutbacks on programs for medical care and unemployment insurance." The resulting reduction in public sector employment, added to the rising unemployment in the private sector. Unemployment in Canada rose from 5.6% in 1973 to 8.5% in 1978 (Ibid.:129) Consumer proves in Canada were rising at 12.5% annually by Decemperator 174 (Ibid.:114).

G. Sanctity of the Welfare State and Government Regulation of the Economy

The post-World War II era, from 1945 to the 1970's, can be characterized as one of pervasive government intervention and regulation of the economy and the stablightent of the welfare state,

This acceptance of Keynesian economics and liberalism signifies public commitment by governments to "...resolve ecomonic and social problems through government action..." (Gonick, 19785:87).

growth and prosperity, when unemployment and inflation are relatively

low and the tax base is increasing. However, programs to combat "social and sconsmic problems cannot expand faster than the tax base without "...impinging on the high incomes of the affluent classes" (Gonick, 1978B:87).

The belief that "big government" is responsible for stagflation has become widespread and there has been sustained pressure from business and the public to return to "free enterprise". De-regulation of the economy, drastic reduction in government expenditures (especially on public sector jobs and pocial programs and services), rejection of welfare liberalism, and a return to emphasis on responsibility of the individual characteriantian mergence of neo-conservatism witnessed throughout the Western capitalist world since the mid-1970's. It is a response to the near collariant the capitalist economy and complete disillusionment with welfare interpretation (Gonick, 1978B:88-89).

. Underemplement or Unemploy the the ly-Educated .

Without getting into a direction of whether or not neo-conservative policies will prevail in the 1980's, it is a reasonable assumption that unemployment and underemployment will continue to remain high, as will inflation. The growth rates of Western capitalist countries alowed dramatically in the 1970's and where is no reason to believe they will increased in the foreseeable future.

Employers will make stained effort to keep wages down and increase their profits, as they have since 1975.

The economic climate is not buoyant enough to absorb the growing numbers of highly educated workers. Lockhart. (1971A:253) reports that Canadian universities increased the number of Ph.D. 'a

graduating in 1959 from 200 by a factor of six by 1969 but the over-all demand for them barely doubled, and in fact had declined to no increase by the end of the decade.

Likewise, the increased demand for scientific manpower in industrial research establishments in Canada between 1968 and 1970 was actually less than one-sixth of that anticipated. Although the Science Council of Canada expected a decrease in demand for scientists and engineers after 1970, there would be a 12% annual increase in the Ph.D. pool to at least 1975 even if no new doctoral candidates were admitted. The University of Toronto cut graduate admissions by 25% in 1970. Lockhart (1971A:253) remarks that

...even this belated response to the supply/demand. logic assures that on the basis of the most optimistic demand projections, there will be two to three new Ph.D.s for every available job over the next decade, at least.

The rush to close the gammin. education with the United States was so intense in Canada during the 1960's that by 1970, the proportion of graduate engineers and scientists in the labor force was 1.9% in Canada, compared to 1.5% in the United States (Lockhart, 1975A: 198).

Lockhart (1979A:229) refers to the 1970's as the decade of "...economic recession and unmet.occupational expectations." As with the economy, the crisis of unemployed highly educated manpower

was world-wide,

Barbara Frum (1971A:20) remarked that

(s)uddenly there is so libtle demand, for university graduates that a frightening and frightened segment of the Class of '71 can't find even the mundane opportunities that in years plat, they might have spaced at. No one knows, how many of 75,000 graduates in the class of '71 have been unable to find the kind of work they have been 6

وهيزا

I trained for, but it could be as many as half...The Class of '71 is the victim of what is coming to be known as The Great Training Robbery.

Contrary so human capital theory, the unemployment rates went up with educational qualifications. A study of 1974 graduates from Ontario's post-secondary institutions indicates that four months after labor force entry, the unemployment rate of university graduates was 11.9%, and 9% for community college graduates, whereas the over-all unemployment rate was 3.7% in Ontario (Lockhart, 1979A:231).

Lockhart (1975A:199) points out that whereas the supply of university maduates increased 10% to 15% annually in Canada⁶ during the late 1960's and early 1970's, there was also a decline in demand. "The resulting gap between supply and demand may thus be as high as one-thirm the annual production, perhaps more if underemployment is considered" (Ibid.):

Freeman (1975A:287) reports that in the United States,

.(t)he college job market underwent an unprecedented downturn in the 1970's when the earnings of graduates relative to other workers, the rate of return to investigg in higher education, and employment opportunities dropped sharply, especially for new graduates.

Freeman (1975A:291) also reports that

(a) ccording to a [U.S.] Bureau of Labor Statistics, survey, a majority of persons in the classes of 1970-71 had earnings below their expectations, with 33 percent of male bachelor traduates earning' "substantially lower" than expected, compared to 31 percent receiving "about the same" income as expected and just 12 percent "higher income".

There was a "... marked slowdown in the rate of growth of pro-

fessional and managerial jobs in the 1969-1974 period..." which suggests "...the possibility of a major market adjustment problem in the near

54

179

Κ,

future due to the greater, increase in the supply of college graduates than in the number of high-level jobs" (Freeman, 1975A:291).⁸

<u>Time</u> reported on May 24, 1971 that "(a) survey of 140 U.S. colleges and universities indicated that between March 1970 and March 1971, job bids for male B.A.'s dropped 61%, and a staggering 78% for Ph.D.'s" (Daniere, 1973A: 147).

Daniere (1973A:147) states that over-investment in education also resulted in a misallocation of the available labor force. Underemployment of university graduates became generalized in the early 1970's. Underemployment of these graduates leads to inefficient allocation in the labor market and to the phenomenon of job-up-grading.

Job-upgrading is the process by which the "annual excess of educated (or trained) members of the labor force over requirements in terms of previous-year qualifications is "wiped out...the excess supply at any level of educational achievement is hired in preference to individuals of slightly lesser qualifications in jobs for which the latter had previously qualified (<u>Ibid</u>.).

When a majority of employees have the new level of education, it becomes a requirement for the job. The job becomes "up-graded" even though the content of the job has not changed. Job-up-grading seems to be the near classical solution to increasing credentialism and rising levels of education.

The sharp drop in demand for engineers in the United States resulted in 25% fewer Ph.D. is darmed during the 1970's than during the 1960's (<u>Edmonton Journal</u>, Sept. 18, 1989). The drop in supply is so severe that American universities may have to lower their standards and hire faculty without doctorates (Ibid.). Demand for most college and graduate degrees, however, is continuing to worsen. <u>The Washington Post</u> cites the prediction that "...by 1985, 2.5 American college graduates will be competing for every 'college' job" (<u>Edmonton Journal</u>, June 5, 1980).

In stark contrast to the 1960's, teaching jobs in post-secondary institutions across North America are in high demand and low supply during the 1970's. The growth rate of full-time faculty in the United States from 1972 to 1977 was 9% (<u>New York Times</u>, quoted in <u>Edmonton Journal</u>, March 20, 1980).

There has been a marked increase in the proportion of part-time faculty members in post-secondary institutions, amounting to 50% rate of growth between 1972 and 1977 in the United States (<u>New York Times</u>, quoted in <u>Edmonton Journal</u>, March 20, 1980). Community colleges employ 51% of their faculty members part-time, compared to 24% for four year colleges (<u>Ibid</u>.). According to the U.S. National Center for Educational Statistics, 32% of the total higher education teaching force of 675,000 are part-time faculty members (Ibid_).

Cantor and Roberts (1979B:1) cite economic recession and sharply declining birth /rates⁹ as two main factors causing the great drop in demand for teachers during the 1970's in England and Wales. As a result, the number of teacher-training places in colleges of education were cut back four times between 1974 and 1977, from 75,000 to 85,000 proposed places in 1977 to 45,000 (Ibid.:89).

Furthermore, the colleges of education themselves, which numbered 155 in 1972, have been reduced -- more than 60 merged with polytechnics and other institutions, at teast 14 have closed, and the rest have been diversified. The change has been so drastic that "...in 1981 not one

of the former colleges of education will be offering the same courses it provided in 1972" (Cantor and Roberts, 1979B:89).

The post-industrial society view that demand for highly educated manpower is infinitely elastic is simply not true for the 1970's. There is no reason to believe the situation will improve in the 1980's. Thus, another basic assumption of both human capital theory and the technical-function theory of education is invalid.

I. Manpower Training Programs

Another important assumption of human apital theory is that inequality in the distribution of personal income can be drastically reduced and poverty eliminated through education, training, and relocation of workers. Income is viewed as a function of productivity which is a function of education and training in both human capital theory and the technical-function theory of education.

Consequently, Canada and the United States established federal manpower training programs in co-operation with the provinces and states during the 1960's to up-grade and/or retrain workers. The problem of poverty was considered serious enough in both countries for the federal governments to intervene directly.

It was generally believed that the social benefits far exceed the social costs of education and training and so huge public expenditures were justified. The creation of a federal Department of Manpower and Immigration in Canada reflected the felt needs for manpower planning and the development of manpower policies at the federal and provincial levels in departments of education as well as labor. Labor market information gathering and dissemination also became an .182

important function of this federal department.

1

The Economic Council of Canada (1971B:98) states the primary w objective of manpower planning to be growth and the secondary objective to be equity and stabilization.¹⁰

Canadian manpower policy since 1966 has been concerned primarily with adults, has been much more comprehensive, and the federal government as played a more dominant role in planning than previously. The Minister of Manpower and Immigration, Allan J. MacEachen, before the House of Commons Standing Committee on Labour, Manpower, and Immigration in 1970, defined

(t)he main objective of the Department of Manpower and Immigration is to further the economic growth of Canada by endeavouring to ensure that the supply of manpower matches the demand qualitatively, quantitatively, and geographically (Economic Council of Canada, 1971B:96).

Similarly, in a paper presented by the Department of Manpower and Immigration for OECD in July 1968, the following statement of policy appeared:

The general aim of Canadian manpower policy is to encourage the effective allocation of manpower resources and the development of the labour force supply and characteristics compatible with the maximum sustainable rate of growth in real per capita income (Economic Council of Canada, 1971B:96).

Canadian federal government involvement in manpower training began with the Technical and Vocational Training Assistance Act of 1960, under which the federal and provincial governments shared the costs of programs developed to cover the training of the following three groups: 1) technical and vocational high school students; 2) youths and adults requiring post-secondary education to become technicians; and 3) adults (employed and unemployed) requiring training for employment or improved employment opportunities (Economic Council of Canada, 1971B:100). Expenditures by the federal government on these programs between 1960 and 1967 totalled \$900 million.

Programs since 1967 are not shared-cost. The Adult Occupational Training Act of 1967 provides for federal long-term loans and capital grants for adults. Living allowances are also included.

The Department of Manpower and Immigration buys training services from private schools, technical institutes, and private industry after consulting with the provinces. Individuals enrolled in apprenticeships and any adults who have been out of school for one year and are considered capable of benefitting from the training (in terms of improved earnings prospects) are eligible for the adult training programs.

Part of the reason for the use of institutional training is because so many technical and vocational institutions were built between 1960 and 1966 under the previous federal policy.¹¹ Another reason is that industry is concentrated in a few places in Canada so that workers outside those areas cannot receive training-in-industry in their own area. An additional reason for the low participation rate with industry in manpower training in Canada is the great slowdown in the economy. Unemployment post-1967 is considerably higher than previously and employers have no incentive to participate in such training programs (Economic Council of Canada, 1971B:107-108).

The trend in Canada is towards adult training programs being another form of social assistance and economic as well as social stabilization in times of severe economic contraction.¹² Certainly the previously stated goals of manpower forecasting and planning for

long-run education and economic development have been abandoned. Such planning is now viewed as hazardous, uncertain, theoretically weak, and politically unfeasible. That view has become more common in the international development and aid agencies as well, though short-run planning is still utilized by such agencies as well as governments in response to technological change, social pressures, and changing tastes (McLeod, 1973B:211).

The strong emphasis in Canadian manpower policies on growth, especially since 1969, is in sharp contrast to American manpower policies which promote programs oriented primarily to help "disadvantaged" groups (Economic Council of Canada, 1971B:98).

American manpower policy stresses equity. The underlying assumption is that "average" workers probably function adequately in the private market but the "disadvantaged" workers will not without federal government intervention on their behalf.

In a message to Congress on August 12, 1969, President Richard M. Nixon`stated that

(m)anpower training means: (1) making it possible for those who are unemployed or on the fringes of the labor force to become permanent, full-time workers; (2) giving those who are now employed at low income the training and the opportunity they need to become more productive and successful; (3) discovering the potential in those people who are now considered unemployable, removing many of the barriers now blocking their way (Economic Council of Canada, 1971B:98).

Even though experts agree that training in industry is preferable to institutional training, there is an almost exclusive reliance on the latter in Canada. The Economic Council of Canada (1971B:104) reports that "(1)ess than 5% of total Canadian federal adult occupational training expenditures (excluding apprenticeships) is directed to training-in-industry."

Canadian Manpower Training Program funds allocated to inindustry training increased from 1% of CMPT funds in 1967-68 to nearly 10% in 1974-75. However, this still falls far below the proportions spent on in-industry training in the United States and most European countries (Goldman, 1976B:38).

In 1972, the Adult Occupational Training Act of 1967 was amended to allow for federal funding of on-the-job training under two new programs. One was to provide for on-the-job training as well as for the creation of new jobs for disadvantaged workers. The other was to train unemployed and underemployed workers for jobs which are vacant due to a shortage of qualified workers. These programs operated in the private and public sectors. They have, however, received substantiality less funding than have other programs (Goldman, 1976B:92).

The U.S. federal manpower policy was developed and pushed in 1961-1962 as a direct result of human capital theory.¹³ During the Johnson administration, there was a shift away from institutional to on-the-job training.¹⁴ A number of different programs were developed as part of Johnson's War on Poverty.

During Nixon's administration, the NAB-JOBS (the National Alliance of Businessmen's training in the business sector) program of subsidizing private employers was established. Later, with the Manpower Revenue(Sharing Act, Nixon decentralized manpower planning and implementation of programs to the state and local levels, though they were federally funded.

 \mathbf{x}

J. The War on Poverty in the United States

From the very beginning, the American War on Poverty focused on developing programs to provide equality of opportunity for all groups in society and depended upon economic growth and prosperity rather than direct or indirect redistribution of money or goods to 'achieve that goal (Plotnick and Skidmore, 1975B:5). In a speech at Yale University on June 11, 1962, President John F. Kennedy stated that."(t)he national interest lies in high employment and steady expansion of output, in stable prices and a strong dollar" (Lampman, 1971B:13).

Since early 1963, when President Kennedy first proposed the development of a comprehensive federal policy to eradicate poverty in the United States, job training programs and community development and participation by the poor in these programs have been stressed. Income maintenance programs have been of peripheral nature. Willard Wirtz and Daniel Moynihan stressed that the main cause of poverty is lack of jobs. The community approach to social problems, which gained credence in the 1950's, includes the idea that any pathology lay in the community not in the individual. Both ideas remained central to the War on Poverty.

Plotnick and Skidmore (1975B:5) report that Kennedy's economic advisors urged

...that expansionary fiscal and monetary policy could produce vigorous and sustainable economic growth (at tolerable price levels). The tight labor markets thus produced would, it was argued, cmpate jobs and continual government revenues (fiscal dividends) to be spent on social programs. This, in turn, led to the view that the poor did not need handouts as much as enrichment programs to render them more employable. The largest portion of the Economic Opportunity Bill, which was signed into law in August 1964, went to human capital programs. Youth opportunity programs and work experience programs constituted 58% of the total with another 33% going to urban and rural community action programs. Although \$800 million was approved for this first effort in the War on Poverty, it was only a very modest percentage of the federal budget and not enough to do justice to the goals of the program. Far more money was spent on social welfare programs (Plotnick and Skidmore, 1975B:6). In fact, Plotnick and Skidmore (<u>Ibid</u>.:74) report that "...special anti-poverty expenditures have mever accounted for more than 6 percent of the total assistance to the poor and their relative importance has declined since 1968."

Federal programs were designed to create new jobs, prepare workers for existing and new jobs, and to provide equal employment opporunnity. Thomas I. Ribich (1968E:51, footnote 28) remarked that "...it might be argued that retraining and tight labor markets are, to some extent, substitute policies."

The efficiency and the long-term visbility of federal job creation programs are highly questionable. Plotnick and Skidmore (1975B:187) report that

> ...the real new employment created by public employment programs, as a percentage of the new jobs created on paper, is estimated at 51 percent after one year, 44 percent after two years, and 4 percent "under a permanent program".

The entire manpower training concept as a means of eradicating poverty has come under heavy criticism. Plotnick and Skidmore (1975B:187) point out that

(i)t has also been hard to demonstrate that manpower programs judged cost effective in themselves have placed previously unemployed workers into jobs that would have been otherwise unfilled. Their effect has probably been, at least partly, to put newly trained workers into jobs previously filled by other workers, who were simply discarded.

Y

The notion of education as capital, like most constituent part of the premises and assumptions of human capital theory, depends upon a continually expanding economy with sustained growth and elastic demand for educated labor. The American War on Poverty, like the Canadian effort, also depended upon these factors.

As soon as the economy began to slow down and unemployment as well as inflation rise, public programs premised on human capital theory fell under heavy criticism. Keynesian economists consider equality and efficiency as antithetical in periods of sustained inflation (Lampman, 1971B:28). However, the view of human capital economists that removing poverty makes society more efficient ¹⁵ prevailed into the early 1970's.

A rather interesting occurrence was the decision by President Richard M. Nixon to fight inflation by sharply increasing unemployment to above 4% in order to keep wages down, which in turn would lessen total spending and hence bring prices down (Lampman, 1971B:156). Unfortunately, unemployment rose to nearly 6% in 1970 while inflation also grew worse.¹⁶

K. Gross Inequality in the Distribution of Personal Income

Modern liberalism has generally prevailed over conservatism in most Western capitalist countries since World War II. One major

goal of these liberal governments has been the elimination of poverty and a lessening of the gross inequality in the distribution of personal income. The liberal belief that inequality is largely a result of man-made, institutional obstacles and socially-determined attitudes reinforces the liberal strategy of purguing social change and reform through the state rather than the individual. Thus, the state has taken on the responsibility and sover to resolve the issue of inequality within the existing institutional framework by use of non-violent, legal, constitutional, and administrative "solutions" (Morton and Watson, 1971A:290).

Human capital theory fits into the mainstream of modern liberalism. Schultz has consistently pointed out the absolute necessity for government intervention and involvement in eradicating poverty in the United States as well as reducing the inequality in the distribution of personal income through increased investment in human capital. "Diminished" inequality is the liberal ideal.

The human capital economic argument is that everyone would benefit from the resulting development of talent and contribution to productivity. Poverty is inefficient and wasteful.

Western capitalist society has moved along a liberal continuum from legal equality to religious equality to political equality (Lampman, 1971B:17). Now, the offensive is directed towards reducing economic inequality. Reducing the inequality in the distribution of personal income has been the goal since World War II.

Human capital economists, like other liberals, do not want to eliminate <u>all</u> inequality, but only those inequalities which are "arbitrary", "functionless", and/or "capricious" (Lampman, 1971B:42).

Inequalities which are destructive and oppressive must be eliminated but some inequality is unavoidable, necessary, and functional, according to this view.

Human capital economists, like educators and sociologists who subscribe to the technical-function theory of education, believe that complex, modern societies require social stratification and differentiation and that some inequality is functional (Lampman, 1971B: 32). Thus, meritococcy as the ideal stratification system and a strong welfare state with centralized power constitute the basic thrust of modern liberalism.

The United States pursued the goal of eliminating poverty by attempting to increase the standard of living of those individuals in the bottom income group. Low income was viewed as a sign of social and political failure in the early and mid-1960's. Thus, the primary goal was to significantly reduce and eventually eliminate the number of families living below the poverty line.

Unfortunately, the question of which group would bear the cost of income redistribution was never addressed (Lampman, 1971B:50). Income inequality in terms of the functional income and shares of income, that is in terms of the division of income between labor and property, was ignored. The size distribution of income, a ranking from rich to poor which expresses inequality in terms of the share of all income going to the top 10% and so on of all income receivers, was also ignored (Jbid.:43).

In the 1960's, 80% of the American labor force were employees (Lampman, 1971B:44).¹⁷ As such, these individuals are dependent upon their employers for work. Therefore, most workers are adversely

affected by high unemployment and inflation.

Attempts to bring people above the poverty line by educating and/or training them fail if those individuals are unable to obtain stable amployment at above poverty level wages.

In aggregate terms, what has been the pattern of income inequality? Human capital economists maintain that there is and has been a sustained lessening of inequality in the distribution of personal income since World War II. The reasons for this, they argue, are varied and complex. One main reason is the increased level of education of workers.¹⁸

Lampman (1971B:46-47) uses U.S. Bureau of Census data to illustrate the size distribution of income between 1947 and 1967 in the United States. He asserts that

...numerous scholars have produced a consensus that there was indeed a lessening of inequality in the United States during 1938-1948 and no clear trend one way or another during 1948-1967. The lessening of inequality seems to have been confined to those groups within the top half of the income distribution, with no great change in the income share of, and no change in the inequality within, the bottom half...the post-war years have seen a slight gain in share of income by the families in the fortieth to ninety-four percentiles, at the expense of the bottom forty and the top six percentiles.

Thus, Lampman (1971B:47) maintains that there has been "...stability in the overall distribution of wealth and income..." in the United States. This means that "...the top 10 percent of income receivers has received about 28 percent of total money income and the lowest 10 percent about 2 percent of income. The share of the lowest 20 percent of income receivers has been about 5 percent of total money income" whereas the top 20 percent of income receivers receive over 40 percent of total money income (Ibid.:46).²⁰ 192

Plotnick and Skidmore (1975B:104) state that the relative income gap between rich and poor increased \$130 per person between 1965 and 1972. Although the absolute income gap between rich and poor fell between 1965 and 1968 in the United States, it has risen since then. They maintain that pretransfer absolute powerty in the United States did not really decrease between 1965 and 1972. It was static, as was relative poverty, for that period. However, since 1972, there has been an increase in income inequality as well as in the distribution of income. The percentage of people living below the poverty line has also increased (Ibid.: 179-180).

Leo A. Johnson (1977B:4) uses data from <u>Taxation Statistics</u> to illustrate the increase in income inequality in Canada between 1946 and 1971.

For example, in 1946, the richest ten per cent of earners received about 20 times as much income as the poorest ten per cent, whereas in 1971 they received 45 times as much. Similarly, in 1946, the income received by the riches decile equalled that received by the poorest 55 per cent of earners, whereas in 1971 their income equalled that of the poorest 64 per cent of earners.

Not only has there been a decline in relative income of the poorer income earners but there has also been a decrease in absolute purchasing power as well. Johnson (1977B:5-6) points out that this

decline

...appears to be occurring at an accelerating rate, and to be reaching higher and higher in the ranks of income earners. Thus while the sixth decile still increased its portion of earnings in the 1951-56 period,' it then levelled off between 1956 and 1961, then declined between 1961 and 1971...If this pattern of earnings change continues, then even earners in the fifth and sixth, deciles are likely to experience declining standards of living in the foreseeable future.

sta.

Johnson (1977B:7-8) provides the following example of the magnitude of inequalities in purchasing power between different occupational groups in Canada between 1946 and 1971:

In a period when average earners received a purchasing power increase of \$1,609 (in 1961 dollars), doctors received an average increase of \$13,954, dentists received \$10,672, and lawyers received \$10,026. Farmers, in contrast, received an increase in purchasing power of only \$637, while pensioners lost \$95.

Furthermore, a survey done in 1971 revealed that more than 25% of Canadians in the 80 to 90 percentile level (\$8,000 to \$10,000) believed that they were poor (Stewart, 1971A:45). Poverty line income, as defined by the Canadian Senate, for a family of four in 1971 was \$5,315 (Johnson, 1977B:21).

There has been a definite increase of poverty in Canada since World War II. Johnson (1977B:20), using poverty line income figures provided by the Senate, states that

... in 1946 almost twenty per cent of earners earned below the poverty line levels for single individuals. In 1971 this proportion had increased to thirty per cent... In spite of the enormous increase in average earnings between 1946 and 1971, maldistribution of income created a situation where fully sixty per cent of all earners still could not adequately support a family of four.

Thus, between 1946 and 1971, "...the richest twenty per cent of Canadian earners have received almost half of all new income, while the poorest fifty per cent received only about twenty per cent" (Johnson, 1977B:7).

The degree of inequality in the distribution of personal income varies among European countries. Some, like the Netherlands and the

6

Scandinavian countries, are more egalitarian than the United States. Others, like Italy, have more inequality in the distribution of personal income than the United States (Lampman, 1971B:46).

The underdeveloped countries in general have more inequality in the distribution of personal income than the United States. In many instances, the inequality in income is the most extreme. However, because of many social and economic differences, it is very difficult to make comparisons between underdeveloped countries as well as between them and the modern, industrialized countries. The Soviet Union and most other communist countries do not publish size-distribution data (Lampman, 1971B:46).

It is commonly believed that government transfer payments to the poor and various social welfare programs provide the poor with at least poverty line income and services and also that this is done at the expense of the wealthy. Minimum wage legislation is also popularly believed to provide workers with at least poverty line income. Such government expenditures along with a "progressive" income tax system are thought to contribute to a reduction in the inequality in the distribution of personal income. These assumptions and beliefs are false (National Council of Welfare, 1976B, 1978B).

Johnson (1977B:21) compared the provincial welfare benefits with the poverty line measures developed by the Senate for the years 1969-1971 and concluded that

...for normal families, welfare in most provinces falls below even the minimum standards for food, clothing, and shelter set by the Senate Committee, and none comes close to providing poverty line incomes. The worst, New Brunswick, provides only 73 per cent of basic necessities and 44 per cent of poverty line requirements.

Minimum wages²¹ are set marginally higher than welfare benefits in some provinces, but in many provinces they are lower. For a family of four, minimum wage employment provides income far below the poverty line.²² Furthermore, full-time, stable employment at minimum wage is not the norm (Johnson, 1977B:21). Minimum wage employment is usually non-unionized, seasonal, and unstable.

Plotnick and Skidmore (1975B) document the fact that poor families in the United States with unemployed able-bodied adults fare worse than the disabled and elderly in terms of welfare benefits.

Contrary to popular belief, then, poor people, whether employed or unemployed, are not adequately provided for by welfare programs or minimum wage legislation. In fact, government transfer programs do not benefit the poor at the expense of the rich. They do not contribute towards a lessening of inequality in the distribution of personal income. Johnson (1977B:23) has shown that "...low wage earners actually receive less money from transfer payments than do those in high income brackets." Thus

(n)ot only has the addition of transfer payments improved the relative position of the lowest-paid wage earners, but in the past twenty years a change for the worse has occurred. And, even when we examine the relative position of all income recipients, no improvement of distribution due to transfer payments has come about. Clearly, then, the professed aim of the welfare system -- to prevent poverty and to overcome disparity -- has almost completely failed. The huge outcry about the crushing burden of taxation on the rich to alleviate the condition of the poor has served only as a smoke screen for the preservation of the incomes and priviledges of the rich (Johnson, 1977B:23-24).

The structure of taxation in Canada was studied by Professor Thomas A. Wilson of the University of Toronto and reported by the Toronto Star on February 11, 1970. Johnson (1977B:25) summarized the

results:

Ľ,

...when all taxes, direct and indirect, were taken into account, families whose total income from all sources was less than \$2,000 per year paid more than 60 per cent of their gross income in taxes. For families at \$10,000 per year, taxes represented only 38 per cent of income, while above \$10,000 level little progression occurred.

Allan M. Maslove conducted a study of the system of taxation for the Economic Council of Canada in 1972. His conclusion was that rather than having a progressive income tax system, Canadian taxes are highly regressive.²³ Johnson (1977B:26) quotes from Maslove's (1972B:64) report that

By far the most striking conclusion to be drawn from an examination of total tax payments is the extreme regressivity of the system at the lower end of the income scale and the lack of any significant progressivity over the remainder of the income range...Indeed, over the lower portion of the income scale, the system tends to contradict the ability-to-pay principle by taxing the poor at a higher rate than those who are better off.

In the United States, the over-all combination of taxes prior to World War II was regressive. It "...compounded the pretax inequality of incomes...By 1929, some measure of equalization had become fixed as an outstanding characteristic of our U.S. fiscal system..." (Lampman, 1971B:95). There was a large increase in income tax during World War II (to over 25%) and federal taxes increased over state and local taxes for the first time. However, the "progressive" income tax system (including corporate taxes) in the United States has not reduced the inequality in the distribution of personal income.²⁴

Chiswick and Mincer (1972A) published results of a research project in which they utilized time-series data to determine changes

in the distribution of personal income of adult males in the United States between 1939 projected to 1985. Their conclusion was that "...relative inequality declined between 1939 and the early postwar years but has subsequently remained almost unchanged" (<u>Ibid</u>.:34). The study is an attempt to apply time-series analysis to investigate the relationship between human capital and income distribution, which previously had been limited to cross-sectional analysis. What is interesting to us about this study is the conclusion that changes in employment conditions are much more important in explaining changes in inequality than is schooling. Chiswick and Mincer (<u>Ibid</u>.) conclude

that

(f)or adult males in the United States, changes in the income distribution are affected mainly by changes in distributions of schooling, age, and employment. During the past 20 years, changes in the level and inequality of age and the inequality of schooling have been small and could not have greatly influenced the overall distribution, especially since they tend to have canceling effects. The stronger influence has been the business cycle through its effects on the dispersion of weeks of employment ... Most of the observed differences in inequality between 1939 and 1965 is explained by changes in employment conditions. The remainder is a decline in the inequality of wage rates, possibly due to a decline in the rate of return to schooling.

L. <u>Using Rate-of-Return Analysis to Balance Economic Efficiency and</u> Social Equity

Schultz has been critical of American tax laws for years because, in general, they do not take into account human capital. Investment in human capital on the money market is very difficult, especially if the borrower has no property to use as collateral. The development of student loan programs by the federal and state governments in the United States as well as by the federal and provincial governments in Canada and other financial assistance for students since the mid-1960's was a result of the lack of opportunities in the private sector and the importance placed on such opportunities by Schultz and other human capital economists.²⁵

By 1969, however, with the publication of a study by Hansen and Weisbrod, Schultz became convinced that the great public subsidies to higher education had a regressive effect on the distribution of personal income. Schultz (1971B:176-177) became convinced that the financing of higher education is quite regressive

... because it increases the lifetime earnings of college graduates in part at the expense of the others, and, closely related, because higher education provides educational services predominantly for students from middle and upper income families, a part of the cost of these educational services being paid for by taxes on poor families...(A) much smaller proportion of the undergraduate students in publically financed institutions receive financial aid for reasons of their having inadequate income than do undergraduate students in private colleges and universities. In either case, the financing is such that substantial amounts of valuable assets are being transferred by society to a particular intellectually elite set of individuals.

Thus, because of the high public subsidies to higher education, Schultz (Ibid.26) believes that it "...is in general both socially inefficient and inequitable."

Because of the tremendous expansion of post-secondary education in the 1960's in North America and the large amount of public subsidies,²⁶ the state, provincial, and federal governments in the United States and Canada began shifting more of the cost directly onto the students.

This was rationalized in human capital terms as a legitimate investment because the students would realize a rate of return on their education equal to most business investments. Therefore, it was felt that they should finance a large part of the investment. Consequently, by the mid-1960's, student loans were an established, accepted way of financing one's post-secondary education.

Another strategy of the state to combat its fiscal crisis was to raise tuition fees.²⁷ This also fell under the rationale of human capital theory, since the private rate of return was considered high enough to warrant the increased private investment in education. Schultz has been a proponent of increasing tuition fees to full cost for a number of years. The real cost of producing services should be the price, according to Schultz. Because there are real differences in quality between various post-secondary educational institutions, there should be corresponding differences in price. Once prices are established which reflect the full cost and differences in quality, tuition will be "...socially efficient prices to which the student can respond" (Schultz, 1972A:45). Only then will the post-secondary educational sector of the economy function efficiently on both the social and individual levels. In Schultz's (<u>Ibid</u>.) own words,

(n) owhere are students confronted by prices for these services that are equal to the real cost of producing them, and therefore the prices to which they respond the not socially efficient prices. As a consequence, no matter how efficient students are privately in their decisions, from the point of view of the economy as a whole, the ellocation of resources to...education with not be socially efficient.

Educational policy changes in Ontario in 1972 included the raising of tuition fees, an increase in student loans, and a decrease in opportunities to earn teaching stipends. The Wright Commission supported these changes and stated that the availability of student grants and loans would ensure equality of educational opportunity to individuals from all groups in society.

Porter, et. al. (1973B), in a study of social class and educational opportunity in Ontario, are critical of the 1972-73 policy changes for post-secondary education in Ontario because they are basically regressive. For instance, the first \$800 of an award must be taken as a loan.

Porter, et. al. (1973B:206) supported a substantial increase in tuition fees (to cost level) but only if accompanied by "...an all grant (no loan) scheme based on parental resources." Otherwise, accessibility to post-secondary education for lower class students would be seriously jeopardized. Student loans to those from middle and upper class families would still be available, however,

By 1977, the Ontario Economic Council advocated that students pay the full cost of their post-secondary education and that the government should not provide any subsidies to colleges and universities (Buttrick, 1977B). Tuition fees have been raised substantially across Canada and the United States, although not to the level of full cost.

The idea of shifting the allocation of public monies from. post-secondary institutions directly to the students themselves developed in the early 1970's. Schultz (1972A:24-25) is one of the advocates of such a policy change because it would be allocatively more efficient and socially equitable. Students from low-income

families would be highly subsidized, whereas those from middle and high-income families would not receive subsidies. Porter, et. al. (1973B:208) also support this proposed policy change. They state that such a policy change would be "...both politically and constitutionally attractive."

While financial aid²⁸ to post-secondary students from lowincome families in the United States has increased, that has generally not been the case in Canada. Tuition fees have escalated in both countries, as have room and board charges. However, Canada does not have a national work-study program providing subsidized employment at universities for students from low-income families, and there is very little money available to undergraduates in the form of bursuries, grants, and scholarships based on need. Student loans are becoming increasingly more difficult to obtain. While based partially on need, the course of study and prospects for repayment are also considered as important factors on a student loan application.

Although the increases in tuition fees have been justified in terms of human capital theory, in that there is a high private rate of return to such investments, there has been a reduction in the value of higher education in the eyes of the public because there has, in fact, been a great shortage of jobs for graduates in the 1970's. The increasing costs of post-secondary education combined with the increasing unemployment and underemployment have contributed to a levelling off and slight decline in post-secondary enrollments, especially at the universities, during the 1970's.

Because the increased private costs of education have not been accompanied by any comprehensive scheme to provide students from low-
income families with public subsidies, post-secondary education in North America has become more socially regressive and inequitable. Equality of educational opportunity has not advanced, and, in fact, may well be farther from reality than it was twenty years ago.

Human capital theory has been severely criticized along theoretical and methodological grounds during the past decade by sociologists, educators, and politicians, as well as economists. Criticism has come from both the left and the right. Conservatives disagree with the basic principles and philosophy of liberal school reforms. Marxists and socialists also disagree with basic principles of liberal school reform, but for different reasons. Even some liberals are upset with the slow pace of reform and the seemingly poor results. The public, likewise, is divided in opinion as to why educational reform has failed to resolve many of the problems it was supposed to resolve. There has been a general decline in public expectations of education and in willingness to support expensive social programs. The chronic world-wide economic recession has added additional pressure on politicians to abandon many of the programs, if not principles, of liberal educational reform. Human capital theory has survived the storms so far by adjusting the mix of efficiency and equity to suit the times. It remains to be seen how long human capital theory can continue to do this.

CHAPTER VI

Footnotes

¹Providing equality of educational opportunity implies less inequality of educational attainment and educational achievement between social groups and classes, and, ultimately, less inequality in society.

²In contrast, programs in Britain designed to improve educational achievement of working class children have neglected the question of educational provision, which continues to be an issue. See Byrne, et. al., 1975, for theoretical and empirical analysis of this problem.

³ However, as Gonick (1978B:66) points out "(t)he function of job training has been almost entirely shifted from business to the school system."

⁴Gonick (1978B:118) reports "(t)he consumer price index rose from 2.9 per cent in 1971 to 4.8 per cent in 1972 and 7.6 per cent in 1973." By 1974, inflation was 10.8% in Canada (Ibid.:119).

⁵For discussion of how wages but not profits and prices were kept down in Canada, see Gonick (1978B:114-128).

⁶ The rate was even higher for post-graduates.

⁷Freeman (1975A:289) reports that

...by 1974 real college starting rates were 11 - 25 percent below those five years earlier and earnings relative to those of other workers 13 - 27 percent below 1969 levels. The figures thus demonstrate the existence of sizable downward alternatives in real college starting salaries during the market turndown -- the classic "price system" response to manpower surpluses.

⁸Freeman (1975A: 305-307) states that the main reason for the "shortage" of -- or high demand for -- college graduates during the 1960's was because of the expansion of graduate education which effectively kept college graduates out of the labor market. In fact, "...the 'net number' of college graduates entering the market declined from the late fifties to 1967, due to the increase in graduate enrollments." Ironically, just as the supply of college graduates increased, the demand for college graduates declined drastically and dropped by over 50% in the following five years (1967-1972).

Ϊ

⁹There were over 832,000 live births in 1967, 784,000 in 1970, and just over 600,000 in 1975 (Cantor and Roberts, 1979B:89).

¹⁰Growth refers to economic growth. Equity refers to distributional goals. Retraining the poor and the unemployed are examples of programs designed for this latter objective. Stabilization refers to economic stabilization. Manpower policy in Canada is used as a contracyclical tool to stabilize the economy. Programs developed for this objective include manpower training programs in periods of economic recession (Economic Council of Canada, 1971B:98).

¹¹Critics of this policy point out that it was developed primarily as a response to pressure from private industry in a period of rapid economic expansion and shortage of skilled workers.

¹²Canadian Government research into the effects of adult occupational training in 1973 on future employment/unemployment patterns of trainees revealed that manpower training programs for the unemployed have not contributed to the permanent reintegration of those individuals into the labor force. Research indicates that recipients of unemployment insurance who had received manpower training had an equal chance as those who had never received training to reappear on claim. The major cause of this has been attributed to the worsening of the economy since 1972 and the "declining labor situation" which has had a "...negative impact on both the trainees and the non-trainees sampled."

13 There is disagreement as to whether or not American manpower policies and programs were based on human capital theory. In narrow terms of employment and earnings or in terms of strict cost-benefit analysis, most of the manpower training programs were not efficient. Many of the investment were questionable from a purely economic viewpoint. (There were, however, social and psychological benefits.) However, they were based on the expectation of a continuing buoyant economy and low unemployment. The sharp increase in unemployment since 1970, when it was nearly 6%, has exasperated attempts to eradicate poverty through manpower training programs. Unemployment between 1964 and 1969 dropped from 5.5% to 3.5% in the United States (Lampman, 1971B:154). The lowest recorded unemployment in the United States was one year during World War II, when it was 2% (Ibid.: 155). However, unemployment was 5.6% in 1972 and between 8% and 9% in 1975 (Plotnick and Skidmore, 1975B:177). Inflation began to accelerate in the United States in 1968 when prices rose 4.7% and 6% in 1969 (Lampman, op. cit.:155). Thus, stagflation has become firmly established in the American economy since 1970. In 1974 and 1975, inflation was running at 11% (Plotnick and Skidmore, op. cit.:111). Inflation weached 12% in 1979 and 1980. Politicians are putting the emphasis on fighting inflation in 1981 even though unemployment rates are higher as well. Most economists concur with the prediction that 1981 will be a year of severe recession.

¹⁴According to the Economic Council of Canada (1971B:104), approximately 80% of federal adult occupational training expenditures in the United States was spent on in-industry training. By comparison, the corresponding figure in Canada was less than 5% (excluding apprentices).

¹⁵ This view states that because the poorest one-fifth of the population in the United States produces 3% of the total product but consumes 5% of the total product, it is economically efficient to eliminate poverty (Lampman, 1971B:29).

¹⁶ In terms of eradicating poverty, inflation complicates the situation because, as Plotnick and Skidmore (1975B:177) point out, "(i)nflation adversely affects the net worth of the poor, whereas it increases the net worth of the middle and upper-middle income groups."

17 In Canada, 76.9% of income earners are employees (Johnson, 1977B:40).

¹⁸Between 1950 and 1957, the median level of education in the United States increased one full year (Lampman, 1971B:123, footnote 6). Schultz (1961A:73) states that the stock of capital formed by education rose 8½ times between 1900 and 1957, whereas the stock of reproducible (non-human) capital rose only 4½ times. In 1940, 15% of 18-21 year old Americans went to college.

Schultz (1971B:177) states that

(i)n retrospect, given the type of growth that has characterized our economy and the remarkable increase in the stock of education per worker in the labor force, the gains in elementary and secondary schooling and in higher education taken as a whole have been instrumental. it seems to me, in reducing the inequality in the distribution of personal income ... The rise in the investment in education relative to that invested in nonhuman capital increases total earnings relative to total property income, and property income is distributed much less equally than the earnings of persons from labor. Therefore, investment in schooling reduced the inequality in the distribution of personal income. The hypothesis proposed here is that these patterns of investment are an important part in the explanation of the observed reduction in the distribution of personal income.

Although Schultz states that there has been a lessening of the inequality in the distribution of personal income, he admits that,

Ø

although "(t)his tendency has become a well-established fact, (but) explanations of this phenomenon are still in their infancy" (Schultz, 1972A:65). Furthermore, Schultz (Ibid.) terms the reduction "modest". Schultz (Ibid.) also believes that progressive taxation and welfare programs have done very little, if anything, to change the distribution of personal income. Rather, Schultz (Ibid.) believes that "(t)he increases in the demand for skills and knowledge and the response of the supply that is under way in modern economies lead to the accumulation of forms of human capital that accounts for most of the observed tendency under consideration" (e.g., a lessening of the inequality in the distribution of personal income).

¹⁹See "Post War Changes in the Size Distribution of Income in the United States," <u>American Economic Association Proceedings</u>, Vol. 60, May 1970:247-260.

²⁰Lampman (1971B:46, footnote 3) points out that "...inequality among non-whites is greater than among whites, and inequality among the old is greater than among the young."

²¹Minimum wage legislation does not cover all workers. Students, part-time employees, workers in certain types of jobs, etc., are not included. Furthermore, women in some provinces receive lower minimum wages than men (Johnson, 1977B:23). Minimum wage legislation was passed by the provinces in Canada and by the states in the United States prior to federal minimum wage legislation. The first minimum wage legislation was passed in 1917 in Alberta. However, minimum wage legislation did not apply to men until 1925 in British Columbia and the 1930's in Saskatchewan, Manitoba, Ontario, and Quebec. Furthermore, New Brunswick, Newfoundland, and Prince Edward Island did not pass⁶ minimum wage legislation until after World War II (Ostry and Zaidi, 1972B:257-258).

²²General minimum wage rates in 1970 varied from \$1.25 per hour to \$1.65 per hour in Canada, whereas the poverty line for a family of four required \$2.48 per hour (Johnson, 1977B:23).

²³Johnson (1977B:25-26) reports some of the details from Maslove's study in the following quotation:

For families receiving less than \$2,000 in earned income, taxation from federal, provincial and municipal sources took up more than one hundred per cent of that income in every province other than Manitoba. In other words, a significant part of welfare -- even though such was far below poverty line standards -- went to pay taxes. The Ontario situation was particularly scandalous. There taxes for low income families were 151.1 per cent of earned income. Above the ្ឋ៍ថ

5.

<u>,</u>

\$2,000 level there was a sharp decline in taxation rate to the \$3,000 to \$3,900 level. then a gradual decline to the \$12,000 - \$14,000 level. Above the \$15,000 level of family income the proportion of earned income that went into taxation rose slightly, although, as Maslove pointed out, capital gains had not been included in his calculations. In all probability, if all income had been included, the tax load above \$15,000 was below that for lower levels of income.

Moreover, when transfer payments of all kinds were included, Maslove discovered that the poorest income recipients still paid a much higher proportion of their income in taxes than did those who were better off. Indeed, for those receiving leas than \$2,000 per year, the total tax load amounted to more than 60 per cent of all income. In contrast, those receiving more than \$15,000 income per year paid less than 35 per cent of all income in taxes.

The National Council of Welfare (1976B, 1978B) supports this view in their more recent research on this topic.

²⁴Læmpmæn (1971B:26-27) points out that he and many other economists seriously doubt that progressive income taxation will reduce income inequality. Rather, such a tax system is seen as an attempt by the state to take over the economic power of the rich. Thus, Bertrand de Jouvenel (1951B:73) wrote that it is "...in effect, far less a redistribution of free income from the rich to the poorer... than a redistribution of power from the individual to the state." Although there is some redistribution of income, it is not enough to fund the social welfare programs for which the state assumes responsibility. Conservatives view the emergence of the liberal state as a "disaster". De Jouvenel (Ibid.:77-78) writes that

(i)nsofar as the state amputates higher incomes it must assume their saving and investment functions, as we come to the centralization of investment. Insofar as the amputated higher incomes fail to sustain certain social activities, the State must step in, subsidize these activities, and preside over them. Insofar as income becomes inadequate for the formation and expense of those people who fulfill the more intricate of specialized social functions, the State must see to the formation and upkeep of this personnel. This results in a transfer of power from individuals to officials, who tend to constitute a new ruling class as against that which is being destroyed. ²⁵ The Canadian Government established the Canadian Student Loans program for post-secondary students in 1964. Some of the provinces developed their own programs to supplement this. Quebec started its own loan and financial aid program for post-secondary students in 1966. Newfoundland abolished all fees for first year resident students at Memorial University in 1965. This was later extended to cover higher years and living allowances.were also paid togstudents. By 1969, however, fees were reinstated.

²⁶Federal government support to higher education in Canada rose from 14.3% of university revenue in 1954/55 to 43.9% in 1969/70 to 48.6% in 1971/72 (Munroe, 1973A:42-44).

²⁷Because education 'is labor-intensive and there is little likelihood of increasing productivity in the educational sector, costs will continue to escalate, vis-a-vis other sectors of the economy, just to keep those employed in education at the same relative standard of living. Graduate, professional, and technical education and training have the highest per unit costs and are the least amenable to mechanization (Buttrick, 1977B).

²⁸Financial aid to post-secondary students includes bursuries, grants, scholarships, loans, and work-study jobs. The United States Government pays 80% of the wages received by the students and the university which provides the jobs pays only 20%. 209

 \sim

CHAPTER VII.

SUMMARY AND CONCLUSION

In the preceeding chapters we have traced the development of an idea which has become so well-entrenched and fashionable in academic and political arenas that it has been enshrined in public policy and become part of our collective consciousness. The idea that knowledge and skills possessed by individual workers enhance their productive capabilities and capacities such that they constitute capital in the same sense that the physical means of production constitute capital, i.e., generate a stream of income, is what has come to be called human capital by some neo-classical economists.

Human capital theory itself, however, is part of the paradigm of functionalism which dominated all the social sciences during the 1950's and 1960's. In sociology, technological functionalism was the most accepted view of our modern, complex society. Within technological functionalism, the technical-function theory of education was the widelyaccepted theoretical framework for sociology of education. The macrolevel image of the "post-industrial"¹ society and the belief that Western industrialized capitalist countries have entered that stage of development since World War II is common to both human capital theory and the technical-function theory of education. These ideas have been voiced publically so much that belief in them (that we are indeed living in the "post-industrial" society) is common to academics, politicians, and the general public.

In this thesis, we examined human capital theory and the influence that it has had on public policies, and on educational policies

in particular, from the sociology of knowledge perspective. Furthermore, we have rejected the functionalist paradigm, and human capital theory in particular, because it has been found lacking on both theoretical and empirical grounds to provide an adequate explanation of social reality.

Some of the issues and questions which human capital theory has failed to adequately explain include: 1) the expansion and subsequent contraction of educational systems; 2) changes in the distribution of personal income; 3) inequality of opportunity for individuals from different groups; 4) the relationship between cognitive and technical skills and jobs; 5) the relationship between education and adequacy of job performance; 6) the high unemployment rates and underemployment of well-educated individuals; 7) the lack of economic growth despite a better educated labor force; and 8) the class structure of society and the importance of ownership of physical capital for class determination.

Education has been identified by Schultz and other human capital economists as the main component of human capital which can be, and therefore should be, increased and improved in the Western industrialized capitalist countries. The situation is more complex in the underdeveloped countries, where other components of human capital are also greatly underdeveloped.

One way of determining the validity of a theory is to determine the effectiveness of policies derived from it. Halsey (quoted in Byrne, et. al., 1975B:32) states that "(t)heories postulate paths to stated ends, therefore to test the validity of a theory is to evaluate the effectiveness of a policy which expresses it."

The expectations of public policies, and especially educational policies, which have been heavily influenced by human capital theory have not been met.

One such expectation was that increased education, and hence income, of workers would result in a significantly more equal (or less unequal) distribution of personal income. Along with this "expectation was the belief that poverty could be greatly reduced, if not eliminated, by an increase in quality and quantity of education and training programs for the poor.² Just as individual productivity and income were attributed to better educated workers, it was believed that national productivity and income were dependent upon a welleducated labor force. Thus, public expectations on all levels of education were vastly increased during the 1960's in the developed and underdeveloped countries alike, with the common expectation that national productivity and growth would continue to increase and create more wealth for both the nation and the individual workers.

Social science research in the 1950's and 1960's was premised on the belief that economic growth rates would continue to increase, that demand for highly educated workers was infinitely elastic, and that the occupational structure would expand and become more differentiated and specialized at the professional and paraprofessional (technician) levels. Demand for unskilled labor was expected to drop significantly, with many unskilled occupations disappearing due to increased automation (capital substitution for labor).

There was, in fact, rapid expansion of the tertiary (service) sector of the economy after World War II. Demand for white collar workers was so great that the proportion of women in the Canadian work

force increased from 19.8% in 1941 to 27.8% in 1961 to 35% in 1971, whereas it stood at 13.3% in 1901 (Scheter, 1977A:398). The increase in demand was greatest for salaried clerical workers and salaried professionals accounted for the second largest increase in demand (<u>Ibid</u>.). Porter (1965%) has documented that much of the demand for professionals and highly skilled workers in Canada during the 1950's was met by immigration from Europe.

Technological changes have affected the production process itself in the advanced capitalist states and this has had repercussions in the social relations of production as well. Advances in science and technology have had the effect of accelerating production, increasing productivity and economic growth as well as cheapening commodities produced. There has been a corresponding change in the nature and structure of occupations as well as in and between social institutions. The changing social, economic, and political reality experienced in the transition to monopoly capitalism, or "mature" capitalism, contains new contradictions. The division of labor has become more specialized, differentiated, hierarchical, fragmented, and more bureaucratic than ever before. There has been an increase in fragmentation of tasks accompanied by loss of control by workers over their work and a subsequent increase in alienation. The petite bourgeoisie has been virtually incorporated into the salaried labor force, many jobs have been eliminated by automation, and the reserve army of surplus labor has been increased.

Another very important idea central to the theory of "postindustrial" society, and integral to both human capital theory and the technical-function theory of education, is that class society is 213

No. of Street, Street,

being transcended by the increased importance of human capital vis-a-vis material capital in the generation of income, and hence wealth. Educated workers are considered "capitalists". Their "ownership" of knowledge and skills necessary for the production process enable educated workers to generate their own stream of income just as the owners of material capital are able to generate a stream of income from the use of their capital in the production process. The fundamental class distinction and potential conflict between capital and labor are ultimately "eliminated" by this human capital conceptualization of "post-industrial" society.

The focus on achieved attributes of individuals as the primary determinant of their socio-economic status has led to the popular belief that the system is meritocratic and as such, equitable. Unequal rewards in society are accepted by most as legitimate because the competition for success is seen to be open to all. Individual talent and motivation are thought to be the only traits necessary for economic success. Economic inequality³ per se is not considered by most to be problematic, as long as it is based on achievement and merit.

One's failure in the educational system and/or one's failure to achieve upward social mobility are considered the failures of individuals, not of the system. The individualization of failure prevents people from seeing the structural determinants of economic inequality. The success of economic individualism as a social myth has been boosted by belief that public policies have been able to provide equality of opportunity.

These expectations were part of the vision of post-industrial society and of the paradigm of functionalism. As such, they had a significant impact on public policy, and on educational policy in particular. The primary influence has been the public subsidy of post-secondary education and significantly greater expenditures on elementary and secondary education, made possible by direct federal intervention in education both in Canada and the United States. The failure of these policies to sustain economic growth, to improve the relative as well as the absolute wealth of workers (e.g., reduce inequality in the distribution of personal income), to significantly reduce poverty, to provide equality of opportunity to all groups in society, and to provide employment commensurate with the education of workers has resulted in political, economic, and social crises. Limitations of human capital theory and the technical-function theory of education are made more apparent in the context of these political, economic, and social crises which have plagued capitalist societies, developed and underdeveloped alike, since the late 1960's.

Direct state intervention in the economy was considered necessary for the resolution of the serious contradictions of advanced capitalism which had reached culmination with the Great Depression. The development of Keynesian economics and the philosophy of corporate liberalism provided the framework for this new function of the state. State responsibility for regulating and mediating the economy was established in the United States by President Roosevelt with the New Deal⁴ during the late 1930's. Acceptance of Keynesian economic principles and state intervention in the economy did not occur in Canada, however, until after, World War II⁵ (Wolfe, 1977A:259).

The Great Depression spawned the need for national economic planning by the state as well as the need for state responsibility for direct regulation of the economy.⁶ According to David Wolfe (1977A), this interventionist and regulative role of the state has led to an expansion of its traditional functions to include responsibility for both accumulation and legitimation. The balancing of these two potentially conflicting functions by the state has led to the development of conflict, and at times, crisis management.

In addition to regulating and mediating the economy, the state has come to regulate and mediate class conflict, and ultimately, social control. The establishment of the liberal welfare state⁷ during the past fifty years in North America and Europe has shifted much of the responsibility for social and economic well-being from the individual to the state. This puts the spate itself in a very vulnerable position; social and economic crises now have the potential to become political crises⁸ and the authority and legitimacy of the state itself is threatened.

State regulation of the economy necessitated direct government involvement in education. In most of Europe and the underdeveloped countries, education has been the responsibility of the centralized state for a long time. In Canada and the United States, however, education is a provincial or state responsibility. Therefore, constitutionally, the federal government in either country does not have jurisdiction over education. Nevertheless, the federal governments in both countries have found it necessary to intervene directly and indirectly in education.⁹ Human capital economists were prominent in a strong lobby by social scientists in the United States during the late 1950's and early 1960's which presented an economic rationale for direct federal government intervention in and funding of education. Because most social scientists believed the advanced capitalist countries of North America and Western Europe were evolving into "post-industrial" society and because of the great political concern over international competition with the Soviet Union during the Cold War era, development of scientific and technical expertise became a national priority. Removal of socio-economic and psychological barriers to the development of talent, and thus, enhancement of upward mobility for those members of the subordinate groups who possessed potential talents valued by society, became the political goal¹⁰ on the basis of its perceived economic value for efficiency and rationality.

The debate over schooling during the 1950's can be viewed as part of the struggle between conservatives and modern liberals over the powers, functions, and roles of the state. By 1960, it was evident that liberalism would prevail. It is in this context that human capital theory can be seen to have had an important impact on both the governmental levels and on the level of public opinion. In fact, much of the product of social science research as well as the functionalist paradigm was utilized by liberals for their own political ends.

Human capital theory as well as the technical-function theory of education focus primarily on the economy, and in particular, on the relationship between the structure of occupations and the structure

of the educational system. The discussion of equality of educational opportunity must be seen within the context of equality of opportunity for economic success in society because the primary aim of providing equality of educational opportunity is to enable "talented" members of subordinate groups (the so-called "minorities" or "disadvantaged") to achieve upward social mobility.¹¹ It is the economic goal of government policies and the economic role of education which human capital theory and the technical-function theory of education address themselves.

While the world capitalist economy has experienced sustained economic and monetary crises since 1970, welfare states have had to contend with a fiscal crisis¹² which threatens the very survival of the welfare state. Attempting to reduce sky-rocketing national debt, liberal and social democratic governments in the advanced capitalist countries have cut back on expenditures on social welfare and education, on those very programs which were the fruit of liberal political reforms since the Great Depression.

Thus, with the problems of increasing inflation¹³ and unemployment, accompanied by a levelling off and decline in economic growth rates, it seemed apparent to many that the previously accepted relationship between education and wealth needed to be re-evaluated. The tremendous expansion of education and subsequent upgrading of the labor force did not seem to have resulted in a sustained increase in economic growth nor in provision of equality of opportunity for all groups in society.

Human capital theory brought education into the realm of economic analysis by treating it as an institution which provides a.

public service by functioning to increase capital accumulation and economic growth. Thus, the "economic value" of education is of central concern to human capital economists.

There were two drastic alterations in the demand for education within a decade. High demand for education in the early 1960's, which human capital theory appeared to explain, dropped significantly in the late 1960's. Since then, it has not appeared to be efficient or rational for governments to continue investing in education. There has been a resurgence of the belief that schools consume capital rather than produce it. Even liberals now think that education is consuming more capital than can be productively utilized.

Thus, education was considered out of step with the economy. during the 1950's and early 1960's because of the shortage of highly educated workers and the high demand for them. By the late 1960's and early 1970's, this was reversed; rather than lagging behind the economy, education became "overheated", consuming a great amount of social resources and "producing" large numbers of highly educated individuals, most of whom could not be employed to their productive capacities. Demand for workers, whether highly educated or not, had dropped significantly as the world economy contracted in stagflation.

In times of fiscal crisis, governments are trying to eliminate as many non-productive investments as possible -- to invest their limited capital in sectors which will yield the highest rate of return. For example, there has been a definite shift away from massive subsidies to post-secondary education and an increase of cost to individuals. Schultz views this shift as leading to more rational decision-making by individuals and a realignment of their expectations with the realities

of the market place. This has also been viewed as a rationalization of public policy, since physical capital formed by public investment is not transferred to individuals as gifts. As such, it is within the scope of human capital theory. Thus, governments have developed student loan programs as well⁶ as other financial aid schemes for students.

The rationale for educational reform then, as it often has been, was to restore social and economic (and one could add, political) equilibrium. Just as expanding education in the previous decade was rationalized in terms of economic needs and efficiency as well as social equity, the contraction of education by austerity budgets and lowering of public subsidies as well as expectations¹⁴ were done for the same reasons when the economy went into the new crisis of stagflation.

It is difficult for individuals, just as it is for governments, to continue to place high expectations on the educational system in light of the lack of economic demand. Without economic demand, and hence employment, it is difficult, if not impossible, to achieve any upward mobility.¹⁵ It is even difficult for many to maintain their current status. Rather, many middle-class white males have been, for the first time in at least twenty-five years, experiencing downward mobility. Underemployment has become a major problem, along with unemployment. One could view labor force stratification in terms of some employed, many underemployed (mostly the well-educated), and many unemployed (mostly those with high school education or less).

Just as the worsening economy presented an economic crisis, the liberal policies of most Western industrialized capitalist countries, including the basic concept of the welfare state, came under criticism 220

for being unable to deliver what they had promised.¹⁶ The political crisis of liberalism, then, was brought about by the economic crises experienced since the late 1960's.

By the mid-1970's, there was strong support for neo-conservatism by the general public as well as by many academics and politicians. With this, governments have changed their economic, social, and educational policies in an attempt to deal with the change in social reality and the many contradictions it has exposed.

0

Since 1970, federal governments in both Canada and the United States began serious consideration of a negative income tax, a form of guaranteed minimum income, to replace all existing social welfare programs and services in an effort to rationalize and economize their expenditures.¹⁷ The fiscal crisis of the state was a major concern of both governments, as it was elsewhere. Though governments had taken responsibility for economic growth, and the power to intervene in the economy to ensure that growth, they were experiencing great frustration at being unable to resolve the new crisis of stagflation.

Government responsibility for maintaining aggregate demand (employment, production, and purchasing power), had itself come under question, even by liberal and social democratic governments. Canadian (Liberal) Finance Minister, Jean Crétien, expressed this new conservative bent[°] when he said that

(p)eople brame the government and put too much faith, at the same time, in the government. The intervention of the government has to be marginal, under the present circumstances...The economy is decided by millions of decisions on a daily basis by individuals. It is a collective mentality, and a Minister can hope, through his taxes, to create a climate that is good (Gonick, 1978B:124).

Ì

Gonick (1978B;124) points out that

(w)hat this amounts to is government abdication from responsibility for directing the economy, a direct denial of the 1945 commitment to maintain full employment. It is also an outright admission that the state operates only at the edge of the economy, able, at best, to nudge it slightly but powerless to alter its basic course. In particular, Crétien is saying that he has no answers for stagflation, that he can solve neither unemployment nor inflation "under present circumstances".

Furthermore, perhaps as much to lower people's expectations so as to be "in touch with the market place" as to appease business,

(t)he government budgets over the past three years have been deliberately designed to sustain a massive amount of unemployment. They are budgets of planned unemployment. Their goal is to deliver a docile, worried, labour force, one that is willing to work for less, that is willing to heed their master's voice, and that is less prone to strike (Gonick, 1978B:124-125).

Increased unemployment affected the public sector as well.

In February 1978, Jean Crétien announced that

×.,

(w)e were able to put all the governments together, from NDP to Socred, and say that the public sector should take less growth than the gross national product, that foreign investment is welcome in Canada, that compensation in the public sector should no longer be the pacesetter (Gonick, 1978B:125).

After the failure of mandatory wage and price controls to control inflation in North America, the Canadian and United States governments have taken a neo-conservative position regarding the responsibility and functions of the state. "...(G)overnment leaders applauded private industry as the creator of jobs and wealth and pledged to get out of the way and let private enterprise show what

it could do" (Gonick, 1978B:125-126).

Social and educational policies, like economic policies, are limited by political considerations, the availability of financial resources, and search available to political decision-makers, often of a conflicting or unclear nature (Byrne, et. al., 1975B:13). There have been significant changes in all three of these variables since World War II.

On the macro-level, Carnoy (1979B:2) reminds us that the state is not

...a benign spectator in the process of economic and social change...(C)hanges may occur because of a particular development process, but the process itself is not unalterable or uniquely optimal; if certain trends toward equality or inequality occur, it is because the type of development chosen generally suits the groups with economic and political power.

The shift of responsibility for economic decision-making and economic well-being from the state to the individual cannot be viewed as politically neutral. Likewise, the shift in responsibility for scholastic success and educational attainment as well as for wastage of talent from the state to individuals is not neutral. These issues are public issues as much as they are private ones (Husén; Byrne, et. al., 1975B:45).

Social science theories, and the paradigms to which they belong, are themselves social products. "(S)pecific theories, and their accompanying methodologies, do not arise in a social vacuum, but rather are rooted in material relations of production and therefore serve specific economic and political interests" (Stirling and Kouri, 1979A:169). The psychological theories of learning and achievement which prevailed prior to the mid-1950's were based on the psychological model of inherited ability as measured primarily by I.Q. tests. The conservative, individualistic position pervaded these theories (Byrne, et. al., 1975B:19).

By the mid-1950's, modern liberal ideas began to permeate psychological theories. The idea that environment shapes and influences behavior and measured intelligence was widely accepted. The distribution of ability and educational attainment were viewed as more dependent upon environmental (social class) factors than on genetic ones. This was all part of a movement in political, economic, and social terms away from individual responsibility to state and social responsibility. Socio-economic structures took precedence over individuals. Attributes of individuals were seen to be shaped by these structures. Thus, during the late 1950's and 1960's, the class-culture model of educational attainment developed and prevailed over the previous more conservative, aggressively individualistic psychological one that viewed individuals as socially and economically "fixed" by genetic endowment.

There has not been agreement about the kind of policies that should be developed, the level of intervention, or even what effects education has on society. There has also been disagreement on how much educational reform will affect society -- or even whether educational reforms will themselves be limited by society. These disagreements have been both political and philosophical (Byrne, et. al., -1975B:28).

(M) any...fail to appreciate the structural nature of the concept of class. As a result they fail to distinguish between those variables relating to educational attainment which describe individuals and those which relate to the properties of the system in which people find themselves. Put another way, no firm distinction has been made between <u>system inputs</u> and <u>personal inputs</u> into the process of educational attainment (Byrne, et. al., 1975B:28-29).

Although socio-economic structures have been recognized as having significant influence on the educational attainment of individuals, educational failure has been personalized, just as the liberal concept of class has been trivialized¹⁸ (Byrne, et. al., 1975B:29;31).

The question whether or not wastage of talent rates are to be viewed "...as the aggregate of thousands of misinformed, disinterested decisions about the worth of education, or the predictable outcome of the structure and operation of education systems at national and local levels" depends on whether or not one considers educational attainment to be more attributable to genetic endowment or the structural inequalities of class society (Byrne, et. al., 1975B:30-31).

Halsey reminds us that "(a)n adequate theory must also attend to those structural inequalities of resource allocation which are integral to class society" (quoted in Byrne, et. al., 1975B:31).

The class-cultural model of educational attainment, like human capital theory, is part of the paradigm of functionalism. In this paradigm, value consensus (or ideological agreement) on basic societal issues is assumed to be the norm. It is considered to provide a moral foundation for society, facilitating stability and harmony between groups. Education is considered a key institution for the maintenance of this consensus. 225

A thorough critique of functionalism is beyond the scope of this thesis. Rather, we are pointing out some faults with the paradigm which we consider to be critical. Human capital theory, the technical-function theory of education, and other sociological and psychological theories which are part of the paradigm of functionalism have faults in theory and in research which contribute to the lack of congruence between the goals of public policies developed under the influence of these theories, and the results.

If legislation is based on inadequate sociological of psychological theories either about learning potential or the interrelationship between systems of education and social structures, it is inevitable that the manifest goals of such legislation will not be realized (Byrne, et. al., 1975B:18).

This does not mean that the fault for the failure of public policies lies entirely with social science research. Politicians often are either not aware of research results or are unable or unwilling for a variety of reasons to utilize them.

> It is a characteristic of public sector decisionmaking in the United Kingdom that it is ad hoc and partial. By this I mean that it is frequently a matter of responding to events rather than anticipating and preparing for them, is usually lacking in basis of empirical knowledge and is always pursued without due regard to many crucial interrelationships (Peston, 1969A:70).

When graduate unemployment and underemployment became widespread in the early 1970's in North America, a manpower planning approach to educational planning and job placement was undertaken by the Canadian and United States governments. With a significant drop in aggregate demand, the economic opportunities provided to individuals by their education alone were revealed to be very poor, with the exception of

a few occupations -- primarily in the health care field.

Human capital theorists stated that the problem of unemployment and underemployment was not due to lack of aggregate demand, but to the quality of the labor supply. In other words, individuals were not making rational decisions "in touch with the market place" when deciding what type of education or training they would pursue. Human capital economists persisted through the 1970's in stating that there was still demand for highly educated people to fill senior positions in the labor market. The problem of unemployment and underemployment was still conceptualized by human capital economists as one of labor quality not lack of jobs.

The limits of educational reform, and of life chances, are well set by the labor market -- especially in times of poor demand. This has been realized by many since opportunities for employment have become much scarcer for most people during the 1970's.

The labor market itself must be seen within the class structure of society. While analysis of the labor market¹⁹ lies outside the scope of this thesis, it is relevant to our analysis to realize that structural characteristics of the labor market are more important than worker attributes in the determination of who gets what jobs and for what rewards. The hierarchical structure of occupations in advanced capitalist societies serves a political function (e.g., enhancement of capital accumulation).²⁰ The political function is obscurred by emphasis placed on economic explanations.

While some human capital economists have conceded that cyclical unemployment, not education, is the most important factor in explaining income inequality in the United States²¹ the emphasis of mainstream human capital analysis has continued to be on the importance of

matching demand with an appropriate quality of labor supply. Human capital theory completely ignores the political functions of education and labor market dynamics. Questions of educational attainment, equality of opportunity, and economic inequalities cannot be adequately understood by focusing on purely economic considerations and ignoring the political ones.

...(I)t is to be expected that education will reflect what key groups in society regard as important educational aims and these in turn will reflect what such groups regard as important social and economic needs. Historically, of course, such needs have always been defined in labour market terms...

It does need to be emphasised, however, that the labour market is a fundamental constraint on the education service of any society. In a given 'ocality the labour market sets the boundates for social-class formation (Byrne, et. al., 1975B:45).

...(L)ife chances in education are deeply implicated in the logic of operation of a social market economy. What can be provided as a framework of opportunity is determined, ultimately, by the character of local labour markets. What is provided as education is the outcome of political compromise. What can ultimately be achieved by any group of children is limited by the hierarchical character of work organisations. In short, educational attainment is a public issue not a private achievement, an artefact of the distribution of power in society rather than the distribution of intelligence (Ibid.:46).

Carnoy (1971B:1/17) reminds us that

(e)conomists first became interested in education because of its role as an input into economic growth...the treatment of education as an investment in human capital and the rate of return analysis associated with it is seen by many economists as the basis of understanding the relationship between education and the economic system. Research by human capital economists indicates that "... average pecuniary earnings of individuals with more schooling are higher than earnings of those with less schooling" (Carnoy, 1971B:1/17).

It is also a well-known fact that if education is publically subsidized, many more people will attend school than would if it was not subsidized. Carnoy (1971B:1/17) points out that "...the contribution of schooling to different groups in the society varies according to their political power or their ability to gain control of resources associated with elite-oriented education."

While human capital theory focuses on an economic analysis of education which appears to be scientifically-based and hence politically neutral (though amenable to modern liberal concepts), sociologists and political economists recognize the political function of education. Carnoy (1971B:1/6) points out, however, that most sociologists have recognized the elite bias of education but have not extended sociological elite analysis²² to the economy.

Social scientists have long recognized this elite bias in the educational system, but they have chosen to ignore that a system with this bias bases its strategy of skill production primarily on political grounds -on income and status considerations -- and only secondarily on considerations of maximizing economic growth or the potential social contribution of all members of the society (see Smyth, 1970). Economists assume, tacitly, that all students (parents) can choose the amount and nature of skill production desired. They do not recognize that the institutions themselves determine students' future roles through tracking, property tax base financing, and the nature of school social relationships (on this last part, see Friedenberg, 1965).

Carnoy (1971B:1/7-1/8) has developed

...a dynamic model of rates of return to different levels of schooling based on the elite-formation school system... The model explains how a monopoly rent to investment in schooling shifts over time to higher and higher levels of schooling and is eliminated when nonelites dominate the enrollment of any schooling level.

The political function of education in underdeveloped countries, e.g., elite-formation, has been recognized by economists but only as a necessary step in modernizing the economies.²³ Most of the literature dealing with elite formation in underdeveloped countries has been written by sociologists, not economists.

Human capital theory ostensibly provides politicians, academics, and the general public with an economic rationale for the political intervention in education which during the 1950's and 1960's resulted in large-scale expansion, mostly at public expense, of the educational system and an increase in participation rates and educational attainment. In the 1970's, human capital theory provided the economic rationale for the political decision to drastically lower public expenditures and subsidies to education and a disassociation of the state from responsibility for individual educational decision-making and eventual occupational attainment.

This economic rationale leaves out the political rationale for such decisions and policies. Although sociologists have indicated the beneficial effects of expanded education on social as well as economic development, and on the contribution of higher education to upward mobility if not entirely to elite formation, they have not paid much attention to their limits. Not much attention has been

1.2

to the problem of conflict between human capital elites and physical capital elites.²⁴

Comprehensive discussion of elite analysis and the political functions of education are beyond the scope of this thesis; however, cursory reference to this issue is necessary in order to highlight the inadequacy of the economic focus of human capital theory.

Ignoring the structure of the system has influenced the type of questions asked by economists and often caused misinterpretation of their results (Carnoy, 1971B:1/6).

This is why, for instance, rates of return to investment in education in the United States, Canada, Western Europe, and underdeveloped countries are similar despite being "...at different stages of development, with significant differences in the rate of expansion and average level of schooling..." (Carnoy, 1971B:1/6). The seemingly a-political nature of human capital theory also accounts for why the human capital explanation for differences in rates of return to different groups in society with identical levels of education are so weak and implausible. Carnoy also state that some of the problems encountered in production function analysis of education may also be due to the lack of consideration of differential education for elites and non-elites.²⁵ Viewed from a political perspective, "(s)chools may be efficiently producing a different set of outputs for elites than for non-elites" (Ibid.11/7).

Although the focus of human capital theory has been primarily . on economic relationships between education and occupation, it did provide support for the political rationale for expansion, democratization, and subsidization of post-secondary education during the late 1950's

and early 1960's. This political rationale was two-fold. On the international level, there was the outright competition for scientific, technological, and ideological dominance. Nationalism was strong during the 1950's and early 1960's. In addition, there was also strong public pressure on politicians to support public subsidization of post-secondary education. This social demand for expanded tertiary education was related to the public perception of post-secondary education as the institution most suitably geared to employment preparation, and hence, as the new vehicle for social mobility. The idea that post-secondary education is necessary for productive employment and as a means of achieving social equity is not new. Subsequently, with the development of the investment model of education by Schultz and the resulting growth and impact of human capital research by economists, politicians were provided with the economic rationale which suited their political rationale for direct public control over and subsidization of post-secondary education.

It was definitely the use of post-secondary education as a social escalator and, for a time, as an indispensable tool in the race with the Russians that justified the unprecedented infusion of resources into post-secondary education both in the United States and in Canada.

...to put it rather bluntly, the introduction of an overwhelming public support has resulted in the politicalization of post-secondary education. Since the government is the main agent in this process, it will, whether it likes it or not, bear the brunt of responsibility for whatever uses and abuses our society makes of post-secondary education (Ontario. Commission on Post-Secondary Education in Ontario, 1971B:13).

Thus, the political role of education supercedes the economic role. Income and status distribution for social reproduction of elites

and non-elites are the primary political goals of education; the economic and social contributions of all members to society are secondary goals (Carnoy, 1971B:1/6).

The expansion of education has led to rising expectations which, for many, have not been met. The role of schooling in society and the benefits to individuals of increased educational attainment have come under critical reassessment by governments as well as by individuals. Research has proven expectations of increased schooling of the work force leading to increased economic growth, a lessening of the inequality in the distribution of personal income, and an increase in social mobility to be false in both the developed and the underdeveloped countries (Carnoy, 1979B:1).

Attempts by governments to balance economic efficiency and social equity since the late 1950's have not resulted in a relative, over-all improvement in the standard of living for most workers. While it was possible for governments to sponsor programs designed to enhance social equity at the expense of some economic efficiency during the 1960's, this has not been the case during the 1970's. The result of this reversal of priorities has been a generalized, political crisis -- a crisis of legitimation.

The legitimation of increasing economic inequality has come under question. It is a political problem because "...legitimation of economic inequality is critical to the political defense of the fundamental institutions which regulate the U.S. economy" (Bowles and Gintis, 1977B:248).

The legitimation of unequal rewards has ostensibly been based on merit, which itself has generally been measured in terms of educational attainment. This way of stratifying society has itself come under question along with the legitimacy of educational credentials.²⁶

There is an attack on credentialism both in academic circles²⁷ as well as in schools themselves, reflected by the high rates of truancy and crime. The authopity and legitimacy of education has declined. During the 1970's, "...schools have lost their authority in the minds of some people" (Squires, 1979B:197).

Affirmative action programs designed to enhance equality of opportunity have come under attack for undermining meritocratic principles,²⁸ and for lowering the standards and quality of education.²⁹ Racism and class conflict have also intensified as education looses its status as legitimator of the social order³⁰ (Squires, 1979B:197).

If we accept that "....schooling is a servant of the economy and society...[and that] the economic and social structure of all but a few societies are dominated by elites which [sic] use schooling to further their own needs and goals" it is not difficult to understand that the "...purpose and effect of school expansion cannot be separated from elite control: the benefits of additional achooling accrue to the elite. Non-elites are left largely with illusions and disillusions" (Carnoy, 1971B:0/2). Likewise, "...the argument that increased schooling leads to improved social mobility and income distribution is ambiguous at best...The more unlike in culture the elites and nonelites are, the less possible it is to reach equality" (Ibid.:0/2;0/4).

Just as human capital theory and the technical-function theory of education provided an economic rationale for the state to intervene in education, providing huge public subsidies and encouraging higher rates of educational participation for the non-elite, so those same

theories are being used as a rationale for reversing these very policies. In the former case, the expansion of education during the 1950's and 1960's was consonant with the social demand for education, with the high demand for well-educated workers, with high rates of economic growth, and with seeming v increasing prosperity for all. In the case of the latter, however, the "cooling-out" function of education has not been working smoothly.³¹ Despite the highly stratified and diversified nature of education nowadays, underemployment and unemployment are problems which chip away at the authority and legitimacy which education and, indeed, the state itself have had in perpetuating the myth of an open society characterized by equality of opportunity for all.

Contradictions which have intensified throughout the 1970's will not be resolved by state abdication of responsibility. The emphasis on individual decision-making and responsibility is illusory and temporary. We share Gonick's (1978B:127-128) opinion that

...some form of central economic planning is inevitable in all industrialized countries. Without it, capitalism as an economic system is unlikely to survive. The strains, contradictions and problems...demand some new rationalizing effort...The question that matters today in Canada is not planning versus no planning, but planning by whom and for what objectives.

Education has taken much of the brunt of the turmoil, controversies, contradictions, and conflicts which have surfaced in most advanced capitalist countries since World War TI. The debate which raged in the political arena as well as in academe between social scientists about the relevance and, indeed, fitness of the educational system in meeting the "needs" of the economy, of society, as well as

Q.

of individual students, was primarily a clash between conservatism and liberalism and intensified by the ideological warfare with the Soviet Union.

That debate of the 1950's and 1960's continues today but under very changed socio-economic and political circumstances as well as under very different terms. The issue of the fitness of the educational system serving the "needs" of the economy continues but a more important issue is that of the function of education. Now we are being told that education is for self-fulfillment and culture, not necessarily for employment preparation; that life-long education means we no longer are denied equality of opportunity if we fail because the doors are always open to go back and try again. It is still too early to know whether or not these new strategies will work, whether expectations will be lowered so that adjustment to more oppressive, alienating conditions is achieved without more turmoil.

One thing is clear. Education has been used, and continues to be used by the political elite of the ruling class as a key institution for the legitimation of changing social, economic, and political realities and to generate ideological support for the kinds of public policies and strategies used to minimize the impact of these crises.

CHAPTER VII

Footnotes

¹Central to the theory of "post-industrial" society are the posited effects of science-based technological innovations on the occupational structure (as well as on the structures of power and authority) and the focus on the university as the key institution in society (because it produced theoretical and practical knowledge and technical skills crucial for the maintenance and advancement of a "post-industrial" society).

In the "post-industrial" society, a new factor of production, expert knowledge, emerges and eliminates the "antagonistic relations of capitalist production". Bell (quoted in Thatcher, 1978B:194) says that "in effect, not labor power (and the working class) but science (and knowledge classes) is the 'decisive factor' in the growth of the productive forces of society."

Thus, "post-industrial" society is the result of a "...gradual evolution from a capitalist to a technocratic type of economy" (Thatcher, or 1978B:195). It is post-capitalist, and as such, class is no longer determined by ownership of the (physical) means of production. The antagonistic categories of capital and labor are rendered anachronistic.

In the technocratic economy, functional efficiency, evaluated in terms of technical skill levels, provides the commanding criteria for the allocation of individuals to authority positions (Thatcher, 1978B:195).

Production and administration in post-industrial society are automated, computerized, and bureaucratized. Expert knowledge is the scarce factor which becomes the single most important and powerful attribute.

Technical advancement in Western industrialized capitalist countries is science-based; so higher education has become essential for the development of technical skills and knowledge which have become the new basis for authority and power. Post-industrial theorists assert that power, which belonged to the owners of physical capital in capitalist society, has shifted to a highly educated, technocratic elite in the post-capitalist, "post-industrial" society.

Furthermore, post-industrial society is posited to be meritocratic. Power, status, and wealth are all to be earned through achievement based on mastery of scientific knowledge and technical skills -- not based on ascriptive or particularistic factors.

Expert knowledge based on science and learned in institutions of higher education and the principle of merit based on achievement rather than ascription are central concepts common to post-industrial theory in sociology, human capital theory in economics, and the technical-function theory of education. These theories focus on the key role of schooling in the production of knowledge, as the institution geared to the production of cognitive and technical skills, as the vehicle for upward mobility, and as the ultimate legitimator of the social order and one's place in it.

Bell (quoted in Karabel and Halsey, 1977B:607) describes the post-industrial society as a meritocracy in which

(d)ifferential status and differential income are based on technical skills and higher education. Without those achievements one cannot fulfill the requirements of the new social division of labor which is a feature of that society. And there are few high places open without those skills.

...Technical skill becomes a condition of operative power, and higher education the means of obtaining technical skill. As a result, there has been a shift in the slope of power, as, in key institutions, technical competence becomes the overriding consideration...Increasingly, the newer professional occupations, particularly engineering and economics, become central to the technical decisions of the society. The post-industrial society, in this dimension of status and power, is the logical extension of the meritocracy; it is the codification of a new social order based, in principle, on the priority of educated talent.

In social fact, the meritocracy is thus the displacement of one principle of stratification by another, of achievement for ascription.

In the words of Daniel Bell (quoted in Karabel and Halsey, 1977B:608), "(t)echnical skill, in the post-industrial society, is what the economists call 'human capital'."

²The poor includes individuals from minority ethnic and racial groups, women, the elderly, the young, the disabled, and white males with less than a high school education -- whether employed or unemployed.

Political and social equality of all individuals in society stems from liberal philosophy and English utilitarian philosophy. The equality to which most politicians and social reformers of the 1960's and 1970's refer is social and political equality. Economic equality for all is not a goal. Economic inequality which is not
based on achievement and merit, however, is seen as illegitimate.

⁴One of the main ideas of the New Deal was that increased government spending would increase effective (aggregate) demand, dampening, if not eliminating, the downward phases of the business cycle. This would stabilize the economy and the resulting economic and social climate would be more favorable to both capital and labor. Unfortunately,

(t)he New Deal managed to push government spending up by more than 70 percent, but this was nowhere near enough to bring the economy to a level at which human and material resources were fully employed... unemployment was still well above 15 percent of the labor force (Baran and Sweezy, 1966B:175).

Only the outbreak of World War II saved the American economy.

^bGanada has a unique position in the Western capitalist world because of its hinterland, dependent status vis-a-vis the United States (and, previously, Great Britain). Canada did not become an industrialized capitalist state until after World War II. Though Canada has since entered the phase of monopoly capitalism, it is a very peculiar form because of the branch plant economy and the fact that nearly all of the large, multi-national corporations are foreign-owned. So, although Canada is a "developed", modern, industrialized, capitalist society, it is very much an "odd" case vis-a-vis other Western capitalist states precisely because of its hinterland status as a nation.

⁶State economic planning to minimize, if not eliminate, severe fluctuations in the business cycle (as well as the resulting socioeconomic and political crises) includes the development of appropriate discretionary fiscal and monetary policies to regulate aggregate demand, provide a favorable invegtment climate (e.g., high profits), high and stable employment, relative market stability, and a rising standard of living, as well as labor and social welfare legislation to provide a minimum level of socio-economic security for all individuals.

Justification for state intervention in the economy was "...based on the Keynesian view that the general level of activity in the economy was governed by the level of investment and the level of effective demand" (Wolfe, 1977A:252). Accordingly, it was argued that the state should maintain a high level of effective demand "...to improve private firms' expectations about the future rate of return on their potential investment projects, and thereby to guarantee a high and stable level of employment and income for the whole economy" (Ibid.:253).

Discretionary fiscal and monetary policies (deficit and surplus budgeting) are used to regulate economic growth -- to stepilize it.

Governments have alternated between expansionary and restrictive fiscal and monetary policies depending on the relative severity of economic and political crises as well as in response to the need for balancing accumulation and legitimation functions in light of pressure from both capital and labor (Wolfe, 1977A:269).

Baran and Sweezy (1966B:145) consider the tremendous surplus of goods and capital to be characteristic of advanced (monopoly) capitalism and that "(i)t has been through changes in the overall total of spending that government has exercised its greatest influence on the magnitude of effective demand and hence on the process of surplus absorption." Government spending in the United States has increased absolutely and as a percentage of GNP since 1903 (Ibid.:146). Indeed, there has been "...a strong tendency to keep pushing up the level of government spending"; "...more government spending means higher incomes out of which increased taxes can be paid" (Ibid.:163).

Growth of the Canadian money supply during the 1970's has more than doubled the rate maintained during the 1960's (Wolfe, 1977A:267). Canadian (federal) government total cash requirements have increased from a deficit of -\$376 million in 1960 to a deficit of -\$1,507,000,000 in 1975, with a deficit budget from 1970 through 1975 (with the exception of the Nov. 18, 1974 budget, which had a surplus of \$250 million)(<u>Ibid</u>.:266-267).

O'Connor and Offe point out that state intervention in the economy and responsibility for its management have reduced the risks for both capital and labor. By taking on the functions of legitimation and accumulation, the state has become answerable to both capital and labor. Capital is subsidized, guaranteed high levels of demand, and provided with relative market stability; and labor is promised employment, and for those who are unemployed, the private costs are reduced and subsidized by the state. "Full" employment policies provide for a sustained, high level of demand which is required by capital for continued investment. "Full" employment also acts as a good legitimator for the system; it minimized class conflict and worker insecurity and inculcates worker loyalty (Wolfe, 1977A:254).

Social welfare programs and services have

...also contributed significantly to maintaining high levels of effective demand throughout the post-war period and thus to the general level of economic buoyancy. The stabilizing effects of income transfer programs are even greater when employment begins to fall because of the sutomatic rise in spending on programs such as unemployment insurance (Wolfe, 1977A:253).

In addition to fluctuations in government spending, the state uses fiscal policy to stimulate or dampen the economy by lowering or raising corporate and/or personal taxes. Tax incentives are developed

... to reduce the effective cost of capital to corporations and thus increase their inducement to invest. The various devices employed, such as accelerated depreciation allowances, investment allowances, lower corporate tax rates, and tax incentives for research and development, have all had the effect of reducing the level of effective taxation on those corporations willing to undertake research or investment projects at the time and of the magnitude desired by their respective governments. In periods of growing pressure on corporate profit rates, tax policy has been used consistently to maintain a stable level of post-tax profit rates. Corporate tax policies have thus played a central role in the maintenance of profitable accumulation in advanced capitalist economies (Wolfe, 1977A:253).

Furthermore, Wolfe (1977A: 253) states that

(t)hese policies have been complimented by a series of others, such as selective nationalizations of less profitable industries, provision of subsidized sources of credit, improved regulation of international financial and monetary relations, and the reduction of international barriers to trade through successive rounds of the General Agreement on Tarrifs and Trade. All of these measures have expanded the role played by the state in the economies of advanced capitalism and have helped to maintain the conditions for profitable capital accumulation.

⁷The underlying belief in the establishment of the welfare state is that economic growth (stimulated or sustained by state fiscal and monetary policies) would enable the state to afford more social benefits which would, in turn, further stabilize the economy.

^oMichel Kalecki notes that in addition to an economic business cycle, there is now a "political business cycle" (see Wolfe, 1977A:256).

⁹There is a legacy of progressive education which goes back nearly a century in North America which has had some influence on educational philosophy and government educational policies. It was primarily a result of the progressive education movement that power (administrative and financial) over schools was centralized and consolidated into provincial and state departments of education 241

J

in Canada and the United States respectively. The new development in the 1950's was the growing acceptance of belief by liberals that the federal government should intervene directly into the affairs of education, which is constitutionally not a federal jurisdiction. This direct intervention by the federal government in education was considered necessary as part of the functioning of the modern liberal welfare state.

¹⁰Increasing economic efficiency was a foremost concern, particularly of the United States, because of the political crisis over Soviet advances in science and technology. Human capital theory provided a key to overcoming this crisis by providing a theoretical rationale for increasing economic growth, productivity, and profits.

¹¹The converse of this, downward mobility of some members of the dominant (or "advantaged") group, is implied but never dealt with explicitly.

¹²The fiscal crisis of the state in advanced capitalist countries began in the late 1960's and has continued to worsen. The state has increased expenditures much faster than its revenues. At the same time, the economic growth rate has dropped significantly. Public debt has reached astronomical heights. There are several factors which have caused this fiscal crisis. A sustained economic slowdown (recession) with high unemployment, the new phenomenon of inflation, increased international competition, declining profits, decreased investment, decreased research and development, a drop in the standard of living, increasing socio-economic inequality, high interest rates, a drop in the exchange value of the dollar (especially the Canadian dollar), balance of payments deficit, and a shift in state revenue from corporate taxes to personal income taxes have all contributed to this crisis.

"According to O'Connor, the modern capitalist state has increasingly socialized many of the costs of production (e.g., education, research, health, etc.), but must continue to rely on taxes to finance its expanded activities" (Silverman and Yanowitch, 1974B: 393). Many of these state expenditures are subsidies to private capital. However, the social and reproductive costs of production have also been socialized (O'Connor, 1974A:434). The fiscal crisis of the state occurs because "the state has socialized many costs of production but has not socialized profits" and because the cost of inflation is passed on to 'the state sector and the competitive sector of the economy; the state has socialized the production costs and has guaranteed profits for private enterprise (Ibid.)

The severity of stagflation and the inability of advanced capitalist states to resolve this economic crisis has resulted in a new political crisis. It is a crisis of legitimation as well as accumulation. By intervening in the economy, the state has attempted "...to reduce the historical costs of the market to both capital and labor..." and in doing so, "...the state has transferred the sphere of class conflict from the economic order to the political order" (Wolfe, 1977A:255).

¹³Inflation is a perplexing problem for economists. Samuelson (1971, quoted in Fry, 1979B:35) states that

(c)reeping inflation is the malaria of the modern mixed economy. Like malaria it is uncomfortable to live with and just will not go away. But unlike malaria, there seems to be no known cure for creeping inflation that is better than the disease.

Inflation became the new and primary economic problem of the late 1960's. Governments concentrated on developing policies to lower inflation even though they resulted in increased unemployment because uncontrolled inflation was deemed to be the most detrimental to the economy in the long-run. The change of focus from unemployment to inflation can also be considered part of the change from competitive laissez-faire capitalism to monopoly capitalism.

State regulation of aggregate demand within a structure of union-management collective bargaining in monopolistic industries has shifted preoccupation from economic depression to the problem of inflation (Silverman and Yanowitch, 1974B:392).

¹⁴High expectations held by the general public during the 1970's presented a particularly problematic political problem to liberal and social democratic governments. Lowering these expectations was a political necessity if the crisis of legitimation facing the state was to be resolved peacefully.

Prime Minister Trudeau of Canada was quoted in <u>Maclean's</u> (Jan. 10, -1977:8) as stating that

...a large part of my message as a politician is to say: we have to put an end to rising expectations. We have to explain to people that we may even there to put an end to our love for our parents or old people in society, even our desire to give more for education and medical research (quoted in Finkel, 1977A:364).

Bell (1973) has aptly remarked that "(t)he revolution of rising expectations is also the revolution in rising <u>ressentiment</u>" (quoted in Karabel and Halsey, 1977B:626).

¹⁵It has been explicitly recognized and publically stated by politicians and economists that the public policies and programs devel-

oped to lower unemployment, eliminate poverty, and improve the standard of living depend upon a high, if not continually increasing, level of economic growth. Hubert H. Humphrey (in Everett, 1966B:8-9) stated that

(f)irst, aggregate demand must be maintained at a high level. A downturn in our economic growth rate would undo all the other programs which might be conducted. Educating and training men for jobs that do not exist is futile.

Since the passage of the Employment Act of 1946, it has been recognized explicitly that the federal government has a primary responsibility for maintaining aggregate demand...[and] for maintaining employment, production, and purchasing power.

Lack of aggregate demand has been the reality since the late 1960's. In addition, automation has contributed substantially to the drop in demand for workers since the late 1950's. It is increased capital substitution for labor (e.g., automation) which is now the greatest threat to workers.

What automation is beginning to do already [1966] is to freeze those conditions in our society which tend to perpetuate the sons and daughters of the poor in the cycle of poverty and to perpetuate the sons and " daughters of the middle-class and wealthy in the cycle of middle-class and wealth. The basic reason for this ossification is that, for the first time in the history of man, education is placed squarely between man and the work which is his means of livelihood... Unless the link between dropouts from/education and young people from low-income homes is broken before the full impact of automation makes itself felt in the world-of-work; automation will virtually eliminate social mobility from one generation to the next (Reid, 1966A:53-56).

While automation was common and widespread during the 1950's, automation did not result in high "technological unemployment" during the late 1950's and early 1960's because the world-wide economic expansion resulted in a rapid expansion of public sector employment. Thus, the public sector was able to absorb the labor which was being replaced in the private sector by capital.

During the late 1960's and 1970's, the world-side economic deterioration caused by stagflation has caused the public sector to

stagnate as well. Governments are cutting back on their public sector expenditures. Consequently, "...concern over the impact of technology on jobs has once again surfaced" (Gonick, 1978B:141).

Jack Shingleton, Director of Michigan State University Placement Services stated that

It is becoming increasingly evident that unemployment is not the number one problem plaguing college graduates in the market place. Rather, it is underemployment. The prospects of underemployment seem to increase with each passing year (MSU Placement Services, 1974B:1).

Barry Bluestone (1977A:102) adds that

The inadequate incomes of most of the working poor are not of their own making. If we are to blame them for anything it must be for not having the good fortune to complete an education topped off by a college degree. Rather we must blame the economic system which in too many instances provides less than an adequate job for those of adequate talents.

The exclusion of well-educated people from productive employment in Canada is lamented by William A. Dyson in 1971 (quoted in Canadian Committee of the International Conference on Social Welfare, 1972B:10):

Quietly our aggregate wealth, owned by someone grows. All views to the contrary, the issue is not that Canada has insufficient wealth;...The central problem is "people", not dollars: people being increasingly excluded from access to our riches, while a few others, here and abroad, are increasingly favored. Even worse, it is the exclusion of these people -- who grow in number and now include businessmen, scientists, and engineers exclusion from work, from status, from identity, from meaningful life.

According to Carnoy (1971B:1/51), increased educational attainment of workers has resulted in greater capital substitution of labor, greater unemploument of the less-educated, and underemployment of many of the better-educated.

If wages for job categories are inflexible downwards, employers substitute higher skills for lower skills in these jobs.

いいたが、いっていたろうの時をないない

The greater the expansion of schooling, the greater the return to physical capital and the greater the average unemployment of lower skilled, less-educated workers.

¹⁶The crisis of the welfare state is a result of the continuing, unresolved fiscal crisis of the state in advanced capitalist countries. Since 1968, when there was a sudden, sharp increase in unemployment and the introduction of the new problem of inflation (the beginning of stagflation), governments have been unable to stabilize prices or lower unemployment rates. Voluntary wage and price controls in 1969 and 1970 were unsuccessful in Canada. The government increased taxes and began cutting back on public spending as well as introducing tight monetary policies during this period in order to slow down the Canadian economy and reduce inflation. The exchange value of the Canadian dollar was unpegged. Though inflation was slowed, unemployment increased. Constituent fiscal and monetary policies created a political crisis

, so that the Canadian government was pushed to reduce tax for income brackets (Wolfe, 1977A:265-268).

to have inflation rates, unemployment rates have continued to lower inflation rates, unemployment rates have continued to the state. Thus unemployment has become a political as well as an economic problem. Continued high unemployment is viewed as a political failure of governments (Wolfe, 1977A:255). This political crisis arises out of inconsistencies "...between legal and political equality guarant by the state and socio-economic inequality protected and maintained by the state" (Ibid.:257). These inconsistencies are revealed and exasperated by inflation. Hence, inflation is viewed as the primary economic problem because it has heightened the contradiction between the accumulation and legitimation functions of the state.

As a result of the political crisis in Canada, the government pursued expansionary monetary and fiscal policies between 1971 and 1973 in an attempt to provide incentives for investment and, hence, more jobs. The economy was stimulated, and in 1973, Canada experienced the "...largest annual increase in GNP since 1966, a moderate fall in the rate of unemployment, and a significant rise in the G[onsumer] P[rice] I[ndex]..." (Wolfe, 1977A:273). Furthermore, by 1974, labor's share of national income dropped from 72.8% in 1970 to 67.9% and that of capital rose from 11.8% to 16.1% (Ibid.:274). Inflation, however, was on the rise again.

The weak recovery experienced during 1971-1973 was followed by a severe recession in late 1974 and 1975 which marked the end of economic buoyancy around the world. There was a massive economic shake-up in the world economy from which there have been only some weak recoveries. Since 1975, the world economy, and the Canadian one in particular, has been characterized by increasing inflation, increasing unemployment, depressed investment, protectionism, and government susterity (Gonick, 1978B:109-112). Gonick (1978B:108) remarks that the 1975 recession was not only "...the most severe since the Great Depression", but also "...the first generalized recession since the end of the Second World War, the most serious recession in forty years precisely because it was synchronized throughout the capitalist world."

Industrial production in Canada fell by 4.6 per cent between 1974 and 1975, in the U.S. it fell by 9 per cent; in Italy and Belgium by 10 per cent; in Japan by 11 per cent. Exports declined for the first time in fifteen years. Unemployment rose to postwar highs (Gonick, 1978B:108).

Thus, the fiscal crisis of the state has deepened to crisis proportions. Liberal, social democratic governments have had to severely cut back public expenditures, particularly funding of social welfare programs.

In 1965, \$755 billion was spent on social welfare by all levels of government in the United States. This accounted, for 39% of all public expenditures. Nearly 42% of this, or \$31 billion, went to the pre-transfer poor (155 million households). By 1968, the total had risen to \$109 billion but its proportion of all public expenditures remained the same (39%). In 1972, the amount spent by all levels of government on social welfare had risen to \$185 billion and its share of all public expenditures had risen to 46%. About \$79 billion of this went to the pre-transfer poor (Plotnick and Skidmore, 1975B:172).

In Canada, widespread poverty amidst affluence continued to be a problem despite increased social welfare expenditures. A Special Senate Committee on Poverty was appointed to study the problem and make recommendations to Parliament. The report was published in 1971. It presented no real change in policy. However, a group of researchers who had resigned from the project because of differences with the Senators published their own report the same year. The Real Poverty <u>Report</u> proposed far-ranging reforms, from the establishment of a comprehensive earnings policy to combat gross income inequalities to policies to curb the independent powers of corporations to make economic policy, the centralization of collective bargaining for workers, and the expansion of manpower training programs. The Canadian government, however, made no change in social welfare policy except to cut back expenditures.

With unemployment rising throughout the 1970's, unemployment insurance and welfare budgets have been exhausted. Because of the continuing fiscal crisis, the state has changed eligibility requirements and benefit levels of those programs. The National Council of Welfare (1971B:24) points out that because the Canadian government pay 50% of provincial welfare costs, conflict between the federal and provincial governments over welfare budgets increased at the same time the need for social welfare benefits has increased.

When a sudden, massive increase in unemployment results from federal Government fiscal and monetary policy to attempt to combat inflation as at present, welfare rolls increase accordingly... The response of this increase in welfare rolls, in provinces and municipalities across Canada, has been to cut back on already inadequate benefits and to apply eligibility requirements even more repressively (National Council of Welfare, 1971B:24).

Another phenomenon which has intensified the fiscal crisis of the state in Canada is the fact that since 1960, there has been a progressive shift in the revenue base of all levels of government in Canada from corporate taxes to personal income taxes. The proportion of government (all levels) revenue derived from corporate taxes fell from 14.8% in 1960 to 9.5% in 1971 and that of personal income taxes rose from 25.5% in 1962 to 38.5% in 1975. Corporate income taxes did rise to 10.7% in 1975, but this rise was due to higher corporate profits and not to higher corporate taxes. At the federal government level, this is even more pronounced. Whereas corporate taxes accounted for 20.1% of federal government revenue in 1960, and personal income taxes for 37.3% in 1962; the former dropped to 15.3% and the latter rose to 49.6% in 1975. Wolfe (1977A:275) attributes this shift to the "...state's strategy to support and promote the conditions for the profitable accumulation of capital in Canada."

Nevertheless, this increase in personal income taxes has fueled inflation. It has also led to increasing labor militancy as workers have been squeezed by rising inflation, rising taxes, and rising unemployment. This has made it necessary for governments to develop crisis. management strategies in order to try to balance the functions of legitimation and accumulation which have increasingly conflicted openly (Wolfe, 1977A:276-277).

One such crisis management policy was the development of wage and price controls in the United States by the Nixon administration. When this program was implemented in August 1971, the inflation rate was 8.5% and unemployment was 5.9%. When the program ended in January 1973, the inflation rate was 3.3% and unemployment was 5%. By the end of 1973, however, the inflation rate had risen to more than 6% and the unemployment rate jumped to over 11% by early 1974 (Plotnick and Skidmore, 1975B:111).

In Canada, a program of selective wage and price controls was introduced in October 1975. The objective was to lower the 10.8% rate of inflation. The unemployment rate was 7.1% (Wolfe, 1977A:259). There was widespread resistance to wage and price controls, especially because the American program had not resulted in any long-term improvement.

Resistance to wage and price controls was particularly strong/ from labor. The struggle between labor and capital for division of the national income had been intensified by the persistence of stagflation and the contraction of the economy. The Canadian Labor Congress maintained militant opposition to wage and price controls. In March 1976, CLC members protested in Otrawa. A National Day of Protest was declared on October 14, 1976, when over one million CLC members went on strike (Wolfe, 1977A:282).

\$

.1

The Canadian government continued a policy of fiscal and monetary restraint between 1975 and 1977. Inflation was reduced but the rate of unemployment continued to rise. There were 300,000 more Canadians unemployed by the end of 1977 (Gonick, 1978B:123). Capital investment and spending increased, but at a low level (Wolfe, 1977A:283). Business was uneasy about the economic slowdown. Wage and price controls were phased out on April 14, 1978, ending on December 31, 1978. The program had been the cause of much conflict and embarrassment for the government.

¹⁷ The first serious proposals for a negative income tax were made by the conservative economist Milton Friedman in the late 1960's (Williamson, et. al., 1975B:207). President Nixon introduced the Family Assistance Plan in 1969, which provided for a negative income tax to poor families with children to replace the "...existing maze of welfare programs" (Plotnick and Skidmore, 1975B:181). The plan would replace all existing federal and state welfare programs as well as housing, education, and manpower programs. The savings in administrative costs were expected to be high.

The Family Assistance Plan passed the House of Representatives, which had a conservative majority, by a two-to-one vote (Williamson, et. al., 1975B 208). Liberals generally accepted the idea of a negative income tax and it was part of Senator George McGovern's presidential campaign in 1972. Liberals supported a negative income tax because they considered it "...less stigmatizing, more efficient, and more equitable than the present welfare system...as a way to deliver more in the way of welfare benefit per dollar of federal welfare spending" (Ibid.).

In 1972, however, the FAP was defeated in the Senate Finance Committee by strong liberal opposition. Although the welfare system was under attack by liberals and conservatives alike and although both liberals and conservatives supported the idea of a negative income tax to replace the existing welfare system, liberals defeated the FAP because they felt it had a number of serious short-comings. Liberals objected to benefits being restricted to families with dependent children, to generally low benefits, to the work provision, and to the fact that it would replace all other welfare programs and services (Williamson, et. al., 1975B:212, footnote 5).

Another attempt was made to restructure social welfare in the United States with a negative income tax int 1974, but it "...never reached the stage of a legislative proposal" (Plotnick and Skidmore's 1975B:181).

There has been no serious consideration of a negative income tax or anyother direct income redistribution plan since then, primarily for political reasons. The fear that a guaranteed minimum income would be a disincentive to work goes against strong belief in the work ethic. There is also strong public resistance to providing the poor with untied money. Many of the poor themselves were against the abolision of existing programs because their benefits would be reduced (Blotnick and Skidmore, 1975B:182). Furthermore, those who would carry the tax burden of financing the program are also the ones with the most political influence. If they believe the distribution of income will be changed significantly by a proposed reform of social welfare, they will resist it (Williamion, 1975B:207).

¹⁸The social literature dealing with the disappearance of class and class differentials which was so characteristic of post-industrial society theory and technological functionalism was based on a conceptualization of class, developed from the work of Warner, which use an aggregate score of a number of individual attributes (such as occupation, income, neighborhood of residence, education, etc.,) to determine class, rather than an analysis of the social structure In making this observation, Mackenzie (1973B:174-175) states that "(p)ogicions in the class structure thus become equated with status score or indices of style of life...(T)his is a fundamentally non-sociological approach to the analysis of class structure..."

In Mackenzie's (1973B:175) opinion, a more appropriate "...view of class structure is one which lays stress upon the economic bases of social class...to position in the division of labor" rather than on income or life-style. The producer function of individuals is more relevant to class determination than is their tonsumer fashion. "...(L) fie chances are determined not simply by level of income, or exposure to television or other suburban dwellers. Rather they are crucially influenced by the individual's situation in the system of production" (Ibid.).

Ĩ.,

"Neo-classical economic analysis of the labor market is based on the principle of marginal productivity. Perfect competition and market equilibrium are part of this concept. The unit of analysis is the individual. It is assumed that individuals try to maximize, the use of their time and energy, referred to as "worker sovereignty". Human capital theory is part of this paradigm.

There are other labor market theory is. According to the dual labor market theory, there are two separate labor markets, the "primary" and the "secondary". There is very little movement of workers between these labor markets. The working conditions, organization of work, wage scales, benefits, and desired worker attributes differ significantly between these two labor markets. (See Doeringer and Piore, 1971B; and Gordon, 1972B, 1977B).

From the dual labor market theory of the late 1960's, radical economists have formulated the radical theory of labor markets. This

draws from the dual labor market theory and the labor market segmentation theory formulated by Reich, et. al. (1975B). • The radical theory of labor markets uses Marxist class analysis and the theory of surplusvalue to explain labor market dynamics and the stratification and segmentation of the labor market by class, sex, race, and age.

²⁰The political function of the hierarchical structure of occupations in advanced capitalist countries is that of enhancing accumulation of capital. By decomposition of tasks and bureaucratic organization in the work place, automation is increased and important changes in the social relations of production as well as in the rate of surplus value occur which are of benefit solely to capital.

Spuires (1979B:91-32) presents this argument from a radical perspective wither well in the following lengthy quotation:

The decomposition of tasks and the bureaucratic organization of work remove knowledge and control of the productive process from those actually involved in production and place that knowledge and control in the hands of those at the top of the pyramid, the capitalists. Stephen Marglin summarized the function of these two changes in the organization of work in the following words:

Rather than providing more output for the same inputs, these innovations in work organization were introduced so that the capitalist got himself a larger share of the pie at the expense of the worker, and it is only the <u>subsequent</u> growth in the size of the pie that has obscured the class interest which was at the root of these innovations. The social function of hierarchical work organization is not technical efficiency, but accumulation. [Marglin, 1974A:62]

The implementation of modern machinery represents one change in the productive process which has evolved, ostensibly, for the perposes of expanding production, improving the efficiency of production, and increasing the total wealth of society in general. But there are important social implications involved in the use of modern machinery, according to the reficiency in the representation. As Braverman [1974B:147] argued:

Machinery comes into the world not as the servant of "humanity", but as the instrument of those to whom the accumulation of capital gives the <u>ownership</u> of the machines. The capacity of humans to control the labor process through machinery is seized upon by management from the beginning of capitalism as the prime means whereby production may controlled not by the direct producer but by the owners and representatives of capital. Thus, in addition to its technical function of increasing the productivity of labor -- which would be a mark of machinery under any social system -- machinery also has the capitalist system the function of divesting the mass of workers of their control over their own labor.

The radical challenge maintains that conventional theory ignores important social dimensions of work, particularly the primacy of capital accomulation and the exploitative relationships which result from that driving force, while accepting the technological justification for changes in the work process along with the notion that all groups of people benefit from those changes. Rather than viewing capitalism as one unique form of a modern industrialized society; conventional analysis, draws conclusions about industrialism in general from its observations of capitalist societies. By failing to take into consideration the social '9 dimensions of work which are inherent in a capitalist society but not in all industrial societies, the conventional analysis misses the most salient defining characteristics of American life.

²¹See Chiswick and Minner, 1972A.

²²Elite analysis conceptualizes, society in terms of elite versus non-elite groups. Carnoy **£19**71B:**1**/2) istates that

> ...elites are much more socially cohensive than non-elites. An elite is a relatively small group which comprehends its policy-making power (Mills, 1959). It is held together by this "gesponsibility", by its small size, and by oligopoly (collusive) arrangements in its use of resources (Baran and Sweezy, 1966). The analysis assumes that the elite is organized and exercises monopoly control over many facets of economic, political, and social life. A society may have competing elites with different interests. Each elite group is organized and attempts to structure institutions to favor its interests. On the other hand, the non-elite

generally does not exhibit social cohesiveness (or is prevented from doing so by the elite), so members of the non-elite behave largely as individuals, competing with each other, their differences rarely include any elimination of the elite/non-elite hierarchy.

²³Economists consider a Westernized, "modern" elite to be necessary for the socio-economic development of underdeveloped countries. Traditional, "unequeated" elites are considered "backward" and an impediment to development. Because access to the new elite is dependent upon one's formal education and merit, it is possible for members of the traditional non-elite to attain elite or sub-elite status. Thus, "...the educational system retains its legitimacy as the vehicle for social mobility in the eyes of the non-elite" (Carnoy, 1971B:1/61, footnote 2).

²⁴Carnoy (1971B:1/51) states that human capital elites and physical capital elites may or may not be the same. In the socialist and communiant states these ^{el}ites are usually the same; however, in "private enterprise" states, they may be different.

Conflict between human capital elites and phy elites in the United States has resulted in vocational education at the secondary school level. Carnoy (1971B:1/54) explains that the

(o)wners of physical Capital desire cheaper skilled labor and would like to avoid bearing the cost of on-the-job training. Owners of secondary and university level human capital wish to keep the wages of academical trained labor high. There is pressure from Ton-elites to gain access to the high wages associated with secondary training. The compromise among the elites with competing goals -- one wanting a rapid expansion of secondary school output, the other a much slower increase -is to increase secondary school trained labor, but labof which is not competitive with academic secondary output. Vocational training shifts the burden of skill development out of the factory into the public sector, subsidizing industry (owners of Physical capital). Although vocational schools attract those private students unable to enter the academic track, owners of physical capital are clearly better off with vocationally trained labor than if they had to train the labor themselves. The human capital elite is better, off because they have reduced the rate of increase of competing labor force.

The non-elites who get vocational training would appear also to be better off relative to a condition in which there were no secondary schooling available for them at all. However, they also appear to be aware that they have been assigned a low status role in the economic hierarchy...

In the United States, the vocational track of the general high school was conceived as a means of permanently assigning non-elites to certain roles in the economy (Lauter and Howe, 1970). Junior colleges, which in many ways serve the same vocational purpose today, have been described by Clark as also a means of "cooling out" students who may have otherwise demanded university training (Clark, 1965).

There artiging the aspects of this compromise between human capital and physical capital elites, especially for the non-elites. Those workers who attend vocational secondary schools become taxpayers who subsidize universities and specific industries. They have low status jobs in factories and often cannot get jobs which match their training.

It is the needs of the elite groups and not those of the nonelites which are met by the compromise of vocational education.

²⁵Carnoy (1971B:1/7) states that

Production function estimates for the school as a firm assume that schools produce education for children of elites and non-elites and are maximizing the achievement output of schools for both groups (Hanushek, 1968; Kiesling, 1967). Yet the combination of resources used seems to be more nearly efficient for the former than for the latter (Levin, 1970). The normal reaction is that the schools are blatantly discriminatory, but the functions estimated by these economists may be misspecified. Schools may be efficiently producing a different set of outputs for elites than for non-elites.

²⁶Since the passage of Title VII of the Civit Rights Act of 1964, it has become very difficult for employers in the United States to use education, sex, age, and race as the basis for employment. Sources (1979B:196-197) sees this erosion of traditional sources of investoricy and legitimacy leading to more conflict as the economy continues to stagnate. This conflict will fall along lines of race and times. ²⁷Rising educational stainment of workers since World War II has been accompanied by an increase in educational qualifications demanded by employers. As more individuals graduate from colleges and universities, hiring standards are upgraded. This educational inflation has been called educational "credentialism".

On a global level; there is a correlation between rising educational attainment levels and rising unemployment (Carnoy, 1975A). What has occurred in both the developed and underdeveloped countries is a decreasing ratio of labor to capital. Employers substitute those with more schooling for those with less schooling. Automation takes over many jobs previously done by the unskilled and less-educated (Spence, 1973A, 1974B).

Skill labelling and the "sheepskin effect" of educational credentials have become common (Leibenstein, 1965A, 1969A). Often, the job obtained has little to do with one's education. Employers use educational credentials as an index of trainability (Blaug, 1972A:71).

The American economist, Kenneth J. Arrow, asserts that education is a screening device for employment, that schooling contributes nothing to one's capabilitian for productive actitity. Educational credentials, however, are used to sort people according to traits which ensure high productivity (Arrow, 1973Aa). See also Taubman and Wales, 1974B; Fleming, 1974B; Sewell, 1971B; Rees, 1965A; Solow, 1965A; Blitz, 1962A; Hansen et. al., 1970A; Keats, 1965B; and Thurow, 1974A.

²⁸This is because individuals have been provided with educational and employment opportunities on the basis of ascribed traits like sex, age, race, and class rather than on merit (as measured by achieved traits a Furthermore, in providing quotas to enable proportional representation, affirmative action programs have denied qualified members of the dominant group. Apportunities.

²⁹Because strategies for affirmative action include the lowering of, and often the elimination of, entrance requirements for post-secondary education, post-secondary institutions are faced with the problem of either lowering academic standards and requirements for graduation and/or massive remedial and up-grading programs for these individuals. Some institutions, such as New York City College, have done both and have been severely criticized as a result (Pincus, 1974A).

³⁰From the point of view of elite analysis, expansion of ed cation to accommodate increased social demand for it presents the elite with a dilemna.

> Since education is held up by the elite as the logitimator of their rule and as the means to gain economic advantage and social

status, access to education becomes a crucial political issue. A dilemna for the elite is, how to satisfy the social demand for more and more schooling without giving away economic and policy-making power. The current structure of the schools solve this dilemna by restricting the high return (in terms of higher income) component of formal education to schooling Tevels accessible almost exclusively to the elite. -Within levels of schooling, the curriculum, teachers, and other inputs are more efficient producers of achievement and modernization for those who enter school with an already higher level of these outputs. The high return component of education is distributed to those with elite characteristics over those without. This leaves most of those who believe in education as a road to economic success holding more education but little absolute increase in income or political power. Potential disillusionment creates a second dilemna: if the elite provides education without economic (or at. least other intangible) benefits, they may be promoting their own overthrow. The elite can deal with this problem by becoming more hierarchical (dictatorial) or attempting to expand the opportunities for employment, or both. Again, however, the first line of defense is the ducational system itself. Schooling socializes the individual to believe that if he fails, it is his fault, not the system's (Kozol, 1967). The system is fair, he is taught. But he, the failure, has not met its needs or requirements. If the individual is convinced of this, even lack of employment will not seem unreasonable to him (see Illich, 1968). (Carnoy, 1971B:1/5-1/6).

깐

ţ.

Not only the poorly educated or the college drop-outs personalize their failure, but even individuals who are very successful in the educational system such as engineers and scientists who are not employed in their area of specialization see this in very a-political terms, often blaming themselves for not working hard enough to prove themselves at research, etc.

Carnoy (1971B:1/61, footnote 3) makes the point that many of the economically successful well-educated individuals are not necessarily members of the elite.

The advent of technology has provided another possibility for the elite to avoid political problems. The tendency to provide education in "neutral" or "objective" fields like engintering or other technological professions allows the non-elite to participate -- if they participate at all -- in the economic gains but makes them very a-political.

³¹The "cooling-out" function of education is not a concept devised by radicals. Rather, it is one reflecting the concern by technological functionalists for the rising tide of public expectations in light of the socio-economic realities. As Clark (1961:513) points out,

(a) major problem of democratic society is inconsistency between encouragement to achieve and the realities of limited opportunity. Democracy asks individuals to act as if social mobility were universally possible; status no to be won by effort; and rewards are to accrue to those who try. But democratic societies also need selective training institutions, and hierarchical work organisations permit increasingly fewer persons to succeed at ascending levels. Situations of opportunity are also situations of denial and failure. Thus democratic societies need not only to motivate achievement but also to mollify those denied it in order to sustain motivation in the face of disappointment and to deflect resentment. In the modern mass democracy, with its large-scale organisation. elaborated ideologies of equal access and participation, and minimal commitment to social origin as basis of status, the task becomes critical.

BIBLIOGRAPHY

Articles.

Abramovitz, Moses. "Economic Growth in the United States; A Review 1962 Article," <u>American Economic Review</u>, Vol. 52, No. 4, Sept.: 762-782.

Ackerman, Frank, et. al. "Income Distribution in the United States," 1971 <u>The Review of Radical Political Economics</u>, Vol. 3, No. 3, Summer: 20-43.

Alexander, Charles. "No Progress Without Pain: M.I.T.'s Lester 1980 Thurow Offers Some Hard Economic Choices," <u>Time</u>, Vol. 116, No. 12, Sept. 22: 78.

Allen, Michael Patrick. "The Structure of Interorganizational Elite 1974 Cooptation: Interlocking Corporate Directorates," <u>American</u> <u>Sociological Review</u>, Vol. 39, No. 3, June: 393-406.

Amin, Samir. "Toward a Structural Crisis of World Capitalism," 1975 Socialist Revolution, Vol. 5, No. 1, Spril: 9-44.

Anderson, C. Arnold. "Literacy and Schooling on the Development 1965 Threshold: Some Historical Cases," in Anderson and Bowman (eds.), Education and Economic Development. Chicago: Aldine Pub. Co.: 347-362.

Andrews, Jane Gribbon. "Off-Campus Education: Academic Colonialiam" 1979 Change, Vol. 11, No. 1, Feb.: 54-55.

Arrow, Kenneth J. "Higher Education As A Filter," Journal of Public 1973a Economics, Vol. 2: 193-216.

-------. "The Theory of Discrimination," in Ashenfelter and Rees 1973b (eds.), <u>Discrimination in Labor Markets</u>. Princeton, New Jersey: Princeton University Press: 3-33.

Assheton-Smith, Marilyn. "John Porter's Sociology: A Theoretical 1979 Basis for Canadian Education," <u>Canadian Jou</u>gal of Education, Vol. 4, No. 2: 43-54.

Astin, Alexander W. "On the Failures of Educational Policy," <u>Change</u>, 1977 Vol."9, No. 9, Sept.: 40-43.

Attiyeh, Richard. "A New Look at the Economics of Higher Education," 1973 <u>Minerva</u>, Vol. 11, No. 3, July: 335-347.

"Baby Boom Generation May Not Create Spending Orgy," Edmonton Journal, 1981 April 6. Bacchus, M.K. "Structural Transformation as a Pre-requisite for the 1979 Success of Non-Formal Education Programmes in Economically Less-Developed Countries," Lead Paper, Commonwealth Specialists Conference on Non-Formal Education for Development. New Dehli, India, Jan.: (Mimeo).

Balogh, Thomas, and P.P. Streeten. "The Coefficient of Ignorance," 1963 <u>Bulletin of the Oxford University Institute of Economics and</u> Statistics, Vol. 25, No. 2, May: 99-107.

Baran, Paul. "The Commitment of the Intellectual," Monthly Review, 1965 Vol. 16, No. 11, March: 1-11.

Barth, Michael C. "<u>Generating Inequality</u>: A Review Article," <u>Journal</u> 1977 of Human Resources, Vol. 12, No. 1, Winter: 92-101.

Becker, Gary S. "Investment in Human Capital: A Theoretical Analysis," 1962 Journal of Political Economy, Vol. 70, No. 1, Feb.: 9-49.

1960 Review, Proceedings, Vol. 50, May: 346-354.

------. "Underinvestment in College Education?" in Wykstra (ed.), 1971 <u>Education and the Economics of Human Capital</u>. New York: The Free Press: 105-115.

Becker, Gary S., and Barry R. Chiswick. "The Economics of Education 1966 and the Distribution of Earnings," <u>American Economic Review</u>, <u>Papers and Proceedings</u>, Part 3, Vol. 56, No. 2, May: 358-369.

Beckerman, Wilfred. "Inflation and the Class Struggle," <u>New Statesman</u>, 1972 Vol. 84, NO. 2177, Dec. 8: 856-858.

Bell, Carolyn Shaw. "The Next Revolution," <u>Social Policy</u>, Vol. 6, 1975 No. 2, Sept./Oct.: 5-11.

Bell, Daniel. "On Meritocracy and Equality," <u>Public Interest</u>, No. 29, 1972 Fall: 29-68.

------ "On Meritocracy and Equality," in Karabel and Halsey (eds.), 1977 <u>Power and Ideology in Education</u>. New York: Oxford University Press: 607-635.

Benton, Ted. "Education and Politics," in Holly (ed.), <u>Education or</u> 1974 <u>Domination? A Critical Look at Educational Problems Today</u>. London: Arrow: 9-37.

Berberoglu, Berch. "The Essence of Inequality: A Critical Analys 1976/77 of Classical and Contemporary Theories of Stratification and Inequality," Social Praxis, Vol. 4, No. 1-2: 47-74.

Berg, Ivar. "The Question of Legitimacy," <u>Change</u>, Val. 3, No. 5, 1971 Sept.: 40-45.

----- "Rich Man & Malifications for Poor Man's Jobs. Are 1969 Emproyers Designing Too Much Education for the Jobs They Offer?" Trans-active col. 6, No. 5, March: 45-50.

Berg, Ivar, and Marcia Freedman. "The American Workplace: Illusions 1977 and Realities," <u>Change</u>, Vol. 9, No. 11, Nov.: 24-30.

Bergmann, Barbara R. "The Effect on White Incomes of Discrimination 1971 in Employment," Journal of Political Economy, Vol. 79, No. 2, March/April: 294-313.

------. "Occupational Segregation, Wages, and Profits When Employers' 1974 Discriminate by Race or Sex," <u>Eastern Economic Journal</u>, Vol. 1, April-July: 103-110.

Berner, Böel. "'Human Capital', Manpower Planning, and Economic 1974 Theory: Some Critical Remarks," <u>Acta Sociologica</u>, Vol. 17, No. 4: 236-255.

Bernstein, Basil. "Social Class and Linguistic Development: A 1961 Theory of Social Learning," in Halsey et. al. (eds.), <u>Education, Economy, and Society</u>. New York: The Free Press: 288-314.

Bhaduri, A. "On the Significance of Recent Controversies on Capital 1972 Theory: A Marxian View," in Harcourt and Laing (eds.), <u>Capital</u> and Growth. Harmondsworth, England: Penguin Books: 251-260.

Bibb, Robert, and William H. Form. "The Enters of Industrial,
1977 Occupational, and Sex Stratification Markets," Social Forces, Vol. 55, Nov 4, June: 974-996.

ð.

-

Billings, Dwight B. "Book Review: <u>Conflict Sociology: Toward An</u> 1976 <u>Explanatory Science</u>, by Randall Collins, New York: Academic Press, 1975, Social Forces, Vol. 55, No. 2, Dec. 533-534.

Binyon, Michael. "The Overeducated Russian," Change, Vol. 11, No. 5, 1979 Aug.: 62-63.

Blau, Francine D., and Carol L. Jusenius. "Economists' Approaches to 1976 Sex Segregation in the Labor Market: An Appraisal," in Blaxall and Reagan (eds.), <u>Women and the Workplace: The Implications</u> of Occupational Segregation. Chicago: University of Chicago Press: 181-199.

Blaug, Mark. "The Correlation Between Education and Earnings: What 1972a Does It Signify?" <u>Higher Education</u>, Vol. 1, No. 1: 53-76.

-----. "The Empirical Statu's of Human Capital Theory: A Slightly 1976a Jaundiced Survey," <u>Journal of Economic Literature</u>, Vol. 14, No. 3: 827-855.

------. "The Private and Social Returns on Investment in Education: 1967 Some Results for Great Britain," Journal of Human Resources, Vol. 2, No. 3, Summer: 330-346.

1976b America: Educational Reform and the Contradictions of Economic Life," Challenge, Vol. 19, No. 3, July/Aug.: 59-61.

 "Review of Jacob Mincer, <u>Schooling, Experience, and Earnings</u>,"
 1976c <u>Economic Development and Cultural Change</u>, Vol. 25, No. 1, Oct.: 166-171.

Blishen, B.R. "Social Class and Opportunity in Canada," <u>Canadian</u> 1970 Review of Sociology and Anthropology, Vol. 7, No. 2: 110-127.

Blitz, Rudolph C. "The Nation's Educational Outlay," in Mushkin (ed.), 1962 <u>Economics of Higher Education</u>. Washington, D.C.: U.S. Government Printing Office: 147-169.

Bluestone, Barry. "The Characteristics of Marginal Industries," in 1977 D.M. Gordon (ed.), <u>Problems in Political Economy: An Urban</u> Perspective. Lexington, Mass.: Lexington Books: 97-102.

Bluestone, Barry, et. al. "Education and Industry," in Carnoy (ed.), 1975 Schooling in a Corporate Society. New York: David McKay: 161-173.

Blunn, Ron. "One Million Good Reasons to be a Boss," Edmonton Journal, 1980 May 27.

Bodstrom, L. "Swedish University Policy: The Financial Support of 1964 University Students," <u>Minerva</u>, Vol. 3, No. 1, Autumn: 91-98.

Bowles, Roy T., and Prudence Craib. "Canada: Economy, Opportunity, 1979 and Class," in Fry (ed.), Economy, Class, and Social Reality. Issues in Contemporary Canadian Society. Toronto: Butterworths: 51-77.

Bowles, Samuel. "Contradictions in Higher Education in the United 1978 States," in Edwards et. al. (eds.), <u>The Capitalist System</u>. Englewood Cliffs, New Jersey: Prentice-Hall: 491-503.

------. "Education, Class Conflict, and Uneven Development," in 1975 Simmons et. al., <u>Investment in Education; National Strategy</u> <u>Options for Developing Countries</u>. Washington, D.C.: International Bank for Reconstruction and Development: 113-127.

------. "The Integration of Higher Education into the Wage Labor 1974a System," The Review of Radical Political Economics, Vol. 6, No. 1, Spring: 100-133.

-----. "Schooling and Inequality From Generation to Generation," 1972 Journal of Political Economy, Vol. 80 No. 3, Part 2, Supplement, May/June: S219-S251.

1977 Come to a Parting of the Ways?" The Progressive, Vol. 41, June: 20-23.

------. "Unequal Education and the Reproduction of the Hierarchical 1974b Division of Labor," in Silverman and Yanowitch (eds.), The Worker in "Post-Industrial" Capitalism: Liberal and Radical Responses. New York: The Free Press: 225-236.

------. "Unequal Education and the Reproduction of the Social 1971 Division of Labor," <u>The Review of Radical Political Economics</u>, Vol. 3, No. 4, Fall/Winter: 1-30.

Bowles, Samuel, and Herbert Gintis. "Capitalism and Education in the 1975 United States," <u>Socialist Revolution</u>, Vol. 5, No. 3: 101-138.

-----. "I.Q. in the W.S. Class Structure," Social Policy, Vol. 3, 1972 Nov./Dec.: 65-96.

------. "The Invisible Fist: Have Capitalism and Democracy Reached 1978 a Parting of the Ways?" <u>American Economic Review, Papers and</u> <u>Proceedings</u>, Vol. 68, No. 2, May: 358-363.

------. "The Problem with Human Capital Theory -- a Marxian Critique," 1975b <u>American Economic Review, Papers and Proceedings</u>, Vol. 65, No. 2, May: 74-82.

Bowles, Samuel, Herbert Gintis, and Peter Meyer. "Education, I.Q., 1975/76 and the Legitimation of the Social Division of Labor," Berkeley Journal of Sociology, Vol. 20: 233-264.

Bowles, Samuel, Herbert Gintis, and John Simmons. "The Impact of 1976 Education on Poverty: The U.S. Experience," <u>International</u> <u>Development Review</u>, Vol. 18, No. 2: 6-10.

Bowman, Mary Jean. "Converging Concerns of Economists and Educators," 1962a Comparative Education Review, Vol. 6, Oct.: 111-119.

- 0

-----. "Economics of Education," <u>Review of Educational Research</u>, 1969 Vol. 39, Dec.: 641-670.

		263
•		
1971	Education: Alternatives for the Federal Government; Iowa City, Iowa: The American College Testing Program: 37-70.	
1970	"Education and Economic Growth," in Johns et. al. (eds.), Economic Factors Affecting the Financing of Higher Education. Gainesville, Florida: National Educational Finance Project: 92-129.	
 1980	"Education and Economic Growth: An Overview," in King (ed.), Education and Income. Washington, D.Ç.: Worl: Bank: 1-71.	
1968	"Human Capital: Concepts and Measures," in Sowman et. al. (eds.), <u>Readings in the Economics of Education</u> . Paris: UNESCO: 246-259.	
1966	"The Human Investment Revolution in Economic Thought," Sociology of Education, Vol. 39, No. 2, Spring: 111-138.	
<u>¥</u> 1964	· ····································	
19626	"The Social Returns to Education," <u>International Social</u> Science Journal, Vol. 14, No. 4: 647-659.	
1976 .	"Through Education to Earnings?" <u>Proceedings of the</u> <u>National Academy of Education</u> , Vol. 3: 221-292.	-1
Bowman 1968	, Mary Jean, and C. Arnold Anderson. "Concerning the Role of Education in Development," in Bowman et. al. (eds.), <u>Readings</u> in the Economics of Education. Paris: UNESCO: 113-134.	
Breton 1970	, Raymond. "Academic Stratification in Secondary Schools and	
" Bri ti 1981	sh Industry Sees Red," <u>Edmonton Journal</u> , April 23.	
Brown, 1977	Byron W., and Daniel H. Saks. "Review of <u>Schooling in Capitalist</u> America: Educational Reform and the Contradictions of Economic Life," Journal of Economic Issues, Vol. 11, March: 158-162.	•
Brown, 1979 •	Robert Edward. "Capitalien in the Classroom," Change, Vol. 11, No. 3, April: 25-30.	
Bulcoci 1979	x, Jeffrey W. "Changes in Society Affecting Schooling, With Some Implications for Teacher Education," McGill Journal of Education, Vol. 14, No. 2, Spring: 133-147.	
1977	"Economic Determinants of Educational Participation," Department of Educational Foundations, Memorial University, Newfoundland, Nov. (Mimeo).	
	۰	

.

263

9. 22

-

Burkhead, Jesse. "Economics Against Education," <u>Teachers College</u> 1973 <u>Record</u>, Vol. 75, No. 2, Dec.: 193-205.

Cain, Glen. "The Challenge of Segmented Labor Market Theories to 1976 Orthodox Theory: A Survey," Journal of Economic Literature, Vol. 14, No. 4, Dec.: 1215-1257.

----- "The Economics of Labor; An Assessment of Parcent Research. 1975 The Challenge of Dual and Radical Theories of the Labor Martiet to Orthodox Theory," <u>American Economic Review</u>, Vol. 85, No. 2: 16-22.

Cain, Glen, and Harold W. Watts. "The Controversy about the Coleman 1968 Report: Comment," Journal of Hughn Resources, Vol. 3, No. 3, Summer: 389-392.

Carchedi, G. "On the Economic Identification of the New Middle Class," 1975a Economy and Society, Vol. 4, No. 1, Feb.: 1-86.

1975b Economy and Society, Vol. 4, No. 4: 361-417.

Carlson, Kenneth. "Equalizing Educational Opportunity," in Miller and 1974 Gordon (eds.), Equality of Microsoftonel Opportunity. A Referrch Handbook. New York: AMS Press: 1994-136.

Carnoy, Martin. "Class Analysis and Theorement in Human Resources, 1971 A Dynamic Model," <u>The Review of Radical Political Economics</u>, Vol. 3, No. 4, Fall/Winter: 56-81.

----- "Comments on Welch's 'Human Capital Theory'," Palo, Alto, n.d. Cal.: Sranford University and the Center for Economic Studies (Mimeo).

------ "Earnings and Schooling in Mexico," Economic Development 1967a and Cultural Change, Vol. 15, No. 4, July: 408-419.

1975 Investment in Education; National Strategy Options.for:Developing Countries. Washington, J.C.: International Bank for Reconstruction and Development: 90-101.

------. "The Political Economy of Education," 1970? 177-215.

------ "Rates of Return to Schooling in Latin America," 1967b Journal of Human Resources, Vol. 2, No. 3, Summer: 359-374.

Carnoy, Martin, and Dieter Marenbach. "The Beturn to Schooling in the 1975 United States, 1939-1969," <u>Journal of Human Resources</u>, Vol. 10, No. 3, Summer: 312-331.

Carter, Michael A., and Martin Carnoy. "Theories of Labor Markets and 1974 Worker Productivity," Palo Alto, Cal.: Center for Economic Studies. Carter, Michael J. "To Abstain or Not to Abstain (Is That the "-1977 Question?) -- A Critique of the Human Capital Concept," in Schwartz (ed.), The Subtle Anatomy of Capitalism: Selected Readings. Santa Monica, Cal.: Goodyear Pub. Co.: 36-50. 265

Cartter, Allan M. "Scientific Manpower for 1970-85," Science, 1971 Vol. 172, No. 3979, April: 132-140."

. . .

4.5

Cher An, Andrew. "Divorcing Welfare from Marriage," Psychology 1979. Today, Vol. 13, No. 2, July: 92. ,

Vol. 11, No. 2, March: 12-17.

 Chiswick, Barry R., and Jacob Mincer. - "Fine-Series Changes in
 1972 Personal Income Inequality in the United States from 1939, with Projections to 1985," in Schultz (ed.), <u>Investment in</u>
 <u>Education.</u> The Equity Efficiency Quandry. Chicago: University of Chicago Press: 34-71; also in <u>Journal of</u> Political Economy, Volue 80, No. 3, Part 2, May/June: 'S34-S66.

Christiansen, Jens. "Marx and the Palling Rate of Profit," American 1976 Economic Review, Papers and Proceedingst Vol. 66, No. 2. May: 20-25.

Clark, Burton R. - "The 'Cooping-Out' Function of Higher Education," 1960 American Journal of Sociology, Vol. 65, No. 6, May: 569-576.

Clark, Samuel Delbert. "Ind American Takeover of Canadian Sociology: 1976a Myth or Reality?" in Clark, <u>Canadian Mciety in Historical</u> <u>Perspective</u>. Toronto: AcGran-Hill Ryerson, Ltd.: 120-132.

Cohen, David K. 3"Politics and Research: Evaluation of Social Action 1974 Programs in Education," in Miller and Gordon (eds.), Equality of Educational Opportunity. A Handrook for Research. New York: AMS Press: 37-62.

Cohen, David K., and Marvin Lazerson. "Education and the Corporate 1972 On Socialist Revolution, Vol. 2, No. 2, March/April: 47-72.

Colentary viewes S. "The Concept of Equality of Educational Opportunity," 1974 in Miller and Gordon (eds.), Equality of Educational Opportunity. A Handbook for Research. New York: AMS Press: 3-16.

-----: "Equal Schools or Equal Students," Public Interest, No. 4, 1966 Summer: 70-75. 1969 Levin," Journal of Human Resources, Vol. 3, No. 2, Spring: 237-246.

s 🖌

266

973 Educational Review, Vol. 43, No. 1, Feb.: 129-137.

-----. "Toward Open Schools," The Bublic Interest, No. 95, 1964 Fall: 20-27.

÷.

1975' Oxford Review of Education, Vol. 1, No.1: 27-29.

"Colleges: Staying On," The Economist, Vol. 277, No. 7160, Nov. 22: 85. 1980

Collins, Rendall., "The Dialectic of Ideology and Technology: A Review," 1978 Theory and Sigiety, Vol. '5, No. 1, Jan.: 135-138:

971 fication, "Storiczo Sciological Review, Vol. 36, No. 6, Dec.: 100

1974 ... Highest?" Sociology of Education, Volu. 47, 1974. 4, Fall: 419-442.

Cordoran, Mary, et al. "The Effects of Family Background on Earnings, 1976 American Economic Review, Papers and Proceedings, Vol. 66, No. 2, May: 430-435

Opercoran, Thomas B. "Community Colleges: The Coming Slums of Education?" 1972 Change, Vol. 4, No. 7, Sept.: 30-35.

Cory, Christôpher T. "Making It. Con the Odds Bellyened?" Psychology 1979, Today, Vol. 13, No. 2, July: 35-39.

Crane, Diane. "Sociel Class: Origin and Academic Success: The 1969 Influence of the Stratification Systems on Academic Careers, *- Sociology of Education; Vol. 42, No. 1, Winter: 1-17.

Daniere, Andre. "Social Class Competition in American Higher 1973 Education," in McLeod (ed.), Post-Secondary Education Technological Society." Montinal: McGill-Queen's University Press: 147-159.

Davis, Kingsley, and Wilbert E. Moore. "Some Principles of Strati-1945 fication," <u>American Sociological Review</u>, Wol. 10, No. 2: 242-249.

Deaton, Rick. "The Fiscal Trisis of the State in Canada," in 1973 Roussopoulos (ed.), The Political Economy of the State, Quebec/Canada/U.S.A. Montreal: Black Rose Books: 18-58.

Debeauvais, Michel. "The Concept of Human Capital," <u>UNESCO Inter-</u> 1962 <u>national Societ Science Journal</u>, Vol. 14, No. 4: 660-675.

1974 Policies: A Cristin Forment," in Ward (ed.), Education and Development Beck Berlin Bellagio Sonfelence Papers. New York: Praegel 5-47.

Dedyna, Catherine. "University Bids to Balance Books: Fees May Hit 1981 \$690 A Year," <u>Edmonton Journal</u>, April 3.

Denison, Edward Fulton. "An Aspect of Inequality of Opportunity," 1970 <u>Journal of Political Economy</u>, Vol. 78, No. 5, Sept./Oct.: 1195-1202.

1969 Comment, "Mmerican Economic Review, Vol. 59, Dec. 1935-943.

-------. "Education, Economic Growth, and Gaps in Information," 1962 Journal of Political Economy, Vol. 70, No. 5, Part 2, Supplement, Oct.: 125,128.

-----. "Education and Economic Productivity," in Martis and 1965 Levensohn (eds.), <u>Education and Public Policy</u>." Berkeley, Cal.: McCutchan: 328-340.

Dennis, Norma. "Sociology, Education, and Equality in the Sociology 1980 of Education: A Review of Halsey, Heath, and Ridge's Origina and Destinations," Oxford Review of Education, Vol. 6, No. 2: 111-131.

DeVille, Philippe, and Tom R. Burns. "Institutional Responses to 1976/77 Crises in Capitalist Development: The Dialectics of Capitalist State Regulation," <u>Social Praxis</u>, Vol. 4, No. 1-2: 5-46.

Dore, Ron P. "Human Capital Theory, The Diversity of Societies, and 1976 The Problem of Quality in Education," <u>Higher Education</u>, Vol. 5, No. 1, Feb.: 79-102.:

Dos Santos, Theotonio, "The Structure of Dependence," <u>American</u> 1970 <u>Economic Review, Papers and Proceedings</u>, Vol. 60, No. 2, Part 2, May: 231-236.

Dresch, Stephen P., and Adair L. Waldenberg. "Labor Market Incentives, 1978 Intellectual Competence, and College Attendance," New Haven, Conn.: Institute for Demographic and Economic Studies; Inc. 1978 Beb. 15: 1-20 (Mimeo). Drummond, Ian. "Labor Markets and Educational Planning," in Kruger 1968 and Meltz (eds.), <u>The Canadian Labour Market; Readings in</u> <u>Manpower Economics</u>. Toronto: Centre for Industrial Relations, University of Toronto: 243-295. 268 🍯

Duncan, Otis Dudley, and Robert W. Hodge. "Education and Occupational 1963 Mobility: A Regression Analysis," <u>American Journal of Sociology</u>, Vol. 68, No. 6, May: 629-644.

Eakins, David. "Critique: Population and the CapitalistaBomb: 1970 A Review of Paul Erlich's <u>The Population Bomb</u>," <u>Socialist</u> Revolution, Vol. 1, No. 2, <u>March/April</u>: 144-152.

Early, Margaret J. "Literature Education in Ten Countries," <u>American</u> 1974⁴ <u>Educational Research Journal</u>, Vol. 11, No. 4, Fall: 419-422.

Eckaus, R.S. "Economic Criteria for Education and Training," <u>Review</u> 1964 of Economics and Statistics, Vol. 46, No. 2, May: 181-190.

Edgeworth, F.T: "Equal Pay to Men and Women for Equal Wark," Economic 1922 Journal, Vol. 32, No. 128, Dec.: 431-457.

Edmonds, Ron, and Evelyn K. Moore. "I.Q., Social Class, and Educational 1973 Policy," Change, Vol. 5, No. 8, Oct.: 12;64.

Edwards, Richard C., "Individual Traits and Organizational Incentives: 1976 What Makes a 'Good's Worker?" Journal of Human Resources, Vol. 11, No., Law Minter: 51-68.

1977 Educational and Psychological Measurement 1977 Spring: 125-48.

"The Social Relational of Production in the Firm and Labor 1975 Market Structure," Politics and Society, Vol. 5, No. 1: 83-108.

Eide, Kjell. "A Comment on Husen," in OECD, Education, Inequality, and 1975 Life Chances, Vol. 1. 'Paris: OECD: 344-350.

"87% in Survey Hate Job; Superior Leading Villain," <u>Edmonton Journal</u>, 1990 Oct. 31".

Epstein, Erwin H. "The Social Control Thesis and Educational Reform 1978 in Dependent Nations," <u>Theory and Society</u>, Vol. 5, No. 2, . March: 255-276.

Faber, Seymour. "Working Class Organisation," Our Generation

Faris, Robert E.L. "Reflections on the Ability Dimension in Human 1961 Society," American Sociological Review, Vol. 26, Dec.: 835-843. Farran, Roy. "Schools are Trying to Be Too Much to Too Many," 1981 <u>Edmonton Journal</u>, Jan. 20.

Fawcett, Millicent Garrett. "Equal Pay for Equal Work," Economic 1918 Journal, Vol. 28, No. 109, March: 1-6.

Feinberg, Walter. "Educational Equality Under Two Conflicting Models 1975 of Educational Development," <u>Theory and Society</u>, Vol. 2, No. 2, Summer: 183-210.

Finkel, Alvin. "Origins of the Welfare State in Canada," in Paniero 1977 (ed.). The Canadian States Political Economy and Political Power. Toronto: University of Toronto Press, 1994-370.

Finn, Jr., Chester E. "Federal Patronage of Universities in them 1976/77 United States: A Rose by Many Other, Names?" <u>Minerva</u>, Vol. 14, No. 4, Winter: 496-529.

1975776 Change, Vol. 7, No. 10, Dec./Jan.: 24-29.

3.0

Fisher, Irving. "Precedents for Defining Capital," Quarterly Journal 1944 of Economics, Vol. 18, March # 386-408.

1895 What is Capital?" Economic Journal, Vol. 6, Dec.: 509-534.

Flude, Michael. "Social mitel Accounts of Differential Educational 1974 Attainment," in Fride and Ahler (eds.), Educability, Schools, 1975 Ideology. London: Grooms Helm: 15-51.

Fluxgold, Howard. "Fools Gold: The Politics of Education in Ontario," 1975 <u>Canadian Forum</u>, Vol. 54, No. 647, Jan.: 21-26.

Form; William H. "The Internal Stratification of the Working Class: 1973 System Involvements of Auto Workers in Four Countries," American Sociological Review, Vol. 38, No. 6, Dec.: 697-711.

Royer, Lars. "Swedish University Policy: The 1955 Commission on the 1964 Swedish Universities," <u>Minerva</u>, Vol. 3, No. 1, Autumn: 83-90.

Frank, Andre Gunder. "World Crisis, Class Struggle, and 1984," 1975 Socialist Revolution, Vol. 5, No. 3: 59-70,

Frankel, Charles "Equality of Opportunity," Ethicat An International 1971 Journal of Social, Political, and Legal PhiloSophy; Vol. 81, No. 3, April: 191-211.

1975 Proletariat: A Martist Analysis," Socialist Revolution, Vol. 5, No. 4, Oct.-Dec.: 41-83. Freeman, Richard B. "Overinvestment in College Training?" Jour 1975 of Human Resources, Vol. 10, No. 3, Summer: 287-311.

Friedman, Milton. "The Role of Government in Education," in Friedman, 1962 <u>Capitalism and Freedom</u>. Chicago: University of Chicago Press 85-107.

Frum, Barbags. "Class of '71:"The Graduates Nobody Wants," <u>Maclean's</u>, 1971 June: 20.

Fryers, Walt, "How to Transcend the Next Depression," Edmonton Journal, 4 1981 April 3.

Gaff, Jerry G. "Book Review: <u>Graduates: The Sociotogy of an Elite</u>, by 1973 R.K. Kelsald, Anne Poole, and Annette Kuhn. London: Methuen and Co., 1972, "<u>Compensorary Sociology: A Journal of Reviews</u>, Vol. 2, No. 5, Sept.: 551-553.

1

Gans, Herbert J. "The Role of Education in the Escape from Poverty," 1976 In Ashline, "Perzulo, and Norris (eds.), Education, Inequality, and National Policy, Lexington, Mass.: Lexington Books: 61-72.

Gintis, Nerbert. "Alienation and Power," The Review of Radical Political 1972: Economics, Vol. 4, No. 5, Mall: 1-344

1971; Productivity," American Economic Review, Papers and Proceedings, Vol. 61, No. 2, May: 266-279.

1970. Revolution, Vol. 1, No. 3, Nay/June: 13-43.

1971b Crisis in the Classroom);" Monthly Review, Vol. 23, No. 7, Part. 2, Dec.: 40-51.

Ginzberg, Eli. "Educational Realism in Less Eveloped Countries;" 1974 Teachers College Boord, Vol. 75, No. 3, Teb.: 317-326.

Gittell, Marilyn. "The Illusion of Affirmative Action," Change, 1975 Vol. 7, 10. 8, Oct.: 39-43.

Glenn, Evelyn Nakano, and Rosłyn Feldberg. "Degraded and Deskilled: 1977 The Proletarianization of Clerical Work," Social Problems, Vol. 25, No. 1, Oct.: 52-64.

Gonick, Cy. "Poverty and Capitalism," <u>Canadian Dimension</u>, Vol. 6, 1969 No. 5, Oct./Nov.: 26-34.

Goodman, Ellen. "The Educated Unemployed," Edmonston Journal, June 5. 1980

Gordon, David M.' "Capitalist Efficiency and Socialist Efficiency," 1976a Monthly Review, Vol. 28, No. 3, July/Aug.: 19-39.

1976b Comment II in Blaxall and Reagand(eds.), Women and the Workplace; the Implications of Occupational Segregation. Chicago: University of Chicago Press: 238-24444

Gordon, Edmund W. "Compensatory Education: Evaluation in Perspective," 1974a Miller and Gordon (eds.), Equality of Educational Opportunity. <u>A Handbook for Research</u>. New York: AMS Press: 308-317.

1974b in Miller and Gordon (edg.), Equality of Educational Opportunity. A Handbook for Research New York: AMS Press: 177-185.

------. "The Political Economics of Effective Schooling," in Miller 1974a and Gordon (eds.), Equality of Educational Opportunity. A Handbook for Research. New York: AMS Press: 445-460.

1974d Miller and Gordon (eds.); Budlity of Educational Opportunity;" in A Harder Kok for Research. New York: AMS Press: 16-27.

Goslin, David A. "The Functions of the School in Modern Society," 1965 in Goslin, <u>School in Contemporary Society</u>. Chicago: Scott, Foresman, and Co.: 1-18.

Gouldner, Alwin W. "The New Class Project, I," Theory and Society, 1978a Vol. 5, No. 2, Sept.: 153-203.

1978b. No. State V.: 343-389.

"Grads Underemployed," Gateway, March 31. 1981

Greenwald, Howard P. "Book Review: The Overeducated American, by 1978 Richard B. Freeman, 1976," American Journal of Sociology, Vol. 20, No. 5, March: 1307-1309.

Griffiths, Martha W. "Can We Still Afford Occupational Segregation? 1976 Some Remarks," in Marcall and Reagan (eds.), Women and the Workplace, Implications of Occupational Segregation. Chicago: University of Chicago Press: 7-14.

Gruneau, Richard S. "Problems of Egalitarianism and Credentialism in 1978/79 Genadian Education," Interchange, Vol. 9, No., 2: 40-44.

Gwyn, Richard. "Is Inflation a 'Good Guy'?" Edmonton Journal, Dec. 10.

Hall, Bobert E. "Book Review: Schooling, Experience, and Earnings, by 1975 Jacob Mincer," Journal of Political Economy, Vol. 83, No. 2, April: 444-446.

Halliday, Terence C. "The Politics of 'Universal Participatory 1975 Democracy': A Canadian Case Study," <u>Minerva</u>, Vol. 13, No. 3, Autumn: 404-427.

Halperin, Samuel. "The Federal Future in Higher Education," Change, 1978 Vol. 10, No. 2, Feb.: 24-26.

Halsey, Albert Henry. "Education and Social Class in 1972," in 1973 Jones (ed.), <u>The Yearbook of Social Policy in Britain, 1972</u>. London: Routledge and Kegan Paul: 39-52.

 "Educational Organization," in Sills (ed.), International
 Encyclopedia of the Social Sciences, Vol. 4. New York: Crowell, Collier, and Macmillan: 525-532, 32

------. "Sociology and the Equility Debate," Oxford Review of 1975b Education, Vol. 1, No. 1: 9-23.

Hanoch, Giora. "An Economic Analysis of Earnings and Schooling," 1967 Journal of Human Resources, Vol. 2, No. 3, Summer: 310-329.

Hansen, W. Lee. "Income Distribution Effects of Higher Education," 1970 American Economic Review, Papers and Proceedings, Vol. 60, No. 2, Part 2, May: 335-346.

----- "Total and Private Rates of Return to Investment in 1963 Schooling," Journal of Political Economy, Vol. 71, No. 1, Feb.: 128-140.

Hansen, W. Lee, and Burton A. Weisbrod. "The Distribution of Cost 1969 and Direct Benefits of Public Higher Education: The Case of California," <u>Journal of Human Resources</u>, Vol. 4, No. 2, Spring: 176-191.

----- "A New, Approach to Higher Education Finance," in Orwig 1971 (ed.), Finance, Higher Education: Alternatives for the Federal Government. Iowa City, Iowa American College Teating Program: 117-142.

Hansen, W. Lee, Burton A. Weisbrod, and William Scanlop. "Schooling 1970 and Earnings of Low Achievers," American Economic Réview, Vol. 60, June: 409-418.

Harbison, Frederick H. "Human Resource Development Planning in 1962 Modernizing Economies," <u>International Labor Review</u>, Vol. 85, No. 5, May: 1-24.

Harp, John. "Book Review: The Conditions for Educational Equality, edited. 1973 by Sterling M. McMurrin, 1971," Contemporary Sociology: A Journal of Reviews, Vol. 2, No. 5, Sept.: 553-554.

Harvey, Barrier B. "Canadian and gher Education and the Seventies," 1995 Freechange, Vol. 5, 199 2: 42-52.

Hause, John C. "Ability an "pooling as Determinants of Lifetime 1975 Earnings, or If You's So Smart, Why Aren't You Rich?" in Juster (ed.), Education, Income, and Human Behavior. New Tark: McGraw-Hill Book Co.: 123-149.

Haverman, Robert H. "Tinbergen's <u>Income Distribution; Analysis and</u> 1977 <u>Policies</u> — a Review Article," <u>Journal of Human Resources</u>, Vol. 12, No. 1, Winter: 103-114.

Heap, James L. "Conceptual and Theoretical Problems in <u>The Vertical</u> 1972 <u>Mosaic</u>," <u>Canadian Review of Sociology and Anthropology</u>, Vol. 9, No. 2: 176-189.

Hechinger, Fred M. "Education's 'New Majority'," Saturday Review, 1975 Sept. 20: 14-18.

-------. "Public Education as the Great Equalizer," in Ashline,
 1976 Pezzulo, and Norris (eds.), Education, Inequality, and
 National Policy. Lexington, Mass.: Lexington Books: 1-13.

Henchey, N. "Quebec Education: The Unfinished Revolution," McGill 1972 Journal of Education, Vol. 7, No. 2: 95-118.

Heyns, Barbaran "Review Energy: Schooling in Capitalist America: 1978 Education in Form any the Contradictions of Economic-Life, by Samue Porces and Herbert Gintis, 1976;" American Journal of Societ 7, Vol. 83, No. 4, Jan.: 999-1006.

Hills, Nicholas. "U.K. Social Immobility Sparks Fear of Class 1980 Warfare," Edmonton Journal, Jan. 24.

 Holly, Douglas. "The Invisible Ruling Class," in Holly (ed.),
 1974 Education or Domination? A Critical Look At Educational Problems Today. London: Arrow: 105-125.

Hitchock, James. "The New Ocasion ism," Change, Vol. 5, No. 3, 1973 April: 46-50.

Horan, Patrick M. "Book Review: Education, Occupation, and Earnings, 1977 by William H. Sewell and Robert M. Hauser. New York: Academic Press, 1975," Social Forces, Vol. 55, No. 3, March: 822.

"1980 Success in America, by C. Jencks et. al.," Harvard Educational Review, Vol. 50, No. 2, May: 299-304.

1978 "Is Status Attainment Research Atheoretical?" American 1978 Theological Review, Vol. 43, Aug.: 534-541. 1974; and Analysis," Social Forces, Vol. 53, No. 1, Sept.: 33-45.

"Horsman to Study Reparks on Job-Training Programs," Edmonton. Journal, 1980 Oct. 17.

Horton, John. "Order and Conflict Theories of Social Problems," * 1966 <u>American Journal of Sociology</u>, Vol. 71, No. 6, May: 701-713.

Humphrey, Hubert H. "The War on Poverty," in Everett (ed.), Anti-1966 <u>Poverty Programs</u>. Dobbs Ferry, New York: Oceana Pubs.: 6-17.

Hurn, Christopher J. "The Reemergence of Liberal Education," Change, 1978 Vol. 10, No. 9, Oct.: 8-9.

Husén, Torsten, , "Ability, Opportunity, and Carder: A 26 Year Follow-Up," 1968 <u>Educational Research</u>, Vol. 10, No. 3, June: 170-184.

------ "The Equality Meritocracy Dilemna in Education," in Ashline, 1976a Pezzulo, and Norris (eds.), Education, Inequality, and National Policy. Lexington, Mass.: Lexington Books: 45-59.

 "Introduction to the Reviews of Three Studies of the
 1974 International Association for the Evaluation of Educational Achievement (IEA), "American Educational Research Journal,
 Vol. 11, No. 4, Faire 37-408.

1979a Education, Vol. 5, 1973; 267-273.

1976b The Dilemna Between For the State of Lence," Higher Education; Yol. 5, No. 4, Novi: 4074

------ "Strategies for Educational Equality," in Other Education, 1975 Inequality, and Life Thances. Vol. 1. Paris: OECD: 308-343.

"Swedish University Research at the Crossroads," Minerva, 1976 7 Vol. 14, No. 4, Winter: 419-446.

Hussain, Athar. "The Economy and the Educational System in Capitalist 1976 Societies," Economy and Society, Vol. 5: 413-434.

Illich, Ivan. "The Futility of Schooling in Latin America;" Saturday 1968 Review, Vol. 51, April 20: 57-59;74-75.

1970 Vol. 15, No. 1, July 2: 9-15
"Irving Kristol: Neo-Conservative Guru to America's New Order," # 1981 <u>Maclean's</u>, Vol. 94, Jan. 19: 9-12.

Jackson, R.W.B., and William Gerald Fleming. "Who Goes to University --1957 English Canada," in Bissell (ed.), <u>Canada's Crisis in Higher</u> <u>Education</u>. 'Toronto: University of Toronto Press: 75-113.

Jencks, Christopher. "Social Stratification and Higher Education," 1968 <u>Harvard Educational Review</u>, Vol. 38, No. 2, Spring: 277-316.

Jensen, Arthur R. "Equality and Diversity in Education," in Ashline, 1976 Pezzulo, and Norris (eds.), <u>Education, Inequality, and National</u> Policy. Lexington, Mass.: Lexington Books: 125-136.

"Jobs Crunch for Women," Edmonton Journal, March 10. 1981

Johnson, Harry G. "Comments on Mr. John Vaizey's Paper; Towards A 1964 Generalized Capital Accumulation Approach to Economic Development," in OECD, <u>The Residual Factor and Economic Growth</u>. Paris: OECD: 219-225.

------. "Towards a Generalized Capital Accumulation Approach to 1968 Economic Development," in Blaug (ed.), <u>Economics of Education</u>. Vol. 1. Harmondsworth, England: Penguin Books: 34-44.

Johnson, George E. "The Demand for Labor by Educational Category," 1970 The Southern Economic Journal, Vol. 37, No. 2, Oct:: 190-204.

Johnson, Leo A. "The Capitalist Labour Market and Income Inequality '1979 in Canada," in Fry (ed.), Economy, Class, and Social Reality; <u>Issues in Contemporary Canadian Society</u>. Toronto: Butterworth and Co. (Oan.), Ltd.: 153-168.

------. "The Development of Class in Canada in the Twentieth 1972' Century," in Teeple (ed.), <u>Capitalism and the National Question</u> in Canada." Toronto: University of Toronto-Press: 141-183.

Johnson, Terence J. "Professions, Class, and Power," Unpublished 1976 paper presented to the Bearned Societies Meetings, Laval University, Quebec City, Quebec, Canada. May.

Judis, John. "The Triumph of Bourgois Hegemony in the Jone of Nothing 1970 That Challenges It," <u>Socialist Revolution</u>, Vol. 1, No. 3, Max/June: 107-125.

Judy, Richard W. "The Income-Redistributive Effects of Aid to Higher 1970 Education," in Officer and Smith (eds.), <u>Canadian Economic</u> <u>Problems and Policies</u>. Toronto: McGraw-Hill Co. of Canada, Ltd.: 302-317.

Kamens, David H: "Legitimating Myths and Educational Organization: 1977 The Relationship Between Organizational Ideology and Formal

Ε,

Structure," <u>American Sociological Review</u>, Vol. 42, No. 2, April: 208-219.

đ

Kanter, Rosabeth Moss. "The Policy Issues: Presentation VI," in 1976 Blaxall and Reagan (eds.), Women and the Workplace. The <u>Implications of Occupational Segregation</u>. Chicago: University of Chicago Press: 282-291.

Karabel, Jerone. "Open Admissions: Toward Merîtocracy or Equality?" 1972 <u>Change</u>, Vol. 4, No. 4, May: 38-43.

Karabel, Jerome, and Alexander W. Astin. "Social Class, Academic 1975 Ability, and College 'Quality'," <u>Social Forces</u>, Vol. 53, No. 3, March: 381-398.

Karier, Clarence J. "Business Values and the Educational State," in 1973 Karier, Violas, and Spring (eds.), <u>Boots of Crisis: American</u> <u>Education in the Twentieth Century</u>. New York: Rand McNally: 6-29.

-----* "Testing for Order and Control in the Corporate Liberal " 1972 State," Educational Theory, Vol. 22, Spring: 154-180.

Rennedy, Garol. "The Upper Classes Have the Upper Hand," Macleon's, 1980 / Vol. 93, No. 4, Jan. 28: 4

Keppel, Francis, "Education in the Eighties," <u>Harvard Educational</u> 1980 <u>Review</u>, Vol. 50, No. 2, May: 149-153.

Kerckhoff, Alan C. "The Status Attainment" Process: Socialization 1976 or Allocation?" Social Forces, Vol. 55, No. 2, Dec.: 368-381.

Kerr, Clark. "Supporting Notations on 'the Relationship Between 1979 School and Work'," Oxford Eview of Education, Vol. 5, No. 3: 2154225.

Kerr, Clark, et. al. "The Logic and Industrialism," in Silverman. and 1974 Yanowitch (eds.), The Worker in "Post-Industrial" Capitalism: Liberal and Radical Responses. New York: The Free Press: 70-77.

Kiker, B.F. "The Historical Roots of the Concept of Human Capital," 1966 Journal of Political Economy, Vol. 74, No. 5, Oct.: 481-499.

9-----. "Marshall on Human Capital: Comment," Journal of Political 1968 Economy, Vol. 76, No. 5, Sept. /Oct.: 1088-1090.

Kilgour, David. "Economics Nowadays," Edmonton Journaf. 1981

King, Alexander. "The Lamontagne Report: An Erudite Approach to 1972 Science Policy Problems," <u>Science Forum</u>, Vol. 5, No. 2, @ April: 4-9. Kuznets, Simon. "Economic Growth and Income Inequality," American 1955 Economic Review, Vol. 45, No.º1, March: 1-28.

Lampman, Robert J. "Recent Changes in Income Inequality Reconsidered," 1954 American Economic Review, Vol. 44, No. 3, June: 251-268.

Lauter, Paul, and Florence Howe. "How the School System is Rigged 1970 for Failure," New York Review of Books, Vol. 14, No. 12, June 18: 14-21.

Lawson, Robert. "Education and Canadian Identity and the Problem of 1975 the U.S.A.," <u>Comparative Education Center Research Series</u> <u>1975/76</u>, No. 2. College Park, Maryland: University of Maryland, College of Education:

 "Who Controls Canadian Education?" <u>Comparative Education</u>
 1976 <u>Center Research Series, 1976/77</u>, No. 1. College Park, Maryland: Wniversity of Maryland, College of Education:

Layard, P.R.G. "Denison and the Contribution of Education to National 1973 Income Growth: A Comment," Journal of Political Economy, Vol. 81, No. 4, July/Aug.: 1013-1016.

Layard, P.R.G., and J.C. Saigal. "Educational and Occupational 1966, Characteristics of Manpower: An International Comparison," British Journal of Industrial Relations, Vol. 4, No. 2, July: 222-266.

3

Layard, Richard, and George Psacharopoulos. "The Screening Hypothesis 1974 and the Returns to Education," Journal of Political Economy, Vol. 82, No. 5, Sec./Oct.: 985-998.

Lazear, Edward. "Academic Achievement and Job Performance: Note," 1977 American Economic Review, Vol. 67, No. 2, March: 252-254.

Leach, Richard H. "The Federal Role in the War on Powerty Program," 1966 in Everett (ed.), <u>Anti-Poverty Programs</u>. Dobbs Ferry, New. York: Oceana Pubs,: 18-38.

Lebowitz, Michael-A. "Marx's Falling Rate pf Profit," Socialist 1978 Review, Vol. 8, No. 2, March/April: 73-97.

Lee, Betty. "Job Prospects for Women in the '80's," Chatelaine, 1980 Nol. 53, No. 4, April: 44-45;176-190

Lehnan, Edward W. "Review Essay: The Limits of Sociology," Contemporary 1973 : Sociology: A Journal of Reviews, Volume, No. 5, Sept.: 468-473.

Leibenstein, Harvey. "Allocative Efficiency vs. 'X-Efficiency'," 1966 : American Economic Review, Vol. 56, No. 3, June: 392-415. ------ "Economics of Skill Labelling," in Lauwerys and Scanlon 1969 (eds.), <u>World Year Book of Education, 1969. Examinations</u>. London: Evans Bros.:

------. "Shortages and Surpluses in Education in Underdeveloped 1965 Countries," in Anderson and Bowman (eds.), Education and Economic Development. Chicago: Aldine Press: 51-62.

Lekachman, Robert. "Proposition 13 and the New Conservatism," <u>Change</u>, 1978 Vol. 10, No. 8, Sept.: 22-27.

Levin, Henry M. "A Decade of Policy Developments in Improving *****. 1977a Education and Training for Low-Income Populations," in Haveman (ed.), <u>A Decade of Antipoverty Programs: Achievements,</u> <u>Failures, and Lessons</u>. New York: Academic Press: 123-188.

------. "Education, Life Chances, and the Courts: The Role of 1976a Social Science Evidence," in Ashline, Pezzulo, and Norris (eds.), Education, Inequality, and National Policy. Lexington, Mass.: Lexington Books: 73-100.

------. "Educational Opportunity and Social Inequality in 1976b Western Europe," Social Problems, Vol. 24, No. 2, Dec.: 148-172.

------. "The Effect of Different Levels of Expenditure on 1970 Educational Output," in Johns et. al. (eds.), <u>Economic</u> <u>Factors Affecting the Financing of Education</u>. Gainesville, Florida: National Educational Finance Project: 182-215.

------. "A Radical Critique of Educational Policy," <u>Journal of</u> 1977b Educational Finance, Vol. 3, No. 1, Summer: 9-31.

Lichtman, Richard. "The Facade of Equality in Liberal Democratic 1970 Theory," Socialist Revolution, Vol. 1, No. 1, Jan./Feb.: 85-126.

----- "The Ideological Functions of the University," <u>Upstart</u>, 1971 Jan.:

Lithwick, Noah H. et. al. "Postwar Production Relationships in 1967 Canada," in Brown (ed.), <u>The Theory and Empirical Analysis</u> of Production. Vol. 31: Studies in Income and Wealth. Now York: National Bureau of Economic Research: 139-257.

Lloyd, Cynthia B. "The Division of Labor Between the Sexes: A Review," 1975 in Lloyd (ed.), <u>Sex, Discrimination, and the Division of Labor</u>. New York: Columbia University Press: 1-24.

Lockhart, Alexander. "Educational Opportunities and Economic Opportun-1979 ities -- The 'New' Liberal Equality Syndrome," in Fry (ed.), Economy, Class, and Social Reality. Issues in Contemporary Canadian Society. Toronto: Butterworths: 224-237. ------ "Educational Policy Development in Canada: A Critique 1977 of the Past and a Case for the Future," in Carlton, Colley, and MacKinnon (eds.), Education, Change, and Society. Toronto: Gage Ed. Pub. Co., Ltd.: 76-88.

------ "Future Failure: The Unanticipated Consequences of L975 Educational Planning," in Pike and Zureik (eds.), <u>Socialization</u> <u>and Values in Canadian Society</u>, Toronto: McClelland and Stewart: 182-208.

------. "Graduate Unemployment and the Myth of Human Capital," in . 1971 Davies and Herman (eds.), <u>Social Space Canadian Perspectives</u>. Toronto: New Press: 251-254.

Lundberg, Erik. "Comments on Mr. Edward F. Dention's Paper," in 1964 OECD, <u>The Residual Factor and Economic Growth</u> Paris: OECD: 67-73.

MacDonald, John B., and Murray G. Ross. "Ontario and Its Universities. 1974 A Statement By the York University-University of Toronto Seminar on Higher Education," <u>Minerva</u>, Vol. 12, No. 4, Oct.: 515-521.

McGrath, Earl J. "The Time Bomb of Technocratic Education," <u>Change</u>, 1974 Vol. 6, No. 7, Sept.: 24-29.

McIntosh, Naomi E. "To Make Continuing Education a Reality," Oxford 1979 Review of Education, Vol. 5, No. 2: 169-182.

Maclure, Stuart. "The Endless Agenda: Matters Arising," Oxford Review 1979a of Education, Vol. 5, No. 2: 111-127.

------ "Envoi," Oxford Review of Education, Vol. 5, No. 2: 195-198. 1979b

-----. "The School in Question: A Comment," Oxford Review of 1979c Education, Vol. 5, No. 3: 245-249.

MacRae, Jr., Duncan. "Review Essay: The Sociological Economics of 1978 Gary S. Becker," <u>American Journal of Sociology</u>, Vol. 83, No. 5, March: 1244-1258.

McCullough, Constance M., et. al. "Reading Comprehension Education 1974 in Fifteen Countries," <u>American Educational Research Journal</u>, Vol. 2, No. 4, Fall: 409-414.

Mackay, Stuart. "Do You Feel Unmotivated at University? You're Not 1981 Alone," Gateway, March 24.

McNamara, William. "Cold Comfort From the White House," <u>Change</u>, 1979 Vol. 11, No. 2, March: 22-23. McQueen, Roderick. "The Dismal Science's Sunny Side: Contradictions 1980 Beset Economists As Their Credibility Takes A Beating," <u>Maclean's</u>, Vol. 93, Dec. 29: 34.

Maddison, Angus. "Education, Inequality, and Life Chances: The 1975 Major Policy Issues," in OECD, Education, Inequality, and Life Chances. Vol. 1. Paris: OECD: 12-30.

Maeroff, Gene I. "Part-time Profs: Better A Little Work Than None 1980 At All," Edmonton Journal, March 20.

Magdorf, Harry. "Economic Myths and Imperialism," <u>Monthly Review</u>, 1971 Vol. 23, No. 7, Dec.: 1-17.

------. "Militarism and Imperialism," <u>American Economic Review</u>, <u>Papers</u> 1970 and Proceedings, Vol. 60, No. 2, Part 2, May: 237-242;243-246.

Magdoff, Harry, and Paul M. Sweezy. "Economic Stagnation and the 1971 Stagnation of Economics," Monthly Review, Vol. 22, No. 11: 1-11.

Malinvaud, Edmond. "Comments on Mr. Edward F. Denison's Paper," in 1964 OECD, <u>The Residual Factor and Economic Growth</u>. Paris: OECD: 57-66.

 Mann, Michael. "The Ideology of Intellectuals and Other People in the
 1975 Development of Capitalism," in Lindberg et. al. (eds.), <u>Stress</u> and Contradictions in Modern Capitalism; Public Policy and the Theory of the State. Lexington, Mass.: Lexington Books: 275-307.

Marchand, Jean. "Recent Institutional Changes for Improved Manpower 1967 Utilisation in Canada," in OECD, <u>Policy Conference on Highly</u> Qualified Manpower. Paris: OECD: 341-349.

Marchand, Philip. "Universities in Trouble. The End of a Dream," 1976 <u>Atlas World Press Review</u>, Dec.: 29-31.

Marglin, Stephen A. "What Do Bosses Do? The Origins and Functions of 1974 Hierarchy in Capitalist Production," <u>The Review of Radical</u> <u>Political Economics</u>, Vol. 6, No. 2, Summer: 60-112.

Marjoribanks, Kevin. "Equal Educational Opportunity: A Definition," 1975 Oxford Review of Education, Vol. 1, No. 1: 25.

Masters, Stanley H. "The Effects of Family Income on Children's 1969 Education: Some Findings on Inequality of Opportunity," Journal of Human Resources, Vol. 4, No. 2, Spring: 158-175.

Mehta, M.M. "Contribution of Human Resources to Economic Growth," in
1976 Mehta, Human Resources Development Planning. Dehli, India: The Macmillan Co. of India, Ltd.: 1-21.

Merrett, Stephen. "The Rate of Return to Education: A Critique," 1966 Oxford Economic Papers, Vol. 18, Nov.: 289-303.

Meyer, John W. "The Effects of Education as an Institution," 1977 American Journal of Sociology, Vol. 83, No. 1, July: 55-77.

Miles, Michael. "The Triumph of <u>Reaction</u>," <u>Change</u>, Vol. 4, No. 10, 1972/73 Winter: 30-36.

Miller, Herman P. "Annual and Lifetime Income in Relation to Education: 1960 1939-1959," American Economic Review, Vol. 50, Dec.: 962-986.

Miller, Jr., H. Laurence. "On the 'Chicago School of Economics'," 1962 Journal of Political Economy, Vol. 70, No. 1, Feb.: 64-69.

Miller, S.M. "Types of Equality: Sorting, Rewarding, Performing," 1976 in Ashline, Pezzulo, and Norris (eds.), Education, Inequality, and National Policy. Lexington, Mass.: Lexington Books: 15-43.

Mincer, Jacob. "Book Review: The Economic Value of Education, by 1964 T.W. Schultz (1963); Economic Aspects of Education: Three Essays, by W.G. Bowen (1964); and External Benefits of Public Education: An Economic Analysis, by B.A. Weisbrod (1964)," American Economic Review, Vol. 55, No. 1, March: 637-640.

------. "Education, Experience, and the Distribution of Earnings and 1975 Employment: An Overview," in Juster (ed.), Education, Income, and Human Behavior. New York: McGraw-Hill Book Co.: 71-93.

------ "Human Capital and Earnings," in National Academy of 1979 Education, <u>Economic Dimensions of Education</u>. Report of a Committee of the National Academy of Education. Washington, D.C.: National Academy of Education, May: 1-31.

------ "Investment in Human Capital and Personal Income Distribution," 1958 Journal of Political Economy, Vol. 66, No. 4, Aug.: 281-302.

------ "On-the-Job Training: Costs, Returns, and Some Implications," 1962 Journal of Political Economy, Vol. 70, No. 5, Part 2, Supplement, Oct.: 50-79.

Minter, W. John. "Current Economic Trends in American Higher Education," 1979 <u>Change</u>, Vol. 11, No. 1, Feb.: 19-25.

Molyneux, Frank. "International Perspectives," in Houghton and 1974 Richardson (eds.), <u>Recurrent Education</u>. London: Ward Lock: 107-121.

Morgan, James N., and C. Lininger. "Education and Income: Comment," 1964 Quarterly Journal of Economics, Vol. 78, May: 346-347.

281

Morgan, James N., and Ismail Sirageldin. "A Note on the Quality 1968 Dimension in Education," Journal of Political Economy, Vol. 76, No. 5, Sept./Oct.: 1069-1077.

Morgan, James N., and M.H. David. "Education and Income," <u>Quarterly</u> 1963 Journal of Economics, Vol. 77, No. 3, Aug.: 423-437.

Morton, D.C., and D.R. Watson. "Compensatory Education and Contemporary 1971 Liberalism, in the United States: A Sociological View," <u>Inter-</u> <u>national Review of Education</u>, Vol. 17, No. 3: 289-307.

Mourat, Lucia. "Ph.D.'s: Missing Link in the Graduation Chain," 1980 Edmonton Journal, Sept. 18.

Moynihan, Daniel Patrick. "On Universal Higher Education," <u>Minerva</u>, 1971 Vol. 9, No. 2, April: 256-257.

Munroe, David. "Post-Secondary Education in Canada: A Survey of 1973" Recent Trends and Developments," in McLeod (ed.), <u>Post-</u> <u>Secondary Education in a Technological Society</u>. Montreal: McGill-Queen's University Press: 33-49.

Mushkin, Selma J. "Health As An Investment," Journal of Political 1962 Economy, Vol. 70, Part 2, Supplement, Oct.: S129-S157.

Nisbet, Robert A. "The Decline of Academic Nationalism," <u>Change</u>, 1974 Vol. 6, No. 6, July/Aug.: 26-31.

Nollen, Stanley D. "The Economics of Education: Research -- Results 1975 and Needs," <u>Teachers College Record</u>, Vol. 77, No. 1, Sept.: 50-77.

O'Connor, James. "The Fiscal Crisis of the State: Part I," <u>Socialist</u> 1970a Revolution, Vol. 1, No. 1, Jan./Feb.: 12-54.

----- "The Fiscal Crisis of the State: Part II," <u>Socialist</u> 1970b Revolution, Vol. 1, No. 2, March/April: 34-94.

------. "Inflation, Fiscal Crisis, and the American Working Class," 1972 Socialist Revolution, Vol. 2, No. 2, March/April: 9-46.

------. "Inflation and the Working Class," in Silverman and Yanowitch 1974 (eds.), The Worker in "Post-Industrial" Capitalism: Liberal and Radical Responses. New York: The Free Press: 433-441.

Ontario. Commission on Post-Secondary Education in Ontario. 1973 "Post-Secondary Education in Ontario: A Statement of Issues," in McLeod (ed.), <u>Post-Secondary Education in a Technological</u> <u>Society</u>. Montreal: McGill-Queen's University Press: 174-198.

O'Toole, James. "The Reserve Army of the Underemployed: I - The 1975a World of Work," **Stange**, Vol. 7, No. 4, May: 26-33;63.

1

----- "The Reserve Army of the Underemployed: II - The Role 1975b of Education," Change, Vol. 7 No. 5, June: 26-33;60-63.

Offe, Claus. "Advanced Capitalism and the Welfare State," <u>Politics</u> 1972 and <u>Society</u>, Vol. 2, No. 4, Summer: 479-488.

Patterson, Orlando. "Inequality, Freedom, and the Equal Opportunity 1978 Doctrine," in Feinberg (ed.), Equality and Social Policy. Urbana, Ill.: University of Illinois Press: 15-41.

Paukert, Felix. "Income Distribution at Different Levels of -1973 Development: A Survey of Evidence," <u>International Labor Review</u>, Vol. 108, No. 2-3, Aug./Sept.:

 Peabody, Gerald E. "Scientific Paradigms and Economics: An Introduction," 1971 <u>The Review of Radical Political Economics</u>, Vol. 3, No. 2, July: 1-16.

Pechman, Joseph A. "The Rich, The Poor, and the Taxes They Pay," 1969 The Public Interest, No. 17, Fall: 21-43.

Peichinis, Stephen G. "The Canadian Occupational Wage Structure," 1974 in Officer and Smith (eds.), <u>Issues in Canadian Economics</u>. Toronto: McGraw-Hill Ryerson, Ltd.: 230-244.

Persky, Joseph. "Radical Political Economics, Poverty, and Income 1977 Distribution," in Coppedge and Davis (eds.), <u>Rural Poverty and the</u> Policy Crisis. Ames, Iowa: Iowa State University Press: 114-125.

Peston, M. Economics and Administration of Education," in Baron and 1969 Taylor (eds.), Educational Administration and the Social Sciences. London: Athlone Press: 60-71.

Pettengill, John. "Income Distribution and Revolutionary Potential 1975 in Underdeveloped Countries," <u>The Review of Radical Political</u> Economics, Vol. 7, No. 1, Spring: 38-51.

Phelps, Edmund S. "Book Review: The Sources of Economic Growth in the 1962 United States and the Alternatives Before Us," Journal of Political Economy, Vol. 70, No. 5, Oct.: 508-509.

Pike, Robert M. "Equality of Educational Opportunity, Dilemnas and 1978/79 Policy Options," Interchange, Vol. 9, No. 2: 30-39.

Pincus, Fred: "Tracking in Community Colleges," Insurgent Sociologist, 1974 Vol. 4, No. 3, Spring: 17-35.

Porter, James N. "Race, Socialization, and Mobility in Educational 1974 and Early Occupational Attainment," <u>American Sociological</u> Review, Vol. 30, No. 3, June: 303-316.

Porter, John Arthur. "The Democratisation of the Canadian Universities 1970 and the Need for a National System," <u>Minerva</u>, Vol. 8, No. 3,

July: 325-356.

"Review Essay: The Limits of Sociology," Contemporary Sociology: A Journal of Reviews, Vol. 2, No. 5, Sept.: 463-467. 1973 "Social Class and Educational Opportunity," in Porter, The ------Vertical Mosaic: An Analysis of Social Class and Power in Canada. 1965 Toronto: University of Toronto Press: 165-198. Pottinger, J. Stanley, and John H. Bunzel. "Race, Sex, and Jobs: The Drive Toward Equality; The Politics of Quotas," Change, 1972 Vol. 4, No. 8, Oct.: 25-35. "Prepare for Burst of Technology, Labor Told," Edmonton Journal, 1981 April, 25. Psacharopoulos, George. "The Perverse Effects of Public Subsidization on Education Or How Equitable is Free Education?" Comparative 1977 Education Review, Vol. 21, No. 1, Feb.: 69-90. "Returns to Education: An Updated International Comparison," in King (ed.), Education and Income. Washington, D.C.: World 1980 Bank: 74-109. "Review of Inequality," Comparative Education Review, 1974 Vol. 18, No. 3, Oct.: 430-450. Radner, Roy, and Leonard S. Milfer. "Demand and Supply in U.S. Higher Education: A Progress Report," American Economic Review, Papers 1970 and Proceedings, Vol. 60, No. 2, Part 2, May: '326-334. Rees, Albert. "Book Review of Human Capital: A Theoretical and Empirical Analysis with Special Reference to Education, by 1965 Gary S. Becker (1964)," American Economic Review, Vol. 55, No. 1, March: 958-960. Reich, Michael, et. al. "A Theory of Labor Market Segmentation," 1973 American Economic Review, Papers and Proceedings, Vol. 63, No. 2, May: 359-365. Reid, Timothy E. "The Mind Beneath the Wheel: The Educationally Handicapped and Automation," in Fotheringham (ed.), <u>Transition:</u> Policies for Social Action. Toronto: McClelland and Stewart, Ltd.: 1966 Resnick, Philip. "The Vertical Mosaic Revisited: The Dynamics of Power in Canada, " in Heap (ed.), Everybody's Canada. Toronto: 1974 Burns and MacEachern, Ltd.: 45-69. Reuber, Grant L., and Ronald G. Brodkin. "Stagflation: The Canadian Experience," in Officer and Smith (eds.), Issues in Canadian 1974 Economics. Toronto: McGraw-Hill Ryerson Ltd.: 25-37.

Ribich, Thomas I. "The Effect of Educational Spending on Poverty 1970 Reduction," in Johns et. al. (eds.), Economic Factors Affecting the Financing of Education. Gainesville, Florida: National Educational Finance Project: 216-243.

Rich, Harvey. "The Vertical Mosaic Revisited: Towards A Macro-sociology 1976 of Canada," Journal of Canadián Studies, Vol. 11: 14-31.

Richardson, C. James. "Education and Social Mobility: Changing 1977 Conceptions of the Role of the Educational Systems," <u>Canadian</u> <u>Journal of Sociology</u>, Vol. 2, No. 4: 417-433.

Richardson, Ray. "Book Review: <u>Investment in Human Capital</u>, by 1973 T.W. Schultz (1971)," <u>Journal of Political Economy</u>, Vol. 81, No. 4, July/Aug.: 1048.

Riesman, David, and Gerald Grant. "Evangelism, Egalitarianism, and 1973 Educational Reform," Minerva, Vol. 11, No. 3, July: 296-317.

Robinson, Albert J. "The Bladen Commission and Graduate Education," 1966 <u>The Canadian Journal of Economics and Political Science</u>, Vol. 32, No. 4, Nov.: 520-525.

"Room At the Top -- If Your I.Q. is Right," <u>The Economist</u>, Vol. 274, 1980 No. 7115, Jan. 12-18: 19-20.

Roosevelt, Frank. "Part I: Controversies in Economic Theory. Cambridge 1975 Economics as Commodity Fetishism," <u>The Review of Radical Political</u> Economics, Vol. 7, No. 4, Winter: 1-32.

Ross, Murray G. "The Dilution of Academic Power in Canada: The 1972 University of Toronto Act," Minerva, Vol. 10, No. 2, April: 242-258.

Rothstein, R. "Down the Up Staircase: Tracking in Schools," <u>This</u> 1971 Magazine Is About Schools, Vol. 5: 103-140.

Rothstein, Stanley W. "The University As People Pupcessor," <u>Change</u>, 1978 Vol. 10, No. 10, Nov.: 8-9.

Rush, G.B. "Occupation and Education in Canada: The Context of Political 1977 Economy," in Carlton, Colley, and MacKinnon (eds.), Education, Change, and Society: A Sociology of Canadian Education. Toronto: Gage Educational Pub.: 24-35.

Russell, Beverly, and Arnold Shore. "Limitations on the Governmental 1976/77 Use of Social Science in the United States," <u>Minerva</u>, Vol. 14, No. 4, Winter: 475-495.

Sacks, Herbert S. "Bloody Monday: The Trauma of College Admissions," 1978 <u>Change</u>, Vol. 10, No. 4, April: 10-15.

Sandee, Jan. "Comments on Mr. Edward F. Denison's Paper," in OECD, 1964 The Residual Factor and Economic Growth. Paris: OECD: 74-76. li. A Sawyer, Darwin O. "Review Essay: Social Roles and Economic Firms: 1978 The Sociology of Human Capital," <u>American Journal of Sociology</u>, Vol. 83, No. 5, March: 1259-1270.

Schaar, John H. "Equality of Opportunity, and Beyond," in Pennock and 1967 Chapman (eds.), Equality, Nomos II. New York: Atherton: 228-249.

Schecter, Stephen. "Capitalism, Class, and Educational Reform in 1977 Canada," in Panitch (ed.), <u>The Canadian State: Political</u> <u>Economy and Political Power</u>. Toronto: University of Toronto Press: 373-416.

Schneider, Joseph W. "Book Review of Women and Work: An International 1973 Comparison, by Margorie Galenson. Ithaca, New York: New York State School of Industrial and Labor Relations, Cornell University, 1973; and <u>Sex Structure</u>, by Anne-Marie Henshel. Don Mills: Longmans, Canada, 1973," <u>Social Forces</u>, Vol. 53, No. 2, Dec.: 348.

Schreiber, E.M. "Cultural Cleavages Between Occupational Categories: 1976 The Case of Canada," Social Forces, Vol. 55, No. 1, Sept.: 16-29.

Schuh, G. Edward. "Neoclassical Economic Theory, Poverty, and Income 1977 Distribution," in Coppedge and Davis (eds.), <u>Rural Poverty and the Policy Crisis</u>. Ames, Iowa: Iowa State University Press: 87-110.

Schultz, Theodore William. "Capital Formation By Education," Journal 1960 of Political Economy, Vol. 68, No. 6, Dec.: 571-583.

------. "Economics of Being Poor," in Coppedge and Davis (eds.), 1977 <u>Rural Poverty and the Policy Crisis</u>. Ames, Iowa: Towa State University Press: 35-42.

------. "The Economics of Poverty. Investing in Poor People: An 1965 Economist's View," <u>American Economic Review, Papers and Proceedings</u>, Vol. 55, No. 2, May: 510-520.

- ------ "Education and Economic Growth," in Henry (ed.), <u>Social</u> 1961a <u>Forces Influencing American Education</u>. Chicago: University of Chicago Press: 46-88.
- ------. "A 'Guide' to Investors in Education with Special Reference 1974 to Developing Countries," in Ward (ed.), Education and Development Reconsidered: The Bellagio Conference Papers. New York: Praeger: 48-57.

-----. "Gross Capital Formation Represented by High School, 1958 College, and University Education in the United States, 1900 to 1956," American Economic Review Paper No. 5807, Department of Economics, University of Chicago, April 29.

-----. "Human Capital," in Sills (ed.), International Encyclopedia 1968a of the Social Sciences, Vol. 2. New York: Crowell, Collier, and

•

Macmillan, Inc.: 278-287.

J .

đ

------. "Human Capital: A Growing Asset," <u>Saturday Review</u>, Vol. 44, 1961b Part 1, Jan. 21%

------ "Human Capital: Policy Issues and Research Opportunities,"' 1972a in National Bureau of Economic Research, Economic Research: <u>Retrospect and Prospect</u>. Vol. 6: <u>Human Resources</u> New York: National Bureau of Economic Research.

------. "The Human Capital Approach to Education," in Johns et. al., 1970a (eds.), <u>Economic Factors Affecting the Financing of Education</u>. Gainesville, Florida: National Education Finance Project: 40-68.

----- "Human Wealth and Economic Growth," The Humanist, Vol. 19, 1959a No. 1, Jan./Feb.: 71-81

-----. "Investment in Human Capital," American Economie Review, 1961c Vol. 51, No. 1, March: 1-17

------. "Investment in Human Capital: Reply," <u>American Economic</u> 1961d Review, Communications, Vol. 51, No. 1, March: 1035-1039.

------. "Investment in Man: An Economist's View," Social Service 1959b Review, Vol. 33, No. 2, June: 109-117.

------. "Optimal Investment in College Instruction: Equity and 1972b Efficiency," Journal of Political Economy, Vol. 80, No. 3, Part 2, May/June: S2-S30.

------. "Public Approaches to Minimize Poverty," in Fishman (ed.), 1966 <u>Poverty Amid Affluence</u>. New Haven, Conn.: Yale University Press: 165-181.

------. "The Rate of Return in Allocating Investment Resources to 1967 Education," <u>Journal of Human Resources</u>, Vol. 2, No. 3, Summer: 293-309.

------. "The Reckoning of Education as Human Capital," in Hansen (ed.), 1970b <u>Education, Income, and Human Capital</u>. New York: National Bureau of Economic Research: 297-306.

------. "Reflections on Investment in Man," Journal of Political 1962 Economy, Vol. 70, No. 5, Part 2, Supplement, Oct.: S1-S8.

------. "Resources for Higher Education: An Economist's View," 1968b Journal of Political Economy, Vol. 76, No. 3, May/June: 327-347.

-----. "Some Economic Aspects of the Northland," in Officer and 1970c Smith (eds.), Canadian Economic Problems and Policies. Toronto: McGraw-Hill Co. of Canada, Ltd.: 352-358.

------. "The Value of the Ability to Deal with Disequilibria," 1975 <u>Journal of Economic Literature</u>, Vol. 13, No. 3: 827-846.

1972c Atomic Scientists, Feb.: 29-32.

Schwartzman, David. "The Contribution of Education to the Quality 1969 of Labor: Reply," <u>American Economic Review</u>, Vol. 59, Dec.: 944-946.

Segre, Monique, et. al. "A New Ideology of Education," Social Forces, 1972 Vol. 50, No. 3, March: 313-332.

Selden, David. "The Mystery of Educational Productivity," <u>Change</u>, 1978 Vol. 10, No. 2, Feb.: 50-51.

Selowsky, Marcelo. "On the Measurement of Education's Contribution 1969 to Growth," <u>Quarterly Journal of Economics</u>, Vol. 83, No. 3, Aug.: 449-463.

"Seniors Hit Hardest By Inflation; Labor Income Suffered in 1970's," 1980 <u>Edmonton Journal</u>, May 28.

"'70's World Jobless Rate Worst in Decades - Study," Edmonton Journal, 1981 March 20.

Sewell, William H. "Book Review: The Illusion of Equality: The Effect 1974 of Education and Opportunity, Inequality and Social Conflict, by Murray Milner, Jr., San Francisco: Jossey-Ross, 1972," Social Forces, Vol. 53, No. 1, Sept.: 134.

Shafer, Walter E. et. al. "Programmed for Social Class: Tracking in 1970 High School," <u>Trans-action</u>, Vol. 7, No. 12, Oct.: 39-46;63.

Shaffer, Harry G. "Investment in Human Capital: Comment," <u>American</u> 1961 <u>Economic Review</u>, Communications, Vol. 51, No. 1, March: 1026-1035.

Shaiken, Harley. "Craftsman into Baby Sitter," in Illich et. al., 1977 <u>Disabling Professions</u>. Don Mills: Burns and MacEachern: 111-127.

Shannon, Robin. "inequality in the Distribution of Personal Income," 1975 in OECD, <u>Education, Inequality, and Life Chances</u>. Vol. 1. Paris: OECD: 109-158.

Shea, Brent Mack. "Two-Year Colleges and Inequality," <u>Integrated</u> 1975 <u>Education</u>, Vol. 13, No. 1, Jan./Feb.: 38-43.

Shils, Edward. "Elitism," <u>Minerva</u>, Vol. 12, No. 1, Jan.: 1-7. -1974a

------. "Identity and Openness in Higher Education," <u>Minerva</u>, Vol. 12, 1974 No. 2, April: 258-276.

-----. "The Reorganisation of Higher Education in Sweden," 1974c Minerva, Vol. 12, No. 1, Jan.: 83-114.

Simon, Brian. "Karl Marx and Education," in Simon, <u>Intelligence</u>, 1971 <u>Psychology</u>, and Education: <u>A Marxist Critique</u>. London: Lawrence and Wishart: 177-199.

"Skilled Workers on Import List," Edmonton Journal, Feb. 2. 1981

Smith, David N. "Capitalism and the Universities," in Smith, 1974 <u>Who Rules the Universities?</u> New York: Monthly Review Press: 139-172.

Smith, Marshall S. "Equal Opportunity -- Some Promise and a Lack of 1976 Vision," in Ashline, Peszulo, and Norris (eds.), Education, <u>Inequality, and National Policy</u>. Lexington, Mass.: Lexington Books: 169-183.

------ "Equality of Educational Opportunity: Comments on Bowles 1968 and Levin," <u>Journal of Human Resources</u>, Vol. 3, No. 3, Summer: 384-389.

Smith, Virginia B. "More for Less: Higher Education's New Priority," 1972 in Wilson and Mills (eds.), <u>Universal Higher Education: Costs</u>, <u>Benefits</u>, <u>Options</u>. Washington, D.C.: American Council on Education: 122-141.

Smolensky, Eugene. "Investment in the Education of the Poor: A 1966 Pessimistic Report," <u>American Economic Review, Papers and</u> Proceedings, Vol. 56, No. 2, Part 3, May: 370-378.

Solow, Robert M. "Beview of Human Capital, by Gary S. Becker," 1965 Journal of Political Economy, Vol. 73, No. 5, Oct.: 552-553.

Spady, William G. "Educational Mobility and Access: Growth and 1967 Paradoxes," <u>American Journal of Sociology</u>, Vol. 73, No. 3, Nov.: 273-286.

Spence, Andrew Michael. "Job Market Signalling," Quarterly Journal of 1973 Economics, Vol. 87, No. 3, Aug.: 355-374.

Stager, David A.A. "Economic Issues in Canadian Education," in Officer 1974 and Smith (eds.), <u>Issues in Canadian Economics</u>. Toronto: McGraw-Hill Ryerson, Ltd.: 311-320.

Stewart, Walter. "The Real Poor in Canada -- and Why We Don't Know 1971 Who They Are," <u>Maclean's</u>, Vol. 84, No. 1, Jan.: 44-46.

Stigler, George J. "Comment," <u>Journal of Political Economy</u>, Vol. 70, 1962a No. 1, Feb.: 70-71. 1962b Economy, Vol. 70, No. 1, Feb.: 94-105.

• 1

Stiglitz, Joseph/E. "The Theory of 'Screening', Education, and the 1975 Distribution of Income," American 'Economic Review, Vol. 65, No. 3, June: 283>300.

Stirling, Robert, and Denise Kouri. "Unemployment Indexing -- The 1979 Canadian Context," in Fry (ed.), Economy, Class, and Social, <u>Reality. Issues in Contemporary Canadian Society</u>. Toronto: Butterworths: 169-205.

Stolzenberg, Ross M. "Occupations, Labor Markets, and the Process of 1975 Wage Attainment," <u>American Sociological Raview</u>, Vol. 40, No. 5, Oct.: 645-665.

Stolzman, James, and Herbert Gamberg. "Marxist Class Analysis Versus 1973/74 Stratification Analysis as General Approaches to Social Inequality," <u>Berkeley Journal of Sociology</u>, Vol. 18: 10,225.

Strumilin, Stanislav G. "The Economics of Education in the U.S.S.R., 1962 International Social Science Journal, Vol. 14, No. 4: 633-659.

Struzik, Ed. "Job Dissatisfaction --- Dilemna or Crisis?" <u>Edmonton</u> 1981 Journal, Jan. 27.

"Subsidies Offered to Hire Disabled," Edmonton Journal, Jan. 23. 1981

Sweet, Lois. "Exploitation of Women is a Matter of Economics," 1981 Edmonton Journal, Jan. 19.

1981 Edmonton Journal, Feb. 9.

Sweazy, Paul M. "On the Theory of Monopoly Capitalism," Monthly 1972 Review, Vol. 23, No. 11, April: 1-23.

Szmanski, Albert. "Racism and Sexism as Functional Substitutes in the 1976 Labor Market," <u>The Sociological Quarterly</u>, Vol. 17, No. 1, Winter: 65-73.

Taubman, Paul, and Terence Wales. "Education as an Investment' and as 1975a a Screening Device," in Juster (ed.), Education, Income, and Human Behavior. New York: McGraw-Hill Book Co.: 95-121.

1975b and as a Screening Device," in Carnegie Commission on the Future of Higher Education, Sponsored Research of the Carnegie Commission on Higher Education. New York: McGraw-Hill Book Co.: 188-190. ------. "Mental Ability and Higher Educational Attainment in the 1975c Twentieth Century," in Juster (ed.), <u>Education, Income, and</u> Human Behavior. New York: McGraw-Hill Book Co.: 47-70.

Tenenbaum, Samuel. "The Teacher, the Middle Class, the Lower Class," 1966 Ph Adams (ed.), <u>Introduction to Education: A Comparative</u> Analysis. Belmont, Cal.: Wadsworth Pub. Co.: 294-301.

Thurow, Lester C. "The Causes of Poverty," <u>Quarterly Journal of</u> 1967 <u>Economics</u>, Vol. 81, **Ho**. 1, Feb.: 39-57.

------. "Disequilibrium and the Marginal Productivity of Capital 1968 and Labor," <u>Review of Economics and Statistics</u>, Vol. 50, No. 1, Feb.: 23-31.

-----. "Education and Economic Inequality," <u>The Public Interest</u>, 1972 No. 28, Summber: 66-81.

------. "Measuring the Economic Benefits of Education," in Gordon 1974 (ed.), <u>Higher Education and the Labor Market</u>. New York: McGraw-Hill Book Co.: 373-418.

------. "Problems Without Solutions: Solutions Without Problems," 1976 in Ashline, Pezzulo, and Norris (eds.), Education, Inequality, and National Policy. Lexington, Mass.: Lexington Books: 157-167.

Trimberger, Ellen Kay. "Open Admissions: A New Form of Tracking?" 1973 Insurgent Sociologist, Vol. 4, No. 1, Fall: 29-43.

Trow, Martin. "'Elite Higher Education': An Endangered Species?" 1976 Minerva, Vol. 14, No. 3, Autumn: 355-376.

Turner, Jonathan H. "A Strategy for Reformulating the Dialectical 1975 and Functional Theories of Conflict," <u>Social Forces</u>, Vol. 53, No. 3, March: 433-444.

Tussing, A. Dale. "Education and the New Unemployment," <u>The Journal of</u> 1976 <u>General Education</u>, Vol. 28, No. 2, Summer: 115-131.

Tyler, Gus. "The University and the Labor Union: Educating the 1979 Proletariat," Change, Vol. 11, No. 1, Feb.: 32-37.

Tyree, Andrea, and Judith Treas. "The Occupational and Marital 1974 Mobility of Women," <u>American Sociological Review</u>, Vol. 39, No. 3, June: 293-302.

Unseem, Michael. "Government Influence on the Social Science 1976 Paradigm," <u>Sociological Quarterly</u>, Vol. 17, No. 2, Spring: 146-161.

 Vaizey, John. "Implications of the IEA Studies for Educational
 1975 Planning with Respect to Organization and Resource Allocation," in Purves and Levine (eds.), <u>Educational Policy and International</u> Assessment. Implications of the IEA Surveys of Achievement.

-Berkeley, Cal.: McCutchan: 153-163.

Voelker, Alan M. "Science Education in Nineteen Countries," 1974 <u>American Educational Research Journal</u>, Vol. 11, No. 4, Fall: 415-418.

Wachtel, Howard M. "Class Consciousness and Stratification in the 1974 Labor Process," <u>The Review of Radical Political Economics</u>, Vol. 6, No. 1, Spring: 1-31.

------. "Looking at Poverty from a Radical Perspective," The 1971 <u>Review of Radical Political Economics</u>, Vol. 3, No. 3, Summer: 1-19.

Walker, Mike. "Grants, Not Loans: FAS," <u>Gateway</u>, March 31. 1981

Wallace, Anthony F.C. "Schools in Revolutionary and Conservative 1968 Societies," in Lloyd-Jones and Rosenau (eds.), <u>Social and</u> <u>Cultural Foundations of Guidance</u>. New York: Holt, Rinehart, and Winston: 193-203.

Watkins, Melville. "Education in the Branch-Plant Economy," 1969 Canadian Dimension, Vol. 6, No. 5, Oct./Nov.: 37-39.

Weiner, Rose. "Credentials and Common Sense," <u>Manpower Report</u>, 1968 Dec.:

Weisbrod, Burton A. "Education and Investment in Human Capital," 1962 Journal of Political Economy, Vol. 70, No. 5, Part 2, Supplement, Oct.: S106-S123.

Welch, Finis. "Human Capital Theory: Education, Discrimination, and 1975 Life Cycles," <u>American Economic Review, Papers and Proceedings</u>, Vol. 65, No. 2, May: 63-73.

------. "Labor-Market Discrimination: An Interpretation of Income 1967 Differences in the Rural South," <u>Journal of Political Economy</u>, Vol. 75, No. 3, June: 225-240.

Weisskopf, Thomas E. "Capitalism, Underdevelopment, and the Future 1972 of the Poor Countries," in Bhagwati (ed.), <u>Economics and World</u> <u>Order From the 1970's to the 1990's</u>. New York: Macmillan: 43-67.

West, E.G. "Private Versus Public Education," Journal of Political 1964 <u>Economy</u>, Vol. 72, No. 5, Oct.: 465-475.

Westoby, Adam. "Economists and 'Human Capital'," in Holly (ed.), 1974 <u>Education or Domination? A Critical Look at Educational</u> Problems Today. London: Arrow; 38-57.

Wharton, Jr., Clifton R. "The New Darwinism of Basic Learning," <u>Change</u>, 1979 Vol. 11, No. 8, Dec.: 38-41.

Wilcox, Kathleen, and Pia Moriarty. "Schooling and Work: Social
1976 Constraints on Equal Educational Opportunity," <u>Social Problems</u>, Vol. 24, No. 2, Dec.: 204-213.

Wilkinson, Bruce W. "Present Values of Lifetime Earnings for 1966 Different Occupations," <u>Journal of Political Economy</u>, Vol. 74, No. 6, Dec.: 556-572.

Wilson, Kenneth L. "Toward an Improved Explanation of Income 1978 Attainment: Recalibrating Education and Occupation," American Journal of Sociology, Vol. 84, No. 3, Nov.: 684-697.

Windham, Douglas M. "Economic Analysis and the Public Support of
 1979 Higher Education: The Divergence of Theory and Policy," in
 National Academy of Education, Economic Dimensions of Education.
 Washington, D.C.: National Academy of Education: 117-139.

Wolf, Frank. "Democracy as Disease?" <u>Change</u>, Vol. 11, No. 3, 1979 April: 8-9;64.

Wolfe, David. "The State and Economic Policy in Canada, 1968-1975," 1977 in Panitch (ed.), <u>The Canadian State: Political Economy and</u> <u>Political Power</u>. Toronto: University of Toronto Press: 251-288.

Wolff, Richard D. "Modern Imperialism: The View from the Metropolis," 1970 <u>American Economic Review, Papers and Proceedings</u>, Vol. 60, No. 2, Part 2, May: 225-230.

Wright, Erik Olin. "Class Boundaries in Advanced Capitalist Societies," 1976 New Left Review, No. 98, July/Aug.: 1-41.

-----. "Race, Class, and Income Inequality," <u>American Journal of</u> 1978 Sociology, Vol. 83, No. 6, May: 1368-1397.

Wycko, 'Bill. "The Work Shortage: Class Struggle and Capital Reproduction," 1975 <u>The Review of Radical Political Economics</u>, Vol. 7, No. 2, Summer: 11-30.

Wynne, Dorothy, et. al. "Should Liberal Arts Colleges Try to Make 1979 Their Graduates More 'Marketable' So That They Can Better Compete For Jobs With Vocationally-Prepared Graduates?" <u>Change</u>, Vol. 11, No. 6, Sept.: 52-55.

Yankelovich, Daniel. "Who Gets Ahead in America," <u>Psychology Today</u>, 1979 Vol. 13, No. 2, July: 24-28;40-43;90-91.

Yarmolinsky, Adam. "Challenges to Legitimacy: Dilemnas and Directions," 1976 <u>Change</u>, Vol. 8, No. 3, April: 18-25.

Zeigler, Harmon, and Wayne Peak. "The Political Functions of the n.d. Educational System," (Mimeo).

Zweig, Michael. "Bourgeois and Radical Paradigms in Economics," 1971a <u>The Review of Radical Political Economics</u>, Vol. 3, No. 2, July: 43-58.

------. "New Left Critique of Economics," <u>The Review of Radical</u> 1971b Political Economics, Vol. 3, No. 2, July: 67-74.

Books.

Adams, Donald K., ed. Introduction to Education: A Comparative 1966 Analysis. Belmont, Cal.: Wadsworth Pub. Co.

Adams, Ian. The Poverty Wall. Toronto: McClelland and Stewart, Ltd. 1970

Adams, Ian, et. al. <u>The Real Poverty Report</u>. Edmonton: Hurtig Pubs. 1971

Adelman, Irma, and Cynthia Taft Morris. Economic Growth and Social 1973 Equity in Developing Countries. Stanford, Cal.: Stanford University Press.

Alberta. Commission on Educational Planning. <u>A Choice of Futures</u>. 1972 Report of the Commission on Educational Planning. Walter H. Worth, Commissioner. Edmonton: Government Printers.

Alberta. Department of Advanced Education. <u>Further Education:</u> 1975 <u>Policy, Guidelines, and Proceedures</u>. Edmonton: Alberta Advanced Education.

Alberta Colleges Commission. <u>First Annual Report</u>, <u>1969-1970</u>: The 1971 Alberta College System.

1972 Second Annual Report, 1970-1971: The Alberta College System.

Alexander, Kern, and K. Forbis Jordan, eds. <u>Educational Needs in the</u> 1976 Public Economy. Gainesville, Florida: University of Florida Press.

Alexander, Leigh, and John Simmons. <u>The Determinants of School</u> 1975 <u>Achievement in Developing Countries: The Educational Production</u> <u>Function</u>. Washington, D.C.: International Bank for Reconstruction and Development.

Anderson, C. Arnold, et. al. <u>Where Colleges Are and Who Attends:</u> 1972 <u>Effects of Accessibility on College Attendance</u>. A Report Prepared for the Carnegie Commission on the Future of Higher Education. New York: McGraw-Hill Book Co.

Armit, Robert E. <u>An Introductory Handbook on Work in Alberta</u>. Edmonton: 1980, Corewest Management Services, Ltd. -----. Measurement and Analysis of Employment and Income in

1971 Alberta: A Starting Point. A Report prepared for the Human Resources Research Council, Government of Alberta, January 29.

Ashenfelter, Orley C., and James Blum, eds. <u>Evaluating the Labor</u> 1976 <u>Market Effects of Social Programs</u>. Princeton, New Jersey: Industrial Relations Section, Department of Economics, Princeton University.

Ashenfelter, Orley C., and William G. Bowen, eds. Labor and the 1975 National Economy. Rev. ed. New York: W.W. Norton.

Ashenfelter, Orley C., and Wallace E. Oates, eds. <u>Essays in Labor</u> 1977 <u>Market Analysis: In Memory of Yochanan Peter Comay</u>. New York: John Wiley and Sons.

Ashenfelter, Orley C., and Albert Rees, eds. <u>Discrimination in</u> 1973 Labor Markets. Princeton, New Jersey: Princeton University Press.

Ashline, Nelson F., et. al., eds. <u>Education, Inequality, and</u> 1976 National Policy. Lexington, Mass.: Lexington Books.

Atkinson, A.B., ed. <u>The Personal Distribution of Incomes</u>. London: 1976 George Allen and Unwin, Ltd.

Averitt, Robert T. The Dual Economy; The Dynamics of American 1968 Industry Structure. New York: W.W. Norton.

Baetz, Reuben C., and Kevin Collins. <u>Manpower Programs: Equity and</u> 1975 Integration. Ottawa: Canadian Council on Social Development.

Baran, Paul A., and Paul M. Sweezy. <u>Monopoly Capital; An Essay on the</u> 1966 <u>American Economic and Social Order</u>. New York: Monthly Review Press.

Barlow, Melvin L. <u>The Philosophy for Quality Vocational Education</u> 1974 Programs. Washington, D.C.: American Vocational Association.

Baron, George, and William Taylor, eds. Educational Administration 1969 and the Social Sciences. London: Athlone Press.

Becker, Gary S. <u>The Economic Approach to Human Behavior</u>. Chicago: 1976 University of Chicago Press.

-----. Economic Theory. New York: Knopf.

1971

------. Human Capital: A Theoretical and Empirical Analysis, With 1964 Special Reference to Education. New York: National Bureau of Economic Research.

1975 Special Reference to Education. 2nd. ed. New York: National

Bureau of Economic Research.

 Human Capital and the Personal Distribution of Income:
 An Analytical Approach. Ann Arbor, Michigan: Department of Economics and the Institute of Public Administration, University of Michigan.

Bell, Daniel. The Coming of Post-Industrial Society. New York: 1973 Basic Books.

Bendix, Reinhard, and Seymour Martin Lipset, eds. <u>Class, Status</u>, 1963 and Power. Social Stratification in Comparative Perspective. New York: The Free Press.

Berg, Ivar. Education and Jobs: The Great Training Robbery. 1970 New York: Praeger.

Bernbaum, Gerald. Knowledge and Ideology in the Sociology of Education. 1977 London: Macmillan.

Bernbaum, Gerald, ed. <u>Schooling in Decline</u>. London: Macmillan. 1979

Bertram, Gordon W. The Contribution of Education to Economia Growth. 1966 (Staff Study No. 12, Science Council of Canada, June 1966). Ottawa: Information Canada.

Bestor, Arthur Eugene. <u>Educational Wastelands; The Retreat From</u> 1953 <u>Learning in Our Public Schools</u>. Urbana, Ill.: University of Illinois Press.

Bird, Caroline. <u>The Case Against College</u>. New York: David McKay. 1975

Birou, Alain, et. al., eds. <u>Towards a Re-Definition of Development</u>. 1977 <u>Essays and Discussion on the Nature of Development in an</u> International Perspective. New York: Pergamon Press for OECD.

Bisconti, Ann Stouffer, et. al. <u>College Education on the Job...The</u> 1976 <u>Graduates' Viewpoint</u>. Bethlehem, Pa.: The CPC Foundation.

Bissell, Claude Thomas, ed. <u>Canada's Crisis in Higher Education</u>. 1957 Toronto: University of Toronto Press.

Blau, Peter Michael. <u>Inequality and Heterogeneity: A Primitive</u> 1977 <u>Theory of Social Structure</u>. New York: The Free Press.

-----. The Organization of Academic-Work. New York: John Wiley 1973 and Sons.

Blau, Peter Michael, and Otis Dudley Duncan. <u>The American Occupational</u> 1967 Structure. New York: John Wiley and Sons. Blaug, Mark. Economics of Education. A Selected Annotated Bibliography. 1970a 2nd ed. Oxford: Pergamon Ptess.

1973 Countries. Geneva: International Labour Office.

-----. An Introduction to the Economics of Education. London: 1970b Allen Lane.

Blaug, Mark, ed. <u>Economics of Education</u>. Vol. 1. Harmondsworth, 1968 England: Penguin Books.

Blaug, Mark, ed. <u>Economics of Education</u>. Vol. 2. Harmondsworth, 1969 England: Penguin Books.

Blaug, Mark, et. al. The Utilisation of Educated Manpower in 1967 Industry; A Preliminary Report. London: Oliver and Boyd.

Blaxall, Martha, and Barbara Reagan, eds. <u>Women and the Workplace;</u> 1976 <u>The Implications of Occupational Segregation</u>. Chicago: University of Chicago Press.

Bonnell, V., and Michael Reich. Workers in the American Economy. 1969 Boston: New England Free Press.

Boocock, Sarane Spence. <u>An Introduction to the Sociology of Learning</u>. 1972 Boston: Houghton Mifflin Co.

Bottomore, Thomas B. <u>Elites and Society</u>. Harmondsworth, England: 1964 Penguin Books.

-----. <u>Social Criticism in North America</u>. Toronto?: CBC. 1966

. . .

-----. Sociology. A Guide to Problems and Literature. 2nd ed. . 1972 London: George Allen and Unwin.

Boudon, Raymond. <u>Education</u>, <u>Opportunity</u>, <u>and Social Inequality</u>; 1973 <u>Changing Prospects in Western Society</u>. New York: John Wiley and Sons.

Bourgault, Pierre L. <u>Innovation and the Structure of Canadian Industry</u>. 1972 (Background Study for the Science Council of Canada. Special Study No. 23, Oct., 1972). Ottawa: Information Canada.

Bowen, Water G. Economic Aspects of Education: Three Essays. 1964 Printon, New Jersey: Department of Economics, Industrial Relations Section, Princeton University.

Bowles, Samuel, and Herbert Gintis. Schooling in Capitalist America. 1976 Educational Reform and the Contradictions of Economic Life. New York, Basic Books. Bowman, Mary Jean, et. al., eds. <u>Readings in the Economics of Education</u>. 1968 Paris: UNESCO.

 Boyd, A.D., and A.C. Gross. <u>Education and Jobs. Career Patterns</u>
 1973 <u>Among Selected Canadian Science Graduates With International</u> <u>Comparisons.</u> (Special Study No. 28, Science Council of Canada, June, 1973). Ottawa: Information Canada.

Braverman, Harry. Labor and Monopoly Capital; The Degradation of 1974 Work in the Twentieth Century. New York: Monthly Review Press.

Breton, Raymond. <u>The Canadian Condition: A Guide to Research in</u> 1977 <u>Public Policy</u>. Montreal: Institute for Research on Public Policy.

Brookover, Wilbur B., and Edsel L. Erickson. <u>Society, Schools, and</u> 1969 <u>Learning</u>. East Lansing, Mich.: Michigan State University Press.

------. <u>Sociology of Education</u>. Homewood, Ill.: Dorsey Press. 1975

Brown, Murray, ed. <u>The Theory and Empirical Analysis of Production</u>. 1967 New York: National Bureau of Economic Research.

Burn, Barbara B. <u>Higher Education in a Changing World; Reflections</u> 1976 <u>on an International Seminar</u>. New York: International Council for Educational Development.

Burn, Barbara B., et. al. <u>Higher Education in Nine Countries: A</u> 1971 <u>Comparative Study of Colleges and Universities Abroad.</u> A Report for the Carnegie Commission on the Future of Higher Education. New York: McGraw-Hill Book Co.

Bushnell, David S. Organizing for Change: New Priorities for 1973 Community Colleges: New York: McGraw-Hill Book Co.

Buttrick, John A. <u>Educational Problems in Ontario and Some Policy</u> 1977 <u>Options</u>. (Occasional Paper 4). Toronto: Ontario Economic Council.

Byrne, David, et. al. The Poverty of Education. A Study in the 1975 Politics of Opportunity. London: Martin Robertson.

Byrne, N., and J. Quarter, eds. <u>Must Schools Fail? The Growing</u> 1972 <u>Debate in Canadian Education</u>. Toronto: McClelland and Stewart.

Cain, Glen, and Harold W. Watts, eds. <u>Income Maintenance and Labor</u> 1973 <u>Supply. Econometric Studies</u>. Chicago: Markham.

Callahan, Raymond E. Education and the Cult of Efficiency; A Study 1962 of the Social Forces That Have Shaped the Administration of the Public Schools. Chicago: University of Chicago Press. Cameron, David Murray. <u>Schools for Ontario: Policy-Making</u>, 1972 <u>Administration, and Finance in the 1960's</u>. Toronto: University of Toronto Press.

Campbell, Gordon. <u>The Community College in Canada</u>. Ottawa: 1971a Association of Universities and Colleges of Canada.

------. <u>Community Colleges in Canada</u>. Toronto: McGraw-Hill 1971b Ryerson, Ltd.

Canada. Department of Labour. <u>Occupational Trends in Canada, 1931</u> 1965 <u>to 1961</u>. (Research Program on the Training of Skilled Manpower. Report No. 11, Sept. 1963). Ottawa: Queen's Printer.

------ Department of Manpower and Immigration. <u>The Canadian</u> 1974 <u>Classification and Dictionary of Occupational Qualification</u> <u>Requirements</u>. Vols. 1 and 2. Ottawa: Information Canada.

------. Parliament. Special Senate Committee on Poverty. 1971 <u>Poverty in Canada</u>. Report of the Special Senate Committee on Poverty. Ottawa: Information Canada.

------. Royal Commission on Canada's Economic Prospects. <u>Report</u> 1957 <u>of the Royal Commission on Canada's Economic Prospects</u>. (Cordon Commission.) Ottawa: Queen's Printer.

Canadian Committee of the International Conference on Social Welfare. , 1972 Developing Social Policy in Conditions of Dynamic Change. Report of the Canadian Committee of the International Conference on Social Welfare to the Biennial Meetings of the Conference at The Hague, August 8, 1972. Ottawa: Canadian Council on Social Development.

Canadian Teachers' Federation. <u>A Summary of Federal Direct and</u> 1973 <u>Indirect Involvement in Canadian Education</u>. Ottawa: Canadian Teachers' Federation.

Canadian Welfare Council. Social Policies for Canada: A Statement 1969 By the Canadian Welfare Council. Ottawa: Canadian Welfare Council.

Cantor, Leonard M., and Iolo F. Roberts. Further Education Today; 1979 A Critical Review. London: Routledge and Kegan Paul. Carnegie Commission on the Future of Higher Education. <u>College</u> 1973a <u>Graduates and Jobs</u>. New York: McGraw-Hill Book Co.

- ------. <u>A Digest of Reports of the Carnegie Commission on Higher</u> 1974 <u>Education With an Index to Recommendations and Suggested</u> <u>Assignments of Responsibility for Action</u>. New York: McGraw-Hill Book Co.
- <u>Quality and Equality: New Levels of Federal Responsibility</u>
 for <u>Higher Education</u>. A Special Report and Recommendations by the Carnegie Commission on Higher Education. Hightstown, New Jersey: McGraw-Hill Book Co.

------. <u>Sponsored Research of the Carnegie Commission on Higher</u> 1975 Education. New York: McGraw-Hill Book Co.

-------. Toward a Learning Society: Alternative Channels to Life, 1973b Work, and Service. New York: McGraw-Hill Book Co.

- Carnoy, Martin. <u>Can Educational Policy Change Income Distribution</u> 1979 <u>in Latin America?</u> (With the collaboration of Jose Lobo, Alejandro Toledo, and Jacques Velloso.) Geneva: International Labour Organisation.
- <u>The Economics of Schooling and International Development</u>.
 (Cuaderno, No. 63.) Cuernavaca, Mexico: Centro Intercultural de Documentacion.
- 1977 UNESCO International Institute for Educational Planning.
 - -----. Education As Cultural Imperialism. New York: David McKay. 1974

Carnoy, Martin, and Derek Shearer. Economic Democracy: The Challenge 1980 of the 1980's. White Plains, New York: M.E. Sharpe.

- Caty, G., et. al. <u>The Research System.</u> Comparative Survey of the 1972 Organization and Financing of Fundamental Research. Vol. 1: France, Germany, and the U.K. Paris: OECD.
- Caty, G., et. al. <u>The Research System. Comparative Study of the</u> 1973 <u>Organization and Financing of Fundamental Research</u>. Vol. 2: Belgium, Netherlands, Norway, Sweden, Switzerland. Paris: OECD.
- Caty, G., et. al. <u>The Research System.</u> Comparative Study of the 1974 <u>Organization and Financing of Fundamental Research</u>. Vol. 3: <u>Canada, U.S., General Conclusions</u>. Paris: OECD.

Cheal, John E. Investment in Canadian Youth. An Analysis of 1963 Input-Output Differences Among Canadian Provincial School Systems. Toronto: Macmillan. Chernick, S.E. Interregional Disparities in Income. (Staff Study 1966 No. 14 for Economic Council of Canada, August 1966.) Ottawa: Queen's Printer.

Chiswick, Barry R., and June A. O'Neill, eds. <u>Human Resources and</u> 1977 Income Distribution: Issues and Policies. New York: W.W. Norton.

Clark, Burton. Educating the Expert Society. San Francisco, Cal.: 1962 Chandler.

Clark, Samuel Delbert. <u>Canadian Society in Historical Perspective</u>. 1976 Toronto: McGraw-Hill Ryerson, Ltd.

Clement, Wallace. <u>The Canadian Corporate Elite</u>. Toronto: McClelland 1975 and Stewart.

Clignet, Remi. Liberty and Equality in the Educational Process; 1974 <u>A Comparative Sociology of Education</u>. New York: John Wiley and Sons.

Clignet, Remi, and Philip Foster. <u>The Fortunate Few: A Study of</u> 1967 <u>Secondary Schools and Students in the Ivory Coast</u>. Evanston, Ill.: Northwestern University Press.

Cohen, Marshall, et. al., eds. <u>Equality and Preferential Treatment</u>. 1977 Princeton, New Jersey: Princeton University Press.

Cohen, Wilbur J. <u>Toward A Social Report</u>. Washington, D.C.: U.S. 1969 Government. Department of Health, Education, and Welfare.

Cohn, Elchanan. <u>The Economics of Education</u>. Lexington, Mass.: 1972 Lexington Books.

-----. <u>The Economics of Education</u>. rev. ed. Lexington, Mass.: 1979 Lexington Books.

Coleman, James S. Equality of Educational Opportunity. Washington, 1966 D.C.: U.S. Government Printing Office.

Commission on the Financing of Higher Education in Canada. <u>Financing</u> 1965 <u>Higher Education in Canada</u>. (Bladen Report.) Toronto: University of Toronto Press.

Coons, John E., et. al. <u>Private Wealth and Public Education</u>. 1970 Cambridge, Mass.: Harvard University Press.

Coppedge, Robert O., and Carlton G. Davis, eds. <u>Rural Poverty and</u> 1977 the Policy Crisis. Ames, Iowa: Iowa State University Press.

Cordell, Arthur J. <u>The Multinational Firm, Foreign Direct Investment,</u> 1971 <u>and Canadian Science Policy</u>. (Special Study No. 22, Background Study for Science Council of Canada.) Ottawa: Information Canada. Cosin, B.R., ed. <u>Education: Structure and Society; Selected Readings</u>. 1972 Harmondsworth, England: Penguin Books.

Cosin, B.R., et. al. <u>School and Society; A Sociological Reader</u>. 1971 London: Routledge and Kegan Paul.

Crozier, Michael, et. al. <u>The Crisis of Democracy</u>. Report on the 1975 Governability of Democracies to the Trilateral Commission. New York: New York University Press.

Curtis, James E., and William G. Scott, eds. <u>Social Stratification</u>: 1973 <u>Canada</u>. Scarborough, Ontario: Prentice-Hall of Canada, Ltd. :

 Davidson, Roger H., and Sar A. Levitan. <u>Antipoverty Housekeeping:</u>
 1968 <u>The Administration of the Economic Opportunity Act</u>. Ann Arbor, Mich.: Institute of Labor and Industrial Relations, University of Michigan.

Davis, J. Ronnie, and John F. Morall III. <u>Evaluating Educational</u> 1974 <u>Investment</u>. Lexington, Mass.: Lexington Books.

Denison, Edward Fulton. Accounting for United States Economic 1974 Growth 1929-1969. Washington, D.C.: The Brookings Institution.

1962 The Sources of Economic Growth in the United States and 1962 the Alternatives Before Us. (A Supplementary Paper of the Committee for Economic Development.) New York: Committee for Economic Development.

------. Why Growth Rates Differ. Postwar Experience in Nine 1967 <u>Western Countries</u>. Washington, D.C.: The Brookings Institution.

Dennison, John D., et. al. <u>The Impact of Community Colleges: A Study</u> 1975 <u>of the College Concept in B.C.</u> Vancouver, B.C.: Vancouver City College, Langara Campus.

Dennison, John D., and Alex Tunner. <u>Socio-economic Survey: Students</u> 1972 <u>Entering Post-Secondary Education in B.C., Fall, 1971;</u> Tabulation of Responses. Vancouver, B.C.: B.C. Research.

Denton, Frank T., and Byron G. Spencer. <u>Does Canada Face a Labor</u> 1976 <u>Shortage in the 1980's?</u> Hamilton, Ontario: Department of Economics, McMaster University.

Doeringer, Peter B., and Michael J. Piore. <u>Internal Labor Markets and</u> 1971 <u>Manpower Analysis</u>. Lexington, Mass.: D.C. Heath.

Dowd, Douglas Fitzgerald. <u>The Twisted Dream: Capitalist Development</u> 1974 <u>in the United States Since 1776</u>. Cambridge, Mass.: Winthrop. Dreeban, Robert. <u>On What Is Learned In Schools</u>. Reading, Mass.: 1968 Addison-Wesley. Dupre, Joseph Stefan, et. al. <u>Federalism and Policy Development:</u> 1973 <u>The Case of Occupational Training in Ontario</u>. Toronto: University of Toronto Press.

Durkheim, Emile. The Evolution of Educational Thought. Lectures 1977 on the Formation and Development of Secondary Education in France. (Peter Collins, translator.) London: Routledge and Kegan Paul.

Dyck, Harold J. <u>Social Futures, Alberta 1970-2005</u>. Edmonton: 1970 Human Resources Research Council.

Eckaus, Richard S. Estimating the Returns to Education: A Disaggregated 1973 Approach. A Technical Report sponsored by the Carnegie Commission on Higher Education. New York: McGraw-Hill Book Co.

Economic Council of Canada, Canadian Higher Education in the Seventies. 1972 Ottawa: Information Canada.

------. Eighth Annual Report, September 1971: Design for Decision-1971a Making. An Application to Human Resources Policies. Ottawa: Information Canada.

1968 September 1968. Ottawa: Queen's Printer.

-----. <u>Report on Intellectual and Industrial Property</u>. Ottawa: 1971b Information Canada.

Edwards, Richard C. <u>Contested Terrain</u>. The Transformation of the 1979 Workplace in the Twentieth Century. New York: Basic Books.

Edwards, Richard C., et. al., eds. <u>The Capitalist State</u>. <u>A Radical</u> 1978 <u>Analysis of American Society</u>. 2nd ed. Englewood Cliffs, New Jersey: Prentice-Hall.

Edwards, Richard L. Breaking the Poverty Cycle: Readings on Income 1972 Maintenance. New York: MSS Information Corp.

Everett, Robinson Ó., ed. <u>Anti-Poverty Programs</u>. Dobbs Ferry, New 1966 York: Oceana Pubs.

Evetts, Julia. The Sociology of Educational Ideas. London: Routledge 1973 and Kegan Paul.

Fagerlind, Ingemar. Formal Education and Adult Earnings. Stockholm, 1975 Sweden: Almqvist and Wiksell.

Feagin, Joe R. Subordinating the Poor: Welfare and American Beliefs. 1975 Englewood Cliffs, New Jersey: Prentice-Hall.

Feinberg, Walter, ed. Equality and Social Policy. Urbana, Ill.: 1978 University of Illinois Press.

Ð

Field, Frank. Unequal Britain: A Report on the Cycle of Inequality. 1974 London: Arrow Books.

Finn, Jr., Cheater. Scholars, Dollars, and Bureaucrats. Washington, 1978 D.C.: The Brookings Institution.

Finnigan, Bryan, and Cy Gonick. <u>Making It: The Canadian Dream</u>. 1972 Toronto: McClelland and Stewart.

Firestone, O.J. Industry and Education: A Century of Canadian Develop-1969 ment. Ottawa: University of Ottawa Press.

Fisher, Irving. The Nature of Capital and Income. New York: Macmillan. 1906

Fishman, Leo, ed. <u>Poverty Amid Affluence</u>. New Haven, Conn.: Yale 1966 University Press.

Fleming, William Gerald. Educational Opportunity, The Pursuit of 1974 Equality. Scarborough, Ontario: Prentice-Mall of Canada.

Floud, Jean E. Social Class and Educational Opportunity. London: 1956 William Heinemann, Ltd.

Fotheringham, John Brooks, ed. <u>Transition: Policies for Social Action</u>. 1966 Toronto: McClelland and Stewart.

Frank, Eric. Human Resource Development: The European Approach. 1974 Houston, Texas: Gulf Pub.

Friedenberg, Edgar Zodiag. Coming of Age in America: Growth and 1965 Acquiescence. New York: Vintage Books.

1975 Garden City, New York: Doubleday.

Friedman, Milton. <u>Capitalism and Freedom</u>. Chicago: University of 1962 Chicago Press.

Froomkin, Joseph. <u>Needed: A New Federal Policy for Higher Education</u>. 1978 Washington, D.C.: Institute for Educational Leadership, The George Washington University.

Fry, John Allan, ed. <u>Economy, Class, and Social Reality: Issues in</u> 1979 <u>Contemporary Canadian Society</u>. Toronto: Butterworths.

Fuchs, Victor R. <u>Productivity Trends in the Goods and Service Sectors</u>, 1964 <u>1929-1961</u>. <u>A Preliminary Survey</u> (Occasional Paper 89.) New York: National Bureau of Economic Research.

Galbraith, John Kenneth. The New Industrial State. New York: Signet. 1968 Books.

Garfinkel, Irwin, and Robert H. Haveman. Earnings Capacity, Poverty, 1977 and Inequality. New York: Academic Press.

Gartner, Alan, et. al., eds. After Deschooling, What? New York: 1973 Harper and Row.

1974 Stratification. New York: Harper and Row.

-----, eds. What Nixon Is Doing To Us. New York: Harper and Row. 1973

Gayfer, Margaret. An Overview of Canadian Education. 2nd ed. 1978 Toronto: The Canadian Education Association.

Giddens, Anthony. The Class Structure of the Advanced Societies. 1973 London: Hutchinson and Co., Ltd.

Gilmour, James M., and John N.H. Britton. <u>The Weakest Link: A</u> 1978 <u>Technological Perspective on Canadian Industrial Underdevelopment</u>. Ottawa: Science Council of Canada.

Ginzberg, Eli. <u>Good Jobs, Bad Jobs, No Jobs</u>. Cambridge, Mass.: 1979 Harvard University Press.

Glass, David. Social Mobility in Britain. London: Routledge and 1954 Kegan Paul.

Glenday, Daniel, et. al., eds. Modernization and the Canadian State. 1978 Toronto: Macmillan of Canada.

Goldman, Barbara. <u>New Directions for Manpower Policy</u>. Ottawa?: 1976 C.D. Howe Research Institute.

Goldthorpe, John H. <u>Social Mobility and Class Structure in Modern</u> 1980 Britain. Oxford: Clarendon Press.

Gonick, Cy. Inflation or Depression: An Analysis of the Continuing 1975 Crisis of the Canadian Economy. Toronto: James Lorimer and Co.

------. Out of Work. Why There's So Much Unemployment, and Why 1978 It's Getting Worse. Toronto: James Lorimer and Co.

Gordon, David M. <u>Economic Theories of Poverty and Underemployment:</u> 1972 Orthodox, Radical, and Dual Labor Market Perspectives. Lexington, Mass.: Lexington Books.

-----, ed. Problems in Political Economy: An Urban Perspective. 1977 2nd ed. Lexington, Mass.: Lexington Books.

Gordon, Margaret S., ed. <u>Higher Education and the Labor Market</u>. 1974 New York: McGraw-Hill Book Co.

Gouldner, Alvin W. The Coming Crisis of Western Sociology. New 1970 York: Avon Books.

Gross, Ronald. The Life-Long Learner. New York: Simon and Schuster. 1977

Guthrie, James W., et. al. <u>Schools and Inequality</u>. Cambridge, Mass.: 1971 The M.I.T. Press.

 Hake, Barry J. Education and Social Emancipation. Some Implications
 1975 for General Secondary Education Towards the Year 2000. The Hague: Martinus Nijhoff.

Hall, Oswald, and Bruce McFarland. <u>Transition From School To Work</u>.
1963 (Research Program on the Training of Skilled Manpower, Report No. 10. Government of Canada. Department of Labor.) Ottawa: Queen's Printer.

Halsey, Albert Henry, et. al. Origins and Destinations. Family, 1980 Class, and Education in Modern Britain. Oxford: Clarendon Press.

Halsey, Albert Henry, et. al., eds. Education, Economy, and Society; 1961 A Reader in the Sociology of Education. New York: The Free Press.

 Handlin, Oscar, and Mary F. Handlin. <u>The American College and American</u>
 <u>1970</u> <u>Culture: Socialization as a Function of Higher Education</u>. A Study sponsored by the Carnegie Commission on Higher Education.

New York: McGraw-Hill Book Co.

Hansen, W. Lee, ed. <u>Education, Income, and Human Capital. Studies</u> 1970 <u>in Income and Wealth</u>. New York: National Bureau of Economic Research.

Hansen, W. Lee, and Burton A. Weisbrod. <u>Benefits</u>, Costs, and Finance 1969 of Higher Education. Chicago: Markham.

Harbison, Frederick. <u>Educational Planning and Human Resource Develop-</u> 1967 <u>ment</u>. Paris: UNESCO International Institute for Educational Planning.

Harbison, Frederick, and Ibrahim Abdelkader Ibrahim. <u>Human Resources</u> 1958 for Egyptian Enterprise. New York: McGraw-Hill Book Co

Harbison, Frederick, and Charles A. Myers. Education, Manpower, and
 1964 Economic Growth; Strategies of Human Resource Development.
 New York: McGraw-Hill Book Co.

Harcourt, Geoffrey Colin. Some Cambridge Controversies in the Theory 1972 of Capital. Cambridge, England: Cambridge University Press.

Harcourt, Geoffrey Colin, and N.F. Laing, eds. <u>Capital and Growth</u>. 1972 Harmondsworth, England: Penguin Books. Harrington, Michael. <u>Decade of Decision; The Crisis of the American</u> 1980 System. New York: Simon and Schuster.

-----. The Other America. Poverty in the United States. Baltimore, 1962 Maryland: Penguin Books.

Harris, Robin S., ed. <u>Changing Patterns of Higher Education in Canada</u>. 1966 Toronto: University of Toronto Press.

Harris, Seymour Edwin. <u>The Market for College Graduates and Some</u> 1969 <u>Related Aspects of Education and Income</u>. 2nd ed. Westport, Conn.: Greenwood Press.

Harris, Seymour Edwin, and Alan Levensohn, eds. Education and Public 1965 Policy. Berkeley, Cal.: McCutchan.

Hartman, Robert W. Credit for College: Public Policy for Student Loans.
1971 A Study sponsored by the Carnegie Commission on Higher Education. New York: McGraw-Hill Book Co.

Haveman, Robert H., ed. A Decade of Federal Antipoverty Programs: 1977 Achievements, Failures, and Lessons. New York: Academic Press.

Havighurst, Robert James, and Daniel U. Levine. Society and Education. 1979 5th ed. Boston: Allyn and Bacon.

Heap, James L., ed. Everybody's Canada: The Vertical Mosaic Reviewed 1974 and Re-examined. Toronto: Burns and MacEachen.

Henry, Nelson B., ed. Social Forces Influencing American Education.
1961 The Sixtieth Yearbook of the National Society for the Study of Education. Chicago: University of Chicago Press.

Hess, Robert David, and Judith V. Torney. <u>The Development of Political</u> 1967 Attitudes in Children. Chicago: Aldine Pubs.

Hettich, Walter. Expenditures, Output, and Productivity in Canadian 1971a University Education. (Special Study No. 14 prepared for the Economic Council of Canada.) Ottawa: Information Canada.

1971b and Efficiency Criteria in Benefit-Cost Analysis. (Special Study No. 19 prepared for the Economic Council of Canada.) Ottawa: Information Canada.

Hodgetts, J.E., ed.⁷ <u>Higher Education in a Changing Canada</u>. (Symposium 1965 presented to the Royal Society of Canada.) Toronto: University of Toronto Press.

Hodgson, Ernest D. <u>Federal Intervention in Education</u>. Toronto: 1976 Canadian Education Association.

Holland, John W., and Michael L. Skolnik. Public Policy and Manpower 1975 Development. Toronto: Ontario Institute for Studies in Education. Holly, Douglas, ed. Education or Domination? A Critical Look at 1974 Educational Problems Today. London: Arrow. Holmes, Richard A. Economic Returns to Education in Canada. Ottawa: Information Canada. 1974 Hughes, John F., and Anne O. Hughes. Equal Education: A New National Strategy. Bloomington, Ind.: Indiana University Press. 1972 Hurn, Christopher J. The Limits and Possibilities of Schooling; An 1978 Introduction to the Sociology of Education. Boston: Allyn and Bacon. Husén, Torsten. The Learning Society. London: Methuen and Co., Ltd. 1974a -----. -Present Trends and Future Developments in Education: A 1971 European Perspective. Toronto: Ontario Institute for Studies in Education. The School in Question. A Comparative Study of the School and Its Future in Western Society. New York: Oxford University 1979 Press. Social Background and Educational Career; Research Perspectives on Equality of Educational Opportunity. Paris: Centre for 1972 Educational Research and Innovation, OECD. ------. Social Influences on Educational Attainment; Research Perspectives on Educational Equality. Paris: Centre for 1975 Educational Research and Innovation, OECD. Talent, Equality, and Meritocracy: The Availability and 1974b Utilisation of Talent. The Hague: Martinus Nijhoff. Husen, Torsten, and Gunnar Boalt. Educational Research and Educational 1967 Change: The Case of Sweden. New York: John Wiley and Sons. Illich, Ivan. Deschooling Society. New York: Harper and Row. 1971 Illich, Ivan, et. al. Disabling Professions. London: Marion Boyars. 1977 Illing, Wolfgang M., and Zoltan E. Zsigmond. Enrolment in Schools and Universities 1951-52 to 1975-76. (Staff Study No. 20 1967 prepared for the Economic Council of Canada.) Ottawa: Queen's Printer.

3

International Association of Universities. Some Economic Aspects of 1960 Educational Development in Europe. (A Provisional Report of a Conference at the Villa Serbelloni, Bellagio, Italy, 1960.) Paris: International Universities Bureau.

Jackson, R.W., et. al. <u>Background Studies in Science Policy:</u> 1969 <u>Projections of Research and Development Manpower and</u> <u>Expenditures.</u> (Special Study No. 6, Science Council of Canada.) Ottawa: Queen's Printer...

Jallade, Jean Pierre. Occupational and Educational Structures of the 1970 Labour Force and Levels of Economic Development; Possibilities and Limitations of an International Comparison Approach. Paris: OECD.

James, Dorothy Buckton, ed. <u>Analyzing Poverty Policy</u>. Lexington, 1975 Mass.: Lexington Books.

Jencks, Christopher, and David Riesman. <u>The Academic Revolution</u>. 1968 New York: Doubleday.

Jencks, Christopher, et. al. <u>Inequality: A Reassessment of the Effect</u> 1972 of Family and Schooling in America. New York: Basic Books.

Jencks, Christopher, et. al. <u>Who Gets Ahead? The Determinants of</u> 1979 <u>Economic Success in America</u>. New York: Basic Books.

Johns, Roe Lyell, et. al., eds. <u>Economic Factors Affecting the</u> 1970 <u>Financing of Education</u>. Vol. 2. Gainesville, Florida: National Education Finance Project.

Johnson, F. Henry. <u>A Brief History of Canadian Education</u>. Toronto: 1968 McGraw-Hill Book Co. of Canada.

Johnson, Leo A. Incomes, Disparity, and Impoverishment in Canada 1973 Since World War II. Toronto: New Hogtown Press.

------ Poverty in Wealth: The Capitalist Labour Market and Income 1977 Distribution in Canada. 2nd, rev. ed. Toronto: New Hogtown Press.

Jones, Kathleen, ed. <u>The Yearbook of Social Policy in Britain, 1972</u>. 1973 London: Routledge and Kegan Paul.

-----, ed. <u>The Yearbook of Social Policy in Britain, 1973</u>. 1974 London: Routledge and Kegan Paul.

Jones, Kathleen, and Sally Baldwin, eds. <u>The Yearbook of Social Policy</u> 1976 in Britain, <u>1975</u>. London: Routledge and Kegan Paul.

Jones, Kathleen, et. al. Issues in Social Policy. London: Routledge 1978 and Kegan Paul.

Jones, Kathleen, et. al., eds. <u>The Yearbook of Social Policy in</u> 1977 Britain, 1976. London: Routledge and Kegan Paul.

Jouvenel, Bertrand de. Ethics of Redistribution. Cambridge, England: 1951 Cambridge University Press.

Juster, F. Thomas, ed. <u>Education</u>, <u>Income</u>, and <u>Human Behavior</u>. 1975 A report prepared for the Carnegie Commission on Higher Education and the National Bureau of Economic Research. New York: McGraw-Hill Book Co.

Kaldor, Mary. <u>The Disintegrating West</u>. London: A. Lane, Ltd. 1978

Kanter, Rosabeth Moss. <u>Men and Women of the Corporations</u>. New York: 1977 Basic Books.

Kanter, Rosabeth Moss, and Barry A. Stein, eds. Life In Organizations: 1979 Workplaces As People Experience Them. New York: Basic Books.

Karabel, Jerome, and Albert Henry Halsey, eds. Power and Ideology in 1977 Education. New York: Oxford University Press.

Katz, Joseph. Education in Canada. Vancouver, B.C.: Douglas, David, 1974 and Charles.

Katz, Michael. <u>The Irony of Early School Reform</u>. Cambridge, Mass.: 1968 Harvard University Press.

Keats, John. <u>Schools Without Scholars</u>. Boston: Houghton, Mifflin. 1958

------. <u>The Sheepskin Psychosis</u>. Philadelphia, Pa.: Lippincott. 1965

Kelly, Frank. Prospects for Scientists and Engineers in Canada.
1971 (Special Study No. 20, Background Study for the Science Council of Canada.) Ottawa: Information Canada.

Kerr, Clark, et. al. <u>Industrialism and Industrial Man</u>. New York: 1964 Oxford University Press.

Kiker, B.F. <u>Human Capital: In Retrospect</u>. Columbia, South Carolina: 1968 University of South Carolina Press.

1971 University of South Carolina Press.

King, Timothy, ed. <u>Education and Income</u>. (World Bank Staff Working 1980 Paper No. 402.) Washington, D.C.: World Bank.

Kolko, Gabriel. <u>Wealth and Power in America; An Analysis of Social</u> 1962 Class and Income Distribution. New York: Praeger.
Kolko, Joyce. <u>America and the Crisis of World Capitalism</u>. Boston: 1974 Beacon Press.

Kozol, Jonathan. Death At An Early Age; The Destruction of the Hearts
 and Minds of Negro Children in the Boston Public Schools.
 Boston: Houghton, Mifflin.

-----. The Night Is Dark and I Am Far From Home. Boston: 1975 Houghton, Mifflin.

Kruger, Arthur M., and Noah H. Meltz, eds. <u>The Canadian Labour</u> 1968 <u>Market: Readings in Manpower Economics</u>. Toronto: Centre for Industrial Relations, University of Toronto.

Lagace, Michael D. Educational Attainment in Canada: Some Regional 1968 and Social Aspects. (Dominion Bureau of Statistics Special Labour Force Studies, No. 7.) Ottawa: Queen's Printer.

 Lamontagne, Maurice (Chairman). A Science Policy for Canada.
 1970 Report of the Senate Special Committee on Science Policy.
 Vol. 1: A Critical Review: Past and Present. Ottawa: Information Canada.

1972 Committee on Science Policy. Vol. 2: <u>Targets and Strategies</u> for the Seventies. Ottawa: Information Canada.

1973 Committee on Science Policy. Vol. 3: <u>A Government Organization</u> for the Seventies. Ottawa: Information Canada.

Lampman, Robert J. Ends and Means of Reducing Income Poverty. 1971 Chicago: Markham.

1962 <u>1956</u>. Princeton, New Jersey: Princeton University Press.

Lasch, Christopher. The Culture of Narcissism: American Life in an 1978 Age of Diminishing Expectations. New York: W.W. Norton.

Lawr, Douglas, and Robert Gidney, eds. <u>Educating Canadians. A</u> 1973 <u>Documentary History of Public Education</u>. Toronto: Van Nostrand, Reinhold, Ltd.

Levine, Daniel U., and Robert James Havighurst, eds. Farewell to 1971 <u>Schools??</u>? Worthington, Ohio: Charles A. Jones Pub.

Leys, Colin. Underdevelopment in Kenya: The Political Economy of 1975 Neo-Colonialism. London: Heinemann.

Lind, Lorn J. The Learning Machine: A Hard Look At Toronto Schools. 1974 Toronto: Anasi. Lindberg, Leon N., et. al. <u>Stress and Contradiction in Modern</u> 1975 <u>Capitalism; Public Policy and the Theory of the State</u>. Lexington, Mass.: Lexington Books.

Lipset, Seymour Martin, and Reinhard Bendix. <u>Social Mobility in</u> 1959 <u>Industrial Society</u>. Berkeley, Cal.: University of California Press.

Lithwick, Noah H. <u>Canada's Science Policy and the Economy</u>. 1969 Toronto: Metheun.

Lloyd, Cynthia B., ed. <u>Sex, Discrimination, and the Division of</u> 1975 <u>Labor</u>. New York: Columbia University Press.

Lloyd, Cynthia B., et. al., eds. <u>Women in the Labor Market</u>. New 1979 York: Columbia University Press.

Lloyd-Jones, Esther McDonald, and Norah Rosenau, eds. Social and 1968 <u>Cultural Foundations of Guidance</u>. New York: Holt, Rinehart, and Winston.

Longstreet, Wilma S. <u>Beyond Jencks: The Myth of Equal Schooling</u>. 1973 Washington, D.C.: Association for Supervision and Curriculum Development.

Macdonald, John B., et. al. The Role of the Federal Government in
 1969 Support of Research in Canadian Universities. (Science Council of Canada and Canada Council Special Study No. 7.) Ottawa:
 Queen's Printer.

Machlup, Fritz. <u>Education and Economic Growth</u>. 2nd ed. Lincoln, 1975 Nebraska: University of Nebraska Press.

McLeod, T.H., ed. <u>Post-Secondary Education in a Technological Society</u>. 1973 Montreal: McGill-Queen's University Press.

Mackenzie, Gavin. <u>The Aristocracy of Labor. The Position of Skilled</u> 1973 <u>Craftsmen in the American Class Structure</u>. Cambridge, England: Cambridge University Press.

Magdoff, Harry, and Paul M. Sweezy. <u>The Dynamics of U.S. Capitalism</u>. 1972 New York: Monthly Review Press.

Marshall, Alfred. Principles of Economics. Vols. 1 and 2. Nineth 1961 (varorium) ed. (C.W. Guillebaud, ed.) London: Macmillan.

Martell, George, ed. <u>The Politics of the Canadian Public School</u>. 1974 Toronto: James Lewis and Samuel.

Marx, Karl. Value, Price, and Profit. New York: International 1935 Pubs.

1933

--. Wage-Labor and Capital. New York: International Pubs.

Maslove, Allan M. <u>The Pattern of Taxation in Canada</u>. Ottawa: 1972 Economic Council of Canada.

Medsker, Leland L., and Dale Tillery. <u>Breaking the Access Barriers</u>. 1971 <u>A Profile of Two-Year Colleges</u>. New York: McGraw-Hill Book Co.

Meltz, Noah M. <u>Changes in the Occupational Composition of the</u> 1965 <u>Canadian Labour Force, 1931-1961</u>. Ottawa: Department of Labour, Government of Canada.

Michigan State University Placement Services. <u>Recruiting Trends</u> 1974 <u>Survey 1974-75</u>. East Lansing, Mich.: Michigan State University Placement Services.

Miles, Michael W. <u>The Radical Probe: The Logic of Student Rebellion</u>. 1971 New York: Antheneum.

Miller, LaMar P., and Edmund W. Gordon, eds. Equal of Educational 1974 Opportunity. A Handbook for Research. New York: AMS Press.

Miller, Seymour Michael, and Frank Riessman. Social Class and Social 1968 Policy. New York: Basic Books.

Miller, Seymour Michael, and Pamela A. Roby. <u>The Future of Inequality</u>. 1970 New York: Basic Books.

Mills, C. Wright. <u>The Power Elite</u>. New York: Oxford University Press. 1956

Milner, Jr., Murray. <u>The Illusion of Equality</u>. San Francisco, Cal.: 1972 Jossey-Bass.

Mincer, Jacob. <u>Schooling, Experience, and Earnings</u>. New York: 1974 National Bureau of Economic Research.

Molgat, Paul, and J.A. MacMillan. <u>Education in Area Economic Development</u>. 1972 Winnipeg, Man.: Center for Settlement Studies, University of Manitoba.

Montague, J. Tait. Labour Markets in Canada: Processes and Institutions. 1970 Scarborough, Ont.: Prentice-Hall of Canada.

Moore, Jr., Barrington. <u>Political Power and Social Theory: Seven</u> -1965 <u>Studies</u>. New York: Harper and Row.

----- Reflections on the Causes of Human Misery and Upon Certain 1972 Proposals to Eliminate Them. Boston: Beacon Press.

Moynihan, Daniel Patrick. <u>Maximum Feasible Misunderstanding; Community</u> 1969 Action in the War on Poverty. New York: The Free Press. 1973 House. The Politics of a Guaranteed Income. New York: Random

Murphy, Raymond. <u>Sociological Theories of Education</u>. Toronto: 1979 McGraw-Hill Ryerson, Ltd.

National Academy of Education. Economic Dimensions of Education.
1979 A Report of a Committee of the National Academy of Education. (Douglas M. Windham, ed.) Washington, D.C.: National Academy of Education.

 National Council of Welfare. <u>Bearing the Burden, Sharing the Benefits</u>.
 1978 A Report by the National Welfare Council on Taxation and the Distribution of Income. Ottawa: National Council of Welfare, March.

------. <u>The Hidden Welfare System</u>. A Report by the National 1976 Council of Welfare on the Personal Income Tax System in Canada. Ottawa: National Welfare Council, Nov.

------. Jobs and Poverty. A Report by the National Council of 1977a Welfare on Canada's Working Poor. Ottawa: National Council of Welfare, June.

------ Poor Kids. A Report by the National Council of Welfare 1975 on Children in Poverty in Canada. Ottawa: National Council of Welfare, March.

1979 Welfare. Ottawa: National Council of Welfare, Oct.

------. <u>The Working Poor</u>. A Statistical Profile prepared by 1977b the National Council of Welfare. Ottawa: National Council of Welfare, June.

------. The Working Poor: People and Programs. A Statistical 1981 Profile prepared by the National Council of Welfare. Ottawa: National Council of Welfare.

Nelsen, Randle W., and David A. Nock, eds. <u>Reading, Writing, and</u> 1978 <u>Riches: Education and the Socio-Economic Order in North</u> <u>America. Kitchener, Ont.: Between the Lines Pubs.</u>

Nisbet, Robert A. <u>Twilight of Authority</u>. New York: Oxford University 1975 Press.

Norman, Victor D. <u>Education</u>, <u>Learning</u>, and <u>Productivity</u>. Bergen, 1976 Norway: Universitetsforlaget.

Officer, Lawrence H., and Lawrence B. Smith, eds. Canadian Economic 1970 Problems and Policies. Toronto: McGraw-Hill Co. of Canada.

----, eds. Issues in Canadian Economics. Toronto: McGraw-Hill

1974 Ryerson, Ltd.

Okun, Arthur M. <u>Equality and Efficiency</u>. The Big Tradeoff. Washington, 1975 D.C.: The Brookings Institution.

----- The Political Economy of Prosperity. Washington, D.C.: 1970 The Brookings Institution.

Okun, Arthur M., et. al. <u>Inflation: The Problems It Creates and the</u> 1970 Policies It Requires. New York: New York University Press.

Okun, Arthur M., and George L. Perry, eds. <u>Brookings Papers on</u> 1978 <u>Economic Activity</u>. Vol. 3: <u>Curing Chronic Inflation</u>. Washington, D.C.: The Brookings Institution.

Ontario. Commission on Post-Secondary Education in Ontario. 1972a <u>Cost-Benefit Study of Post-Secondary Education in Ontario,</u> School Year 1968-69, Vol. 1. Toronto: Queen's Printer.

------. Financing Post-Secondary Education. Toronto: Queen's 1972c Printer.

------. <u>The Learning Society</u>. Final Report of the Wright Commission 1972d on Post-Secondary Education in Ontario. Toronto: Ministry of Government Services.

------. The Production of Scientific Knowledge in Ontario Univer-1972e sities: An Overview of Problems. Toronto: Queen's Printer.

Ontario. Provincial Committee on Aims and Objectives of Education in 1968 the Schools of Ontario. <u>Living and Learning</u>. Report of the Provincial Committee on Aims and Objectives of Education in the Schools of Ontario. Toronto: Newton.

Organization for Economic Co-Operation and Development. Ability and 1961 <u>Educational Opportunity</u>. Report on the Conference organized by the Office for Scientific and Technical Personnel, in collaboration with the Swedish Ministry of Education, in Kungalv, Sweden, June 1961. (Albert Henry Halsey, ed. and rapporteur general.) Paris: OECD.

-------. <u>Classification of Educational Systems: Canada, Greece,</u> 1973a <u>Yugoslavia</u>. Paris: OECD.

------. Development of Higher Education 1950-1967: A Statistical 1970a Survey. Paris: OECD.

-----. Education and Working Life. Paris: OECD.

1977a

	Education and Working Life in Modern Societies. Paris:
1975 a	OECD.
	Education in OECD Developing Countries. Paris: OECD.
1974a	
	<u>Education, Inequality, and Life Chances</u> . Vols. 1 and 2. Paris: OECD.
	Educational Policies and Trends in the Context of Social and Economic Development Perspectives: A Report. Paris:
17770	OECD.
	Educational Policies for the 1970's. (General Report, Conference on Policies for Educational Growth, Paris, June
,	1970.) Paris: OECD.
یر روند خون مند مند ا	The Educational Situation in OECD Countries. Paris:
1974Ъ	OE CD.
1975c	External Examiners' Report on Educational Policy in Canada. Toronto: University of Toronto Students' Administrative Council
	and Canadian Association for Adult Education.
	Financing of Education for Economic Growth. (Lucille
1964	Reifman, ed.) Paris: OECD.
	'Labour Force Statistics 1960-1971. Paris: OECD.
1973Ъ	
	Long-Range Policy Planning in Education. Paris: OECD:
1973c	. Long wange folicy flamming in Education. Fails: OECD.
	Policy Conference on Economic Growth and Investment in Education. Paris: OECD.
- 0	
1967	Policy Conference on Highly Qualified Manpower. Paris: OECD.
	Study Group in the Economics of Education. <u>The Residual</u> Factor and Economic Growth. Paris: OECD.
1,2040	ractor and heomonal orowin. Targa, obop.
1976	<u>Reviews of National Policies for Education: Canada</u> . Paris: OECD.
19/0	raris; <u>OECU</u> .
	Reviews of National Science Policy: Canada. Paris:
1969	OECD.
1977c	Paris: OECD.
1974 c	(Report on the Conference on Future Structures of Post-Secondary

يدر هده

.....

,

.

.....

• •

~

٠,

Education, June 1973.) Paris: OECD.

-----. 'Training of and Demand for High-Level Scientific and 1966 Technical Personnel in Canada. Paris: OECD.

Orwig, M.D., ed. <u>Financing Higher Education: Alternatives for the</u> 1971 <u>Federal Government</u>. Iowa City, Iowa: American College Testing Program.

Ossenberg, Richard J. <u>Canadian Society: Pluralism, Change, and</u> 1971 Conflict. Scarborough, Ont.: Prentice-Hall of Canada.

-----, ed. <u>Power and Change in Canada</u>. Toronto: McClelland and 1980 Stewart.

Ostry, Sylvia, ed. <u>Canadian Higher Education in the Seventies</u>. 1972 (Abridged papers presented at the Economic Council of Canada Seminar at Montebello, Quebec, Oct. 1971.) Ottawa: Information Canada.

1967 Force. Ottawa: Dominion Bureau of Statistics.

Ostry, Sylvia, and Mahmood A. Zaidi. <u>Labour Policy and Labour Economics</u> 1972 in Canada. Vols. 1 and 2. 2nd ed. Toronto: Macmillan of Canada.

Panitch, Leo, ed. <u>The Canadian State: Political Economy and Political</u> 1977 Power. Toronto: University of Toronto Press.

Parai, Louis. <u>Immigration and Emigration of Professional and Skilled</u> 1965 <u>Manpower During the Post-War Period</u>. (Special Study No. 1, Economic Council of Canada.) Ottawa: Queen's Printer.

Parkin, Frank. <u>Class, Inequality, and Political Order. Social</u>
1972 <u>Stratification in Capitalist and Communist Societies</u>.
St. Albans, England: Paladin.

-----, ed. The Social Analysis of Class Structure. London: 1974 Tavistock.

Parsons, Talcott. Essays in Sociological Theory. 2nd, rev. ed. 1954 Glencoe, Ill.: The Free Press.

Peters, Lawrence J., and Raymond Hull. The Peter Principle. New 1969 York: William Morrow and Co.

Pike, Robert M. Who Doesn't Get to University — and Why; A Study 1970 on Accessibility to Higher Education in Canada. Ottawa: Association of Universities and Colleges of Canada.

Piven, Francis Fox, and Richard A. Cloward. <u>Regulating the Poor:</u> 1971 The Functions of Public Welfare. New York: Pantheon Books. Plotnick, Robert D., and Felicity Skidmore. Progress Against Poverty. 1975 <u>A Review of the 1964-1974 Decade</u>. New York: Academic Press.

Podoluk, Jenny R. Earnings and Education. Ottawa: Queen's Printer. 1965

Ponsioen, J.A. <u>National Development</u>. A Sociological Contribution. 1968 The Hague: Mouton.

Porter, John Arthur. The Measure of Canadian Society: Education, 1979 Equality, and Opportunity. Toronto: Gage.

1965 Power in Canada. Toronto: University of Toronto Press.

Porter, Marion R., et. al. Does Money Matter? Prospects for Higher 1973 Education. Toronto: Institute for Behavioral Research, York University.

Postner, Harry H. An Analysis of Canadian Manufacturing Productivity; 1971 <u>Some Preliminary Results</u>. (Staff Study No. 31, Economic Council of Canada.) Ottawa: Information Canada.

Poulantzas, Nicos A. <u>Classes in Contemporary Capitalism</u>. London: 1978a Verso.

1978b Political Power and Social Classes. London: Verso.

Prewitt, Kenneth, and Alan Stone. The Roling Elites; Elite Theory, 1973 Power, and American Democracy. New York: Harper and Row.

Psacharopoulos, George. Earnings and Education in OECD Countries. 1975. Paris: OECD.

1913 (Assissted by Keith Hinchliffe.) Amsterdam: Elsevier Scientific Pub.

...

Purves, Alan C., and Daniel U. Levine, eds. <u>Educational Policy and</u> 1975 <u>International Assessment; Implications of the IEA Surveys of</u> <u>Achievement.</u> Berkeley, Cal.: McCutchan.

Purves, Douglas D. Human Capital and the Financial Portfolio. 1975 Kingston, Ont.: Institute for Economic Research.

Rabinovitch, Robert. An Analysis of the Canadian Post-Secondary 1966 Student Population. Part 1: A Report on Canadian Undergraduate Students. Ottawa: Canadian Union of Students.

Radner, Roy, and Leonard Miller. <u>Demand and Supply in U.S. Higher</u> 1975 <u>Education</u>. (With the collaboration of Douglas L. Adkins and Frederick E. Balderston.) New York: McGraw-Hill Book Co.

á ·

C

0

.

1

Raskin, Marcus G. <u>Being and Doing</u>. New York: Random House. 1971

------. <u>Notes on the Old System: To Transform American Politics</u>. 1974 New York: David McKay.

Rea, K.J. <u>The Political Economy of Northern Development</u>. (Science 1976 Council of Canada Background Study No. 36.) Ottawa: Information Canada.

Rea, Jr., Samuel A. <u>Investment in Human Capital and Income Maintenance</u> 1974 <u>Programs</u>. Toronto: Institute for the Quantitative Analysis of Social and Economic Policy, University of Toronto.

Rehberg, Richard A., and Evelyn R. Rosenthal. Class and Merit in the 1978 American High School; An Assessment of the Revisionist and Meritocratic Arguments. New York: Longman.

Reich, Michael, et. al., eds. Labor Market Segmentation. Lexington, 1975 Mass.: D.C. Heath.

Reuchlin, M. Individual Orientation in Education. The Hague: 1972 Martinus Nijhoff.

Reynolds, Larry T., and Janice M. Reynolds, eds. <u>The Sociology of</u> 1970 <u>Sociology</u>. New York: David McKay.

Ribich, Thomas I. <u>Education and Poverty</u>. Washington, D.C.: The 1968 Brookings Institution.

Richmond, D.R. <u>Poverty and Institutional Reform</u>. Report of the 1969 Ontario Economic Council. Toronto: Ontario Economic Council.

Rickover, Hyman George. American Education, A National Failure; 1963 The Problem of Our Schools and What We Can Learn From England. New York: Dutton.

-----. '<u>Education and Freedom</u>. New York: Dutton. 1959

r

Rogers, Daniel C., and Hirsch S. Ruchlin. <u>Economics and Education:</u> 1971 <u>Principles and Applications</u>. New York: The Free Press.

Roueche, John E.^{*} Salvage, Redirection, or Custody: Remedial Education 1958 in the Community Junior College. Washington, D.C.: American Association of Junior Colleges.

Roussopoulos, Dimitrios I., ed. <u>Canada and Radical Social Change</u>. 1973a Montreal: Black Rose Books.

-----, ed. The Political Economy of the State: Quebec/Canada/U.S.A. - 1973b Montreal: Black Rose Books. 1974 Rose Books. .

Routh, G. Occupation and Pay in Great Britain 1906-1960. Cambridge, 1965 England: Cambridge University Press.

Sarup, Madan. <u>Marxism and Education</u>. London: Routledge and Kegan 1978 Paul.

Schultz, Theodore William. <u>The Economic Value of Education</u>. New 1963 York: Columbia University Press.

1971 <u>Research</u>. New York: The Free Press.

------. <u>Transforming Traditional Agriculture</u>. New Haven, Conn.: 1964 Yale University Press.

-----, ed. Investment in Education. The Equity-Efficiency Quandry. 1972 Chicago: University of Chicago Press.

Schwartz, Jesse G., ed. <u>The Subtle Anatomy of Capitalism: Selected</u> 1977 <u>Readings</u>. Santa Monica, Cal.: Goodyear Pub.

Science Council of Canada. Issues in Canadian Science Policy. A 1972a Commentary on Some Aspects of Vol. 2 of the Report of the Senate Committee on Science Policy. Ottawa: Information Canada.

1972b Information Canada.

Scott, Peter. <u>Strategies for Postsecondary Education</u>. New York: 1975 John Wiley and Sons.

Scoville, James G. The Job Content of the U.S. Economy, 1940-1970. 1969 New York: McGraw-Hill Book Co.

1972 Lexington, Mass.: Lexington Books.

Selowsky, Marcelo. <u>A Note on Preschool-Age Investment in Human Capital</u> 1976 in Developing Countries. Washington, D.C.: World Bern.

Sewell, D.O. <u>Training the Poor. A Cost-Benefit Analysis of Manpower</u> 1971 <u>Programs in the U.S. Antipoverty Program</u>. Kingston, Ont.: Industrial Relations Centre, Queen's University.

Sewell, William Hamilton, and Robert M. Hauser. Education, Occupation, 1975 and Earnings: Achievement in the Early Career. New York: Academic Press.

Sexton, Patricia Cayo. <u>Education and Income: Inequalities in Our</u> 1961 Public Schools. New York: Viking Press. Sharp, Laure Metzger. Education and Employment: The Early Careers 1970 of College Graduates. Baltimore, Maryland: Johns Hopkins Press.

Sheehan, John. <u>The Economics of Education</u>. London: George Allen 1973 and Unwin, Ltd.

Sheffield, Edward, et. al. Systems of Higher Education in Canada. 1978 New York: International Council for Educational Development.

Sheppard, Harold L., and Neal Q. Herrick. Where Have All The Robots 1972 <u>Gone? Worker Dissatisfaction in the '70's</u>. New York: The Free Press.

Shields, James J., and Colin Greer, eds. Foundations of Education: 1974 Dissenting Views. New York: John Wiley and Sons.

Sills, David L., ed. International Encyclopedia of the Social Sciences. 1968 New York: Crowell, Collier, and Macmillan.

Silverman, Bertram, and Murray Yanowitch, eds. <u>The Worker in "Post-</u> 1974 <u>Industrial" Capitalism: Liberal and Radical Responses</u>. New York: The Free Press.

Simmons, John, ed. <u>The Education Dilemna; Policy Issues for Developing</u> 1980 <u>Countries in the 1980's</u>. Oxford: Pergamon Press.

Simmons, John, et. al. <u>Investment in Education: National Strategy</u> 1975 <u>Options for Developing Countries</u>. Washington, D.C.: International Bank for Reconstruction and Development.

Singleton, W.T., and P. Spurgeon, eds. <u>Measurement of Human Resources</u>. 1975 London: Taylor and Francis, Ltd.

Smith, Adam. An Inquiry Into the Nature and Causes of the Wealth of 1965 Nations. Books I-V. (Edwin Cannan, ed.) New York: The Modern Library.

Smith, Cynthia J., ed. <u>Advancing Equality of Opportunity: A Matter</u> 1978 <u>of Justice</u>. Washington, D.C.: Institute for the Study of Educational Policy.

Solmon, Lewis C., and Paul J. Taubman, eds. <u>Does College Matter?</u> 1973 <u>Some Evidence on the Impacts of Higher Education</u>. New York: Academic Press.

Spence, Andrew Michael. <u>Market Signalling: Informational Transfer</u> 1974 <u>in Hiring and Related Screening Processes</u>. Cambridge, Mass.: Harvard University Press.

Squires, Gregory D. Education and Jobs. The Imbalancing of the Social 1979 Machinery. New Brunswick, New Jersey: Transaction Books. 1976 <u>Action</u>. East Lansing, Mich.: Michigan State University Press.

Stager, David A. Some Economic Aspects of Alternative Systems of 1969 Post-Secondary Education. (A paper presented at the Seventh Canadian Conference on Educational Research, Victoria, Jan. 28.) Toronto?: Department of Political Economy, University of Toronto.

Stager, David A., and Alan M. Thomas. <u>Continuing Education in Canada</u>. 1972 Toronto: Ontario Institute for Studies in Education.

Stein, Bruno, and Seymour Michael Miller, eds. Incentives and 1973 Planning in Social Policy. Chicago: Aldine.

Stoikov, Vladimir. <u>The Economics of Recurrent Education and Training</u>. 1975 Geneva: International Labor Office.

Symons, Thomas H.B. The Symons Report. Toronto: McClelland and Stewart. 1978

1975 Studies. Vols. 1 and 2. Ottawa: Association of Universities and Colleges.

Taubman, Paul J., and Terrence Wales. <u>Higher Education and Earnings:</u> 1974 <u>College As An Investment and a Screening Device</u>. New York: McGraw-Hill Book Co.

Tepperman, Lorne. Social Mobility in Canada. Toronto: McGraw-Hill 1975 Ryerson, Ltd.

Thatcher, Richard William. Professionally-Credentialled Workers and 1978 Monopoly Capital: A Marxist Response to Post-Industrial Theory: (Unpublished Ph.D. thesis, Department of Sociology, University of Alberta.)

Thurow, Lester C. <u>Generating Inequality; Mechanisms of Distribution</u> 1975 in the U.S. Economy. New York: Basic Books.

1971 Præger.

1970 Investment in Human Capital. Belmont, Cal.: Wadsworth.

1980 for Economic Change. New York: Basic Books.

Thurow, Lester C., and Robert E.B. Lucas. <u>The American Distribution</u> 1972 of Income: A Structural Problem. (Hearings before the Joint Economic Committee of the Congress of the U.S.) Washington, D.C.: U.S. Government Printing Office. Tinbergen, Jan. <u>Income Distribution</u>. Analysis and Policies. 1975 Amsterdam: North Holland Pub.

United Nations Educational, Scientific, and Cultural Organization. 1970 Educational Planning: A World Survey of Problems and Prospects. Paris: UNESCO.

1972 Report of the International Commission on the Development of Education. (Edgar Faure, Chairman.) Paris: UNESCO.

Upited Nations Research Institute for Social Development. <u>Contents</u> 1972 <u>and Measurement of Socioeconomic Development</u>. New York: Praeger.

United States. Commission on Civil Rights. The Federal Civil Rights 1975 Enforcement Effort - 1974. Vol. 3: To Ensure Equal Educational Opportunity. Washington, D.C.: U.S. Commission on Civil Rights.

-----. Department of Health, Education, and Welfare. <u>Work in</u> 1972? <u>America</u>. Report of a Special Task Force to the Secretary of Health, Education, and Welfare. Cambridge, Mass.: The M.I.T. Press.

------. Department of Health, Education, and Welfare and the Office 1962 of Education. Economics of Higher Education. (Selma J. Mushkin, ed.) Washington, D.C.: U.S. Government Printing Office.

------ Department of Labor. <u>Manpower Report of the President</u>, 1972b 1972. Washington, D.C.: U.S. Government Printing Office.

------. Department of Labor and Department of Health, Education, and 1974a Welfare. <u>Manpower Report of the President, 1974</u>. Washington, - D.C.: U.S. Government Printing Office.

------. Equal Employment Opportunity Commission. Affirmative 1974b Action and Equal Employment: A Guidebook for Employers. Vol. 1. Washington, D.C.: U.S. Equal Employment Opportunity Commission.

------. National Advisory Commission on Rural Poverty. <u>The People</u> 1967 <u>Left Behind</u>. A Report by the President's National Advisory Commission on Rural Povetty. Washington, D.C.: U.S. Government Printing Office.

Vaizey, John: <u>Education in the Modern World</u>. new ed. London: 1975 Weidenfeld and Nicolson.

Vaizey, John, et. al. <u>The Political Economy of Education</u>. London: 1972 Duckworth and Co., Ltd.

Venn, Grant. Man, Education, and Manpower. Washington, D.C.: The 1970 American Association of School Administrators. Visalberghi, Aldo. Education and the Division of Labor; Middle-1973 and Long-Term Perspectives in European Technical and Vocational Education. The Hague: Nijhoff.

von Zur-Muchlen, Max. The New Crisis in the Universities. Ottawa: 1976 Statistics Canada.

-----. <u>Profile of Ph.D.'s in Canada</u>. Ottawa: Department of 1975 the Secretary of State.

Waines, W.J. Federal Support of Universities and Colleges of Canada. 1970 Ottawa: Association of Universities and Colleges of Canada.

Walters, Dorothy. <u>Canadian Growth Revisited</u>, 1950-1967. (Staff 1970 Study No. 28, Economic Council of Canada.) Ottawa: Queen's Printer:

------. <u>Canadian Income Levels and Growth: An International</u> 1968 <u>Perspective</u>. (Staff Study No. 23, Economic Council of Canada.) Ottawa: Queen's Printer.

Ward, F. Champion, ed. <u>Education and Development Reconsidered:</u> 1974 The Bellagio Conference Papers. New York: Praeger.

Watson, Cicely. <u>New College Systems in Canada</u>. Paris: OECD. 1973

Watson, dicely, and Joseph Butorac. <u>Qualified Manpower in Ontario</u>, 1968 1961-1986. Toronto: Ontario Institute for Studies in Education.

Weaver, Janice F., ed. <u>Educational Policy</u>. Highlights of the Lyndon 1975 B. Johnson Memorial Symposium on Educational Policy. Danville, 111.: The Interstate Printers and Publishers.

Weaver, James, ed. <u>Modern Political Economy: Radical Vs. Orthodox</u> 1973 <u>Approaches</u>. Boston: Allyn and Bacon.

West, E.G. <u>Economics of Education and the Politician</u>. London: 1968 Institute of Economic Affairs.

1965 Affairs. Education and the State. London: Institute of Economic

Wexler, Philip. The Sociology of Education: Beyond Equality. 1976 Indianapolis, Ind.: Bobbs-Merrill.

White, Derek A. <u>Business Cycles in Canada</u>. (Staff Study No. 17, 1967 Economic Council of Canada.) Ottawa: Queen's Printer.

Wilkinson, Bruce W. <u>Studies in the Economics of Education</u>. 1966 Ottawa: Queen's Printer. Williams, C. Brian. <u>Manpower Management in Canada. A Selected</u> 1968 <u>Bibliography</u>. Kingston, Ont.: Industrial Relations Centre, Queen's University.

Williams, Peter, ed. <u>Prescription for Progress? A Commentary on the</u> 1976 <u>Educational Policy of the World Bank</u>. London: University of London, Institute of Education.

Williamson, John B., et. al. <u>Strategies Against Poverty in America</u>. 1975 New York: John Wiley and Sons.

Wilson, J. Donald, et. al., eds. <u>Canadian Education: A History</u>. 1970 Scarborough, Ont.: Prentice-Hall of Canada.

Wilson, Logan, and Olive Mills, eds. Universal Higher Education: 1972 Costs, Benefits, Options. Washington, D.C.: American Council on Education.

Windham, Douglas M. Education, Equality, and Income Redistribution. 1970 Lexington, Mass.: Lexington Books.

Wirtz, William Willard and the National Manpower Institute. 1975 <u>The Boundless Resource. A Prospectus for an Education-Work</u> Policy. Washington, D.C.: The New Republic Book Co.

Wolfle, Dael. The Uses of Talent Princeton, New Jersey: Princeton 1971 University Press.

Wood, W. Donald. <u>Occupational Trends and Their Implications</u>. 1961 Kingston, Ont.: Industrial Relations Centre, Queen's University.

Woodhall, Maureen. Economic Aspects of Education. A Review of 1972 <u>Research in Britain. Windsor, England: National Foundation</u> for Educational Research in England and Wales.

Wright, Erik Olin. <u>Class, Crisis, and the State</u>. London: New Left 1978 Books.

1979 Academic Press.

Wykstra, Ronald A., ed. <u>Education and the Economics of Human Capital</u>. 1971 New York: The Free Press.

Wylie, Torrance. <u>Government Support of Universities and Colleges</u>. 1964 Ottawa: Canadian Universities Foundation.

Yotopoulos, Pan A. <u>Allocative Efficiency in Economic Development</u>. 1967 Athens: Center of Planning and Economic Research.

Yotopoulos, Pan A., and Jeffrey B. Nugent. <u>Economics of Development;</u> 1976 <u>Empirical Investigations</u>. New York: Harper and Row. Young, Darius, and A.V. Machinski. <u>An Historical Survey of Vocational</u> 1972 <u>Education in Canada</u>. Ottawa: Algonquin College.

Young, Michael. <u>The Rise of the Meritocracy, 1870-2033; An Essay</u> 1958 <u>on Education and Equality</u>. Harmondsworth, England: Penguin Books.

Ziel, H.R., ed. Education and Productive Society. Toronto: Gage. 1965

Interviews.

Bulcock, Dr. Jeffrey W. Professor, Department of Educational Foundations, Memorial University, Newfoundland. Interviewed several times in the fall and winter of 1980-1981 at the University of Alberta.

Leadbeater, David. Ph.D. student and Instructor, Department of Economics, University of Alberta. Interviewed in fall and winter of 1978-1979.

Shaffer, Dr. Edward H. Professor, Department of Economics, University of Alberta. Interviewed in the spring of 1979.

Sharir, Dr. S. Professor, Department of Economics, University of Alberta. Interviewed in the spring of 1979.